NASA Contractor Report 3146

N82-15456

(NASA-CR-3146) NASTRAN USER'S GUIDE (LEVEL 17.5) (Computer Sciences Corp., Hampton, Va.) 552 p dC A24/ME A01 CSCL 20K Unclas

H1/39 08739

NASTRAN[®] User's Guide (Level 17.5)

E. I. Field, D. N. Herting, and M. J. Morgan

CONTRACT NAS1-12969 JUNE 1979





NASA Contractor Report 3146

A THE R. P. LEWIS

NASTRAN[®] User's Guide (Level 17.5)

E. I. Field, D. N. Herting, and M. J. Morgan Universal Analytics, Inc. Playa Del Rey, California

Prepared for Langley Research Center under Contract NAS1-12969



Scientific and Technical

Information Office

PRECEDING PAGE BLANK NOT FILMED

INTRODUCTION

The User's Guide is designed to assist the current and potential users of NASTRAN¹ (NASA STRUCTURAL ANALYSIS) to make full use of the program's capabilities. NASTRAN is intended to serve the engineering community by integrating the state-of-the-art in engineering and numerical methods into a single program for the analysis of large and complex structures.

Four technical manuals serve to document the NASTRAN program. The <u>Theoretical Manual</u> presents a topic-by-topic discourse of the theory, assumptions, and methods of analysis for those interested in the analytical techniques. The <u>User's Manual</u> is the primary reference document which describes those items related to the use of NASTRAN that are independent of the computing systems being used. The third document is the <u>Programmer's Manual</u> which presents the details of the program, its organization, its file structure, and its implementation on each of the three computer systems, IBM 360/ 370, UNIVAC 1100 Series, and CDC 6000/CYBER Series for which NASTRAN is maintained. The fourth document is the <u>Demonstration Problem Manual</u> which serves to describe and illustrate sample problems used to verify the proper functioning of the program. These problems are selected to demonstrate the program capabilities with its many options for modeling, loading, selection of output and control of analyses.

Each of these four manuals provides a wealth of information useful to the understanding, use and development of NASTRAN. However, in that they were designed for reference purposes primarily, the organization of each does not readily lend itself to easy access except by experienced users and programmers. Therefore, the <u>User's Guide</u> was developed and organized so as to present the many options available in the logical sequence in which the input data would be prepared. These options are briefly described and tabulated according to purpose to help the user select the alternative best suited to his needs. References are given to the location in the standard documentation where detail descriptions may be found. The User's Guide also contains many examples illustrating typical applications of NASTRAN modeling and control features which may be used as models for input data preparation.

The user is expected to be familiar with the rudiments of finite element structural analysis and to be aware of basic modeling concepts. The Theoretical Manual, of course, can be referred

PURCEARED PAGE HEANK NOT FILMED

NASTRAN: Registered trademark of the National Aeronautics and Space Administration.

199 (N. 49) 199 199 199 199 199

to for a detail discussion of each of NASTRAN's analytical capabilities. The Guide directs the user to the application of this knowledge and suggests extensions of it to new problem areas. The User's Manual defines all the input and control options, outlines the steps taken for each type of analysis, and summarizes the error messages which may then be issued by NASTRAN. The sophisticated user will want to understand NASTRAN's organization, its controlling functions, and its data handling techniques which are fully described in the Programmer's Manual. Execution of the sample problems set out in the Demonstration Problem Manual will give the user invaluable exposure and practice in actual operation of the program.

TABLE OF CONTENTS

۰,

С

ì

Secti	lon		Page No.			
1.	INTE	ODUCTION				
	1.1	ORIENTATION OF THE USER'S GUIDE	1.1-1			
	1.2	ORIENTATION TO NASTRAN	1.2-1			
	1.3	NASTRAN AND STRUCTURAL ANALYSIS	1.3-1			
	1.4	NASTRAN AND STRUCTURAL DESIGN	1.4-1			
	1.5	PLANNING A NASTRAN ANALYSIS	1.5-1			
	1.6	MANAGEMENT OF NASTRAN PROCESSING	1.6-1			
2.	USE OF NASTRAN DOCUMENTATION					
	2.1	INTRODUCTION	2.1-1			
	2.2	USER'S GUIDE (UG)	2.2-1			
	2.3	USER'S MANUAL	2.3-1			
	2.4	THEORETICAL MANUAL	2.4-1			
	2.5	PROGRAMMER'S MANUAL	2.5-1			
	2.6	DEMONSTRATION PROBLEM MANUAL	2.6-1			
3.	MODELING WITH NASTRAN					
	3.1	INTRODUCTION	3.1-1			
	3.2	STRUCTURES	3.2-1			
	3.3	SUBSTRUCTURES	3.3-1			
	3.4	AERODYMAMICS	3.4-1			
	3.5	FLUIDS	3.5-1			
	3.6	ACOUSTIC CAVITIES	3.6-1			
	3.7	HEAT TRANSFER	3.7-1			
	3.8	OTHER MODELING OPTIONS	3.8-1			
	3.9	MODELING RESTRICTIONS	3.9-1			
4.	ENGI	EERING ANALYSES WITH NASTRAN				
	4.1	INTRODUCTION	4.1-1			
	4.2	LINEAR STATIC ANALYSIS (RIGID FORMATS 1, 2, and 14, APP DISP)	4.2-1			
	4.3	NONLINEAR STATIC ANALYSIS AND BUCKLING (RIGID FORMATS 4, 5, and 6, APP DISP)	4.3-1			
	4.4	NORMAL MODES ANALYSIS (RIGID FORMATS 3, 13, and 15, APP DISP)	4.4-1			
	4.5	COMPLEX EIGENVALUE ANALYSIS (RIGID FORMATS 7 and 10, APP DISP)	4.5-1			

Section			Page No.
4.6	DYNAMIC	RESPONSE ANALYSIS (RIGID FORMATS 8, 9, 11 and 12, APP DISP)	4.6-1
4.7	LINEAR S	TATIC HEAT TRANSFER ANALYSIS (RIGID FORMAT 1, APP HEAT)	4.7-1
4.8	NONLINEA	R STATIC HEAT TRANSFER ANALYSIS (RIGID FORMAT 3, APP HEAT)	4.8-1
4.9	TRANSIEN	T HEAT TRANSFER ANALYSIS (RIGID FORMAT 9, APP HEAT)	4.9-1
4.1	O AERODYNA	MIC FLUTTER ANALYSIS (RIGID FORMAT 10, APP AERO)	4.10-1
4.1	AERODYNA	MIC RESPONSE ANALYSIS (RIGID FORMAT 11, APP AERO)	4.11-1
4.1	2 CORRELAT	ION OF ANALYSIS VERSUS MODEL CAPABILITIES	4.12-1
5. INT	RODUCTION 1	O THE NASTRAN SYSTEM	
5.1	PROGRAM	DESIGN CRITERIA	5.1-1
	5.1.1	Modular Design and Executive Control	5.1-2
	5.1.2	Design for User Control	5.1-3
5.2	PROGRAM	SYSTEM REQUIREMENTS	5.2-1
	5.2.1	NASTRAN Computers	5.2-1
	5.2.2	Alphanumeric Data	5.2-2
	5.2.3	Word Packing	5.2-2
	5.2.4	The Open Core Concept	5.2.
		5.2.4.1 Definition of Open Core	5.2-3
		5.2.4.2 Example of an Application of Open Core	5.2-3
	5.2.5	NASTRAN Input and Output Files	5.2-3
		5.2.5.1 Use of the Operating System Input File	. 5.2-4
		5.2.5.2 Use of the Subroutine XRCARD	. 5.2-4
		5.2.5.3 Use of the Subroutine RCARD	. 5.2-4
		5.2.5.4 Use of the Operating System Output File	. 5.2-5
		5.2.5.5 GINØ	. 5.2-5
5.	. 3 NASTRA	N DATA DECF ORGANIZATION	. 5.3-1
	5.3.1	The NASTRAN Card	. 5.3-1
	5.3.2	Function of the Executive Control Deck	. 5.3-3
	5.3.3	Function of the Substructure Control Deck	. 5.3-4
	5.3.4	Function of the Case Control Deck	. 5.3-4
	5 3 5	Function of the Bulk Data Deck	5,3-4

Secti	on		Page No.
		5.3.6 INPUT Module Data	5.3-5
		5.3.7 NASTRAN Character Set	5.3-5
6.	E XE C	UTIVE CONTROL DESCRIPTION	
	6.1	INTRODUC (ION	6.1-1
	6.2	INPUT SEQUENCE AND FORMAT	6.2-1
	6.3	EXECUTIVE CONTROL OPTIONS	6.3-1
		6.3.1 Use of Executive Control Cards	6.3-1
	6.4	EXAMPLES OF EXECUTIVE CONTROL DECKS	6.4-1
		6.4.1 Example 1 - Cold Start, No Checkpoint	6.4-1
		6.4.2 Example 2 - Cold Start, Checkpoint, Modified System Parameters	6.4-1
		6.4.3 Example 3 - Restart, Checkpoint, Rigid Format Switch	6.4-2
		6.4.4 Example 4 - Pestart, No Checkpoint, Altered Rigid Format	6.4-2
7.	SUBS	TRUCTURE CONTROL DESCRIPTION	
	7.1	INTRODUCTION	7.1-1
	7.2	INPUT SEQUENCE AND FORMAT	7.2-1
	7.3	EXAMPLES OF SUBSTRUCTURE CONTROL DECKS	7.3-1
		7.3.1 Example 1 - Phase 1, Create Substructure WING	7.3-1
		7.3.2 Example 2 - Phase 2, Operations on Substructure WING	7.3-2
8.	CASE	CONTROL DESCRIPTION	
	8.1	INTRODUCTION	8.1-1
	8.2	INPUT SEQUENCE AND FORMAT	8.2-1
	8.3	CASE CONTROL OPTIONS	8.3-1
		8.3.1 Bulk Data Selection	8.3-1
		8.3.2 Output Selection (User's Manual, Section 2.3)	8.3-1
		8.3.3 Subcase Specification (User's Manual, Section 2.3)	8.3-3
	8.4	EXAMPLES OF CASE CONTROL DECKS	8.4-1
		3.4.1 Example 1 - Multiple Static Loadings	8.4-1
		8.4.2 Example 2 - Linear Combination of Subcases	8.4-2
		8.4.3 Example 3 - Statics Problem with Symmetry	8.4-3
		8.4.4 Example 4 - Use of REPCASE in Static Analysis	8.4-4

H

•

ţ

Secti	on		Page No.
		8.4.5 Example 5 - Use of MØDES and PUNCH Output Options	8.4-5
		8.4.6 Example 6 - Use of SØRT2 Output Option	8.4-6
		8.4.7 Example 7 - Use of XYØUT with Frequency and Random Response	8.4-6
9.	BULK [DATA DESCRIPTION	
	9.1	INTRODUCTION	9.1-1
	9.2	PROBLEM DEFINITION	9.2-1
	9.3	OPTIONS AND FORMAT	9.3-1
	9.4	USER MASTER FILE	9.4-1
	9.5	AUTOMATED DATA GENERATION	9.5-1
		9.5.1 Utility Module INPUT Usage	9.5-1
10.	PRINT	ING AND PLOTTING FEATURES	
	10.1	INTRODUCTION	10.1-1
	10.2	INPUT DATA ECHO AND DIAGNOSIS	10.2-1
	10.3	USER-REQUESTED DIAGNOSTIC MESSAGES	10.3-1
	10.4	ERROR AND WARNING MESSAGES	10.4-1
	10.5	CASE CONTROL DECK OUTPUT CONTROL	10.5-1
	10.6	SUBSTRUCTURE CONTROL DECK OUTPUT CONTROL	10.6-1
		10.6.1 SØFTPRINT Command	10.6-1
		10.6.2 ØUTPUT Subcommand	10.6-2
		10.6.3 PRINT Subcommand	10.6-2
		10.6.4 PLØT Command	10.6-2
	10.7	MISCELLANEOUS PRINTOUT	10.7-1
		10.7.1 Parameter-Controlled Output	10.7-1
		10.7.2 DMAP-Controlled Output	10.7-1
		10.7.3 Automatic Output	10.7-2
	10.8	STRUCTURAL AND CONTOUR PLOTTING	10.8-1
		10.8.1 Structural Plotting	10.8-1
		10.3.2 Contour Plotting	10.8-2
	10.9	XY PLOTTING AND PRINTING	10.9-1
	10.10	MATRIX TOPOLOGY	10.10-1

1

••

~

wow day had been

ection	CVCT-	VETEN CONTROL CARD DECKE			Page No.
11.	2421F	M CONTRO	L CARD DECK	S	
	11.1	INTRODU	CTION	• • • • • • • • • • • • • • • • • • • •	11.1-1
	11.2	NASTRAN	ON THE IBN	SYSTEM 360/370 (ØS)	11.2-1
		11.2.1	Deck Orgar	ization Summary	11.2-1
		11.2.2	Descriptio	ns of the Input Deck	11.2-1
			11.2.2.1	JØB Card	11.2-1
			11.2.2.2	NASTRAN Instream Procedure	11.2-1
			11.2.2.3	EXEC Card	11.2-8
			11.2.2.4	Control Cards for Optional Tape and Disk Files	11.2-9
			11.2.2.5	NASTRAN Data Deck	11.2-10
			11.2.2.6	/* Card	11.2-10
		11.2.3	Examples .		11.2-10
			11.2.3.1	Example 1 - Standard	11.2-10
			11.2.3.2	Example 2 - Standard, with Plotting	11.2-12
			11.2.3.3	Example 3 - Very Large Model, with Checkpoint and Plotting	11.2-12
			11.2.3.4	Example 4 - Restart	11.2-12
			11.2.3.5	Example 5 - Substructuring, Initial SØF Creation Checkpoint	11.2-13
			11.2.3.6	Example 6 - Substructuring, Multiple SØFs	11.2-13
			11.2.3.7	Example 7 - User Tape Input	11.2-14
			11.2.3.8	Example 8 - User Tape Output	11.2-14
			11.2.3.9	Example 9 - Demo Driver Decks and UNF Tape	11.2-14
			11.2.3.10	Example 10 - UMF File on Disk	11.2-15
1	1.3	NASTRAN	ON THE UNI	VAC 1108/1110 (EXEC 8)	11.3-1
		11.3.1	Deck Organ	ization Summary	11.3-1
		11.3.2	Descriptio	n of the Input Deck	11.3-1
			11.3.2.1	RUN Card	11.3-1
			11.3.2.2	QUAL Card	11, 3-2
			1.3.2.3	Control Cards for Tapes	11.3-2
			2.4	Link 1 Card	11 2 2

24

÷ .

Section			<u>Page No.</u>
	11.3.2.5	The NASTRAN Data Deck	11.3-3
	11.3.2.6	Other Links	11.3-3
	11.3.2.7	The FIN Card	11.3-3
11.3.3	Cataloging	the NASTRAN Program	11.3-3
11.3.4	Examples .		11.3-4
	11.3.4.1	Example 1 - Options = none	11.3-4
	11.3.4.2	Example 2 - Options = plot	11.3-6
	11.3.4.3	Example 3 - Options = checkpoint	11.3-6
	11.3.4.4	Example 4 - Options = checkpoint, plot	11.3-6
	11.3.4.5	Example 5 - Options = restart	11.3-7
	11.3.4.6	Example 6 - Options = restart, plot	11.3-7
	11.3.4.7	Example 7 - Options = restart, checkpoint	11.3-7
	11.3.4.8	Example 8 - Options = restart, checkpoint, plot	11.3-8
	11.3.4.9	Example 9 - Options = create a UMF	11.3-8
	11.3.4.10	Example 10 - Options = edit the old UNF and create a new UMF \dots	11.3-8
	11.3.4.11	Example 11 - Options = UMF	11.3-9
	11.3.4.12	Example 12 - Options = UMF, plot	11.3-9
	11.3.4.13	Example 13 - Options = UMF, checkpoint	11.3-9
	11.3.4.14	Example 14 - Options = UMF, checkpoint, plot	11.3-10
	11.3.4.15	Example 15 - Options = user tape input	11.3-10
	11.3.4.16	Example 16 - Options = user tape output	11.3-10
	11.3.4.17	Example 17 - Options = user tape input and output	11.3-11
	11.3.4.18	Example 18 - Options = new SØF on disk	11.3-11
	11.3.4.19	Example 19 - Options = old SØF on tape	11.3-12
11.4 NASTRAN	ON THE CDC	6000/CYBER (NØS AND NØS/BE)	11.4-1
11.4.1	Organizati	on of the Input Deck	11.4-1
11.4.2	Descriptio	ns of the Input Deck	11. 4-1
	11.4.2.1	JØB Description Cards	11.4-1
	11.4.2.2	Control Cards to Load NASTRAN	11.4-1
	11.4.2.3	Optional Tape Setup Cards	11.4-2

Section	n	Pa	age No.
		11.4.2.4 Execution Procedure Cards	11.4-3
		11.4.2.5 NASTRAN Data Deck	11.4-3
		11.4.2.6 EØR and EØF Cards	11.4-3
		11.4.3 Examples	11.4-3
		11.4.3.1 Example 1 - Simple Load and Execute (NØS)	11.4-4
		11.4.3.2 Example 2 - Checkpoint and Plot	11.4-5
		11.4.3.3 Example 3 - Static Analysis Restart	11.4-6
		11.4.3.4 Example 4 - Restart and Checkpoint Using Disk	11.4-6
		11.4.3.5 Example 5 - Substructuring, Initiate SØF	11.4-7
		11.4.3.6 Example 6 - Substructuring, Extend SØF	11.4-8
		11.4.3.7 Example 7 - UMF Run	11.4-8
		11.4.3.8 Example 8 - Executive Demonstration Problems	11.4-9
		11.4.3.9 Example 9 - Data Transmission	11.4-10
		11.4.3.10 Example 10 - Execution under NØS/BE	11.4-11
12.	NASTR	AN INFORMATION SOURCES	
	12.1	NASTRAN NEWSLETTER	12.1-1
	12.2	SOFTWARE PROBLEM REPORT (SPR) LOG	12.2-1
	12.3	NASTRAN USER'S COLLOQUIUM	12.3-1
	12.4	SPR SUBMITTAL	12.4-1
	12.5	DER SUBMITTAL	12.5-1
13.	OPE RA	TING PRINCIPLES OF NASTRAN	
	13.1	INTRODUCTION	13.1-1
	13.2	PROGRAM DESIGN OVERVIEW	13.2-1
	13.3	THE EXECUTIVE SYSTEM	13.3-1
	13.4	FUNCTIONAL MODULES	13.4-1
		13.4.1 Module Types	13.4-1
		13.4.2 Module Data Interfaces	13.4-1
		13.4.3 Module Overlays and Open Core	13.4-2
		13.4.4 Module Restrictions	13.4-3
	13.5	THE NASTRAN 1/0 SYSTEM	13.5-1

. الا

Section	<u>1</u>			Page No.
		13.5.1	External User Interfaces	13.5-1
		13.5.2	NASTRAN Data Blocks	13.5-1
		13.5.3	Checkpoint Files	13.5-2
		13.5.4	Substructure Operating Files	13.5-3
	13.6	NASTRAN	CHECKPOINT/RESTART LOGIC	13.6-1
		13.6.1	Restart Types	13.6-1
		13.6.2	Restart Decision Process	13.6-2
		13.6.3	Restart Rules	13. 6-3
	13.7	NASTRAN	PROGRAMMING CONVENTIONS	13. 7-1
		13.7.1	Diagnostic Outputs	13.7-1
		13.7.2	Matrix/Vector Degree of Freedom Sets	13.7-2
		13.7.3	Error Messages	13.7-4
14.	ESTIN	WATING RE	SOURCE REQUIREMENTS FOR NASTRAN	
	14.1	INTRODU	JCTION	14.1 -1
	14.2	METHODS	FOR PREDICTING NASTRAN COST FACTORS	14.2-1
		14.2.1	Past NASTRAN Experience	14.2-1
		14.2.2	Testing for Cost with NASTRAN	14.2-1
		14.2.3	Evaluation of the Computerized Algorithms	14.2.2
	14.3	REQUIR	EMENTS FOR THE RIGID FORMATS	14.3-1
		14.3.1	Static Condensation and Guyan Reduction	14.3-1
		14.3.2	Element Matrix Generation	14.3-3
		14.3.3	Determination of Open Core Size	14.3-4
	14.4	ESTIMA	TING CPU TIME REQUIREMENTS	14.4-1
		14.4.1	Matrix Multiplication and Addition - MPYAD	. 14.4-1
		14.4.2	Symmetric Triangular Decomposition (SDCØMP)	. 14.4-2
		14.4.3	Symmetric Decomposition with Spill	. 14.4-3
		14.4.4	Forward-Backward Substitution - FBS, GFBS	. 14.4-5
		14.4.5	Mode Extraction (Module READ)	. 14.4-5
	14.5	ESTIMA	TING OTHER HARDWARE REQUIREMENTS	. 14.5-1
		14.5.1	Input/Output Data File Factors	. 14.5-1

.

.

Section	on		Page No.
		14.5.2 Tape and Permanent File Requirements	14.5-2
		14.5.3 Substructure Operating File (SØF) Requirements	14.5-3
15.	DIREC	CT MATRIX ABSTRACTION PROGRAM (DMAP)	
	15.1	DMAP OVERVIEW	15.1-1
	15.2	FUNDAMENTAL DMAP RULES	15.2-1
	15.3	EXECUTIVE OPERATIONS	15.3-1
	15.4	UTILITY OPERATIONS	15.4-1
	15.5	MATRIX OPERATIONS	15.5-1
	15.6	STRUCTURAL OPERATIONS	15.6-1
16.	NASTE	VAN EXTERNAL INTERFACES	
	16.1	INTRODUCTION	16.1-1
	16.2	PRE- AND POSTPROCESSORS	16.2-1
	16.3	NASTRAN MODULES FOR DATA TRANSFER	16.3-1
		16.3.1 NASTRAN-GIN# Files	1 6 . 3-1
		16.3.2 NASTRAN FØRTRAN Files	16.3-1
		16.3.3 NASTRAN Punch Files	16.3-2
		16.3.4 NASTRAN SØF Files	16.3-2
17.	INTRO	DUCTION TO NASTRAN APPLICATIONS	
	17.1	INTRO DUCTION	17.1-1
	17.2	SUMMARY OF APPLICATIONS	17.2-1
		17.2.1 Static Analysis	17.2-1
		17.2.2 Dynamic Analysis	17.2-1
		17.2.3 Substructuring and Modal Synthesis	17.2-1
		17.2.4 Aeroelastic Analysis	17.2-1
		17.2.5 Heat Transfe r Analysis	17.2-2
		17.2.6 Cyclic Symmetry Modeling	17.2-2
		17.2.7 Hydroelastic Analysis	17.2-2
		17.2.8 Acoustic Cavity Analysis	17.2-3
		17.2.9 Fully Stressed Design	17.2-3
	17.3	PLANNING FOR SPECIAL APPLICATIONS	17.3-1

`

1.1

<u>Sectio</u>	Pag			
18.	STATIC ANALYSIS			
	18.1	INTRODUCTION	18.1-1	
	18.2	DEFINING THE STRUCT'JRAL MODEL	18.2-1	
		18.2.1 Finite Elements	18.2-3	
		18.2.2 Constraints and Boundary Conditions	18.2-4	
		18.2.3 Loading Constraints	18.2-6	
	18.3	STATIC ANALYSIS CONTROL DATA		
		18.3.1 Executive Control Data	18.3-1	
		18.3.2 Case Control Deck	18.3-1	
		18.3.3 Control Parameters	18.3-2	
	18.4	OTHER STATIC SOLUTIONS	18.4-1	
		18.4.1 Property Optimization	18.4-1	
		18.4.2 Cyclic Symmetry	18.4-1	
		18.4.3 Nonlinear Static Solutions	18.4-2	
	18.5	STATIC ANALYSIS EXAMPLES	18.5-1	
		18.5.1 Rigid Format 1 - Shell with Symmetry Boundary Conditions	18.5-1	
		18.5.1.1 Input Cards - Unclamped Boundary and Symmetry Plots	18.5-1	
		18.5.1.2 Notes on Data Cards	18.5-2	
		18.5.1.3 Input Cards - Restart with Clamped Boundary	18.5-4	
		18.5.1.4 Notes on Data Cards for Restart	18.5-4	
		18.5.2 Rigid Format 1 - Nonsymmetric Bending of a Cylinder	18.5-6	
		18.5.2.1 Input Cards - Axisymmetric Model	18.5-7	
		18.5.2.2 Notes on Data Cards	18.5-7	
		18.5.3 Rigid Format 2 - Free Body Ring	18.5 -9	
		18.5.3.1 Input Cards - Inertia Relief	18.5-10	
		18.5.3.2 Notes on Data Cards	18.5-10	
		18.5.4 Rigid Format 5 - Symmetric Buckling of a Cylinder	18.5-12	
		18.5.4.1 Input Cards - With Structural Plotting	18.5-14	
		18.5.4.2 Notes on Data Cards	18.5-15	
		18.5.5 Rigid Format 6 - Piecewise Linear Analysis with Rod Elements	18.5-17	

·~. .

•

Secti	on	<u>Pa</u>	ge No.
		18.5.5.1 Input Cards - Stress-Dependent Materials 1	8.5-17
		18.5.5.2 Notes on Data Cards	8.5-19
19.	DYNAN	MJC ANALYSES	
	19.1	INTRODUCTION	9.1-1
	19.2	NASS MODELING	9.2-1
		19.2.1 Inertia Properties of Structural Elements	9.2-1
		19.2.2 Concentrated Mass Elements	9. 2-1
		19.2.3 Other Mass Options	9.2-2
	19.3	DAMPING EFFECTS	9.3-1
	19.4	SPECIAL DYNAMIC EFFECTS	9.4-1
	19.5	DIRECT VS. MODAL FORMULATION	9.5-1
	19.6	EIGENVALUE ANALYSIS	9.6-1
		19.6.1 Normal Modes Analysis	9.6-1
		19.6.2 Complex Eigenvalue Analysis	9.6-2
	19.7	FREQUENCY RESPONSE ANALYSIS	9.7-1
	19.8	TRANSIENT RESPONSE ANALYSIS	9.8-1
	19.9	EXAMPLES OF DYNAMIC ANALYSES	9,9-1
		19.9.1 Rigid Format 3 - Normal Modes Analysis of a Plate	9.9-1
		19.9.1.1 Input Cards - With Structural Plotting	9.9-3
		19.9.1.2 Notes on Data Cards	9.9-4
		19.9.2 Rigid Format 7 - Damped Vibration of a String	9.9-5
		19.9.2.1 Input Cards - Scalar Elements	9.9-6
		19.9.2.2 Notes on Data Cards	9.9~6
		19.9.3 Rigid Format 10 - Rocket Guidance and Control	9.9-7
		19.9.3.1 Input Cards - DMAP Alter, Transfer Function and Resequencing	9.9-8
		19.9.3.2 Notes on Data Cards	9.9-9
		19.9.4 Rigid Format 8 - Direct Frequency Response of a Plate	9.9-1 0
		19.9.4.1 Input Cards - Frequency Dependent Loading	9.9-11
		19.9.4.2 Notes on Data Cards	9.9-11
		19.9.5 Rigid Format 11 - Modal Frequency and Random Response of	
		a Beam	9.9.11

-

h

٠.

Section			Page No.
		19.9.5.1 Input Cards - "Complex" Loading and Random Response Output	10.0.13
		19.9.5.2 Notes on Data Cards	10 0-13
	19.9.6	Rigid Format 9 - Direct Transient Analysis with Direct Matrix Input	10 0-10
		19.9.6.1 Input Cards - Direct Matrix Input and XYØUT Plotting	10.0.10
		19.9.6.2 Notes on Data Cards	19.9-10
	19.9.7	Rigid Format 12 - Hodal Transient Response of a Beam	19.9-19
		19.9.7.1 Input Cards - Both Structural and XY@UT Plotting	10 0.91
		19.9.7.2 Notes on Data Cards	14 9 22
20. SUBS	TRUCTURIN	G AND MODAL SYNTHESIS	
20.1	INTRODU	CTION	20.1.1
20.2	BASIC (ONCEPTS	20.1-1
20.3	SUMMARY	OF OPERATIONAL COMMANES	
	20.3.1	The Substructure Analysis Tree	20.3*1
	20, 3, 2	Equivalence Operation	
	20.3.3	Combine Operation	20.3-1
	20, 3, 4	Reduce Operation	
	20, 3, 5	MREDUCE or CREDUCE Operation	20.3-5
	20.3.6	Solve Operation	20.3-5
	20.3.7	Recover Operations	20.3-6
	20.3.8	Plot Operations	20.3-6
20 4	CURSTRE		20.3-6
20.4		Duckler Advaltsis EXAMPLES	20-4-1
		Problem 1 - Substructure Analysis Using Rigid Format 1 and 2	20.4-1
	20.4.2	Synthesis and Rigid Format 9	
		20.4.2.1 Description	
		20.4.2.2 Input	20.4-1
		20.4.2.3 Results	
21. AE ROE	LASTIC A	ALYSTS	
21.1	INTRODUC	TION TO NASTRAN AEROELASTICITY	

• 1

Ξ,

.

2

Sec. 2

Section		Page No.
21.2	DATA INPUT AND GENERATION	21.2-1
21.3	STRUCTURAL-AERODYNAMIC INTERCONNECTIONS	21.3-1
21.4	AERODYNAMIC MODELING	21.4-1
	21.4.1 Modal Flutter Analysis	21.4-1
	21.4.1.1 The K-Method	21.4-2
	21.4.1.2 The KE-Method	21.4-2
	21.4.1.3 The PK-Method	21.4-3
	21.4.2 Aeroelastic Response Analysis	21.4-3
21.5	MODAL FLUTTER ANALYSIS EXAMPLE	21.5-1
	21.5.1 Input	21.5-1
	21.5.2 Theory	21.5-1
	21.5.3 Results	21.5-1
	21.5.4 Driver Deck and Sample Bulk Data	21.5-2
21.6	AEROELASTIC RESPONSE EXAMPLE	21.6-1
	21.6.1 Frequency Response Input	21.6-1
	21.6.2 Frequency Response Results	21.6-2
	21.6.3 Frequency Response Driver Deck and Sample Data	21.6-2
	21.6.4 Transient Response Input	21.6-4
	21.6.5 Transient Response Results	21.6-4
	21.6.6 Transient Response Driver Deck and Sample Bulk Data	21.6-5
22. HEA	T TRANSFER ANALYSIS	
22.	INTRODUCTION TO NASTRAN HEAT TRANSFER	. 22.1-1
22.3	2 HEAT IRANSFER ELEMENTS	. 22.2-1
22.	3 CONSTRAINTS AND PARTITIONING	. 22.3-1
22.	4 THERMAL LOADS	. 22.4-1
22.	5 STATIC ANALYSIS	. 22.5-1
22.	6 TRANSIENT ANALYSIS	. 22.6-1
22.	7 HEAT TRANSFER EXAMPLE	. 22.7-1
	22.7.1 Input	. 22.7-1
	22.7.2 Theory	. 22.7-1

TABLE OF CONTENTS (Contir	nued)
---------------------------	-------

Section	<u>on</u>		Page No.	
		22.7.3 Results	22.7-2	
		22.7.4 Driver Deck and Sample Bulk Data	22.7-2	
23.	CYCLIC SYMMETRY MODELING			
	23.1	CYCLIC SYMMETRY MODELS	23.1-1	
	23.2	STATIC ANALYSIS	23.2-1	
	23.3	NORMAL MODES ANALYSIS	23.3-1	
	23.4	CYCLIC SYMMETRY EXAMPLE	23.4-1	
		23.4.1 Description	23.4-1	
		23.4.2 Input	23.4-1	
		23.4.3 Results	23.4-2	
		23.4.4 Driver and Sample Bulk Data	23.4-2	
24.	HYDROELASTIC ANALYSIS			
	24.1	SOLUTION OF THE FLUID MODEL	24.1-1	
	24.2	HYDROELASTIC INPUT DATA	24.2-1	
	24.3	RIGID FORMATS	24.3-1	
	24.4	HYDROELASTIC DATA PROCESSING	24.4-1	
	24.5	HYDROELASTIC SAMPLE PROBLEM	24.5-1	
		24.5.1 Description	24.5-1	
		24.5.2 Driver Deck and Bulk Data Listing	24.5-2	
25.	ACOUS	STIC CAVITY ANALYSIS		
	25.1	MODELING OF THE ACOUSTIC CAVITY	25.1-1	
	25.2	ASSUMPTIONS AND LIMITATIONS	25.2-1	
	25.3	ACOUSTIC CAVITY SAMPLE PROBLEM	25.3-1	
		25.3.1 Description	25.3-1	
		25.3.2 Driver Deck and Bulk Data Listing	25.3-3	
26.	FULLY	STRESSED DESIGN ANALYSIS		
	26.1	FULLY STRESSED DESIGN THEORY	26.1-1	
	26.2	APPLICATION	26.2-1	
	26.3	SAMPLE PROBLEM	26.3-1	
		26.3.1 Description	26.3-1	

Sect	ion

ſ

		Page No.
26.3.2	Input	26.3-2
26.3.3	Theory	26. 3- 3
26.3.4	Results	26.3-3
26.3.5	Driver Deck and Bulk Data	26. 3-4

. . . .

•••

• 4

1. INTRODUCTION

Contraction and a

1.1 ORIENTATION TO THE USER'S GUIDE

The overall organization of the User's Guide is subdivided into two parts. The first part begins with a review of NASTRAN design philosophy, specific instructions and suggestions on how to use the program documentation, data descriptions for model and load definition and descriptions of the program control options. The description of these basic tools of NASTRAN is followed by examples and guidelines for the execution of each of the standard forms of analysis available in NASTRAN and a summary review of the computer and plotter output that can be obtained with NASTRAN. The Guide then discusses some of the utility functions available for user convenience, provides examples of machine-dependent card deck setup for execution of NASTRAN on the three computer systems for which NASTRAN is maintained. The communication devices designed to help keep the user up to date with the current developments and for him to share this experience with other NASTRAN users are outlined. The operating principles and means of estimating computer resource requirements, the special NASTRAN matrix abstraction language (DMAP) for controlling execution, and the means of interfacing NASTRAN with other programs are presented. All three sections of the Guide serve to introduce the capabilities which are then discussed in terms of their applications in the second part of the Guide. The second part of the Guide describes in separate sections the analytical capabilities for which NASTRAN is especially designed: static and dynamic analyses, substructuring, aerodynamics, heat transfer, and a variety of other special purpose modeling facilities.

The User's Guide is not intended to represent the full detail of NASTRAN's capabilities. NASTRAN provides such a flexible array of analytical options, the user will find unending combinations of alternatives. As the user becomes familiar and comfortable with the use of NASTRAN, he will want to use the more sophisticated features of NASTRAN. The User's Guide will serve to help direct these users to effectively apply their imagination and analytical abilities. However, to keep the volume of this Guide within manageable proportions, the user is expected and encouraged to make full use of the references provided to the standard NASTRAN manuals.

1.1-1

1.2 ORIENTATION TO NASTRAN

NASTRAN is a general purpose computer program designed to analyze the behavior of structures under a wide variety of operating environments. Its nature is general purpose which makes the program useable for structures of any size, shape, class, or configuration, for any geometric representations which can be identified by any convenient coordinate system, for elastic relations which may range from isotropic to general anisotrophy, for non-linear behavior that can be represented by piecewise linear approximations, for complex as well as real matrix operations, for vibration frequency and mode determination, and for a wide variety of loading conditions. The various loading conditions may be concentrated or distributed, transient, steady-state sinusoidal; static temperature profiles; enforced deformations; time-varying as well as static surface and body forces; and stationary Gaussian random excitation. NASTRAN will also perform aerodynamic, hydroelastic, and heat transfer analyses.

NASTRAN, as a computer program, is organized to permit future modification and continued expansion to new disciplines. MASTRAN is maintained and supported by the National Aeronautics and Space Administration (NASA). New releases of the program are made periodically to meet the rapidly expanding needs of government agencies and the aerospace industry. It also has received wide acceptance in the petro-chemical, nuclear reactor, transportation, and construction industries both domestically and abroad.

NASTRAN may be obtained from COSMIC, 112 Barrow Hall, University of Georgia, Athens, Georgia 30602. Versions of NASTRAN are available for installation on each of three computer systems: the CDC 6000 Series, the IBM 360/370 Series, and the UNIVAC 1100 Series computers.

The following sections provide an introduction to NASTRAN as a computer program system for structural analysis and structural design. The final two sections focus on the management topics of planning and controlling an analysis activity using NASTRAN.

1.2-1

1.3 NASTRAN AND STRUCTURAL ANALYSIS

NASTRAN analyzes models by the displacement method of the finite element approach. With this approach, the distributed physical properties of a structure are represented by a model which consists of a finite number of idealized components or structural elements. These elements are defined by, and serve to interconnect a finite number of grid points to which loads may be applied. The grid point is the basic framework for the structural model and all aspects of the model definition are referenced either directly or indirectly to these grid points. The structural element (which may be a line, a surface, or a solid element) is a convenient means for specifying many of the properties of the structure, including material properties, mass distribution, and some types of applied loads. In addition, various kinds of constraints can be applied to grid points. Single-point constraints are used to specify boundary support conditions and multipoint constraints may be used to specify a linear relationship among selected degrees of freedom.

Structural elements are provided in NASTRAN for the specific representation of the more common types of construction including rods, beams, solids, shear panels, plates, and shells of revolution. More general types of construction are treated by combinations of these elements and/or the use of "general" or user specified elements. For sophisticated dynamic and control systems analyses, aerodynamic transfer functions and other non-structural features also may be incorporated into the analytical model.

NASTRAN has been specifically designed to treat large problems with many degrees of freedom. The only limitations on problem size are those imposed by practical considerations of running time and by the ultimate capacity of auxilliary storage devices provided by the computer facility being used. Computational procedures implemented in NASTRAN have been selected to provide the maximum obtainable efficiency for large problems with particular attention given to processing sparce matrices. Though NASTRAN is decidedly <u>not</u> constrained to in-core processing only, the more core memory that can be made available, the more efficient that processing will be.

NASTRAN also provides an automated multi-stage substructuring capability for static and dynamic analyses. This capability is particularly useful for handling large or complex models and for modeling structures with repetitive components. The capability may be used to realize significant processing and model development efficiencies for very large models.

1.3-1

INTRODUCTION

The needs of the structural analyst have been considered in all aspects of the design of the program. However, in view of the wide range of possible applications of the program now and in the future, all the analytical and user needs could not be anticipated in advance. For this reason, a high degree of flexibility and generality has been incorporated into the program. For example, in addition to the usual list of structural elements that refer to specific types of construction, the user may elect to define a "general" element to represent part of a structure by providing deflection influence coefficients, or to represent part of a structure by its vibration modes. The user is provided with a number of convenience features, including plotting, selective output, checkpoint/restart, and diagnostic facilities which are definite necessities for large problems.

A major difficulty that the user faces in the solution of large problems is the avoidance of errors in the preparation of input data. Card formats and card ordering are made as simple and flexible as possible in order to avoid errors caused by trivial violations of format rules. A number of aids such as automatic data generation, extensive plotter capability, grid point singularity tests, and grid point force balance printout have been provided to help the user develop and validate his model. The sophisticated checkpoint/restart capability is designed for recovery from data or system error conditions, to allow for examination of intermediate results, and to allow for retrieval of additional selective output following completion of the analysis.

1.3-2

1.4 NASTRAN AND STRUCTURAL DESIGN

NASTRAN may be effectively used to support the $t \ge 0$ of structural design as well as perform detailed structural analyses.

The automated multi-stage substructuring capability is an ideal tool for the support of design activities. The substructuring capability may be used to model as separate components those segments of the overall structure which may require design iterations. The substructure least likely to be changed would then be incomporated in the early stages of model development since it would not need to be regenerated in order to study the effects of design changes made to components added in the later stages of model development.

NASTRAN also provides a capability for an iterative fully stressed property design option with static analysis. This capability automatically minimizes cross sectional area thickness, or moments of inertia to permit the structure to carry prescribed loads as maximum stress.

Special facilities have been incorporated to help a user interface NASTRAN with other programs for pre- and post-processing. Thus, automated data generation and/or output interpretation can be utilized with special user programs designed to meet his special needs.

Whether for design or for analysis, exploiting NASTRAN's capabilities requires planning and control to obtain reliable results efficiently.

PLANNING A NASTRAN ANALYSIS

And the second second

1.5 PLANNING A NASTRAN ANALYSIS

Detailed modeling of complex structures for analysis can be expensive in terms of both human and computer resources. Control of these costs is usually required, and control can be best maintained by adherence to an analysis project plan. In this section, guidelines to the preparation of a NASTRAN analysis plan are presented.

A suggested approach to planning a NASTRAN analysis involves separating the activity into three phases: the problem definition phase, the analysis set-up phase, and the analysis execution phase.

The problem definition phase provides the framework for the remaining NASTRAN planning activity. Key activities in this phase include:

- Identify the structure to be modeled using references to drawings, layouts, or other documents.
- 2. Carefully define the purpose and objectives of the analysis.
- 3. Identify the desired results to be obtained from the analysis.
- 4. Define the budget, computer resources, and timeframe for the analysis.
- 5. Define the engineering assumptions which are to be made.

The analysis set-up phase involves a variety of tasks. Some pertinent questions to be an swered are also indicated for some tasks.

- Locate and collect descriptive data of the model. Can automatic data generation be used?
- 2. Define normal NASTRAN output requirements. What theoretical or experimental date are available for model checkout and validation?
- 3. Identify external input or output interfaces with NASTRAN. Are pre- or post-processors to be used or developed?
- 4. Select the required Rigid Format and note any special requirements of that Rigid Format. Will several Rigid Formats be used at different stages in the analysis?

1.5-1

INTRODUCTION

- 5. Plan for documentation at this stage of the analysis. What structural plots are required for documentation?
- 6. Prepare a detailed cost resource requirement and schedule plans for the analysis. Are there plans consistent with the budget computer resources, and timeframe established in the problem definition phase?

The plan for execution phase should include a description of the tasks and provide answers to the questions which follow.

- Specify data base management procedures. Will the User's Master File (UMF) or Substructure Operating File (SØF) be used, or will other computer system file utilities be used?
- 2. Specify methods for checking the model and/or its components. Can plotting be effectively used? Can simplified loadings and boundary conditions be used at intermediate stages of model development and checkout?
- 3. Plan for tracking the analysis progress in terms of cost, model validity, and model accuracy. How can NASTRAN diagnostics or special Cost Control output be used?
- 4. If the analysis will involve using an unfamiliar Rigid Format, DMAP alters, or other new modeling procedures, plan a "pilot model" analysis. With the pilot model, test all NASTRAN features to be exercised in the final analysis. This will assure the planned analysis methods will satisfy the analysis objectives.

1.6 MANAGEMENT OF NASTRAN PROCESSING

The organization of the NASTRAN system provides many means for management control and monitoring of NASTRAN processing. NASTRAN provides standard solution techniques called Rigid Formats for executing different types of static, dynamic, stability, or heat transfer analyses. The Rigid Formats provide the principal form of management control because they provide a proven, well documented procedure for a given analysis sequence. In conjunction with each Rigid Format, NASTRAN provides a checkpoint/restart capability. This capability allows running a problem in multiple steps to obtain intermediate results, as required, for checking the model and its response. Also, the checkpoint/restart capability can be used for efficient error recovery. With this capability, costly and unnecessary repetition of valid data calculations may be avoided when recovering from errors.

NASTRAN also provides an extensive diagnostic output capability. The diagnostic messages can be used in several ways to monitor the progress of an analysis.

The concept of NASTRAN Rigid Formats is described in Section 4 of this guide. The use of checkpoints, restarts, and diagnostics is presented in Section 6.

NASTRAN documentation also contains other information which can be used to effectively manage a structural analysis project with NASTRAN. For example, Section 14 presents information to help the user estimate resource requirements (machine time and core) prior to beginning processing. These estimates can also be used to support cost p¹ jections for the analysis.

2. USE OF NASTRAN DOCUMENTATION

Sec. 1

1000

2.1 INTRODUCTION

Documentation for NASTRAN consists of five manuals; the User's Guide, the User's Manual, the Theoretical Manual, the Programmer's Manual, and the Demonstration Problem Manual. Each document's purpose, content, and use is described in this section. Other sources of NASTRAN information pertaining to reported errors or corrections, and user's experience, for example, are described in Section 12.

2.2 USER'S GUIDE (UG)

The User's Guide is designed to assist both the current and potential user of NASTRAN to make full, efficient application of the program's capabilities. An important function of the Guide is to provide engineering management with an overview of NASTRAN's technical capabilities. With this concise overview, management may better direct NASTRAN activities from an informed position not previously possible without extensive reading of NASTRAN documentation or personal hands-on experience with the program.

The first thirteen sections of the guide are designed to provide the new and occasional user with an awareness of the access to standard NASTRAN analysis capabilities. These sections provide much of the information required and the reader will be directed to the appropriate NASTRAN reference manuals for technical details of model building, selection of analysis to be performed, data deck and system control deck creation.

The remaining sections of the User's Guide are designed to provide the more sophisticated NASTRAN user with information needed to execute large, complex, or unusual analyses of structures and coupled structural/non-structural component systems. For example, material presented in "Operating Principles of NASTRAN" and a section on the introduction to the matrix abstraction language (DMAP) provide an understanding of program design and guidance needed to exploit many non-standard NASTRAN capabilities.

A significant portion of the User's Guide, beginning with Section 17, consists of independent sections describing specialized NASTRAN capabilities in engineering terms. These are intended to aid the sophisticated user with substructuring with modal synthesis, aeroelasticity, heat transfer analyses, and other capabilities including acoustic analysis, cyclic symmetry, hydroelastic analysis, and fully stressed design.

Examples are presented throughout the Guide which can be carefully examined to help the user set up his own problem for execution. Also, references are given to locations in the NASTRAN manuals where additional or more detailed information is available. The contents of the manuals provided with NASTRAN are summarized below.

2.3 USER'S MANUAL

The User's Manual presents the more practical subjects, in a reference manual style on how to input data to NASTRAN and on how to control its processing. This manual begins with a general discussion of structural modeling with a description, in physical terms, of each modeling tool available.

Many structural elements are provided in NASTRAN. Each type of element is discussed in terms of how it can be used, what input is required, and what type of output can be obtained as results. The reader is shown how to use the constraints and partitioning features of NASTRAN. The many types of static and dynamic load capabilities are summarized. Both the direct and modal formulations of the dynamic equations of motion are presented.

The specific detail of format and card content for each input and control option of the Executive Control, Case Control, and Bulk Data Decks is presented separately in Sections 2.2, 2.3, and 2.4, respectively. In each section, the card mnemonics are listed alphabetically for easy reference. Also, since large problems may require sizable input decks, a description of how to use the User's Master File feature for storing the data on tapes is given in Section 2.5. The automated data generating feature and substructuring with modal synthesis are presented in Sections 2.6 and 2.7, respectively. The extensive plotting capability and its control options are discussed separately in Section 4.

Section 3 presents a general description on how to use the current Rigid Formats which direct the flow of all NASTRAN's standard analysis. Here is also discussed the input data for processing, how to use the checkpoint/restart feature, and how to alter the Rigid Format for special purposes. A subsection is devoted for each available Rigid Format. Each section contains a detail description of the processing steps, a summary of the Case Control requirements, a list of the output options, and a list of the special parameters provided for execution control and special output requests.

Section 5 contains all the detail instructions with examples for using the Direct Matrix Abstraction Program (DMAP) commands. These DMAP commands are used to control all of NASTRAN processing. They are used to build and to alter the standard Rigid Formats or to create entirely new sequences for matrix processing with NASTRAN.

Section 6 lists and explains all of the error and diagnostic messages issued by NASTRAN. Section 7 contains a very useful dictionary of the common terms and mnemonics used throughout the NASTRAN documentation.

2.3-1

THEORETICAL MANUAL

2.4 THEORETICAL MANUAL

The Theoretical Manual presents the design concepts and the mathematical basis for all of NASTRAN's capabilities. The user will find an overall view of the basic philosophy of NASTRAN, a summary definition of how the program is organized, and a mathematical description of the operations it performs. The assumptions and approximations are identified. The available finite element modeling tools such as grid points, coordinate systems, element connectivities, direct matrix input, multipoint constraints, modal coordinates, transfer functions are described.

Suggestions are made on how to use these capabilities. Sections are included on special topics such as heat transfer analysis, a@roelasticity, plotter graphics, special modeling techniques, error analysis, and interaction between structures and fluids. An index for the Theoretical Manual is included to help the NASTRAN user locate additional references to the specific subjects of interest.

2.5 PROGRAMMER'S MANUAL

The Programmer's Manual contains the detailed descriptions of the many modules and subroutines of NASTRAN and the data blocks on which they operate. The frequent user of NASTRAN would do well to read the first section. It contains information on how NASTRAN processes data and how it controls that processing. Though understanding these processes is not required, it will help to visualize what happens during the execution of an analysis and may help the user thereby to efficiently organize his processing requests. Also, NASTRAN offers such a plethora of options, not all combinations of which can be adequately described. Hence, an understanding of how the program executes will help the user interpret the documentation as needed to exploit the system.

Section 2 is also of value to an experienced user. Here he will find a detailed description of the many data pools generated by and maintained within NASTRAN. Occasionally the user will encounter the need for some of this data to help him locate an error in his model. NASTRAN provides the facilities for retrieving this mata and printing it out with special DMAP utilities. An alphabetical index of the data block names is provided to help locate the pertinent data descriptions.

The remaining sections are devoted to describing the utility subroutines, the functional modules, the matrix processing modules, the routines that generate the element matrices, and the interfaces of NASTRAN with resident computer operating systems. Also, instructions are given for adding new program capabilities and new input cards to NASTRAN. Finally, attention is given to describing the utility programs available to help support and maintain NASTRAN.

DEMONSTRATION PROBLEM MANUAL

a say

2.6 DEMONSTRATION PROBLEM MANUAL

The Demonstration Problem Manual contains a discussion of each of the demonstration problems supplied with the NASTRAN program. The manual is subdivided into separate sections for each of the major analytic disciplines addressed by NASTRAN. Each section describes the problems to be solved, the models selected, the loads applied, and a discussion of the NASTRAN results with a comparison to the theoretical solution, where possible. Each Rigid Format analysis is demonstrated at least once, and practically all of the NASTRAN input options are exemplified. Listings of the Executive Control, Substructure Control, and Case Control decks for each problem are included. Additionally, there is at least one example of how to use each Bulk Data card type. The complete data decks for all of the demonstration problems are supplied with the NASTRAN system delivery package so that they may be executed by the user if he so desires.

Whenever a user contemplates exercising a new capability, he is encouraged to execute the demonstration problem exemplifying that feature.

3. MODELING WITH NASTRAN

3.1 INTRODUCTION

The basic methods and options for creating NASTRAN models are summarized in this section. The purpose of a model is to represent the properties of an actual physical system by a system of matric equations which provide accurate approximations to the expected system response. This mathematical model may be used to simulate many types of physical phenomena including static and dynamic response, stability, and heat transfer in structures as well as fluid dynamics, aerodynamics, and control system interactions with structures.

The basis for most NASTRAN modeling is the finite element. The finite element method assumes the physical system is subdivided geometrically into elementary components. In structural analysis, these components are represented by typical structural elements such as beams, plates, shear panels, and solids. Because each element mathematically represents only a simple response pattern, those sections of the physical system which are expected to undergo complex response patterns must be subdivided into many elements. When the required number of elements becomes large, user aids such as automated data generators (User's Guide, Section 9.4; User's Manual, Section 2.6) are helpful in preparing the NASTRAN data. Also, NASTRAN's automated multi-stage substructuring (AMSS) facilities (User's Guide, Section 20) help build and analyze these large and complex models.

From the data describing the finite element representation of the physical system, NASTRAN computes the finite element matrix coefficients defining the stiffness, mass, conduction, etc., properties of each element. These finite element matrices are combined together to form the overall system matrices which are used to represent the behavior of the basic model. Special features of the physical system may be defined and then added to complete the model using direct input matrix terms and general elements. Further processing is then undertaken to impose environmental constraints, loads, and/or perform matrix transofmrations in preparation for the solution phase (User's Guide, Section 4.0).

Once the physical system is defined by the finite element model, NASTRAN may be used to process that data for several different types of analyses. The basic finite element mathematical model used for structural static analysis may also be used as the basis for all types of dynamic analyses, heat transfer analyses, buckling, etc. The basic structural model matrices must be augmented with additional data appropriate to each specific type of analysis to define mass inertias, damping, radiation effects, control system input, and/or aerodynamic surface effects, etc.

3.1-1

MODELING WITH NASTRAN

In many cases, Guyan reduction techniques, modal formulations, and substructuring methods may be used to reduce the problem size to more manageable proportions. These topics are discussed later in Section 4.

Described below are basic considerations and general options for formulating a mathematical model in NASTRAN. The topics discussed start with building the basic finite element structural model and how to use substructuring techniques. Non-structural considerations such as for aerodynamics and fluid/structure interaction are then presented followed by available options for thermal heat transfer analysis. General capabilities of NASTRAN for research and development are then illustrated in the discussions of options for control systems and direct matrix input for special effects. The final section lists a number of the modeling restrictions imposed by NASTRAN. The methods of obtaining solutions for models defined by these data are summarized later in Section 4.

3.2 STRUCTURES

In NASTRAN structural analysis, finite elements are used to represent the stiffness, mass, and damping properties of the physical system. Although the elements are pure mathematical entities, they have been chosen to simulate frequently-occurring structural components such as beams, plates, shear panels, solids, etc. The more complex structural components, themselves, may be subdivided into sets of elementary finite elements to obtain sufficient detail to represent their desired physical properties and anticipated stress patterns.

Table 1 lists available NASTRAN elements, their characteristics, and their application. Some groups of elements are computationally similar and are listed together. For instance, IHEX1 represents three solid elements, IHEX1, IHEX2, and IHEX3, with different numbers of connected points. Individual descriptions of all these elements are given in the User's Manual, Section 1.3.

Four types of input data are used to define the finite elements: grid points, connection cards, property cards, and material data. The grid point data card (GRID) defines the location of the points connected by the elements and the orientation of the resulting displacement components defining the translation and rotation of these points. The connection data card (CBAR, CQUAD1, etc.) also points to data cards defining material and properties of the element. Material data (MAT1, MAT2, etc.) are used to define elastic, mass, and damping properties. Physical data define element cross-sectional area and other properties. Many elements may reference the same set of property data and many sets of property data may reference the same material data.
	Element Name Perties	ensions	Points	lar Points	cational DØF	isymmetric	ermal Luads	iffness	nped Mass	isist. Mass	scous	f. Stiffness	isticity	it Transfer	otted	operty Opt.	droelastic	oustic	roelastic	te: 1 = 1,2,
	iØ β∃Α	2	8																•	
	i ai xa	- m	ব			٠									٠			٠		
	9A 8	-	2		٠		•	٠	٠	•		٠	٠	٠	٠	٠				
	CONEAX	5	2		٠	•	•	٠	•						٠					
	i gmag	0	2	٠							•									
	I SA13	0	2	٠				•												
	FUIUJ	۳	ব			٠									٠		•			
	YO8 H	5	8	٠										٠						
	i AX3H	۳	8				٠	٠	٠					•	٠					
	IX3HI	- m	32				•	•	٠	٠		•			٠					
	ISSAM	0	2	•					٠											
	LWNOD	0			•				•											
	CONROD	-	2		•		•	٠	•	•		•	•		•					
	ODWEW	2	4				•	•	•	٠		•	•	•	•	•				
	11900	2	4		•		•	•	•	•					•					
	IGAUD	5	4		•		•	•	•	•		•	•	•	•	•				
	RØD	_	2		•		•	•	•	•		•	•	•	•	•				
	SHEAR	5	4					•	•						•					
	LIØIS	5	4			•												•		
`	TRTRA		4				•	٠	•					•	•					
	INKPRG	5	2			•	•	•		•					•					
	AAYAXI	m	4			•	•	•		•					•					
	TOPEC	e e e e e e e e e e e e e e e e e e e	4		-	•	•	•		•				•	•					
^	JCGAI	~	 		•	-	•	•	•	-					•					
		~	 С			•	-	•		•				-	•					
	TDIAN	~	~ ~		•	•	•	•	•	•				•	•	•				
]	TONEN		m		•			•	•	•		-		•	•	•				
ļ	т і аат	~	с, , , , , , , , , , , , , , , , , , ,		•		•	•	•	•		•	•	•	•	•				
	ואגרו	~	с. С		•		•	•	•	•		•	•	•	•	•				
		_	~		-		•	•	•			•	•	•	•					
	JSIN	-	~		•			•	•		•	•	•		-					
	MEDGE	e	9				•	•	•					•	-					
							_	•	•					•	•					

Table 1. NASTRAN Finite Element Characteristics

. 8

SUBSTRUCTURES

3.3 SUBSTRUCTURES

In cases when finite element models become too large for efficient NASTRAN processing, substructuring may be used as a method to subdivide the model into more convenient and/or manageable components. Each component substructure itself may be further subdivided into even more basic components. These basic components may be modeled independently and combined directly with other components at the matrix data level to build the final desired model for solution analysis. This procedure is also convenient for redesign and remodeling problems. Changes in one section of the structure would not necessarily require reformulation of the entire model.

The automated multi-level substructure system (AMSS), described in Section 20, provides a three-phase system for modeling with substructures. Each elementary or basic substructure component model is submitted in a Phase 1 run. The only restrictions on the model are that boundary grid points have compatable locations with adjoining components. Grid point and element identification numbers and responses are unrestricted. All required data for the models are automatically generated and permanently stored in files provided and maintained by the user.

In the second phase of substructuring, the substructures may be combined; they may be reduced in matrix size using the Guyan reduction or modal synthesis methods; and they may be combined again to form even more complex models. At any stage the user maintains direct control over each operation using simple linguistic commands (CØMBINE, REDUCE, SØLVE, etc.) All data is stored on the user provided substructure operating file at each stage providing each of restarting and recovering from data errors. Any substructure may be analyzed using standard NASTRAN Rigid Formats modified automatically for substructuring to provide for a solution and data recovery.

For detail data recovery at the level of the original basic substructure, the Phase 3 execution can be made from a restart tape or from the original model input data itself. Again, the user-provided substructure operating file is accessed by NASTRAN to automatically retrieve the required solution vectors needed for computing detail element stresses and preparing deformed plots, etc.

3.3-1

AERODYNAMICS

3.4 AERODYNAMICS

Aerodynamic elements are strips, boxes, or bodies which represent the surfaces for computation of aerodynamic interactions with the structure. These elements, like structural elements, are defined by their geometry. Their motions are defined by degrees of freedom at associated aerodynamic grid points. Often, requirements of aerodynamic theory will dictate the geometry of the boxes. For example, the doublet lattice theory requires trapezoidal boxes with their parallel edges aligned to the streamlines. Also, if one box lies in the wake of another, their edges must be on the same two streamlines. Aerodynamic elements and grid points are generated to reduce the labor for the user and to enforce the theoretical restrictions on the geometry.

Structural and aerodynamic degrees of freedom are connected to the structure by interpolation. This feature allows selection of grid points and elements for structures and for aerodynamics, each to be defined in a manner which is well suited to the particular problem. The structural model for a wing may consist of a one-, two-, or three-dimensional array of grid points. The aerodynamic theory may utilize a strip theory or lifting surface theory. For further details on the aeroelastic theory and modeling procedures, refer to Theoretical Manual, Section 17, to User's Manual, Section 1.11 and to User's Guide, Section 21.

.

3.5 FLUIDS

The hydroelastic capability of NASTRAN is designed primarily to solve for small motion dynamic response of models with combined structure and fluid effects. The options include both rigid and flexible container boundaries, free surface effects and compressibility. The fluid is described with axisymmetric finite elements. The structure, however, is described with conventional non-axisymmetric elements to form matching boundaries with the fluid. The analysis assumes small perturbations about static equilibrium. Second-order velocity terms are ignored. See Theoretical Manual, Section 16.1; and User's Manual, Section 1.7, for full explanation.

3.6 ACOUSTIC CAVITIES

An acoustic cavity analysis may be performed to obtain the stationary waves in the steadystate flow of a gas through an axisymmetric chamber with radial slots. The width and depth of the slots and the diameter of the center volume may vary along the axis of the chamber. The boundaries of the chamber are assumed to be rigid. A Fourier series expansion is used to analyze each series harmonic as a separate NASTRAN run. See Theoretical Manual, Section 16.2; and User's Manual, Section 1.9 for details.

HEAT TRANSFER

3.7 HEAT TRANSFER

Modeling procedures using finite elements for heat transfer analysis closely parallel standard structural modeling methods. Normal NASTRAN rod, plate, and solid elements may be used to model conduction and capacitance for heat transfer analysis in structures. At each grid point, temeprature is the generalized degree of freedom rather than displacement. Heat input rates on grid points, surfaces, and volumes are the generalized loads.

A set of specialized "boundary elements" (HBDY) are also provided for heat transfer modeling. These elements are used to define convection between structural surfaces and fluids or other structural surfaces. They are also used in the definition of radiation "view factor" terms and as receivers of thermal flux inputs. Other non-convective boundaries may be specified using conventional constraints (SPC or MPC).

For further details, refer to descriptions given in Section 22 of the User's Guide, in Section 8 of the Theoretical Manual, and Section 1.8 of the User's Manual.

3.8 OTHER MODELING OPTIONS

In addition to the conventional finite elements, NASTRAN offers several alternative procedures for modeling special effects. Non-elastic structures and mechanisms may be modeled using multipoint constraint equations or rigid elements. Portions of structures may be modeled with general elements (GENEL) defined by experiment or by analysis using other programs. Direct matrix inputs (DMI, DMIG) may be used to model specific phenomena using real or complex stiffness, mass, or damping matrices.

^ccalar points, extra points, scalar elements, and transfer functions are available for modeling non-structural phenomena and defining non-structural degrees of freedom. Use of scalar quantities in NASTRAN is illustrated in the NASTRAN Demonstration Problem Manual and in the User's Guide, Sections 18 and 19. Used in conjunction with solution techniques described in Section 4, a wide variety of advanced problems may be analyzed.

3.9 MODELING RESTRICTIONS

Aside from theoretical restrictions due to finite element formulation, very few general modeling restrictions are imposed by NASTRAN. The most frequently encountered restrictions are limits on the model size which occur due to cost and computer resource restrictions. These factors are discussed in Section 14. Some of the more general modeling restrictions:

a. Element Connection Order

Most element geometries require that the connected grid points be defined in a specific order. This order is used to define the element coordinate system which, in turn, is used to define the orientation for the input of loads and for the output of stresses.

b. Element Geometry

The accuracy of some elements such as membranes and plates may be seriously effected by their geometry. Large aspect ratios and severely warped surfaces should be avoided. See Theoretical Manual, Section 15.2.

c. Engineering Units

No unit conversions are built into MASTRAN. All models must be specified using a consistent set of units for force, mass, displacement, and time. A common user error is the use of force units (pounds) for defining mass. In the metric system a common mistake is the use of mass units (kilograms) for the definition of forces or elastic constraints.

d. Numerical Roundoff

The potential errors caused by numerical roundoff require some modeling considerations. For instance, elements having stiffness terms many orders of magnitude larger than those for adjacent elements may cause numerical problems. The use of rigid elements or multipoint constrains in these cases is recommended. See Theoretical Manual, Section 15.1.

e. Uniqueness of Element ID's

Element identification numbers must be unique within each type. Although different element types may use the same identifications, it is recommended that all elements use unique IDs for the purpose of specifying loads and selecting output print and plot sets.

f. Axisymmetric Problems

Axisymmetric formulations use Forier series expansions (CONEAX, TRAPAX, TRIAAX elements) are thus limited in their scope. Because of this they are incompatable with many of the general three-dimensional capabilities of other NASTRAN elements. See User's Manual, Section 1.3 for details.

4. ENGINEERING ANALYSES WITH NASTRAN

an the state of the state of the state

4.1 INTRODUCTION

The NASTRAN system provides capabilities briefly described here for many types of engineering analyses which can be performed with a particular finite element model. NASTRAN performs these analyses using the Direct Matrix Abstraction Program (DMAP) that is an integral part of the system to define and control the required sequence of matrix generation, assembly, and processing operations.

The NASTRAN user may write his own solution sequence using DMAP, or as is most often the case, he may use one or more of the twenty DMAP sequences which are stored in NASTRAN. The, standard DMAP solution sequences are called Rigid Formats. They relieve the user of having to know details and/or how to control each step of an analysis. Execution of a structural problem may proceed to final solution all in one run or, at the option of the user, the DMAP sequence may be terminated at any desired intermediate point. Restart capabilities are provided forboth scheduled and unscheduled interruptions in processing.

DMAP instructions may also be used to modify the existing Rigid Formats in order to change the processing sequence, to obtain printouts of selected data blocks, to provide for additional input which was generated by other programs, to output results to be used as input to other programs, and to construct entirely independent sequences or DMAP instructions to solve nearly any problem that can be formatted in matrix notation.

The different engineering analysis approaches available with NASTRAN Rigi Formats are described here. Also included are relationships between different structural and non-structural modeling options and Rigid Format applications.

Listings of each Rigid Format are presented in Section 3 of the User's Manual with descriptions and equations for all the key steps. The rules and formats for using DMAP statements are presented in Section 15 of the Guide and Section 5 of the User's Manual.

Rigid Formats are identified with an integer number, i.e., 1, 2, etc., and an analysis approach. There are three approaches: displacement (APProach DISP), heat transfer (APProach HEAT), and aerodynamic (APProach AERD). The user selects the Rigid Format and approach by using commands in the NASTRAN Executive Control Deck described in Section 6.

LINEAR STATIC ANALYSIS (RIGID FORMATS 1, 2, and 14, APP DISP)

a a chinh tibl an a sa sa sa

4.2 LINEAR STATIC ANALYSIS (RIGID FORMATS 1, 2, and 14, APP DISP)

٦

The Linear Static Analyses provided by Rigid Fromats 1, 2 and 14 use most of the features available for model definition in NASTRAN. These include specification of special coordinate systems, grid point locations, element connectivities, material properties, constraints, and temperature distributions as well as direct grid point force and element loadings. The results available for output consist of weight and balance information, displacements, element forces and stresses, grid point loadings, and support reactions. The user may also request plots of the undeformed as well as the deformed structural model.

Rigid Format 2, which provides all the features of Rigid Format 1, also treats structural models which are not fully constrained. It computes the inertia effects of unconstrained rigid body accelerations. These effects are added to the user-specified loadings; the rigid body motions are constrained; and then the normal static analysis is performed.

Rigid Format 14 is a special case of Rigid Format 1 and provides for efficient modeling and analysis of structures exhibiting cyclic symmetry. Many structures, including pressure vessels, rotating machines and antennae for space communications, are made up of virtually identical segments that are symmetrically arranged with respect to a centroidal axis. There are two types of cyclic symmetry: simple rotational symmetry, in which the segments do not have planes of reflective symmetry and the boundaries between segments may be general doubly-curved surfaces; and dihedral symmetry, in which each segment has a plane of reflective symmetry and the boundaries between segments are planar. The use of cyclic symmetry allows the user to model only one of the identical substructures. There will also be a large saving of computer time for most problems.

Precedures for using the static analysis Rigid Formats are outlined in Section 18. The procedure for using cyclic symmetry in NASTRAN is described in detail in Section 23 of the Guide. The theoretical treatment for cyclic symmetry is given in Section 4.5 of the Theoretical Manual.

NONLINEAR STATIC ANALYSIS AND BUCKLING (RIGID FORMATS 4, 5 and 6, APP DISP) 4.3 NONLINEAR STATIC ANALYSIS AND BUCKLING (RIGID FORMATS 4, 5 AND 6, APP DISP)

NASTRAN provides for two types of static nonlinearities, differential stiffness and material plasticity. The differential stiffness effects treated by Rigid Formats 4 and 5 provide the user with a second order approximation to the nonlinear effects of large deflections. The differential stiffness matrix is computed as a correction to the original stiffness matrix from the work done by each element in response to a user-specified preload. Rigid Format 4 adjusts the differential stiffness matrix as a load correction in accordance with user specified convergence criteria. In Rigid Format 4, the applied loads are not linearly related to the internal loads. In Rigid Format 5, the applied loads are assumed to move with their points of application and they remain fixed in magnitude and direction.

Rigid Format 5 adds the differential stiffness matrix to the original stiffness matrix, and solves an eigenvalue problem to obtain the critical load factor that would produce buckling. This critical load factor, or the eigenvalue, multiplied into the preload vector gives the critical loading at which the structure would go unstable. The eigenvector obtained is the mode shape in which the structure would buckle.

Rigid Format 6, on the other hand, is used for problems involving non-linear materials. That is, the material properties are assumed to be stress dependent. The loading is applied in piecewise linear increments. After each increment, a new stiffness matrix is generated based on the current state of stress in each element. This new matrix is used to solve for a corresponding increment of displacement and stress. These increments are accumulated to produce the final nonlinear results. Since each increment in load requires a new solution, the user is faced with the compromise between accuracy, which requires small increments, and efficiency, which suggests using the largest possible increments. All static load options are allowed except for those due to temperature and enforced element displacements.

Section 18 of the Guide includes descriptive detail on how to use each of these three Rigid Formats. The theoretical treatments for material non-linearities and differential stiffness can be found in the Theoretical Manual, Sections 3.8 and 7.0, respectively.

4.3-1

NORMAL MODES ANALYSIS (RIGID FORMATS 3, 13, and 15, APP DISP)

4.4 NORMAL MODES ANALYSIS (RIGID FORMATS 3, 13, and 15, APP DISP)

Rigid Formats 3, 13, and 15 provide for a normal mode analysis of undamped systems with symmetric matrices. Rigid Format 3 is for use with elastic, linear structural models. Rigid Format 13 allows the use of differential stiffness effects. The applied loads are assumed to move with their points of application and they remain fixed in magnitude and direction. Rigid Format 13 adds the differential stiffness matrix to the original stiffness matrix in proportion to a user-specified increment for the deflections due to the preload plus incremental load which is the total load applied to the model. Rigid Format 15 computes normal modes for models utilizing cyclic symmetry. Additionally, normal modes calculations are included in the Rigid Formats for Complex Eigenvalue Transient Response and Modal Flutter analyses summarized in the next Section.

Rigid Formats 3, 13, and 15 generate both mass and stiffness matrices and offer a choice of methods for eigenvalue extraction. If only a limited number of mode shapes are required within a user specified range of frequencies, then the FEER, Inverse Power or Determinant method would be the appropriate choices. The Givens method of tridiagonalization is available and is recommended if all or a large proportion of the natural frequencies and mode shapes are desired. A Guyan reduction, which is available in all Rigid Formats for eliminating specified degrees of freedom, is recommended for reducing the size of the matrices to be used by the Givens method. (Massless degrees of freedom must be eliminated for this method.)

The available output of results includes the eigenvalues and natural frequencies, mode shape deflections, modal forces and stresses for selected elements, modal reaction forces, modal mass, and plots of the structural model for each mode shape.

The detailed theoretical basis for these calculations is presented in the Theoretical Manual, Section 10. Section 19 of the Guide describes how to apply these three normal modes analysis Rigid Formats.

4.4-1

COMPLEX EIGENVALUE ANALYSIS (RIGID FORMATS 7 and 10, APP DISP)

4.5 COMPLEX EIGENVALUE ANALYSIS (RIGID FORMATS 7 AND 10, APP DISP)

Rigid Formats 7 and 10 offer the user a capability to compute complex eigenvalues and vectors of non-conservative systems. These Rigid Formats provide for an analysis of systems which include user-specified input of real or complex mass, damping and stiffness matrices or matrix transfer function equations. The assembled matrices may be real or complex, symmetric or nonsymmetric, singular or nonsingular.

The two formulations provided are the direct formulation of Rigid Format 7 and the modal formulation of Rigid Format 10. The direct formulation provides for the most general description of a model which also includes program generated viscous and structural damping. The unknowns are the degrees of freedom related to grid points, scalar points, and any special extra points which may be used to specify non-structural quantities as are encountered in control system, aerodynamic, or hydrodynamic modeling.

The modal approach, in contrast, employs the free vibration modes of the undamped model as independent degrees of freedom to be assembled with the user-specified direct input matrices and extra point control parameters. Proportional viscous damping on the modal coordinates is included.

The selection of approach would depend on the particular problem to be solved and the number of roots to be extracted. The direct approach offers the greatest flexibility if only a few roots are desired. If many roots are to be extracted, the modal approach may be more economical. A summary of features comparing direct versus modal approaches for the dynamics Rigid Formats is presented in Table 1 to help the user select the Rigid Format best suited to his needs.

The available output includes frequencies and mode shape displacements, element forces and stresses, and support reactions. These may be printed in either SØRT1 or SØRT2 in either real/imaginary or magnitude/phase angle format. SØRT1 prints all the requested output for each frequency. SØRT2 prints each component of output for all frequencies. The theoretical basis for these Rigid Formats is presented in the Theoretical Manual, Section 10, and their use is presented in Section 19 of the Guide.

4.5-1

DYNAMIC RESPONSE ANALYSIS (RIGID FORMATS 8, 9, 11 and 12, APP DISP) 4.6 DYNAMIC RESPONSE ANALYSIS (RIGID FORMATS 8, 9, 11 and 12, APP DISP)

and the second second second with the second s

and the

The NASTRAN dynamic analysis Rigid Formats provide a maximum amount of flexibility and capability. Rigid Formats 8 and 11 offer Frequency and Random Response capability and Rigid Formats 9 and 12 offer Transient Analysis capability. Rigid Formats 8 and 9 use a direct approach; the other two use a modal approach. Table 1 summarizes the various options available to help the user select the Rigid Format best suited to his particular problem.

These capabilities can be used to solve problems with thousands of degrees of freedom with many time steps, or many frequencies and mode shapes. Provision has been made to allow many types of special modeling techniques. Care should be taken, however, to avoid the dynamic analysis of models that may be unnecessarily large because of the lengthy (and expensive) computing times required for solution. Every effort should be made to reduce the size of the problem either by careful design of the original model or by using the Guyan method to reduce the total degrees of freedom. The user should pay special attention to the selection of the most economical method for solving his particular problem.

The modal approach for dynamic analyses differs from the direct approach in that it uses the undamped mode shapes as independent degrees of freedom replacing the original model. In either approach, special extra points may be added for special modeling purposes. The input loads for the modal approach are transformed to the modal coordinates, and the resulting modal displacements are transformed back to the physical displacements for output purposes.

The choice of direct or modal approach is dependent on the specific problem. The direct approach solves a more general problem and does not require the extra steps of extracting the normal modes and of performing the transformations to form modal coordinates. For transient analysis, however, if many time steps are required, the direct formulation may be inefficient compared with the modal approach.

Also available with the modal approach (Rigid Formats 11 and 12) is a mode acceleration method of improving the solution accuracy. This procedure replaces the unconstrained rigid body accelerations with equivalent inertial and damping forces. The augmented loading is then solved statically to obtain improved element forces and stresses. The user is cautioned, however, that this method is costly and should be used selectively.

4.6-1

ENGINEERING ANALYSES WITH NASTRAN

The Frequency and Random Response analysis of Rigid Formats 8 and 11 can be used to analyze the steady state response of systems due to sinusoidally varying loads at specified frequencies. The model may include direct input matrices and transfer functions. The assembled matrices may be non-symmetric, real or complex, and singular or nonsingular. Loads must be input on specific degrees of freedom and may have phase angles, time lags, and frequency dependent magnitudes. The output may be sorted by frequency (SØRT1) or by component for all frequencies (SØRT2) and printed in either a real/imaginary or a magnitude/phase angle format. Curve plotting of these results is also available for each output component or a function of frequency.

The Random Response analysis is treated as a data reduction procedure that is applied to the results of a Frequency Response analysis. The user inputs a power spectral density function, which is assumed to be stationary with respect to time, and the program computes the power spectral density, mean deviation and autocorrelation function for the response of the selected output components. The output from the Random Analysis must be selected one component at a time, not via the Case Control options, but via the control cards of the X-Y plot package. The random output will be real numbers automatically printed and may be plotted versus Frequency if so requested.

The Transient Analyses (Rigid Formats 9 and 12) perform a step-by-step integration of the structural response in a time domain. The user provides the initial conditions, time step requirements, and a time varying load. The available output for specified integration time intervals includes displacements, velocities, accelerations, forces, and stresses for selected elements and support reactions.

Some special features available to Transient Analysis include the capacity to incorporate user-specified direct input matrices, nonsymmetric control function equations, and a limited nonlinear analysis capability. The nonlinear capability is available through displacement and velocity dependent load components applied to specific degrees of freedom. For purposes of economy, the user may change the integration time step. However, for each time step change, the coefficient matrices must be reassembled.

The theoretical basis for these analyses is presented in the Theoretical Manual, Section 9 through 12 and their use is described in Section 19 of the Guide.

4.6-2

LINEAR STATIC HEAT TRANSFER ANALYSIS (RIGID FORMAT 1, APP HEAT)

4.7 LINEAR STATIC HEAT TRANSFER ANALYSIS (RIGID FORMAT 1, APP HEAT)

The linear heat transfer formulation is analogous to the structural formulation with temperatures replacing deflections as the unknowns. Rigid Format 1 is available for determining steadystate temperatures with no radiation effects. The same modeling features for specifying grid point geometry, element connectivity and material properties are provided. In addition, special features are available for the specification of fixed temperature, convection, and heat flux boundary conditions. The available output consists of resultant grid point temperatures throughout the model. This capability is described in detail in Section 22.

NONLINEAR STATIC HEAT TRANSFER ANALYSIS (RIGID FORMAT 3, APP HEAT)

4.8 NONLINEAR STATIC HEAT TRANSFER ANALYSIS (RIGID FORMAT 3, APP HEAT)

1.5

مالية المالية. مالية المالية

tine.

Rigid Format 3, APProach HEAT, which provides all the features of Rigid Format 1, also treats heat transfer analyses with temperature dependent conductivities of the elements and nonlinear radiation exchange. The solution of thermal equilibrium is obtained using an iterative technique that is user controlled. Section 22 describes in detail how to apply this capability.

TRANSIENT HEAT TRANSFER ANALYSIS (RIGID FORMAT 9, APP HEAT)

4.9 TRANSIENT HEAT TRANSFER ANALYSIS (RIGID FORMAT 9, APP HEAT)

Rigid Format 9, APProach HEAT, provides the capability of time domain analysis for models that include linear conduction, film heat transfer, nonlinear radiation, and any special effects modeled with NASTRAN nonlinear elements. User controls for integration procedures and options for linearizing radiation effects are provided. More detail on how to apply this capability is presented in Section 22.

AERODYNAMIC FLUTTER ANALYSIS (RIGID FORMAT 10, APP AERØ)

4.10 AERODYNAMIC FLUTTER ANALYSIS (RIGID FORMAT 10, APP AERØ)

Rigid Format 10, APProach AERØ, provides the capability for modal flutter analysis for stability studies of aeroelastic systems with a minimum number of degrees of freedom. This Rigid Format includes structural mode calculations, aerodynamic matrix calculations, and transformation of the aerodynamic matrix to modal coordinates. The flutter stability analysis is performed using one of three available user selected methods: the K-method, the KE-method, and the PK-method. Solutions may be obtained for subsonic and supersonic flow conditions. How to access this capability is described in Section 21.

AERODYNAMIC RESPONSE ANALYSIS (RIGID FORMAT 11, APP AERØ)

4.11 AERODYNAMIC RESPONSE ANALYSIS (RIGID FORMAT 11, APP AERØ)

and the second

Rigid Format 11, APProach AERØ, provides the capability to compute responses of an aeroelastic system due to applied loads and gusts. This Rigid Format may be used to solve frequency response, random response, and transient response problems. Transient analyses are performed using a Fourier transform method. The structural model is reduced to normal mode format prior to coupling with the aerodynamic matrix. This Rigid Format may be used for both subsonic and supersonic flow conditions. Details are presented to access this Rigid Format in Section 21.

CORRELATION OF ANALYSIS VERSUS MODEL CAPABILITIES

100.00

4.12 CORRELATION OF ANALYSIS VERSUS MODEL CAPABILITIES

A summary of NASTRAN modeling options versus Rigid Format application is presented in Table 2. Modeling options available in NASTRAN were introduced earlier in Section 3; the Rigid Formats are introduced here in Section 4, and the relationships between the two are summarized in the table.

ENGINEERING ANALYSES WITH NASTRAN

Table 1. Summary of Analysis Options for Dynamic Rigid Formats

		RIGID FORMAT NUMBER, APP DISP													
OPTIONS		DIRECT		MODAL											
OPTIONS Proportional Structural Damping Proportional Viscous Damping Damping Damping Damping Elements Real Direct Input Marrices Or Transfer Functions Complex Direct Input Matrices Non-linear Functions Initial Conditions Loads	Complex Eigenvalue 7	Frequency Response 8	Transient 9	Complex Eigenvalue 10	Frequency Response 11	Transient 12									
Proportional Structural Damping	1	1													
Proportional Viscous Damping			۱	2	2	2									
Damping Elements	1	1	۱												
Real Direct Input Marrices or Transfer Functions	3	3	3	3	3	3									
Complex Direct Input Matrices	3	3		3	3										
Non-linear Functions			3			4									
Initial Conditions			3			4									
Loads		3	3		3	3									

Optional input constraints:

- Applied to structure only
 Applied to modal coordinates only
 Applied to structure or extra points
 Applied to extra points only

4.12-2

CORRELATION OF ANALYSIS VERSUS MODEL CAPABILITIES

100 C

www.statingto

Éλ

MODELING		DISP												ASPO						
OPTIONS	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	"ו	3	9	10	11
STRUCTURES/NONSTRUCTURES																				
1,2,3D Structures	•	٠	•	•	•	•	•	•	•	•	•	•	•	•	•		•	•		•
Substructures	•	٠	•					•	•						-		-	-		-
Cyclic Structures														•	•					
Axisymmetric Structures	•	•	•			•	•	•	•	•		•			-		•	•		
Aerodynamic Modeling																	-	-	•	•
Acoustic Modeling			•					•	•	•	•	•								•
Hydroelastic Modeling							•	•	•											
NON-LINEARITIES																				
Large Deflection Modeling				•	•								•							
Non-linear Materials						•														
HEAT TRANSFER																	•			1
Conduction & Convection																	•	•		
Radiation																	•	•		
LCADS																	•	•		
Static Loads	•	•		•	•	•		•	•		•	•	•	•			•	•		•
Frequency Dependent Loads								•			•	-	-	-			•	•		
Time Dependent Loads									•			•								
Non-Linear Loads									•											•
CONSTRAINTS									-			•						•		·
Guyan Reduction	•	•	•		•	•	•	•	•	•	•	•	•	•	•	•		•	•	•
Multi-Point Constraints	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•		•			
Single Point Constraints	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•		•	•		
Element Deformation	•	•		•	•	•							•	•	-	-	-		-	-
Rigid Elements	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•		•	•	•	
Free Body Supports		•	•		-	-	_	-	-	•	•	•	-	-	-	•	-	•	•	•

Table 2. Summary of Modeling Options Versus Rigid Format

į

5. INTRODUCTION TO THE NASTRAN SYSTEM

5.1 PROGRAM DESIGN CRITERIA

The NASTRAN program has been designed according to two classes of criteria. The first class related to functional requirements for the solution of an extremely wide range of large and complex problems in structural analysis with high accuracy and computational efficiency. These criteria are achieved by developing and incorporating the most advanced mathematical models and computational algorithms that have been proven in practice. In particular, they are achieved by providing such features as the bandwidth-with-active-column technique in matrix decomposition; packing routines to take maximum advantage of matrix aparsity so as to conserve input/output time; highly stable and efficient algorithms for the solution of problems in eigenvalue analysis and transient response; and an elegant approach to modeling the effects of control systems and other nonstructural components.

The second class of criteria relates to the operational and organizational aspects of the program. These aspects are somewhat divorced from structural analysis itselt; yet they are of equal importance in determining the usefulness and quality of the program. Chief among these criteria are:

- 1. Simplicity of program input deck preparation.
- 2. Minimization of chances for human error in problem preparation.
- 3. Minimization of need for manual intervention during program execution.
- 4. Ease of program modification and extension to new functional capability.
- 5. Ease of program extension to new computer configurations and operation systems, and generality in ability to operate efficiently under a wide set of configuration capabilities.
- 6. Capability for step-by-step problem solution, without penalty of repeated problem set up.
- 7. Capability for problem restart following unplanned interruptions or problem preparation error.
- 8. Minimization of system overhead, in the three vital areas:
 - a. Diversion of core storage from functional use in problem solution.
 - b. Diversion of auxiliary storage units from functional to system usage.
 - c. System housekeeping time for performing executive functions that do not directly further problem solution.

INTRODUCTION TO THE NASTRAN SYSTEM

To achieve a program design to meet these criteria, NASTRAN was developed as a modular system with its own executive control capability. Specific attention was given to opening the program to offer maximum user control over its operations. These design objectives are described next, followed by an overview of the computer system requirements imposed by the program and a preview of how the NASTRAN data deck is organized.

5.1.1 Modular Design and Executive Control

S,

Ļ

The criteria stated above have been achieved in NASTRAN by using modular separation of functional capabilities in designing the software system. Each module is independent of all other modules in the sense that modification of a module or addition of a new module will not in general require modification of other modules.

Furthermore, control of module executions is performed by an efficient, problem-independent Executive System. The concepts of functional modules and an Executive System providing control are illustrated by the software constraints specified for module development and operation.

The restrictions on modules are as follows:

- 1. Modules may interface with other modules only through auxiliary storage files that contain data blocks.
- 2. Since the availability of the auxiliary files required for the execution of a module depends on the execution of other modules, no module can specify or allocate files for its input or output data. All auxiliary sturage allocation is reserved as an executive function.
- 3. Modules operate as independent subprograms and may not call or be called by other modules. They may be entered only from the executive routine.
- 4. Modules may interface with the executive routine through a parameter table that is maintained by the executive routine. User-specified options and parameters are communicated to modules in this way. The major line of communication is one-way, from user to executive routine to module. However, in addition, an appreciable two-way communication from module back to executive routine (and, therefore, to other modules) is permitted via the parameter table.

PROGRAM DESIGN CRITERIA

No other constraints, except those imposed by the resident compilers and operating systems, are required for functional modules.

The essential functions of the executive system are:

- To establish and control the sequence of module executions according to options specified by the user.
- 2. To establish and communicate values of parameters for each module.
- To allocate files for all data blocks generated during program execution and perform input/output to auxiliary files for each module.
- To maintain a full restart capability for restoring a program execution after either a scheduled or unscheduled interruption.

Each of these functions is essentially independent of any particular feature of structural analysis and applies to the operational control of any complex multimodule, multifile application program. The Executive System is open-ended in the sense that it can accommodate an essentially unlimited number of functional modules, files, and parameters. Modification of the executive system necessary for modification or extension of functional modules is restricted to changes in entries in control tables stored within the executive routine.

5.1.2 Design for User Control

The needs of the structural analyst have also been considered in all aspects of the design of the program. The first thing to be remembered is that, in view of the wide range of possible applications of the program, it is not known exactly what these needs may be. For this reason a high degree of flexibility and generality has been incorporated into certain areas of the program. For example, in addition to the usual list of structural elements that refer to specific types of construction, the user is provided with more general elements that may be used to construct any type of special element, to represent part of a structure by deflection influence coefficients, or to represent part of a structure by its vibration modes. For the more conventional types of structural analysis, the user is presented with a large number of convenience features, including plotting routines, which are definite necessities for analyzing large problems.

A major difficulty that a user faces in the solution of large problems is the avoidance of errors in the preparation of input data. Card formats and card ordering are made as simple and

INTRODUCTION TO THE NASTRAN SYSTEM

flexible as possible in NASTRAN in order to avoid errors caused by trivial violations of format rules. A number of aids for the detection of legal but incorrect data are also provided.

The problems that can be solved by NASTRAN include the following general classes:

- 1. Static Structural Problems
- 2. Elastic Stability Problems
- 3. Dynamic Structural Problems
- 4. General Matrix Problems

- 5. Heat Transfer Problems
- 6. Aeroelasticity Problems

Each general problem class is further subdivided into case types which differ with regard to the type of information desired, the environmental factors considered, or the method of analysis. The mathematical computations required to solve problems are performed by the subprogram units called functional modules. Each case type requires a distinct sequence of functional module calls that are scheduled by the Executive System.

General methods of problem solution for all classes listed above except general matrix problems are established internally to NASTRAN. This is accomplished with coded sequences of macro instructions, called DMAP (for Direct Matrix Abstraction Program) commands, which are stored in the Executive System. Each particular sequence is called a Rigid Format and is requested by the user for the particular analysis to be performed (statics, elastic stability, dynamics, etc.).

Each Rigid Format consists of two parts. The first is a sequence of instructions (including instructions for executive operations as well as for functional module operations) that is stored in tables maintained by the executive system. The second part is a set of restart tables that automatically modify the sequence of instructions to account for any changes in the input data when a restart is made after partial or complete execution of a problem. The restart tables can accommodate a change of Rigid Format such as occurs, for example, when vibration modes are requested for a structure that was previously analyzed statically. The restart tables are, as can be imagined, quite extensive, and their generation constitutes a significant part of the effort expended in developing a Rigid Format. They are, however, one of the more important cost-saving features of NASTRAN.

It is also possible for the user to modify a Rigid Format via an ALTER feature, which is described in Section 2.2 of the User's Manual. Typical uses of the ALTER feature are to schedule

PROGRAM DESIGN CRITERIA

an exit at an intermediate point in a solution for the purpose of checking intermediate output, to schedule the printing of a table or a matrix for diagnostic purposes, and to add or delete a functional module from the sequence of operating instructions.

A flexible procedure is provided for the solution of general matrix problems. All of the matrix operations (such as addition, multiplication, triangular decomposition, and eigenvalue extraction) used in the program can be directly addressed by the user with DMAP. The user constructs a chain of DMAP instructions in order to effect the solution of general matrix problems.

The rules for generating a DMAP sequence are explained in Section 5 of the User's Manual. The DMAP sequence itself consists of a series of statements consisting of executive operation instructions and functional module calls. Each statement contains the name of the instruction (or functional module), the names of the input data blocks, the names of the output data blocks, and the names and values of parameters. Typical examples of parameter usage are to indicate whether an operation is to be performed with single- or double-precision arithmetic, which mathematical method will be used (when there are options), or the desired format of the output.

The names of some of the executive operations are BEGIN, CHKPNT (used when it is desired to copy data blocks onto the Problem Tape in case an unscheduled restart is necessary), FILE (used to save an intermediate data block), REPT (used to provide looping capability), PURGE (used to prevent storage of data blocks), and END.

The functional modules belong to one of the following categories: structural modules, matrix operations, utility modules, and user modules. The structural modules are the main subprograms of NASTRAN. Some examples of structural modules, taken from dynamic analysis, are: READ (Real Eigenvalue Analysis), GKAM (Modal Dynamic Matrix Assembler), TRD (Transient Response, Dynamic), and DDR (Dynamic Data Recovery). Typical matrix operations provided are: add, multiply, transpose, etc. The utility modules mainly provide capability for formatting output data. The user modules are dummy modules that provide the user with the ability to write new functional capability that will automatically be recognized by the Executive System.

The usual methods of output for NASTRAN are the operating system print or punch files and the NASTRAN plot tapes. Procedures for normal output selection are described in Section 2.3 of the User's Manual. The printing of tables or matrices generated by NASTRAN is controlled by a group of utility modules described in Section 5.3 of the User's Manual. In many cases, it is desirable to save matrices and tables for use in restart operations. When using Rigid Formacs,

INTRODUCTION TO THE NASTRAN SYSTEM

it is possible to save preselected tables and matrices by using the checkpoint option described in Section 2.2 of the User's Manual. Checkpointed files are written on the New Problem Tape. It is also possible for the user to save selected matrices on tape by inserting one of the user modules described in Section 5.3 of the User's Manual into the DMAP sequence by means of the ALTER option.

The usual method of input for NASTRAN is the operating system card reader. When performing restarts, the New Problem Tape from a previous run is redesignated as the Old Problem Tape and used as an additional source of input. Tapes that have been prepared with user modules on previous runs can also be used as additional input sources by inserting one of the input user modules into the DMAP sequence by means of the ALTER option.

PROGRAM SYSTEM REQUIREMENTS

5.2 PROGRAM SYSTEM REQUIREMENTS

NASTRAN is a very large, FØRTRAN computer program. It consists of more than 1600 subroutines and over 200 individual overlay segments. The program is designed for batch mode operation on each of three computer configurations: IBM System 360/370 series, CDC 6000/CYBER series, and UNIVAC 1108/1110 series computers.

Two key concepts in the structure of NASTRAN are the open core and file concepts. All data used or created in NASTRAN calculations are stored in an area of computer memory which begins where program instructions end and terminates where the user has specified the limit of memory resources for the particular job. This area of memory is called "open core." Data are only temporarily stored in open core. While the data are not actually needed for computations, they are stored on either input, output, or scratch files assigned to the particular module. Thus, the file is the primary source of data storage, and FCRTRAN file requirements are a major system resource requirement of NASTRAN. Typical NASTRAN runs require as many as 50 FORTRAN files for execution.

The open core concept is described in detail with an example in Section 5.2.4. NASTRAN file usage is described in Section 5.2.5. Additional information about NASTRAN and its characteristics peculiar to installation on different computers is presented in Sections 5.2.1 through 5.2.3.

5.2.1 NASTRAN Computers

Although NASTRAN is a FØRTRAN-oriented system, considerable effort was required to develop programming and word-handling techniques applicable to three separate computer configurations. These computers exhibit wide differences in their binary word sizes and integer representation methods. The current computer configurations for NASTRAN and their significant differences follow:

1. IBM System 360/370 Series

Word Size - 32 Bits

Character Capacity - 8 bits/character and 4 characters/word (character ≡ byte) Integer Representation - Twos complement for negative integers

2. UNIVAC 1108/1110

Word Size - 36 Bils

INTRODUCTION TO THE NASTRAN SYSTEM

Character Capacity - 6 bits/character and 6 characters/word Integer Representation - Ones complement for negative integers

3. CDC 6000/CYBER

Word Size - 60 Bits

Character Capacity - 6 bits/character and 10 characters/word Integer Representation - Ones complement for negative integers

5.2.2 Alphanumeric Data

Data stored within a computer as binary-coded-decimal (BCD) characters must be represented by the proper hardware-defined bit codes. These character codes (and in the case of the IBM System/360, the number of bits representing the code) vary among the NASTRAN computer types. Although the number of characters-per-word could have been obtained from the SYSTEM table, various data blocks and buffers within NASTRAN required firm entry sizes, regardless of computer type, to facilitate indexing. For these reasons, the minimum number of characters-per-word (4) among the three computer types was chosen as a program design standard. Computer types with a word capacity of greater than four characters will have the unused low-order character positions filled with BCD blanks.

5.2.3 Word Packing

Standard FØRTRAN compilers do not provide the capability for storing or retrieving data that occupies less than a full computer word. Through the Machine Word Functions (MAPFNS) routine, some limited word packing (not to be confused with matrix packing) is performed within the Executive System and a few utility subroutines. Packing provides an efficient use of memory space at the expense of the additional operating time needed to combine or separate the elements of the packed words. The Machine Word Function ØRF is generally used for combining elements, while ANDF with a suitable mask is used for separating them.

5.2.4 The Open Core Concept

The design philosophy of the NASTRAN system dictated a completely open-ended design whenever possible. NASTRAN was to have the flexibility to operate on a second-generation machine with a 32K core (such as the now unsupported IBM 7094/7040 DCS) as well as the largest of the IBM S/360 series of computers, and take complete advantage of the additional core storage without major

PROGRAM SYSTEM REQUIREMENTS

program changes. The use of a fixed dimension for large arrays was rejected, since this automatically restricted the size of a problem that could be solved. Instead, modules were programmed to allocate space as required and to use spill logic to transfer data to scratch files when complete core allocation is impossible. In this manner, a problem might cause spill logic to be used on a computer with limited core storage, but not on a computer with a large core storage capacity.

5.2.4.1 Definition of Open Core

The definition of open core in NASTRAN is: a contiguous block of working storage defined by a labeled common block whose length is a variable determined by the NASTRAN executive function, CØRSZ. The implementation of this definition by the module writer consists of the origining of a labeled common block at the end of his overlay segment. This labeled common block contains a dimensioned variable of length 1. CØRSZ returns the number of words of core available between his open core origin and the end of core. The module writer can now write his program as if he had dimensioned his array by that number. In actuality, he is extending beyond the area reserved for the array into an area reserved for the job but not currently used by the segment. When implementing this concept, care must be taken to assure that the system does not use this area.

5.2.4.2 Example of an Application of Open Core

Figure 1 demonstrates the use of open core by two subroutines, A and B. By some means, which are machine dependent and are discussed in Section 5, an end point is established for open core. The length of open core is then the difference between this end point and the labeled common block. In the example shown, subroutine A will have more open core available to it than B does.

5.2.5 NASTRAN Input and Output Files

The particular (IBM S/360, UNIVAC 1100, CDC 6000) operating system input and output files provide the required data connection between NASTRAN, the input data decks, and the printed output. Utility subroutines XRCARD and RCARD convert special NASTRAN input card formats to standard FDRTRAN data words easily handled by all NASTRAN input processors. Printed output is generated through FDRTRAN-formatted write statements. All internal data block input/output is handled by GIND, the system of NASTRAN general purpose input/output routines. GIND provides the required manipulation

INTRODUCTION TO THE WASTRAN SYSTEM

to tailor the variable-length logical data records needed by most NASTRAN modules to fixed-length records available on all direct access mass storage hardware.

5.2.5.1 Use of the Operating System Input File

and the second second

The system input file is read only by the following routines within the NASTRAN Preface:

- SEMINT reads the first card and processes it using utility XRCARD if it is the NASTRAN card.
- The Executive Control Deck containing free-field cards is read and processed by XCSA using the XRCARD utility.
- The Case Control Deck containing free-field cards is read and processed by IFP1 using the XRCARD utility.
- 4. The Bulk Data Deck containing fixed-field cards is read by XSØRT. This data is subsequently processed by IFP using the RCARD utility.

These card conversion utilities (XRCARD and RCARD) provide, respectively, all the free-field and fixed-field data card processing required by NASTRAN.

5.2.5.2 Use of the Subroutine XRCARD

XRCARD interprets NASTRAN free-field data cards and processes the fields into a sequential buffer than can be easily handled by subsequent modules. Free-field data consist of series of data items separated by suitable delimiters and punched in non-specific card columns. Data items may include alphanumeric, integer, and various types of real variables. Field delimiters may include the comma, slash, parnethesis, and blanks. For details regarding data and delimiter usage and the format of the sequential output buffer, see the XRCARD subroutine description in the Programmer's Manual, Section 3.4.19.

5.2.5.3 Use of the Subroutine RCARD

RCARD interprets NASTRAN fixed-field data cards and processes the fields into a sequential buffer than can be easily handled by subsequent modules. Fixed-field data consist of data items cunched within specific card fields. Each eighty-column card is divided into an eight-column ID field (for the card mnemonic) fullowed by either eight eight-column data fields or four sixteen-column data fields. A special character (asterisk or plus) within the ID field determines

PROGRAM SYSTEM REQUIREMENTS

ر و الداري دي دي جي الحج المحج الدي

when the card is to be interpreted as containing sixteen-column fields. The last eight columns of the card are for continuation mnemonics used by XSØRT and are not processed by RCARD. The data item within the ID field must be alphanumeric. The data items within all other fields may include alphanumeric, integer, and various types of real variables. For details regarding data and the format of the sequential output buffer, see the RCARD subroutine description in the Programmer's Manual, Section 3.4.20.

5.2.5.4 Use of the Operating System Output File

Although NASTRAN printed output is formed and placed onto the system output file through use of standard FØRTRAN-formatted write statements, two basic NASTRAN design concepts prohibit every operating module from generating printed output. First, since the FØRTRAN I/Ø package for output generation occupies a sizable block of computer memory, this package is generally positioned by loader directives within specific output-oriented segments, rather than within the root segment of the overlay, to reduce the total memory requirement. Second, because many functional modules generate the same or similar diagnostic and information messages, a NASTRAN message writer (MSGWRT) was developed to centralize message text and thus prevent duplications within many separate modules.

For the reasons previously discussed, NASTRAN output generation is restricted to a specific class of modules which can reside within an output-oriented segment below the link root segment. These segments contain the output-producing modules such as the Output File Processor (ØFP), the Message Writer (MSGWRT), and the various table and matrix printers (TABPT, MATPRT, etc.) along with the output titling (PAGE) and necessary FØRTRAN L/Ø packages.

5.2.5.5 GINØ

GINØ is a collection of subroutines which provide for all input and output operations within NASTRAN except reading data from the resident system input file and writing data on the resident system output and punch files. These latter operations are accomplished through FØRTRAN-formatted read/write statements. NASTRAN performs input/output operations by making the following calls to GINØ entry points:

1. ØPEN

a conservation and a start and the start

@PEN initiates activity for a file (unless the data block assigned to the file is purged, in which case an alternate return is given). A working storage area (GIN# buffer), for use by GIN#, is assigned (allocated) by the calling program, thus providing optimum allocation of storage by the calling program. This working storage area is reserved for use by GIN# until activity on the file is terminated by a call to CL#SE (see paragraph 4, below). On each of the NASTRAN computers, the working storage area allocated by the calling program is separate from the actual physical I/# buffer.

2. WRITE

WRITE writes a specified (by the calling program) number of words on a file. The block of words to be written may comprise an entire logical record or portion of a logical record.

3. READ

READ returns to the calling program a specified (by the calling program) number of words from the logical record at which the file is currently positioned. READ may be used to transmit an entire logical record or portion of a logical record.

4. CLOSE

CLØSE terminates activity for a file. The working storage area assigned at ØPEN is released. The file is repositioned to the load point is requested.

5. REWIND

REWIND repositions the requested file to the load point. The file must be "open," i.e., a REWIND operation is requested subsequent to a call to @PEN and prior to a call to CL@SE.

6. FWDREC

RWDREC repositions the requested file one logical record forward.

7. BCKREC

BCKREC repositions the requested file one logical record backwards.

8. SKPFIL

SKPFIL repositions the requested file forward or backward N logical files, where N is specified by the calling program.
A REAL AND A

9. EØF

EPF writes a logical end-of-file on the requested file.

The basic unit of I/Ø in NASTRAN is a logical record. The length of a logical record is completely variable and may range from one word to an arbitrarily large number of words. For NASTRAN matrix data blocks, the convention was adopted that each column of the matrix would comprise one logical record. For NASTRAN data blocks containing tables, no rigid convention exists. Typically, each logical record contains one table of a specific type.

The logical record concept provides greatest ease in programming. However, since these records must be stored on a physical device such as a drum, disk, or tape, the characteristics of the device must be taken into consideration. The bulk of NASTRAN data is stored on drums or disks. For both these devices, the common unit of organization is a track, which stores a fixed number of words. Thus, there is a conflict between the variable-length GINØ records and the fixed-length tracks.

This conflict is resolved by blocking. GINØ acts as the interface between the device and the NASTRAN program. Using this technique, the program itself need not be concerned with device considerations (which would create machine-dependent code). GINØ has been parameterized so that different devices may be easily accommodated.

Basically, blocking provides for the reading and writing of fixed-length blocks. The length of a block is a function of the device. It may be one track, one-half track, or other integral division of a track (but never more than one track). Because of the relatively large time to access a given position on a track (due to the rotational speed of the device and/or mechanical movement of the head to the track), a block size equal to one full track is the most desirable. However, limitations in the amount of storage available to hold the blocks in core is a second consideration.

Since the logical record lengths are variable but the length of records physically read or written is fixed, logic must be provided to accommodate this situation. This logic is provided in the GIND routine, which allows for the following cases:

- 1. Multiple logical records per block
- 2. Multiple blocks per logical record

INTRODUCTION TO THE NASTRAN SYSTEM

The method by which physical input and output of blocks is accomplished by GINØ is machine dependent. On the IBM S/360, BSAM macros are used; on the UNIVAC 1108, NTRAN is used; on the CDC 6600, NØS macros are used. These implementation differences are transparent to the NASTRAN applications programmer (functional module writer). The systems programmer who is interested in implementation details on the various machines is referred to Section 5 of the Programmer's Manual.

The names of files input as arguments to the GINØ routines listed above may be alphabetic (BCD, of the form 4HXXXX) or integer.

A GINØ file name is BCD if the file contains permanent executive tables or data blocks. A list of these files for a particular NASTRAN run resides in the permanent portion of the FIST executive table. The following list presents all current executive files with their BCD file names:

File	BCD File Name
Data Pool File	POOL
New Problem Tape	NPTP
Old Problem Tape	ØPTP
BCD Plot Tape	PLT1
Binary Plot Tape	PLT2
User's Master File	UMF
New User's Master File	NUMF
User Input File	INPT
User Input Files	INP1 - INP9

Functional modules should not access these permanent executive files. Functional modules access the files on which their input, output, and scratch data blocks reside by internal integer GINØ file names. Prior to calling a functional module, the link driver routine, XSEMi, calls subroutine GNFIST (GNFIST is called exclusively by XSEMi) to generate the FIST executive table. For each input, output, or scratch data block required for operation of a module (this information being contained in the ØSCAR entry), GNFIST searches the FIAT to find the data block. If the data block is in the FIAT and a file has been assigned to it, an internal GINØ file number denoting the data block and a pointer (index) to the entry in the FIAT is placed in the FIST. The following convention is used for internal GINØ file numbers: input data blocks -- 100 + position in the

5.2-8

PROGRAM SYSTEM REQUIREMENTS

and a second second

#SCAR entry; output data blocks -- 200 + position in the **#SCAR** entry; scratch data blocks -- 301 through 300 + n, where n = number of scratch data blocks as defined in the MPL. (The position in the **#SCAR** entry is the position in the DMAP instruction). If the data block is in the FIAT and is purged, no entry is placed in the FIST for example, consider the following DMAP calling sequence for functional module XYZ:

XYZ A,B,C/D,E,F,G/V,N,PARM1/V,N,PARM2 \$

The data blocks input to the module are A, B, and C; the data blocks output from the module are D, E, F, and G; the module's parameters are PARM1 and PARM2. Note that internal scratch files are not mentioned in the DMAP calling sequence. The number of scratch files for a module is defined in the Module Property List (MPL) executive table (see Section 2.4) and is communicated to the Executive System via the ØSCAR. Details on the syntactical rules of DMAP are given in Section 5 of the User's Manual.

In order to read the input data block B, the GINØ file number interval to XYZ is 102; in order to write data block D, the GINØ file number is 201. The third of, say, five scratch data blocks is referenced by XYZ through the GINØ file number 303.

5.2-9

INTRODUCTION TO THE NASTRAN SYSTEM

and a serie of the approximation of the management of the series of the series of the series of the series of the



Figure 1. An Example of the Use of Open Core.

5.3 NASTRAN DATA DECK ORGANIZATION

NASTRAN is designed for batch mode operation. As such, job submittal takes the form of a card deck or card images on file submitted as a file. The input deck begins with the required resident operating system control cards. The type and number of these cards will vary with the installation. Instructions for the preparation of these control cards are presented in Section 11. Also, additional details regarding system control cards should be obtained from the programming staff at each installation.

The operating system control cards are followed by the NASTRAN Data Deck, shown in Figure 1, which is constructed in the following order (depending on the particular job requirements):

- 1. The NASTRAN Card
- 2. The Executive Control Deck
- 3. The Substructure Control Deck
- 4. The Case Control Deck
- 5. The Bulk Data Deck
- 6. The INPUT Module Data Card(s)

5.3.1 The NASTRAN Card

The NASTRAN card is used to change the default values for certain operational parameters, such as buffer size and machine model number. The NASTRAN card is optional, but if present, it must be the first card of the NASTRAN Data Deck. The NASTRAN card is a free-field card (similar to cards in the Executive Control Deck) Its format is as follows:

NASTRAN keyword₁ = value, keyword₂ = value, ...

The most frequently used keywords are as follows:

- BUFFSIZE Defines the number of words in a GINØ buffer. Usually this value is standardized at any particular installation. However, the desired value may be different from the default value of 1803 (IBM), 1183 (CDC), and 1795 (UNIVAC). In any event, related runs, such as restarts and User Master File runs, must use the same BUFFSIZE for all parts of the run
- CØNFIG Defines the model number of the configuration for use in timing equations for matrix operations. Entries exist for the following configurations:

INTRODUCTION TO THE NASTRAN SYSTEM

المحاج والمراجع والمحمول الأملاف فالمتحاج المتحاج المتحاج المحاج المحاج المحاج

MACHINE	CONFIG.	MØDEL NØ.
IBM 360	0 (default) 3	91, 95 50
	4 5 6 7	65 75 85
IBM 370 (no default)	9 10	195 155 165
	11 12 13	145 158 168
CDC 6000 (ne default)	6 16	6400/6500 6600
CDC CYBER 170	0 (default) 12 14 15	175 174 173
UNIVAC 1100	0 (default) 14	1110 1108

and the second sec

The machine type is automatically determined by NASTRAN. If the model number is the default, the CØNFIG keyword is not needed on the NASTRAN card. It is important to indicate the proper configuration: otherwise, all time-dependent matrix decisions will be incorrect.

- 3. MØDCØM(i) Defines a nine-word array for module communications. Currently, only MØDCØM(1) is supported. If MØDCØM(1) = 1, diagnostic statistics from subroutine SDCØMP are printed; if -1, time and core estimates for module SDCMPS are made without doing the decomposition.
- 4. HICØRE Defines the amount of open core available to the user on the UNIVAC 1100 series machines. The user area default is nominally 65K decimal words. The ability to increase this value may be installation limited.
- 5. TRACKS Establishes the format for the number of tracks (7 or 9) required for the NASTRAN permanent plot file which will be copied to tape. This parameter is not required if a physical plot tape is used, since system control cards define the tracks. The parameters TRACKS = 7, FILES = PLT2 cause plot data to be written to a disk file formatted for a plotter that reads 7-track tapes.
- 6. FILES Establishes NASTRAN permanent files as being disk files rather than tape files. The FILES are PBOL, OPTP, NPTP, UMF, NUMF, PLT1, PLT2, INPT, INP1, INP2, ... INP9. Multiple file names must be enclosed with parentheses such as FILES = (UMF,NPTP). The

NASTRAN DATA DECK ORGANIZATION

FILES parameter(s) must be last on the NASTRAN card. Note the plot files, PLT1 and PLT2, cannot be copied from disk to tape on the UNIVAC, therefore, a physical tape must be assigned prior to job execution if plotting from disk is not available.

7. STST - Defines the singularity tolerance in EMG and SMA1. The default value is .01. Singularities are written to GPST or (if altered into the Rigid Format) GPSPC. If module GPSPC is used, it may, on option, automatically constrain the strongest singularity (weakest stiffness) combination.

Additional information for all NASTRAN card options is given in Section 6.3 of the Programmer's Manual.

5.3.2 Function of the Executive Control Deck

The first purpose of executive control is to provide a level of regulation for the many options within NASTRAN. At this level the executive distinguishes between the broad approaches to problem solution, e.g., between a matrix abstraction approach by the analyst or a Rigid Format approach according to problem class. Also at this level, the executive distinguishes between several operational modes, e.g., a first attempt, a continuation, or a modification. Certain other functions of a general nature are convenient to include with the executive control such as problem identification, selection of a level of diagnostics, and the estimation of solution time.

The Executive Control Deck includes cards which describe the nature and type of the solution to be performed. These include an identification of the problem, an estimated time for solution of the problem, a selection of an approach to the solution of the problem, a restart deck from a previous run if the solution is to be restarted, an indication of any special diagnostic printout to be made, and a specification of whether execution of the problem is to be completed in a single run, or whether execution will be stopped (checkpointed) at some intermediate step.

Each of the cards comprising the Executive Control Deck is read and analyzed. Depending on the card, information is either stored in various executive tables maintained in core storage or written in a control table on the New Problem Tape for further processing during a later phase of the Preface.

The Executive Control Deck is described in detail with examples in Section 6.

5.3.3 Function of the Substructure Control Deck

The user places commands needed to control the execution of NASTRAN for automated multi-stage substructuring analyses via the Substructure Control Deck. These commands are automatically translated into DMAP alter packages which are then automatically inserted into the Executive Control Deck.

The Substructure Control Deck is an optional deck. It is used only when the substructuring system has been requested through the APProach card in the Executive Control Deck. Further details and examples for the Substructure Control Deck are found in Sections 7 and 20.

5.3.4 Function of the Case Control Deck

When the Rigid Format solution route is selected, further details of control are provided by the Case Control portion of the executive. In effect, the analyst can manipulate his problem by means of entries he inserts in the Case Control Deck. Fundamental to the method of control in this section is the notion of sets. Boundary conditions, loading cases, and output selections are identified by set numbers and selected accordingly. The analyst can make choices amongst the sets of data representing different physical situations which are allowed to be assembled in the bulk data portion of the problem input. Here also the analyst can regulate his output.

The Case Control Deck includes cards that indicate the following options: selection of specific sets of data from the Bulk Data Deck (i.e., from the data deck that describes the modeling details and parameters of a problem), selection of printed or punched output, definition of sub-cases, and the definition of plots to be made.

The information defining data set selection, output format selection, and subcase definition is read from the Case Control Deck and written into a case control data block. Information defining plot requests is written into a plot control data block.

If the problem is a restart, a comparison with the case control data block from the previous run (stored on the Old Problem Tape) is made. Differences are noted in an exectuive restart table.

5.3.5 Function of the Bulk Data Deck

In NASTRAN the input supplied to the mathematical operations performed in functional modules is provided in the form of previously organized data blocks. Data blocks derive from two sources: those that derive from the bulk input data and those that are generated as output from previous

NASTRAN DATA DECK ORGANIZATION

functional modules. Those that derive from the bulk data are organized into data blocks by the IFP routine, but prior to the execution of IFP, XSØRT sorts the bulk data. Operation of the XSØRT routine is influenced by the type of run. If the run is a cold start (that is, an initial submittal for a given job), the bulk data is read from the system input unit or the User's Master File, is sorted, and is written on magnetic files in preparation for problem execution. If the analyst wants to provide for a future restart, the SØRT routine prepares a file on the New Problem Tape which contains the sorted bulk data. If the run is a restart, the bulk data are copies from the Old Problem Tape with the addition of any changes from the system input unit.

An echo of the unsorted bulk data is given if requested. Similarly, printing of the sorted bulk data may be suppressed on request.

Since the collating sequence of alphanumeric characters varies from computer to computer, the sort routine converts all characters to an internal code prior to sorting. Following the sort, the characters are reconverted. In this way, the collating sequence is made computer independent.

The algorithm used by the sort routine is biased toward the case where the data is in sort or nearly in sort. Consequently, Bulk Data Decks which are nearly in sort will be processed efficiently by the routine.

5.3.6 INPUT Module Data

NASTRAN provides the capability to automatically generate bulk data cards for a number of selected test problems. This capability is fully described in the User's Manual, Section 2.6. The capability is accessed by using a DMAP alter package in the Executive Control Deck and by placing specially formatted data cards following the Bulk Data Deck, as shown in Figure 1.

5.3.7 MASTRAN Character Set

Except for the IBM 360/370 series, all NASTRAN data cards must be punched using the character set shown in Table 1, below. The EBCDIC character set may be used on the IBM 360/370 series. Any EBCDIC characters are automatically translated into the character set shown in the table below. The EBCDIC character card punch configurations are shown in parenthesis for the five characters that differ from the standard character set.

INTRODUCTION TO THE NASTRAN SYSTEM

Ľ,



Figure 1. General construction of NASTRAN data deck.

C. Frank

Service Service

L.V.

A STATE OF A

:

Table 1. NASTRAN Character Set.

.

Character	Card Punch(s)	Character	Card Punch(s)	EBCDIC Punch(s)
blank	blank	N	11-5	
o	o	9	11-6	
1	1	Р	11-7	
2	2	Q	11-8	
3	3	R	11-9	
4	4	S	0-2	
5	5	Т	0-3	
6	6	υ υ	0-4	
7	7	V V	0-5	
8	8	w	0-6	
9	9	X X	0-7	
Å	12-1	Y Y	0-8	
8	12-2	Z	0-9	
c	12-3	\$	11-3-8	
D	12-4		0-1	
E	12-5	+	12	(12-6-8)*
F	12-6	-	11	
G	12-7	(0-4-8	(12-5-8)*
H	12-8		12-4-8	(11-5-8)*
	12-9		4-8	(5-8)*
3	11-1	-	3-8	(6-8)*
ĸ	11-2		0-3-8	
	11-3	11 .	12-3-8	
H	11-4	•	11-4-8	

*IBM 360/370 only.

THE REPORT OF A STATE AND A MARKET OF

and the second second

6.1 INTRODUCTION

The Executive Control Deck serves to identify the job and the type of solution to be performed. It also declares the general conditions under which the job is to be executed, such as, maximum time allowed, type of system diagnostics desired, restart conditions, and whether or not the job is to be checkpointed.

If the job is to be executed with one of NASTRAN's standard Rigid Formats, the number of that Rigid Format is declared. Any alterations to the Rigid Format that may be desired would also be part of the Executive Control Deck. These alterations would be specified as a set of DMAP instructions to be inserted into the Rigid Format sequence (User's Manual, Section 3).

If Direct Matrix Abstraction (User's Guide, Section 15; and User's Manual, Section 5) is used, the complete DMAP sequence must appear in the Executive Control Deck. These DMAP instructions allow the user to specify his own sequence of matrix operations independent of the standard sequences provided by the Rigid Formats.

The organization of the NASTRAN input deck is described next along with the card format in Section 6.2. The Executive Control options are described in Section 6.3 and summarized in Table 1. Examples are also provided in Section 6.4.

EXECUTIVE CONTROL DESCRIPTION

Table 1. Executive Control Options

The following Executive Control cards are mandatory:

and a second second second and the second second

ιì

- 1. <u>APP</u> selects a Rigid Format approach or a user provided Direct Matrix Abstraction Program (DMAP).
- 2. <u>CEND</u> defines the end of the Executive Control deck.
- 3. ID defines the beginning of the Executive Control deck.
- 4. <u>TIME</u> defines the maximum time in minutes allotted to the execution of the NASTRAN program.

The following Executive Control cards are required under certain circumstances:

- 1. BEGIN\$ defines the beginning of user provided DMAP statements.
- 2. <u>END\$</u> defines the end of user provided DMAP statements.
- 3. ENDALTER defines the end of user provided changes to a Rigid Format.
- 4. <u>RESTART</u> defines the beginning of a restart dictionary.
- 5. <u>SPL</u> selects the solution number of a Rigid Format.
- 6. UMF selects a data deck from a User Master File.
- 7. <u>UMFEDIT</u> controls execution of a UMF editor.

The following Executive Control cards are optional:

- 1. ALTER defines the Rigid Format statement(s) at which the user makes alterations.
- 2. CHKPNT requests the execution to be checkpointed.
- 3. DIAG requests diagnostic output to be provided or operations to be effected.
- 4. NUMF requests a User Master File to be created.
- 5. \leq defines a non-executable comment.

6.2 INPUT SEQUENCE AND FORMAT

The sequence of input cards for execution of a NASTRAN run, see Figure 1, begins with the required resident operating system control cards. The type and number of these cards varies with the conventions used at each computer installation. See the programming staff at that installation for instruction in the preparation of these control cards. Typical examples of the operating system control cards used by NASTRAN are illustrated in Section 11 for all three computer systems on which NASTRAN is maintained (Programmer's Manual, Section 5).

These operating system control cards are followed by the NASTRAN Data Deck which consists of four sections:

1. Executive Control Deck (described here).

Substructure Control Deck (Optional, Section 7).

3. Case Control Deck (Section 8).

4. Bulk Data Deck (Section 9).

The Executive Control Cards are summarized in Table 1 and illustrated with examples in Section 6.4. They may occur in any sequence with only the following exceptions:

NASTRAN

This is an optional card and is described in Section 5 of the Guide and Section 2 of the User's Manual.

ID A1,A2

This is the first card of the Executive Control Deck. It may be preceded only by the NASTRAN card, if used, and the operating system control cards.

CEND

This card is the last card of the Executive Control Deck and is followed by the Case Control Deck or the Substructure Control Deck.

The Executive Control Card format is free field. The name of the operation (e.g., CHKPNT, TIME, CEND, etc.) may begin in any card column and is followed by the operand. The operand must be separated from the name by one or more blanks. The fields of the operand are separated by commas and may be integers or alphanumeric. The *first* character of an alphanumeric field must be alphabetic, followed by up to 7 additional alphanumeric characters. If desired, blanks may be placed adjacent to separating commas for legibility. If a card ends with a comma, a continuation card will be expected. Comment cards (\$ in column 1) may be inserted anywhere in the deck following the ID card. The individual options are described next.

6.2-1

EXECUTIVE CONTROL OPTIONS

an and the second s

6.3 EXECUTIVE CONTROL OPTIONS

The Executive Control options are summarized in Table 1. Note that the DMAP options, which are included as part of the Executive Control Deck on input to NASTRAN, are not summarized here. Instead, see Section 15 of this Guide for examples, and Section 5 of the User's Manual for full detail on the individual DMAP instructions. The full explanation and examples for each of the Executive Control options and the operands Ai and Ki is found in Section 2.2 of the User's Manual.

6.3.1 Use of Executive Control Cards

As already noted in Section 6.2, the NASTRAN card, when used, is always the first card in the input stream following the operating system control cards. Otherwise, the ID card is the first card and CEND is the last card of the Executive Control Deck.

If a restart run is being made, the restart dictionary punched from the preceding run must be included. The RESTART card is the first card of the dictionary which contains the user problem identifiers provided for the run being checkpointed. The last card contains the comment \$ END OF CHECKPØINT DICTIØNARY. The information on the RESTART card is used by NASTRAN to check that the correct restart tape (Old Problem Tape) has been provided. If a new Rigid Format execution is to be made with the same model from a previous run, a restart with Rigid Format switch could save processing time by not requiring the regeneration of matrices saved from the earlier run. If desired, the restart dictionary may also be modified to alter the reentry point used for the current problem (see User's Manual, Section 3.1.3, for detail).

If DIAGnostics are desired, the "sense switches" K must be specified. See the list of options in Section 2.2.1. If long runs are being made with the user monitoring the operator's console, a status report can be requested by inputting DIAG 5,6 to get BEGIN and END times for each functional module printed on the operator's console. Most of the options, however, are designed for use in program development and debugging of NASTRAN. Use of DHAP print utilities (Section 15 of the User's Guide), and NASTRAN's plot capabilities (Section 4 of the User's Manual) are suggested for obtaining detailed information which would be helpful in finding errors in a structural model.

If a CHKPNT is requested (A1 = YES), a New Problem Tape must be provided (Section 11 of the User's Guide).

EXECUTIVE CONTROL DESCRIPTION

يې د اور مېرونو د درونه مېرونو د د د د مېرو د د د

. . . .

To modify a Rigid Format sequence, use the ALTER card to add the desired DMAP instructions. The sequence numbers of the Rigid Formats are provided in the User's Manual, Section 3. More than one alter package may be used; however, the last package must terminate with ENDALTER. See Section 15 of the Guide for examples of typical packages.

To be assured that a checkpoint is taken before the operating system terminates a job for insufficient time, set the TIME operand K to less than the time limit specified on the operating system run, job, or job step control card. That way NASTRAN will provide a normal exit, and the punching of the checkpoint dictionary can be completed.

If the UMFEDIT editing feature is to be used, the appropriate input (UMF) and output (NUMF) tapes must be mounted (User's Manual, Section 2.5; and User's Guide, Section 11).

If the approach being used is APP DMAP, the desired DMAP instructions must be initiated with BEGIN\$ or XDNAP\$ (User's Manual, Section 5.2) and terminated with END\$. Note that any alphanumeric commentary (except \$) may be inserted between BEGIN and the dollar sign (\$).

6.4 EXAMPLES OF EXECUTIVE CONTROL DECKS

a construction of the Application of the

The following are examples of commonly employed Executive Control Deck arrangements. Additional examples can be found in Sections 18-26.

6.4.1 Example 1 - Cold Start, No Checkpoint

Card No.	<u>'-</u>	
1	ID	MYNAME, BRIDGE23
2	АРР	DISP
3	SØL	2,0
4	TIME	5
5	DIAG	1,2
6	CEND	

Notes: 2. Execute a displacement approach.

- 3. Use the unmodified (K2=0), Rigid Format 2 Static analysis with inertia relief.
- 5. Request core memory dump if a nonpreface fatal message is generated, and print File Allocation Table (FIAT) (Programmer's Manual, Section 2.4.1) following each call to the File Allocator.

6.4.2 Example 2 - Cold Start, Checkpoint, Modified System Parameters

Card No.

ì

1 NASTRAN	BUFFSIZE=878
-----------	--------------

- 2 ID PERSONZZ, SPACECFT
- 3 CHKPNT YES
- 4 APP DISP
- 5 SØL 1,3
- 6 TIME 15
- 7 CEND
- Notes: 1. Requests a change in GINØ Luffer size for NASTRAN's own read/write routine.
 - 2. Requests a checkpoint tape be written (New Problem Tape).
 - 3. Remove instructions for loop control which process additional sets of constraints (K2=3), Rigid Format 1 Static Analysis.

EXECUTIVE CONTROL DESCRIPTION

6.4.3 Example 3 - Restart, Checkpoint, Rigid Format Switch

ىر بىم مەپ مەرىپ قاقلار ئىيمۇنلارمۇرلارلار ئاپ ئۆچ ئارغا بارغا يۇرىيە بىلەر يەرىپ بىرى بىرى بىر

Card No.

6.4.4

Ì

١	NASTRAN	BUFFSIZE=878
2	ID	PERSØNZZ, CASEB
3	RESTART	PERSØNZZ,SPACECFT,2/21/72, 3353,
4	1,	XVPS, FLAG=0, REEL=1, FILE=6
5	2,	REENTER AT DMAP SEQUENCE NUMBER 7
6	3,	GPL, FLAG=O, REEL=1 FILE=7
7	\$ END Ø	F CHECKPØINT DICTIØNARY
8	ΑΡΡ	DISP
9	SØL	3,3
10	T IME	10
11	CHKPNT	YES
12	CEND	
Notes:	1. The	e same NASTRAN card is used to retain system parameters of the original Id start run.
	2. Nev pul	ID parameters specified to help identify the restart dictionary to be nched by this run.
	3. The in ad wr to	e first card of the restart dictionary punched by the previous run which the CHKPNT option was selected. Date and time of execution is ded to the checkpointed run ID to prevent inadvertent restart from the ong tape. (See User's Manual, Section 2.2.1, for explanation of cards follow.)
	4-7. Th (0 vi	e restart dictionary identifying the data stored on the restart tape Id Problem Tape) which was created as the New Problem Tape on the pre- ous run.
	9. Re ti (1	equests a switch to a normal modes analysis (Rigid Format 3), and removes ne looping instructions for processing additional sets of constraints (2=3).
	11. Re	equests a new checkpoint tape be written, a New Problem Tape for this run.
.4 <u>Exa</u> m	ple 4 - R	estart, No Checkpoint, Altered Rigid Format
Card N	ю.	
<u></u>	10	DEDSANYY PRAJECTY

1	ID	PERSONAY, PROJECTA	
2	RESTART	PERSONZZ, SPACECRAFT, 3/1/73, 53	151,
3	1,	XVPS, FLAGS=0, REEL=1, FILE=6	\$

EXAMPLES OF EXECUTIVE CONTROL DECKS

an an a caracterization and the second s

į

4	2,	REENTER AT DMAP SEQUENCE NUMBER 7
5	3,	GPL, FLAGS=0, REEL=1, FILE=7
6	\$ END ØF	CHECKPØINT DICTIØNARY
7	CHKPNT	NØ
8	APP	DISP
9	SØL	3,3
10	TIME	15
11	ALTER	35
12	MATPRN	KGGX,,,,// \$
13	TABPT	GPST.,,,// \$
14	ENDALTER	
15	CEND	
Notes:	7. СНКР	NT card is opticnal here since <u>N</u> 0 is the default.
	11. Begi 3 se	nning of ALTER package for inserting DMAP instruct quence number 35.

12. Request printout of stiffness matrix KGGX (User's Manual, Section 5.3.2; Programmer's Manual, Section 2.2.1 and 2.3.9.1).

instructions after Rigid Format

- Request printout of Grid Point Singularity Table GPST (User's Manual, Section 5.3.2; Programmer's Manual, Sections 2.2.1 and 2.3.9.3).
- 14. ENDALTER concludes the alter package.

7. SUBSTRUCTURE CONTROL DESCRIPTION

7.1 INTRODUCTION

The Substructure Control Deck (User's Manual, Section 2.7) is an optional deck in the NASTRAN input data stream. Its presence involves the use of the automated multi-stage substructuring system with modal synthesis. When this deck is used, it is always placed immediately following the Executive Control Deck.

Use of the Substructure Control Deck also requires using either of the following three forms of the APProach card in the Executive Control Deck:

APP DISP, SUBS APP DISP, SUBS, 1 APP DMAP, SUBS

The APP DISP, SUBS Executive Control command causes NASTRAN to automatically create ALTERs to the displacement approach Rigid Format specified. The APP DISP, SUBS, 1 command indicates NASTRAN is <u>not</u> to automatically create the ALTERs. In this case, the user will provide an ALTER package for the substructure operation to be performed. This package is placed directly in the Executive Control Deck input stream. The APP DMAP, SUBS command indicates that a user supplied DMAP sequence for substructuring operations will be provided in the Executive Control Deck independent of any specified Rigid Format. These Executive Control commands, their use and control over the substructuring system, are fully explained later in Section 20.

7.2 INPUT SEQUENCE AND FORMAT

The Substructure Control Deck is input immediately following the last card (CEND) of the Executive Control Deck (User's Guide, Section 6). The first card of the Substructure Control Deck must be SUBSTRUCTURE PHASEi, where i = 1, 2, or 3, according to the phase of execution being performed. The last card must be ENDSUBS.

The Substructure Control cards are free format. Blanks are used to separate the control words. The equal sign (=) is used in an assignment statement. A comma (,) as the last entry on a physical card signals that a continuation card will be used. Comment cards, signalled by a dollar sign (\$) in card column one, can be inserted anywhere in the Substructure Control Deck and thus may contain any alphanumeric characters the user desires. Only the first four characters of each control word need be used so long as that option is uniquely identified. A summary of Substructure Control cards is given in Table 1. Typical examples of their use are given in Section 7.4 and additional examples which are related to actual problems can be found in Section 20.

The input sequence of Substructure Control cards is critical to successful execution. All substructure commands except SUBSTRUCTURE, ENDSUBS, and SØF cause DMAP alters to be generated. These alter packages are built by NASTRAN as the Substructure Control Deck commands are encountered. NASTRAN then executes the analysis specified in the same sequence as these DMAP alters are generated. By convention, the SØF command and its subcommand PASSMØRD are placed before other executable commands. The user should also note that the order of <u>subcommands</u> following a given command is <u>not</u> critical.

The function of each Substructure Control command is described in Section 20 of this Guide, and in Sections 1.10.2 and 2.7 of the User's Manual. The DMAP alters automatically generated from each substructuring command are presented in Section 5.9 of the User's Manual. A description of how the user may interface with these NASTRAN-generated substructuring DMAP alters is presented in Section 2.7.2 of the User's Manual where the reader will learn what restrictions are imposed by this procedure on other user-supplied DMAP alters for the Executive Control Deck.

The relationship between the Substructure Control Deck and the Executive Control Deck was given in Section 7.1. This relationship and others among the Substructure Control Deck, the Case Control Deck, and the Bulk Data Deck are fully developed with examples in Section 20.

7.2-1

SUBSTRUCTURE CONTROL DESCRIPTION

Table 1. Summary of Substructure Commands.

A. Phase and Mode Control

and a second for a the constraint of a second s

Ħ	SUBSTRUCTURE	- Defines execution phase (1, 2, or 3)
	NAME*	- Specifies Phase 1 or Phase 3 substructure name (not used in Phase 2)
	SAVEPLOT	- Requests plot data be saved in Phase l
	OPTIONS	- Defines matrix options (K. B. K4, M. P. or PA)
	RUN	- Limits mode of execution (DRY, GØ, DRYGØ, STEP)
ŧ	ENDSUBS	- Terminates Substructure Control Deck

B. SOF Controls

#	SØF	- Assigns physical file for storage of the SØF
	PASSHORD*	- Protects and ensures access to correct file
	SØFØUT or SØFIN	- Copies SØF data to or from an external file
	PØSITIØN	- Specifies initial position of input file
	NAMES	- Specifies substructure name used for input
	ITEMS	- Specifies data items to be copied in or out
	SØFPRINT	- Prints selected items from the SØF
	DUMP	- Dumps entire SØF to a backup file
	RESTØRE	- Restores entire SØF from a previous DUMP operation
	CHECK	- Checks contents of external file created by SØFØUT
	DELETE	- Deletes out selected groups of items from the SØF
	EDIT	- Edits out selected groups of items from the SØF
	DESTROY	- Destroys <u>all</u> data for a named substructure and <u>all</u> the substructures of which it is a component

C. <u>Substructure Operations</u>

COMBINE	- Combines sets of substructures
NAME*	- Names the resulting substructure
TØLERANCE*	- Limits distance between automatically connected grids
CONNECT	- Defines sets for manually connected grids and releases
OUTPUT	- Specifies optional output results
COMPONENT	- Identifies component substructure for special processing
TRANS FØRM	- Defines transformations for named component substructures
SYMTRANSFØRM	- Specifies symmetry transformation
SEARCH	- Limits search for automatic connects
EQUIV	- Creates a new equivalent substructure
PREFIX*	- Prefix to rename equivalenced lower level substructures

Mandatory Control Cards

* Required Subcommand

INPUT SEQUENCE AND FORMAT

1.19

edar a

Table 1 Summary of Substructure Commands (continued).

C. <u>Substructure Operations (continued</u>)

ē.

Ì

REDUCE	- Reduces substructure matrices				
NAME*	- Names the resulting substructure				
BOUNDARY*	- Defines set of retained degrees of freedom				
RSAVE	 Indicates the decomposition product of the interior point stiffness matrix is to be saved on the SØF 				
autput	- Specifies optional output requests				
MREILICE	 Reduces substructure matrices using a normal modes transformation 				
NAME*	- Names the resulting substructure				
RAUNDARY*	- Defines set of retained degrees of freedom				
FIXED	- Defines set of constrained degrees of freedom for modes calculation				
RNAME	 Specifies basic substructure to define reference point for inertia relief shapes 				
RGRID	 Specifies grid point in the basic substructure to define reference point for inertia relief shapes. Defaults to origin of basic substructure coordinate system 				
METHOD	- Identifies EIGR Bulk Data card				
RANGE	- Identifies frequency range for retained modal coordinates				
NMAX	 Identifies number of lowest frequency modes for retained modal coordinates 				
GLDMØDES	- Flag to identify rerunning problem with previously computed modal data				
GLDBGUND	- Flag to identify rerunning problem with previously defined boundary set				
USERMODES	- Flag to indicate modal data have been input on bulk data				
OUTPUT	- Specifies optional output requests				
RSAVE	 Indicates the decomposition product of the interior point stiffness matrix is to be stored on the SØF 				
CREDUCE	- Reduces substructure matrices using a complex modes transformation				
NAME*	- Names the resulting substructure				
BOUNDARY *	- Defines set of retained degrees of freedom				
FIXED	- Defines set of constrained degrees of freedom for modes calculation				
METHØD	- Identifies EIGC Bulk Data card				
RANGE	 Identifies frequency range of imaginary part of the root for retained modal coordinates 				
NMAX	- Identifies number of lowest frequency modes for retained model data coordinates				
GLDMØDE S	- Flag to identify rerunning problem with previously computed model data				
GPARAM	- Specifies structural damping parameter				
B UTPUT	- Specifies optional output requests				
RSAVE	 Indicates the decomposition product of the interior point stillness matrix is to be stored on the SØF 				

* Required Subcommand

SUBSTRUCTURE CONTROL DESCRIPTION

Table 1. Summary of Substructure Commands (continued).

C. Substructure Operations (continued)

1

ì

 $g_{\lambda}(a_{j,\lambda},a_{j,\lambda}^{*})^{2}a_{j,\lambda}^{*}(a_{j,\lambda},a_{j,\lambda},a_{j,\lambda},a_{j,\lambda},a_{j,\lambda}^{*})^{2}\delta_{\lambda}^{*}\delta$

MRECØVER	- Recovers mode shape data from an MREDUCE or CREDUCE operation				
SAVE	- Stores modal data on SØF				
PRINT	- Stores moda! Jata and prints data requested				
SØLVE	 Initiates substructure solution (statics, normal modes, frequency response, or transient analysis) 				
RECOVER	- Recovers Phase 2 solution data				
SAVE	- Stores solution data on SØF				
PRINT	- Stores solution and prints data requested				
BRECØVER	- Basic substructure data recovery, Phase 3				
PLØT	- Initiates substructure undeformed plots				

المربي المهاد ومقام دولالا الأخط الأرابي ال

7.3 EXAMPLES OF SUBSTRUCTURE CONTROL DECKS

The following are examples of Substructure Control Decks. Additional examples are found in Section 20 of this Guide and in Section 1.10.2 of the User's Manual as they apply to specific problems. The notes are intended to highlight the specific intent of the particular card illustrated.

7.3.1 Example 1 - Phase 1, Create Substructure WING

Card No.

Ļ

- 1 SUBSTRUCTURE PHASE1
- 2 SØF(1) = SØF1, 1000, NEW \$ (CDC)
 - or SØF(1) = FT02,,NEW \$ (IBM)
 - or SØF(1) = INP1, 1000, NEW \$ (UNIVAC)
- 3 PASSWØRD = PRØJECTX
- 4 NAME = WING
- 5 SAVEPLØT = 1
- 6 SØFPRINT TØC
- 7 ENDSUBS
- Notes: 2. This deck is the <u>first</u> substructure execution for Project X. It introduces and, hence, initializes the SØF. Three versions of this card are shown, one for each of the NASTRAN computers. Refer to Section 2.7 of User's Manual for further details and instructions on the use of the SØF command. Note that for the life of this SØF file, this is the <u>only</u> time the "NEW" parameter is used.
 - 3. Assigns the password PRØJECTX to this SØF. This password must always be used in later executions to gain access to the SØF.
 - Assigns the name WING to all matrices and tables created by this NASTRAN execution which is to generate a basic phase 1 model.
 - 5. Requests data be saved for undeformed plotting of this structure. To define these data, the following cards <u>must</u> appear at the end of the Case Control Deck:

@UTPUT(PLØT) SET 1 = (any valid set definition)

Note that only one set may be saved for a given Phase 1 substructure.

6. Requests a print of the substructure operating file (SØF) table of contents.

SUBSTRUCTURE CONTROL DESCRIPTION

7.3.2 Example 2 - Phase 2, Operations on Substructure WING

Card No.

- 1 SUBSTRUCTURE PHASE2
- $2 \quad SOF(1) = SOF(1), 1000$
- 3 PASSWØRD = PRØJECTX
- 4 SØFPRINT TØC
- 5 REDUCE WING
- 6 NAME = WINGR
- 7 BOUNDARY = 10
- 8 MREDUCE WINGR
- 9 NAME = MRWING
- 10 BØUNDARY = 20
- 11 FIXED = 30
- 12 METHØD = 40
- 13 NMAX = 5
- 14 ØUTPUT = 1, 10
- 15 SØLVE MRWING
- 16 RECOVER MRWING
- 17 PRINT MRWING
- 18 DISP = ALL
- 19 ENDSUBS
- Notes:

5-7. Requests a Guyan reduction for the matrices associated with substructure WING. The new model will be named WINGR and will have degrees of freedom identified by boundary set 10 specified on BDYC and BDYSi Bulk Data cards. The other degrees of freedom from WING not referenced on these cards will be eliminated or reduced out.

8-14. Requests a normal mode reduction using the matrices associated with WINGR. The new model will be named MRWING. It will contain physical coordinate degrees of freedom defined by boundary set 20, and will have 5 (NMAX) additional modal coordinates computed using the eigenvalue extraction method with set ID 40 identified by the EIGR card. When the modes are computed, degrees of freedom specified by boundary set 30 (FIXED) will be constrained to zero displacement. Diagnostic and summary @UTPUT is requested.

- Requests a solution for model MRWING using the appropriate DMAP sequence specified on the Rigid Format used, i.e., normal modes for RF 3, frequence response for RF 8, etc.
- Requests the process be initiated for recovery of data from the solution obtained for MRWING.

EXAMPLES OF SUBSTRUCTURE CONTROL DECKS

,

Manual

Ż

1

17,18. Requests the displacements be printed for all degrees of freedom from the solution to MRWING.

8. CASE CONTROL DESCRIPTION

8.1 INTRODUCTION

The Case Control Deck is used to provide titling information, to select sets of data from the Bulk Data Deck, to generate the subcase structure, and to make output requests for printing, punching, and/or plotting.

The Case Control data are placed on a file which is read by the NASTRAN functional modules. This information is used by a module to control is processing and to select the specific data on which it is to operate. The sequence in which these functional modules themselves are called is controlled by user input to the NASTRAN Executive Control Deck (User's Guide, Section 6). The Case Control options (User's Manual, Section 2.3) are listed in Table 1. Typical examples of their use are given in Section 8.4. Additional examples related to actual problems can be found in Sections 18 through 26.

INPUT SEQUENCE AND FORMAT

8.2 INPUT SEQUENCE AND FORMAT

The Case Control Deck is input immediately following the last card (CEND) of the Executive Control Deck (User's Guide, Section 6) or, when present, the last card (ENDSUBS) of the Substructure Control Deck (User's Guide, Section 7). It terminates with the BEGIN BULK card which initiates the input of the Bulk Data Deck. This card and the ENDDATA card, which signals the end of Bulk Data, must <u>always</u> be used even if there are no Bulk Data cards.

The Case Control card is free format. Blanks are used to separate the control words. The equal sign (=) is used in an assignment statement. A comma (,) at the end of a physical card signals that a continuation card will be used. Comment cards, signalled by a dollar sign (\$) in card column one, can be inserted anywhere in the Case Control Deck and thus may contain any alphanumeric characters the user desires. Only the first four characters of each control word need be used so long as that option is uniquely identified.

The input sequence of Case Control cards is arbitrary with the exceptions of the locations of SUBCASE, SJBCOM, etc.; OUTPUT(PLOT) and OUTPLOT(XYPLOT) commands; and the set of plot commands. The effects of Case Control card order is discussed in Section 2.3 of the User's Manual.

8.2-

a constant a constant

8.3 CASE CONTROL OPTIONS

The Case Control options are listed in Table 1. These options are divided into functional groups for Bulk Data selection, output selection, and subcase specification as described below. A brief summary of the subcase specification options is presented in Table 2. 'Examples, with explanations, are presented in Section 8.4. Additional examples as they apply to specific problems can be found in Sections 18 through 26. Refer to the discussion of each available Rigid Format (User's Manual, Section 3) for a detailed description of the applicable Case Control options.

8.3.1 Bulk Data Selection

In general, element connectivities, element properties, gridpoint coordinates, and related geometry data used to describe a structural model are <u>not</u> selected from Bulk Data by Case Control. <u>All</u> of these data that are found in the Bulk Data Deck are used in defining the model. However, Case Control <u>is</u> used to select from Bulk Data by specification of the appropriate set identifiers, those load sets, constraint sets, temperature data, direct input matrix data, and method of analysis which are to be used in solving a particular problem. The concept of Case Control allows the user to analyze one model under many different loading and constraint conditions all in one run. Each unique combination of conditions is identified in Case Control as a separate subcase within that one run.

Additional control, it should be noted, is provided directly in Bulk Data via the PARAM card (User's Manual, Section 3.1). Each Rigid Format (User's Manual, Section 3) allows for selected parameters to be specified on this card. These parameters may be used in directing NASTRAN to perform additional processing and to provide for additional output.

8.3.2 Output Selection (User's Manual, Section 2.3)

The Case Control options for control of NASTRAN output allow the user to select the results he wishes to have printed, punched, or plotted. General structural analyses, as performed by NASTRAN, can produce such overwhelming volumes of computer printout that this selective feature is one of the most important functions of the Case Control Deck. A full description of the specialized output available is presented at the conclusion of the discussion of each Rigid Format (User's Manual, Section 3).

CASE CONTROL DESCRIPTION

Certain categories of NASTRAN output are automatically provided (User's Manual, Section 3.1). The automatic output includes the NASTRAN title page, echo of the Executive Control, Case Control, and sorted Bulk Data Decks, DMAP sequence listing on restarts and the Check-point Dictionary when a checkpoint has been requested. Additional output can be requested for diagnostic purposes, ϵ .g., see Executive Control DIAG card (User's Manual, Section 2) and DMAP alters (User's Guide, Section 15).

The specific output which may be requested through Case Control can be divided into categories as follows:

- 1. Output Control provide for titles, organization, pagination and mode of output.
- Output Amounts provide for definition of sets and intervals to limit volume of output.
- 3. Output Categories provide for selection of data types to be output.

These requests may be applied for all subcases or selectively to individual subcases to further limit the volume of desired output. Within a subcase, the user also has the option to suppress output which had been requested above the subcase level. The output requests for specific types of data provide the parameters (User's Manual Section 2.3) with which the user may select special sorting of the output, e.g., printed or punched. The user should refer to the explanations of the Case Control examples in Section 8.4 where certain idiosyncrasies in the use of the output options are described.

The output available from a Random Response Analysis, Rigid Formats 8 and 11 should be especially noted. Unlike the results from other available analyses, the printout of the power spectral density and autocorrelation functions is available only through the plotting control options provided for **BUTPUT**(XYBUT) (User's Manual, Sections 10.4 and 19.7).

Plotting requests, which always appear <u>after</u> the last subcase, are sufficiently specialized that separate sections (User's Guide, Sections 10.3 through 10.5) are devoted to a discussion of the four forms of plotting available in NASTRAN. These forms are:

- Structural illustrates the actual model in an undeformed and/or in a deformed state (User's Manual, Section 4.2).
- X-Y Plot provides curve plotting of selected response data in functional form (User's Manual, Section 4.3).

CASE CONTROL OPTIONS

ير محمدة فيكار بدر وتحارى

Þ

· ***, - ,

- Matrix Plot provides a topographical display of matrix coefficients in graphical form (User's Manual, Section 5.3.2, "SEEMAT").
- 4. Contour Plots displays stress and displacement contours (User's Manual, Section 4.2).

8.3.3 Subcase Specification (User's Manual, Section 2.3)

الرارية العمد محارز

ÿ

- 1. Static Analyses, Rigid Formats 1 and 2
 - Separate subcases for each unique combination of loadings and constraint sets.
 - Loading conditions, selected from Bulk Data with LØAD, DEFØRM, or TEMPERATURE (LØAD) cards, which are associated with the same sets of constraints should be called for in contiguous subcases to avoid unnecessary computations. The SPC constraint sets may include gridpoint displacements as a loading condition which will be superimposed on any other loading selected in that subcase.
- 2. Static Analysis with Differential Stiffness, Rigid Format 4
 - Case Control must contain only two subcases. The loading and constraint sets must be specified above the subcase level. The first subcase is used to specify the output requests that apply only to the linear solution. The second subcase is used to specify output requests for the nonlinear solution.
- 3. Buckling Analysis, Rigid Format 5
 - Same as #2 above except that the second subcase contains the METHBD card selecting the EIGB card from Bulk Data which defines the method of eigenvalue extraction.
- 4. Piecewise Linear Analysis, Rigid Format 6
 - Case Control must contain one and only one subcase.
 - The PLCBEF card which must be used to select the sequence of loading factors to be applied.
- 5. Normal Modes Analysis, Rigid Format 3
 - Multiple subcases are used only to control output requests.
 - The METH#D card must be used to select the EIGR card from Bulk Data.

- 6. Direct Complex Eigenvalue Analysis, Rigid Format 7
 - At least one subcase must be defined for each unique set of direct input matrices (K2PP, M2PP, B2PP).
 - The CMETHØD card must be used to select the EIGC card from Bulk Data for each set of direct input matrices.
 - Multiple subcases for each set of direct input matrices are used only for output control.
- 7. Direct Frequency and Random Response, Rigid Format 8
 - At least one subcase must be defined for each unique set of direct input matrices (K2PP, 32PP, B2PP) or frequencies and only one FREQUENCY card and one @FREQUENCY card can be used for each such set.
 - Consecutive subcases for each set of direct input matrices or frequencies are used to define thr loading conditions - one subcase for each dynamic loading condition selected from Bulk Data by the DLØAD card.
 - Constraints must be specified directly in Bulk Data (GRID or GRDSET) or selected above the subcase level. Grid point displacements specified with these constraint sets will be ignored.
 - Duput from a Random Response Analysis may be selected only with the XYØUT options (User's Guide, Section 10.4)
- 8. Direct Transient Response, Rigid Format 9
 - One subcase must be defined for each dynamic loading condition selected from Bulk Data by the DLOAD card.
 - Const sints must be specified directly in Bulk Data (GRID or GRIDSET) or specified bove the subcase level. Grid point displacements specified with these con sint sets will be <u>ignored</u>.
- 9. Modal Complex Eigenvilue, Frequency and Random Response, and Transient Response Analyses, Rigid Formats 10, 11 and 12
 - METHØD must appear above the subcase level to select the method of eigenvalue extraction.
 - Constraints must be specified airectly in Bulk Data (GRID or GRDSET) or specified above the subcase level. Grid point displacements specified with these constraint sets will be ignored.
- 10. Mormal Modes with Differential Stiffness, Rigid Format 13
 - The Case Control Deck must contain three subcases.
 - The first subcase is used for output selection for the linear case and to define static loading condition with a LØAD, TEMPERATURE(LØAD), or DEFØRM selection, unless all loading is specified by grid point displacements on SPC cards.
 - DSC0EFFICIENT must appear in the second subcase, either to select 7 DSFACT set from the Bulk Data Deck, or to explicitly select the default value of unity.
 - METHØD must appear above the subcase level to select an EIGR bulk data card.
 - The static differential stiffness solution is output from the second subcase.
 The eigenvector solution is output from the third subcase.
 - An SPC set must be selected above the subcase level unless all constraints are specified on GRID cards.

CASE CONTROL OPTIONS

11. Static Analysis Using Cyclic Symmetry, Rigid Format 14

12

 A separate group of subcases must be defined for each symmetric segment. For dihedral symmetry, a separate group of subcases are defined for each half. in Particular

- The different loading conditions are defined within each group of subcases. The loads on each symmetric segment and the selected output requests may be independent. The number of loading cases is specified on the PARAM card NLØAD.
- The SPC and MPC requests must appear above the subcase level and may not be changed.
- An alternate loading method is to define a separate group of subcases for each harmonic index, k. The parameter CYCID is included and the load components for each index are defined directly within each group for the various loading conditions.
- 12. Normal Modes Analysis Using Cyclic Symmetry, Rigid Format 15
 - METHØD must be used to select an EIGR card that exists in the Bulk Data Deck.
 - An SPC set must be selected unless the model is a free body or all constraints are specified on GRID cards, Scalar Connection cards or with General Elements.
 - Multiple subcases are used only to control output requests. A single subcase is sufficient if the same output is desired for all modes. If multiple subcases are present, the output requests will be honored in succession for increasing mode numbers. MØDES may be used to repeat subcases in order to make the same output requests for several consecutive modes.
- 13. Static Heat Transfer Analysis, Rigid Format 1, APP HEAT
 - A separate subcase must be defined for each unique combination of constraints and static loads.
 - A static loading condition must be defined for (not necessarily within) each subcase with a LØAD selection, unless all loading is specified with grid point temperatures on SPC cards.
 - An SPC set must be selected for (not necessarily within) each subcase, unless all constraints are specified on GRID cards or Scalar Connection cards.
 - Loading conditions associated with the same sets of constraints should be in contiguous subcases, in order to avoid unnecessary looping.
 - REPCASE may be used to repeat subcases in order to allow multiple sets of the same output item.
- 14. Nonlinear Static Heat Transfer Analysis, Rigid Format 3, APP HEAT
 - A single subcase must be defined with a single loading condition (LØAD) and a single constraint condition (SPC).
 - An estimated temperature distribution vector must be defined on TEMP cards and selected with a TEMP(MATERIAL) request. Temperatures for constrained components are taken from these TEMP cards and entries on SPC cards are ignored.

CASE CONTROL DESCRIPTION

- 15. Transient Heat Transfer Analysis, Rigid Format 9, APP HEAT
 - A single subcase must be defined with a single constraint condition.
 - DLØAD and/or NØNLINEAR must be used to define a single time-dependent loading condition. The static load cards (QVECT, QVØL, QHBDY, QBDY1, and QBDY2) can be used to define a dynamic load by using these cards with, or instead of, the DAREA cards. The set identification number on the static load cards (Field 2) is used in the same manner as the set identification number on the DAREA cards.
 - TSTEP must be used to select the time-step intervals to be used for integration and output.
 - If nonzero initial conditions are desired, IC must be used to select a TEMP set in the Bulk Data Deck.
 - An estimated temperature distribution vector must be defined or TEMP cards and selected with a TEMP(MATERIAL) request if radiation effects are included.
- 16. Modal Flutter Analysis, Rigid Format 10, APP AER®
 - Only one subcase is allowed.
 - Desired direct input matrices for stiffness $[K_{pp}^2]$, mass $[M_{pp}^2]$, and damping $[B_{pp}^2]$ must be selected via the keywords K2PP, M2PP, or B2PP.
 - CMETHØD must be used to select an EIGC card from the Bulk Data Deck.
 - FMETHØD must be used to select a FLI'TTER card from the Buik Data Deck.
 - METH@D must be used to select an EIGR card that exists in the Bulk Data Deck.
 - SDAMPING must be used to select a TABDMP1 table if structural damping is desired.
- 17. Modal Aeroelastic Response, Rigid Format 11, APP AERØ
 - METHØD must appear above the subcase level to select eigenvalue extraction method.
 - At least one subcase must be defined for each unique set of direct input matrices (K2PP, M2PP, B2PP) or frequencies.
 - Consecutive subcases for each set of direct input matrices or frequencies are used to define the loading condition - one subcase for each dynamic loading condition.

Ŷ

- Constraints must be defined above the subcase level
- DLØAD must be used to define a frequency-dependent loading condition for each subcase. If transient loads are selected, a Fourier Transform is used to compute frequency-dependent loads. All loads in one run must be of the same type.
- FREQUENCY must be used to select one, and only one, FREQ, FREQ1, or FREQ2 card from the Bulk Data Deck. If TLØADs are selected, a TSTEP must be selected.
- ØFREQUENCY(ØTIME) may be used above the subcase level or within each subcase to select a subset of the solution frequencies (times) for output requests. The default is to use all solution frequencies (times).
- If Random Response calculations are desired, RANDØM must be used to select RANDPS and RANDTi cards from the Bulk Data Deck. Only one ØFREQUENCY and FREQUENCY card can be used for each set of direct input matrices.

Table 1. Case Control Options

G

BULK DATA SELECTION

6.5

K.

1

į

Loads:	<u>Constraints</u> :	Direct Input Matrices:	Dynamic Analyses <u>Conditions</u> :	Thermal Fields:
DEFORM DLØAD DSCØEFFICIENT	AXISYMMETRIC MPC SPC	82PP K2PP M2PP TEI	CMETHØD FREQUENCY IC METHØD	TEMPERATURE TEMPERATURE(LØAD) TEMPERATURE(MATERIAL)
LØAD NONLINEAR PLCØEFFICIENT			RANDOM SDAMPING	Aeroelasticity:
			15164	FMETHØD Gust
OUTPUT SELECTION Output Control: ECHØ LABEL LINE MAXLINES ØUTPUT ØUTPUT(PLØT) ØUTPUT(XYØUT) ØUTPUT(XYØUT) SUBTITLE TITLE	Specific Output <u>Amounts:</u> HARMØNICS MØDES ØFREQUENCY SET	SØLUTIØN Set Response: NLLØAD SACCELERATIØN SDISPLACEMENT SVECTØR SVELØCITY	Stress, Force, Degree-of- Freedom Response: ACCELERATIØN DISPLACEMENT ELFØRCE ELSTRESS ESE FØRCE GPFØRCE NCHECK ØLØAD PRESSURE SPCFØRCES STRESS THERMAL VECTØR VECTØR	<u>Aeroelasticity:</u> AERØF

SUBCASE SPECIFICATION

General Case:	Symmetric <u>Case:</u>	Combination Coefficients	
SUBCASE	SYM	General:	Symmetric:
REPCASE SUBCØM	SYMCOM	SUBSEQ	SYMSEQ
CASE C TROL DESCRIPTION

Table 2. Case Control - Subcase Specification

- SUBCASE Defines the beginning of a subcase that is terminated by the next subcase delimiters encountered.
- SUBCØM Defines a combination of two or more immediately preceding subcases in static or inertia relief problems only. Output requests above the subcase level are used.
- SUBSEQ Must appear in a subcase defined by SUBCØM to give the coefficients for making the linear combination of the preceding subcases. The first coefficient applies to the first subcase, the second to the next subcase, etc., for all subcases except combination subcases.
- SYM Defines a subcase in static or inertia relief problems for which only output requests within the subcase will be honored. Primarily for use with symmetry problems where the individual parts of the solution may not be of interest.
- SYMCGM Defines a combination of two or more immediately preceding SYM subcases in static or inertia relief problems only. Output requests above the subcase level are used.
- SYMESEQ May appear in a subcase defined by SYMCØM to give the coefficient, one for each subcase, for making the linear combination of the preceding SYM subcases. A default value of 1.0 is used if no SYMSEQ card appears, otherwise, a coefficient must be specified for all subcases except combination subcases.
- REPCASE Defines a subcase in static or inertia relief problems only that is used to make additional output requests for the previous real subcase. This card is required because multiple output requests for the same item are not permitted in the same subcase.
- MØDES Repeats the subcase in which it appears MØDES times for eigenvalue problems. Used to repeat the same output request for several consecutive modes.

0

なるというというないの

8.4 EXAMPLES OF CASE CONTROL DECKS

The following are examples of commonly employed Case Control Deck arrangements. Additional examples can be found in Sections 18 through 26, as they apply to specific problems. The notes that are provided are intended to highlight the particular intent of the current example.

Details already explained in preceding examples are not repeated.

8.4.1 Example 1 - Multiple Static Loadings

Card No.

1

1	TITLE = STATIC ANALYSIS OF DOMOR FRAME
2	SUBTITLE = CLØSED PØSITIØN
3	ØUTPUT
4	DISPLACEMENTS = ALL
5	MPC = 3
6	SPC = 15
7	SET 1 = 256 THRU 289
8	SET 5 = 15 THRU 20, 23, 25, 101 THRU 123
9	SUBCASE 1
10	LABEL = DECOMPRESSION
11	LØAD = 561
12	ØLØAD = 1
13	SUBCASE 2
14	LABEL = REENTRY THERMAL
15	TEMPERATURE (LØAD) = 761
16	STRESS = ALL
17	SUBCASE 3
18	LABEL = IMPACT
19	SPC = 25
20	LØAD = 1061
21	STRESS = 5
22	EEGIN BULK

8.4-1

and the second second

Notes:

-41-5 A. F. T. M. P. C.

1-2. Identified basic problem and special conditions

MIRES Y+10 - ----

- 3-4. Overall output request for all gridpoint displacements
 - 5. Selects multipoint constraint set from Bulk Data Deck
 - 6. Selects single point constraint set from Bulk Data Deck
- 9-10. Defines sets of grid points and elements for selective output to be specified later
- 11-14. First load subcase, label information, loading selected from Bulk Data and special output request for applied loads on grid point set #1.
- 15-18. Second load subcase, label information, thermal load set selection and special output request for stresses for all elements.
- 19-21. Third load subcase, label information, change in single point constraint set, load selection and limited stress output for elements defined by set #5.
 - 22. Signals beginning of Bulk Data Deck co follow

8.4.2 Example 2 - Linear Combination of Subcases

Card No.

2

1 TITLE - BUILDING ASG637-111-PG 2 \$ SET 1 GROUND FLOOR AND FIRST FLOOR BEAMS 3 SET = 1 THRU 25, 51 THRU 75, 4 101 THRU 125, 151 THRU 175 5 \$ SET 2 SUPPORT GRID POINTS 6 SET 2 = 1 THRU 26 7 DISPLACEMENTS = ALL 8 ELFØRCE = 1 9 **ØLØAD = ALL** 10 SPCFØRCE = 2 11 SUBCASE 10 12 ELFORCE = NONE 13 DEFORM = 10 SPC = 1014 SUBCASE 20 15 ELFORCE - NONE 16

8.4-2

EXAMPLES OF CASE CONTROL DECKS

an and the state of the state o

Card No.

- 17 SPC = 20
- 18 SUBCON 31
- 19 LABEL = PRETENSION PLUS TWICE SUPPORT SETTLEMENT
- 20 SUBSEQ = 1.0, 2.0
- 21 SUBCOM 32
- 22 LABEL = PRETENSION MINUS TWICE SUPPORT SETTLEMENT
- 23 SUBSEQ = 1.0, -2.0
- 24 BEGIN BULK
- Notes: 4,7. Comment card to facilitate interpretation of set definition.
 - 12. Output request for support reactions on single point constraints and printout of gridpoint loading vector.
 - 13-15. First subcase, element deformation loading, suppression of element force output request above subcase level, and selection of single point constraint set.
 - 17-19. Second subcase, enforced displacements at support specified by single point constraint set and suppression of element force output. A null load vector will be output due to absense of direct applied gridpoint loads.
 - 20-22. Combination subcase adding two previous subcases, label information and coefficients for combination with output for displacements, element forces, reactions and applied loads.
 - 23-25. Combination subcase taking subcase 10 and subtraction twice subcase 20. Same output with new label.

Special note should be made of the fact that NASTRAN applies the combination coefficients to the element stress and force results <u>before</u> they are corrected for the DEFBRM or TEMP(LBAD) loading effects. These corrective effects are <u>not</u> multiplied by the combination coefficient when they are added to the combination loading results.

8.4.3 Example 3 - Statics Problem with Symmetry

Card No.

 1
 TITLE = GROUND LEVEL STORAGE TANK

 2
 SUBTITLE = NON-SYMMETRIC LOADS

 3
 SET 1 = 1, 11, 21, 31, 51

 4
 SET 2 = THRU 10, 101 THRU 110

CASE CONTROL DESCRIPTION

Card No. 5

DISPLACEMENTS = ALL

where growthe end of the

- 6 ELSTRESS = 2
- 7 ØLØADS = 1
- 8 SYM 1
- 9 SPC = 11
- 10 LØAD = 21
- 11 SYM 2
- 12 SPC = 12
- 13 LØAD = 22
- 14 SYMEOM 3
- 15 SYMCOM 4
- 16 SYMSEQ 1.0, -1.0
- 17 BEGIN BULK

Notes: 10-15. Definition of symmetric and anti-symmetric loads and support conditions.

16. Symmetric summation of subcases 1 and 2.

17-18. Anti-symmetric combination of subcases 1 and 2. The output selected will be printed for all four subcases.

8.4.4 Example 4 - Use of REPCASE in Static Analysis

Card No.

1 TITLE = FRAMING FOR BLDG 673-7A 2 SUBTITLE = HEAVY EQUIPMENT LØADING 3 SET 1 . 1 THRU 10, 101 THRU 110, 201 THRU 210 SET 2 = 21 THRU 30, 121 THRU 130, 221 THRU 230 4 SET 3 - 41 THRU 100 EXCEPT 55, 57, 59, 61, 65, 72, 78, 95 5 SUBCASE 1 6 1049 = 10 7 SPC = 11 8 9 DISP # ALL ELFORCE - 1 10

8.4-4

EXAMPLES OF CASE CONTROL DECKS

g - 1.75 all all the same against

Card No.

11 REPCASE 2

Concern the second

- 12 ELFORCE = 2
- 13 REPCASE 3
- 14 ELFORCE = 13
- 15 BEGIN BULK

Notes: 8-11. Specifies load and support conditions to be analyzed and requests output for displacements at all grid points and element forces for the elements in set #1.

12-15. Specified two subcases for output control for printing element forces for sets #2 and #3.

8.4.5 Example 5 - Use of MBDES and PUNCH Output Options

Card No.

- 1 TITLE = ØXTANK SUPPORT SYSTEM WITH CONSTRAINED MODES
- 2 LINE = 40
- 3 METHOD = 2
- 4 SPC = 10
- 5 SUBCASE 1
- 6 DISP(PUNCH) = ALL
- 7 STRESS = ALL
- 8 MØDES = 2
- 9 SUBCASE 3
- 10 DISP = ALL
- 11 BEGIN BULK

Notes:

Specified 40 lines maximum per page of printout.

- 5. Selects EIGR card from Bulk Data for specification of method for eigenvalue extraction.
- 7-10. Specified displacements be punched and element stresses be printed for the first two modes.
- 11-12. Specified displacements be printed for all remaining modes.

CASE CONTROL DESCRIPTION

8.4.6 Example 6 - Use of SORT2 Output Option

an energy and a second second and the second s

<u>Card No</u>.

1 TITLE = TRANSIENT ANALYSIS OF AXLE

2 SUBTITLE = TRAVELLING WAVE

- 3 TSTEP = 9
- 4 IC ≈ 9
- 5 SET 1 = 2, 4, 6, 8, 10, 12, 14, 16, 18, 20
- 6 **PUTPUT**
- 7 DISP(SQRT2) = 1
- 8 VELØ(SØRT2) = 1
- 9 BEGIN BULK

Notes: 5. Selects timestep definition from Bulk Data.

- 6. Selects initial conditions from Bulk Data.
- 8-10. Specifies output requests for displacements and velocities for all output time steps (SØRT2) at each grid point selected by set #1. The excitation is produced by initial conditions only.

Special note should be taken that there is only one subcase here and, therefore, it need not be explicitely defined.

8.4.7 Example 7 - Use of XYOUT with Frequency and Random Response

Card No.

1	TITLE = DIRECT RANDOM RESPONSE TO PSD102-A
2	SUBTITLE = CABINET 5A
3	SPC = 11
4	DLØAD = 102
5	FREQ = 101
6	RANDØM = 2
7	B UTPUT
8	SET 1 = 1 THRU 10, 101 THRU 110, 120, 130, 150
9	SET 2 = 0.1, 0.3, 0.6, 1.2, 1.8, 2.4, 3.0
10	ØFREQ = 2

8.4-6

EXAMPLES OF CASE CONTROL DECKS

Card No.

- 11 ACCEL(SØRT2, PHASE) = 1
- 12 SUBCASE 15
- 13 LABEL = DAMPING SOURCE TOP
- 14 B2PP = TIEDAMP
- 15 **PUTPUT(XYPUT)**
- 16 TCURVE = PSD FOR TOP DAMPING
- 17 XYPRINT ACCE PSDF 15/1(T1)/1(T2)/101(R3)
- 18 BEGIN BULK
- Notes: 5-8. Specifies environment for the analysis including support conditions. frequency dependent dynamic load, the set of frequencies for solution and the power spectral density function.
 - 12. Specifies frequency set #2 for outputting results.
 - 13. Requests the acceleration for all output frequencies for each grid point of set #1 in magnitude/phase angle format.
 - 16. Specifies the direct input damping matrix.
 - 19. Requests output from the Random Response analysis consisting of the power spectral density function for the first subcase (#15) at gridpoint 1 (X-acceleration), gridpoint 1 (Y-acceleration) and gridpoint 101 (Z-angular acceleration).

8.4-7

9.1 INTRODUCTION

The primary NASTRAN input medium for definition of the data required for a structural analysis is the Bulk Data Deck. The cards in this deck are used to define the structural model and the various other sets of data used to specify the environment to which the model is subjected for analysis. Also, Bulk Data cards may be used to specify the parameters and limits required to control the analysis. A brief summary of the available options are presented in Table 1. An alphabetically ordered, detailed description of all Bulk Data cards is found in the User's Manual, Section 2.4.

In general, all the data submitted describing the physical properties of the model are used directly by NASTRAN. Those data used to specify conditions of the environment such as constraints, temperatures, and loadings must be selected via Case Control options if they are to be applied to the model. Methods of analysis, including the associated data and limits must also be selected from Bulk Data via the Case Control options. Because these last two categories of information are fully under the control of the user, the Bulk Data Deck may include several sets of these data at once. The "subcase" structure of the Case Control deck provides the user with the facility to define each unique combination of these data as separate subcases to be analyzed since more than one subcase may be solved in a single run with NASTRAN.

For large problems, the Bulk Data Deck may consist of several thousand cards. In order to minimize the handling of large numbers of cards, provision has been made in NASTRAN to alphabetically sort and store the Bulk Data on a checkpoint tape and/or on a User's Master File tape (Section 9.4). Data can then be added or selectively deleted during subsequent runs.

The Bulk Data Deck may be submitted with the cards in any order as a sort is performed prior to the execution of an analysis. It should be noted that the machine time to perform this sort is minimized for a deck that is already sorted. The sort time for a badly sorted deck will become significant for large decks. The user may obtain a printed copy of the unsorted and/or the sorted bulk data by selection in the Case Control Deck (User's Manual, Section 2.3, ECHD). A sorted listing is necessary in order to make modifications during a subsequent execution from a checkpoint/restart. This listing is automatically provided as an echo of the Bulk Data Deck unless specifically suppressed by the user.

9.1-1

سابه جمعانيم يار

. 1

Input data preparation for large finite elements models is very often the largest single cost in an analysis, as well as a major source of error. Automated data generation is one means for reducing both cost and input error through the use of design oriented models that cover more of the structure than a single finite element. NASTRAN provides a specialized utility module, called INPUT, to create the majority of Bulk Data cards required for seven selected kinds of problems. This automated data generation is described in Section 9.5.

To examine the use of Bulk Data Cards, refer to the Demonstration Problem Hanual which contains an extensive index to problems illustrating application of each Bulk Data Card option.

9.2 PROBLEM DEFINITION

NASTRAN embodies a lumped element approach, wherein the distributed physical properties of a structure are represented by a model consisting of a finite number of idealized substructures or elements that are interconnected at a finite number of grid points, to which loads are applied. All input and output data pertain to this idealized structural model.

The grid point definition forms the basic framework for the structural model. All other parts of the structural model are referenced either directly or indirectly to the grid points.

Three general types of grid points are used in defining the structural model. They are:

- Geometric grid point A point in three-dimensional space at which three components of translation and three components of rotation are defined. The coordinates of each grid point are specified by the user.
- Scalar point A point in vector space to which one degree of freedom is assigned. Scalar points can be coupled to geometric grid points by means of scalar elements and by constraint relationships.
- 3. For axisymmetric analyses, the geometric grid point becomes a ring--a point in space rotated about the structural axis of symmetry.

The structural element is a convenient means for specifying many of the properties of the structure, including material properties, mass distribution and some types of applied loads. In static analysis by the displacement method, stiffness properties are input by means of structural elements, scalar stiffness elements, or stiffness matrices. Mass properties (used in the generation of gravity and inertia loads) are input either as properties of structural elements or as properties of grid points. In dynamic analysis, mass, damping, and stiffness properties may be input either as the properties of structural elements or as the properties of grid points (direct input matrices).

Structural elements are defined on connection cards by referencing grid points. In a few cases, all of the information required to generate the structural matrices for the element is given on the connection card. In most cases the connection card refers to a property card, on which the cross-sectional properties of the element are given. The property card in turn refers to a material card which gives the material properties. If some of the material properties are stress-dependent or temperature-deptrident, a further reference is made to tables for this information.

Various kinds of constraints can be applied to the grid points. Single-point constraints are used to specify boundary conditions, including enforced displacements of grid points. Multipoint constraints are used to specify a linear relationship among selected degrees of freedom, including the definition of infinitely rigid elements. Additionally, rigid elements are available for automatic generation of multipoint constraint equations. Omitted points are used to partition matrices and to reduce the number of degrees of freedom used in the analysis. Free-body supports are used to remove stress-free motions in static analysis and to evaluate the free-body inertia properties of the structural model.

Static loads may be applied to the structural model by concentrated loads at grid points, pressure loads on surfaces, or indirectly, by means of the mass and thermal expansion properties of structural elements and enforced deformations of one-dimensional structural elements. Due to the great variety of possible sources for dynamic loading, general functional and tabular forms of loads are provided to the user in dynamic analysis.

General procedures for defining structural models are described in the User's Manual, Section 1. Some useful and specific ideas for modeling using NASTRAN are presented in Sections 18, 19 and 21-26. The user is encouraged to become familiar with the many suggestions and guidelines presented in these sections. They include how to use the special coordinate systems, how to number grid points to avoid inefficient processing, what elements can be used for which purpose, how to load and support a structure, and how to exploit the unusual modeling features of NASTRAN. Additional technical detail, of course, can be found throughout the Theoretical Manual.

NASTRAN provides features for automatic Bulk Data generation. A description of the capability is found in Section 9.5.

9.2-2

OPTIONS AND FORMAT

and the second secon

ORIGINAL PAGE IS OF POOR QUALITY

9.3 OPTIONS AND FORMAT

There are nearly two hundred Bulk Data options currently available in NASTRAN. Because there are so many and varied alternatives, these options are summarized in the following Table 1, where they are organized according to data types for easy reference by the user. A scan of these options will provide the user with an overview of his alternatives for model specification, for load definition and for control of the execution. In the User's Manual, Section 2, each option is listed in alphabetical sort by Bulk Data Card mnemonic. Here, the user will find the full description of the input card format, the applicable restrictions, and the references to other Bulk Data Cards required to complete the input. Here also, the user will find which Case Control option must be used, if any, to selectively apply these data to the problem at hand.

Each category of data found in Table 1 includes references to the standard NASTRAN manuals for the related theory and application of the input shown. The order of the categories was selected to follow a natural sequence for model development. As such, a user may use these tables as a checklist to be assured that he has considered every aspect of his input.

It should be noted that some options appear in more than one category. This is done to help avoid overlooking an option that may serve several purposes.

Specific procedures and the formats for Bulk Data input are clearly spelled out in the User's Manual, Section 2.4. Examples of many of the Bulk Data options can be found throughout this User's Guide where actual problems for each of the standard and special analysis capabilities of NASTRAN are illustrated and explained.

To summarize, each Bulk Data Card is divided into ten fields; each eight columns wide. The first field contains either the card mnemonic or the continuation code from the preceding card if more than one card is required for definition of the data. These continuation codes must be unique in order to facilitate the sorting of the Bulk Data Deck. The last seven columns (74-80) of field 10 on each card are used only for specification of that unique continuation code. The card mnemonic always starts in card column 1. The continuation code from the preceding card starts in card column 2 with a + in card column 1. Each of the remaining data entries to be supplied may be placed without imbedded blanks anywhere within the appropriate field. These entries may be integer, real, or alphanumeric data.

9.3-1

However, occasionally more than eight-card columns are required for the user to specify his data with sufficient accuracy. In these situations, each Bulk Data Card may be expanded to cover two separate cards with 16-column fields for the data entries (fields 2-9). When this arrangement is used, an * must appear following the card mnemonic and/or in card column 1 of the continuation card.

The second s

and the second

and a starter

The type of data to be expected in each field is checked by NASTRAN on input, and a message is issued in the event of an error. To help detect possible modeling errors, NASTRAN also performs many other checks as the input data is used by the program.

USER MASTER FILE

9.4 USER MASTER FILE

NASTRAN provides the User's Master File (UMF) capability to assist the user in handling large. (several boxes of cards) Bulk Data Decks which are often typical for NASTRAN problems. This capability is completely described with examples in the User's Manual, Section 2.5. The capability is briefly summarized here.

Prior to the first run of NASTRAN for a new model, the entire Bulk Data Deck may be submitted to create a User's Master File. On a subsequent run, the user may create either a User's Master File or a Standard checkpoint to a tape called the New Problem Tape (NPTP) on which will be stored the Bulk Data Deck in sorted order. Any other subsequent runs can then be made with the old problem tape as a source for that data deck. When a run is a restart from a checkpoint, the old checkpoint tape, now called the Old Problem Tape (@PTP), must be provided. Any Bulk Data cards submitted with the restart run will be added to the old Bulk Data Deck for access during processing. If a checkpoint for this run is requested, another New Problem Tape will be written on which will be stored the new, augmented Bulk Data Deck. The UMF and @PTP may not be used at the same time, since the bulk data would be duplicated.

Thus, a user can add Bulk Data cards with each subsequent run, and he can also delete cards by inserting Bulk Data cards with a / in column one, followed by one or two of the sequence numbers assigned to each card of the sorted Bulk Data Deck from the previous run. All cards in the range of these sequence numbers will be deleted. If only one number is input, only that Bulk Data card is removed. These features are also available for editing and updating the User's Master File.

AUTOMATED DATA GENERATION

and the superscription of the Statist States and

ensors the part of

9.5 AUTOMATED DATA GENERATION

ì

9.5.1 Utility Module INPUT Usage

NASTRAN provides a special utility module, INPUT, for creating the majority of Bulk Data cards for seven selected kinds of problems. This capability is fully described in Section 2.6 of the User's Manual, where examples of its use are found. A summary of the key INPUT features is provided here:

- INPUT allows the user of NASTRAN to generate the majority of the bulk data cards for a number of selected test problems without having to actually input the physical cards into the Bulk Data Deck.
- 2. The test problems for which partial data are generated by INPUT are:
 - a. N x N Laplace Network from scalar elements
 - b. W x L Rectangular Frame from BAR elements or R@O elements
 - c. W x L Rectangular Array of QUAD1 elements
 - d. W x L Rectangular Array of TRIAl elements
 - e. N segment string from scalar elements
 - f. N cell beam made from BAR elements
 - g. N scalar point full matrix with optional unit loading
 - h. N spoke wheel made from BAR elements

Table 1. Bulk Data Options

A. Spatial Geometry

Axisymmetric Approach (UM 1.3.7, 1.7.1, TM 4.1, 5.9)

AXIC Required to define existence of conical shell problem

والمراجع والمراجع

AXIF Required to define existence of fluid coordinate system, includes default parameters

AXSLOT Required to define existence of acoustic slot analysis, includes default parameters.

PPINTAX Location on shell ring (conical shell) at which loads can be applied and displacements are requested.

RINGAX Ring for conical shell.

a a segur pro- and early proceeding

SECTAX Sector of a conical shell.

Coordinate Systems (UM 1.2.2, TM 3.4.1)

BARBR · Simple beam orientation and property default for CBAR.

- CORDIC Cylindrical coordinate system by reference to (i=1, three grid points, and i=2, coordinates of three points).
- C@RDiR Rectangular coordinate system by reference to (i=1, three grid points, and i=2, coordinates of three points).
- CØRDIS Spherical coordinate system by reference to (i=1, three grid points, and i=2, coordinates of three points).
- GTRAN Redefines output coordinate system displacement sets for substructuring.

TRANS Transformation of basic coordinate systems for substructuring.

Fluid Points (UM 1.7.1, TM 16.1.1)

FREEPT	Fluid surface points for recovery of displacements.			
FSLIST	Declares fluid points (RINGFL) which lie on free surface toundary.			
GRIDB.	Grid location on RINGFL for structure boundary.			
GRIDF	Scalar degree of freedom for acoustic analysis of a fluid.			
GRIDS	Scalar degree of freedom on acoustic slots.			
PRESPT	Pressure points for data recovery in fluid analyses.			
RINGFL	Circle (fluid point) in fluid model.			
SLBDY	List of slot points (GRIDS) on interface between fluid and radial slots.			

AUTOMATED DATA GENERATION

Table 1. Bulk Data Options (cont'd)

100 - 1 100 Eur 100 - 20 20 20 20 100 100 100 100 100 100

A. Spatial Geometry (cont'd)

<u>Grid Points</u> (UM 1.1, 1.2, TM 3.1)

GRID	Grid point location, direction of displacement and constraints.
GRIDB	Alternate GRID definition for hydroelastic analysis.

GRDSET Default options for all GRID cards.

SEQGP Grid and scalar point number resequencing.

Scalar Points (UM 1.1, TM 3.1, 14.2)

EPØINT	Extra points defined for dynamics.		
SEQEP	Extra point number resequencing.		
GRIDF	Scalar degree of freedom for acoustic fluid analysis.		
GRIDS	Scalar degree of freedom on acoustic slot boundary.		
SPØINT	Scalar points defined.		
SEQGP	Grid and scalar point number resequencing.		

Aerodynamics (UM 1.11, TM 17.0)

AEFACT Aerodynamic Spanwise Divisions.

AER® Aerodynamic Physical Data.

B. Element and Properties

Axisymmetric (UM 1.3.7, 1.7.1, TM 4.1, 5.9)

AXIC Required to define existence of conical shell problems.

CCONEAX Concial shell element connection.

PCONEAX Concial shell property.

CTORDRG Toroidal ring element condition.

PT@RDRG Toroidal ring property.

CTRAPRG Trapezoidal ring element connection and property - axisymmetric response.

CTRIARG Triangular ring element connection and property - axisymmetric response.

DMIAX Direct matrix input for axisymmetric problems.

CTRAPAX Trapezoidal ring element connection and property - nonaxisymmetric response.

CTRIAAX Triangular ring element connection and property - nonaxisymmetric response.

Table 1. Bulk Data Options (cont'd)

B. Element and Properties (cont'd)

Dummy (UM 1.3.1, TM 5.1)

ADUMi Atrributes for DCUMi AND PDUMi dummy elements (for user-defined element types i=1 thru 9).

CDUMi Dummy element connection.

PDUMi Dummy element property.

PLOTEL Dummy element definition (for plotting only).

Fluid (UM 1.7.1, TM 16.1.4)

AXIF Required to define existence of fluid analysis and default parameters.

AXSLØT Required to define existence of axisymmetric slot analysis and default parameters.

CAXFi Fluid element connection i=2, 3 or 4 fluid points in an acoustic slot enalysis.

CFLUIDi Fluid element connection for axisymmetric fluid (for user-defined fluid points i=2, 3 or 4).

CSLØTi Slot element connecting i=3 or i=4 GRIDS points for evenly spaced radial slots.

General (UM 1.3.1, TM 5.7)

CNGRNT Indicates identical elements to another specified primary element.

GENEL General element in terms of its transformation matrices.

CGNNi Concentrated mass element connection (i=1 for mass matrix, or i=2 for offset mass inertia properties).

Line (UM 1.3.2, TM 5.2)

CBAR	Simple beam element connection.		
PBAR	Simple beam property.		
BAROR	Simple beam orientation default for CBAR.		
CONROD	Rod element property and connection.		
CRØD	Rod tension-compression-torsion element connection.		
PR B D	Rod element property.		
CTUBE	Tube tension-compression-torsion element connection.		
PTUBE	Tube element property.		
CVISC	Viscous damping connection, extensional and rotational.		
PV ISC	Viscous damper property, extensional and rotational.		

AUTOMATED DATA GENERATION

Table 1. Bulk Data Options (cont'd)

B. Element and Properties (cont'd)

Plate (UM 1.3.5, TM 5.8)

1

CODNEN	Quadrilateral membrane element connection.
PODNEM	Guadrilateral membrane element property.
COOPLT	Quadrilateral bending element connection.
PQDPLT	Quadrilateral bending element property.
CQUAD1	Quadrilateral membrane and bending element connection (i=1 for general and i=2 for homogeneous behavior).
PQUADI	Quadrilateral membrane and bending element property (i=1 for general and i=2 for homogeneous properties).
CSHEAR	Shear panel element connection.
PSHEAR	Shear panel property.
CTRBSC	Basic triangular bending element connection.
PTRBSC	Basic triangular bending element property.
CTRPLT1	Higher order triangular plate element connection.
PTRPLTI	Higher order triangular plate element property.
CTRIA	Triangular membrane and bending element connection (i=) for general and i=2 for homogeneous behavior).
PTRIAI	Triangular membrane and bending element property (i=1 for general and i=2 for homogeneous properties).
CTRMEM	Triangular membrane element connection.
PTRMEM	Triangular membrane element connection.
CTRIM5	Higher order triangular membrane element connection.
PTRIM6	Higher order triangular membrane element property.
CTRPLT	Triangular bending element (TRPLT) connection.
PTRPLT	Triangular bending element (TRPLT) property.
CTWEST	Twist panel element connection.
PTWIST	Twist panel property.

.. .

Table 1. Bulk Data Options (cont'd)

B. Element and Properties (cont'd)

....

..... **1**

= = \

Rigid Connectors (UM 1.4.2, TM 3.5.1)

MPC	Equations of constraint which may be used to simulate a rigid element.
MPCS	Multipoint constraints for substructuring.
MPCADD	Union of multipoint constraint sets defined by MPC cards.
CONCT	Substructuring connectivities
CONCTI	Substructuring connectivities
CRIGDR	Rigid rod element.
CRIGDI	Rigid element for 6 DOF.
CRIGD2	Rigid element for selected DOF.
CR1GD3	General rigid element.
CYJØIN	Connectivities for cyclic symmetry models.
RELES	Releases rigid connections for substructuring.

<u>Scalar</u> (UN 1.3.8, TH 5.6)

CDAMPi	Scalar viscous damper connection (i=2 or 4 without reference to property).
PDAMP	Scalar viscous damper property.
CELASI	Scalar spring connection (i=2 or 4 without reference to property).
PELAS	Scalar spring property.
CHASSI	Scalar mass connection (i=2 or 4 without reference to property).
PMASS	Scalar mass property.
CBNHI	Concentrated mass element connection (i=1 for mass matrix, or i=2 for offset mass inertia properties).
CVISC	Viscous damping connection, extensional and rotational.
PVISC	Viscous damper property, extensional and rotational.

<u>Solid</u> (UH 1.3.9, TH 5.12)

- CHEXA: Hexahedron constant strain element connection (1=1 uses 5 tetrahedra, 1=2 uses 10 overlapping tetrahedra).
- CIMEXi Isoparametric solid hexahedron element connections (i=1 for linear, i=2 for quadratic, and i=3 for cubic functions).

AUTOMATED DATA GENERATION

and the contract of the second restriction and the second restriction of the second second second second second

Table 1. Bulk Data Options (cont'd)

B. Element and Properties (cont'd)

Solid (cont'd)

PIHEX	Isoparametric	solid	hexahedron	element	property.
-------	---------------	-------	------------	---------	-----------

CTETRA Tetrahedron constant strain element connection.

CWEDGE Wedge constant strain element connection.

Surface Heat (UM 1.8.3, TM 8.3)

CHBDY Heat convection boundary for steady-state heat condition.

C. Materials

Anistropic (UM 1.3.1, TM 4.2)

MAT2	Anisotropic material properties.
MATT2	Temperature-dependent table references for MAT2 properties.
MAT3	Orthotropic material properties.
MATT3	Temperature-dependent table references for MAT3 properties.
MAT5	Anisotropic thermal material properties.
MATT5	Temperature-dependent table references for thermal conductivity matrix (MAT5)

Fluid (UM 1.7, TM 16.1.1)

AXIF Defines default parameters and existence of fluid analysis.

AXSLOT Default parameters and the existence of axisymmetric slot analysis.

- BDYLIST Fluid mass density at fluid boundary.
- CFLUIDi Fluid mass density and bulk modulus.

CSLOTi Slot fluid element mass density and bulk modulus.

FSLIST Fluid mass density at free surface.

SLBOY Fluid mass density at boundary between axisymmetric fluid and evenly spaced radial slots.

Isotropic (UM 1.3.1, TM 4.2)

- MAT1 Isotropic material properties.
- MATT1 Temperature-dependent table references for MAT1 properties.

MATTA Temperature-dependent table reference for thermal conductivity or convective film coefficient (MAT4).

Table 1. Bulk Data Options (cont'd)

C. Materials (cont'd)

Stress Dependent (UM 1.3.1, TM 3.8.1)

MATSI Stress-dependent table references for MATI properties.

TABLES1 Tabular stress-strain function referenced by MATS1.

Temperature Dependent (UM 1.8.3)

- MATT: Table references for MAT: type material and thermal properties (i=1, 4 for isotropic, i=2, 3, 5 for anisotropic).
- TABLEMiTabular function for generating material properties and parameter data (i=1thru 4 for specific algorithm)
- TEMP Grid point temperature field.
- TEMPD Grid point temperature field default.

TEMPAX Temperature sets for conical shell problem.

TEMPPi Plate element temperature field (i=1 includes gradient, i=2 includes thermal moments, and i=3 includes tabular description over cross section).

TEMPRB Temperature field for line elements.

D. Constraints and Partitioning

Fluid Boundary (UM 1.7.2, TM 16.1.3)

- BDYLIST Defines boundary (RINGFL points) between fluid and structure.
- FLSYM Axisymmetric symmetry control.
- FSLIST Required to define fluid points which lie on free surface.

SLBDY List of slot points (GRIDS) between fluid and radial slots.

Heat Boundary (UM 1.8.3, TM 8.3)

CHBDY Heat convection boundary element for steady-state heat condition.

Hultipoint Constraints (UM 1.4.2, TM 3.5.1)

- MPC Defines linearly dependent constraint relations between displacements.
- MPCADD Union of multipoint constraint sets defined by MPC cards.

MPCAX Required to define MPC-type relations for conical shell coordinates.

AUTUMATED DATA GENERATION

THE PARTY IS WERE THE PARTY PARTY

Table 1. Bulk Data Options (cont'd)

D. Constraints and Partitioning (cont'd)

1

÷.

Partitioning (UM 1.4.4, TM 3.5.3)

ACET	Defines independent degrees of freedom to be placed in analysis set.
ACET 1	Defines independent degrees of freedom to be placed in analyses set.
RDYC	Defines boundary sets with substructuring
BDYSI	Defines boundary sets with substructuring.
MIT	Defines degrees of freedom omitted from analysis set (reduces independent degrees of freedom).
gmIT1	Defines degrees of freedom omitted from analysis set (reduces independent degrees of freedom).
G MITAX	Omitteu conical shell degrees of freedom.
<u>Rigid Boc</u>	iy Motion Constraints (UM 1.4.3, TM 3.5.5)
SUPAX	Concial shell fictitious supports for determinate reactions for free body analysis.
SUPØRT	Fictitious supports for determinate reactions applied to free body analysis.
Single-P	oint Constraints (UM 1.4.1, TM 3.5.2)
GRID	Grid point location, direction of displacement and constraints.
GRIDB	Grid location on RINGFL for fluid boundary and constraints.
GRDSET	Default options for all GRID cards.
SPC	Defines single-point constraints and enforced displacements.
SPC1	Defines sets of single-point constraints.
SPCADD	Union of singular point constraint sets defined by SPC and SPC1 cards.

SPCAX Defines single-point constraints for conical shell coordinates.

SPCS Defines single-point constraints with substructuring.

SPCS) Defines single-point constraints with substructuring.

SPCD Defines enforced displacements.

SPCSD Defines enforced displacements with substructuring.

Table 1. Bulk Data Options (cont'd)

E. Loads

a line to a second

Combinations (UM 1.5.1, TM 11.1)

DLGAD Linear combination of dynamic load sets.

LGAD Linear combination of static load sets (except temperature and displacement loads).

Linear combination of load sets with substructuring.

(1999),1991年,1993年,1993年,1993年(1993年)(1993年)(1994年)(1994年)(1994年)(1994年)(1994年)(1994年)(1994年)(1994年)(1994年)(19

Dynamic (UM 1.52, 1.53, TM 11.1, 12.1)

DAREA	Dynamic load coole contraction
	Syndamic road scale factor and grid point location.
DELAY	Dynamic load function time delay and grid point location
DLØAD	Linear combination of dynamic load sets.
DPHASE	Dynamic load function phase lead term 0 (theta).
NONLINI	Nonlinear displacement dependent transient forcing functions (i=1 thru 4 defines function type).
PRESPT	Fluid pressure point locations for output and control devices
RANDPS	Frequency dependent self and cross-power crosses devices.
RANDTI	Autocorrelation function time lag for random analysis
TABRNDI	Tabular function of power spectral density versus frequency for months
RLØADI	Frequency dependent dynamic loads for frequency response (i=1, 2 defines function type).
TABLEDI	Tabular function for generating frequency or time-dependent loads (1=1 thru 4 for specific algorithm).
TLMADI	Time-dependent dynamic load for transient response (i=1, 2 for four to the
TSTEP	Time step intervals for solution and output in transform card in the
DAREAS	Dynamic load scale factor with substructuring
DELAYS	Dynamic load function time delay with substructuring
DPHASES	Dynamic load function phase load to substructuring,

DPHASES Dynamic load function phase lead term with substructuring.

AUTOMATED DATA GENERATION

warming and the second second second

Table 1. Bulk Data Options (cont'd)

E. Loads (cont'd)

Heat Transfer (UM 1.5, TM 8.0)

QHBDY	Thermal load at boundary for steady-state heat conduction.
Q8DY1	Uniform heat flux into HBDY elements.
QBDY2	Distributed heat flux into HBDY elements.
QVECT	Vector heat flux into HBDY elements.
QVØL	Heat flux per volume generated by elements.
PLSAD3	Uniform static pressure on surface of hexahedron element.
TEMP	Grid point temperature field.
TEMPD	Grid point temperature field default.
TEMPAX	Temperature sets for conical shell problem.
TEMPPi	Plate element temperature field (only average temperature used for heat transfer analysis).
TEMPRB	Temperature field for line elements (only average grid point temperature used for heat transfer analysis).
<u>Static</u> (UM	1.5.1, TM 3.6)
DEFORM	Envorced axial deformation for line elements.
DSFACT	Defines scale factors for loads and stiffness matrices in differential stiff- ness analysis.
FORCE	Force load specified at grid point by vector.
FØRCEI	Force load specified at grid point (i=1, two grid points, or i=2, four grid points)
FORCEAN	Force loading on an axisymmetric ring.
GRAV	Gravity vector definition (uses mass matrix).
L BA D	Linear combination of static load sets (not including thermal or deformation loadings).
MONAX	Moment specified on an axisymmetric ring.
MUMENTI	Moment specified at grid point (i=1, two grid points, or i=2, four grid points)
PLFACT	Defines scale factors for load increments in piecewise linear analysis.
PLMD	Defines a static pressure load on an area.
PLOAD2	Defines a pressure load applied to two-dimensional (plate) elements.
PLBAD3	Uniform static pressure on surface of hexahedron element.

.

Table 1. Bulk Data Options (cont'd)

E. Loads (cont'd)

Static (cont'd)

P\$INTAX Defines location on axisymmetric conical shell ring (RINGAX) at which loads may be applied.

PRESAX Defines pressure loading of conical shell element.

and the second second

RFØRCE Static loading condition due to centrifugal force field.

SLØAD Defines static loads on scalar points.

TEMP Grid point temperature field.

TEMPD Grid point temperature field default.

TEMPAX • Temperature sets for conical shell problems.

TEMPPi Plate element temperature field (i=1 includes gradient, i=2 includes thermal moment, and i=3 includes tabular description of temperature over cross section).

TEMPRB Temperature field for line elements including gradient.

F. Methods

Buckling (TM 10.3, 10.4)

EIGB Required data for eigenvalue extraction for buckling analysis.

Control Systems (TM 1.4.2)

TF Dynamic transfer function as equation or direct matrix input.

Differential Stiffness (TM 7.1)

DSFACT Define scale factors for applied loads and stiffness matrices in differential stiffness approach.

Eigenvalue (TM 10.0)

- EIGC Data required to perform complex eigenvalue analysis.
- EIGP Defines poles used in complex eigenvalue analysis.

EIGR Data required to perform real eigenvalue analysis.

TABDKP1 Tabular function of structural damping versus frequency for modal formulations only.

Frequency Response (TM 12.1)

- FREQ Set of frequencies used in frequency response analysis.
- FREQ1 Set of frequencies specified by starting frequency, increment, and number of increments desired.
- FREQ2 Set of frequencies specified by starting frequency, final frequency, and number of logrithmic increments desired.

AUTOMATED DATA GENERATION

Table 1. Bulk Data Options (cont'd)

F. Methods (cont'd)

è

1

Frequency Response (cont'd)

TABDMP1 Tabular function of structural damping versus frequency for modal formulations only.

Hydroelastic (UM 1.7, TM 16.1, 16.2)

FREEPT Location of fluid free surface points for displacement recovery.

AXSLØT Required to define existence of axisymmetric slot analysis and default parameters.

Piecewise Linear (TM 3.8)

PLFACT Required to define scale factors for load increments in piecewise linear analysis.

Random Response (TM 12.2)

RANDPS Frequency dependent self and cross-power spectral density factors of load sets.

RANDT1 Autocorrelation function time lag for random analysis.

TABRND1 Tabular function of power spectral density versus frequency for random analysis.

Transient Response (UM 1.6, TM 11.0)

TABDMP1 Tabular function of power spectral density versus frequency and modal formulations only.

TIC Initial conditions for both displacement and velocity for direct formulations only.

TSTEP Time step intervals used in transient analysis solution.

TICS Initial conditions for both displacement and velocity for substructuring (RF 8 and 9 only).

Aerodynamics (UM 1.11, TM 17.0)

- FLUTTER Aerodynamic Flutter Data.
- MKAER@i Mach Number Frequency Tables.
- FLFACT Aerodynamic Physical Data.

G. Miscellaneous

Comment (UM 2.4.2)

\$

For inserting commentary material into unsorted echo deck.

Delete (UM 2.4.2)

For removing cards from either Old Problem Tape or the User's Master File.

· · · · · · ·

Table 1. Bulk Data Options (cont'd)

G. Miscellaneous (cont'd)

and a sum a summary and an a survey of the second second second second second second second second second second

Direct Matrix Input

CØNMI	Concentrated mass element connection (i=1 for mass matrix, i=2 for offset mass inertia properties).
DMI	User-defined direct matrix input, real or complex (general).
DMIG	User-defined direct structural matrix input, real or complex at grid points.
DMIAX	Axisymmetric related user-defined direct matrix input (fluid or structural).
TF	Direct matrix input for dynamic transfer function.

Output Control

FREEPT Location of free surface fluid points for displacement recovery.

PLOTEL Dummy element definition for plotting purposes only.

POINTAX Defines location on axisymmetric conical shell ring (RINGAX) for recovery of displacements.

PRESPT Pressure point on RINGFL for pressure data recovery in fluid.

TSTEP Time step intervals for which solutions and output quantities to transient analysis are desired.

Parameters (UM 3.1.5)

PARAM Specifies parameter values used in DMAP sequences (including rigid formats).

Resequencing (UM 1.2.2)

SEQEP Resequences extra point numbering to optimize bandwidth.

SEQGP Resequences grid and scalar point numbering to optimize bandwidth.

Tabular Input (UM 1.8.3)

DTI Input table data blocks directly.

 TABDMP1
 Tabular function of structural damping versus frequency.

- TABLED1Tabular function for generating frequency or time-dependent loads (user prescribes
i=1 thru 4 for specific algorithm).
- TABLEMiTabular function for temperature-dependent material properties and parametric
data (user prescribes i=1 thru 4 for specific algorithm).
- TABLES1 Tabular stress-strain function referenced by MATS1 card.
- TABRND1 Tabular function of power spectral density versus frequency.

10. PRINTING AND PLOTTING FEATURES

and the second second

10.1 INTRODUCTION

The possible options for printing and/or plotting the results obtained with NASTRAN are many and varied. In most cases the labeling of this output is self-explanatory, or it is deductively obvious. In spite of the wide variety of possible output, certain uniformity has been provided to expedite its interpretation. This section, therefore, illustrates only the characteristic sets of output and some of the less obvious interpretations, where necessary.

The types of NASTRAN printed output may be subdivided into five general categories: (1) input echo, (2) error messages, (3) substructure control requests, (4) case control requests and (5) miscellaneous. The last category itself may be segregated into special output requests controlled other than from case control and automatic output. The location of this output will vary with the sequence of operations performed by NASTRAN, and some will vary with the particular computer system being used. Each of these output categories will be discussed in the sections that follow.

INPUT DATA ECHO AND DIAGNOSIS

in contransition of

10.2 INPUT DATA ECHO AND DIAGNOSIS

3

The input data to NASTRAN undergoes extensive checking, rearranging, and numerical processing in the initial phases of execution. The Executive Control Deck, the Substructure Control Deck, the Case Control Deck, the Bulk Data Deck, and the DMAP sequence are processed in five phases, each phase having its own procedure for processing and diagnosis.

The Executive Control Deck is printed as shown in Figure 1 on every execution of NASTRAN, exactly as it is input on cards, except when approach DMAP is used. In this case, the DMAP deck is echoed later. Diagnostic messages, if any, follow the printout of each data card.

The Substructure Control Deck printout (if substructuring is used) is shown in Figure 2. The deck is reproduced exactly as input, and diagnostic messages, if any, follow the printout of each data card. An echo of the automatically generated DMAP alters may be requested by DIAG 23 in the Executive Control Deck. This echo will follow immediately and precede the Case Control Deck echo described next.

The Case Control Deck printout is shown in Figure 3. All input cards are printed with no modifications. The diagnostic messages deal primarily with proper data format. In certain levels of NASTRAN, a complete list of all legal case control requests are printed if the user misspells a request. The plot control cards, a subset of the Case Control Deck, are processed in a similar manner.

The Bulk Data Deck may be printed in two forms, depending on the ECHØ option selected. The unsorted bulk data echo option (Figure 4) is an unprocessed printout of the input Bulk Data Deck. On a restart from a previous run, only the new cards and deletion cards are printed along with a sorted echo of the modified Bulk Data Deck, including these new cards that were added.

The default printout is a sorted echo (Figure 5) of all the bulk data cards to be used in the computer run. They are arranged in alphabetical and numerical order. All entries in fields 2-9 are shifted and left-justified by NASTRAN for proper alignment to facilitate the sorting. The continuation cards are printed following the parent card. No comment cards are printed in the sorted echo. Each card is assigned a sequence number on the left-hand side to allow for modification of the Bulk Data Deck on restart from a checkpoint.

The diagnostics are printed with the sorted bulk data echo. If errors occurred in the data format, the card containing the effor is identified as shown in Figure 5. For some errors it is

10.2-1

PRINTING AND PLOTTING FEATURES

printed again, and the specific data field containing the error is underlined. The Bulk Data Deck is processed one card at a time, and errors occurring from a combination of different data cards may not be detected at the same time.

Following the bulk data echo is the DMAP compiler source listing (UM 3.2 to 3.20) if requested by DIAG 14 in the Executive Control Deck. In normal structural analyses, this is a listing of the Rigid Format selected in the Executive Control Deck. This listing will include any usersupplied DMAP alters and alters created by NASTRAN for automated substructuring, if used (see Figure 6). Section 3 of the User's Manual describes the Rigid Formats, and Section 5 of the User's Manual describes the details of the DMAP language.

A special procedure is used on a restart from a previous run. The DMAP operations to be executed on restart are identified with an asterisk (*). Immediately preceding the DMAP listing is a summary of modified bulk data cards by data card name. Also, the types of data which are to be modified are marked with a dollar sign (\$). Only those users who are interested in the mathematical operations performed during a restart need to be concerned with these details.

USER-REQUESTED DIAGNOSTIC MESSAGES

10.3 USER-REQUESTED DIAGNOSTIC MESSAGES

The DIAG card in the Executive Control Deck is a convenient tool for printing information on the status of a NASTRAN run. More than thirty diagnostic requests are available. For example, the DIAG card is used to print matrix trailers, the contents of certain arrays and tables, DNAP statements, etc. The method of requesting this type of print is illustrated in Figure 1 where eight DIAG numbers are specified. The information obtained in an unsuccessful run may be used to locate an area where there may be a program error or an engineering modeling error. A few of the more common requests are discussed and illustrated below.

DIAG 8 prints the trailers of matrix and table data blocks. The six words describe the dimensions, form, type and density of a matrix, but may not contain relevant information for a table. Shown below is an example and a description of each item in a matrix trailer.

*** DIAG 8 MESSAGE -- TRAILER FOR DATA BLOCK SCRATCH4 = 9 8 6 1 4 4861

MATRIX NAME	=	SCRATCH4
NØ. ØF CØLUMNS	=	9
N9. 9F RØWS	=	8
MATRIX FØRM	z	6
MATRIX TYPE		1
MAXIMUM NO. OF TERMS IN ANY COLUMN	2	4
MATRIX DENSITY	=	.4861 (* 48.61%)

A complete discussion of data block trailers, including definitions of matrix form and type is presented in the Programmer's Manual, Section 2.2.

DIAG 13 prints in the dayfile the current length of open core each time the calculation routine is called throughout the NASTRAN execution. The length is given in both octal and decimal numbers. An example is shown in Figure 7.

DIAG 14 prints the DMAP statements for the Rigid Format selected. The beginning of the Rigid Format 1 listing is presented in Figure 6.

DIAG 15 displays in the dayfile the GINØ open and close operations for subroutines, data blocks, and scratch files. The numbers from 0 to 3 shown to the right of the open and close statements in Figure 7 are the codes which indicate file conditions (read, write, rewind, end

PRINTING AND PLOTTING FEATURES

of file). The codes are described in Section 3.4 of the Programmer's Manual.

States and a second second second

2. **4.** 21. 21. 200

1310.000.000

* DIAG 19 prints data used by NASTRAN to determine which of three methods will be most efficient for matrix multiply and add or triple multiply (see Figure 8).

DIAG 21 provides a table showing the vector sets to which each degree of freedom of grid, scalar, and extra points belong. Such a table is illustrated in Figure 9.

DIAG 22 provides as many as five tables showing the assignment of degrees of freedom of grid, scalar, and extra points to Analysis, Omit, SPC, and MPC sets. One type of table is shown in Figure 10.

DIAG 23 prints a listing of the DMAP alters generated during an automated multistage substructuring run. The first part of a typical listing is shown in Figure 11.

10.3-2

10.4 ERROR AND WARNING MESSAGES

-

ORIGINAL PAGE IS OF POOR QUALITY

The NASTRAN system will automatically print out messages to the user for several reasons: user information messages, user warning and fatal messages, and system error messages. Most of these messages are listed with a brief description in Section 6 of the User's Manual. Each message is given a number and may refer to a specific data item, mathematical operation, or internal operation. Fatal messages will cause the program to exit.

Of specific user interest are the user information messages which provide data regarding the performance of a particular job. User information messages 3023 through 3028 provide data on the matrix decomposition. These include such data as the semi-bandwidth (B), the active columns (C), the number of columns held in core (R), and the estimated CPU running time in seconds. Message number 3035 prints out the error ratio for a solution. Message number 2073 prints the method selected for a matrix multiply operation. These messages aid the user in detecting problems with running time and accuracy.

Another type of message is the warning of gridpoint singularities (Figure 12) which provides useful information. This printout lists the grid points which have a possible singularity in the stiffness matrix. The number of singularities and the specific unconstrained degrees of freedom (gridpoint displacements or rotations) are listed in the order that they are detected. This is only a warning message and may not be significant in certain special cases.

CASE CONTROL DECK OUTPUT CONTROL

10.5 CASE CONTROL DECK OUTPUT CONTROL

The output data processing for presentation of results is performed by NASTRAN with a single, general purpose routine which prints titles, labels, column headings, and the numerical results. It is therefore obvious that many types of data will share similar formats and have the same general appearance.

The Case Control intput contains specifications for the type and contents of printout (UG 8). Each output request defines a physical type of variable and specifies a set of points or elements to which the request applies. Options are available to print out the results of a dynamic analysis in either of two types of sort. The default option, SØRT1, will print the results for all points or elements for each time step or frequency (Figure 13). SØRT2 will print the results for each point or element for all time steps or frequencies (Figure 14).

In certain dynamic problems (Figure 13) for Complex Eigenvalues or Frequency Response, the output consists of complex numbers. The user has the option of specifying that the output be printed in a real/imaginary format or in a magnitude/phase angle format. Each pair of numbers is written with the real value or magnitude directly above the imaginary value or phase angle. (Note: The phase angles are in degrees.)

The most common format of printed output relates to the grid, scalar, or extra points (Figure 13). The following quantities are printed in this format:

DISPLACEMENT	ACCELERATION
ØLØAD	VECTØR
SPCFØRCE	PRESSURE
VELØCITY	THERMAL

In dynamic analysis, the following types of output are available as SØLUTIØN set output (Figure 14):

SDISPLACEMENT	SACCELERATION	
SVELØCITY	NLLØAD	

This output provides the results only for the degrees of freedom used in the solution of the dynamic problem. If a direct formulation is used, the quantities printed relate to the motion of
PRINTING AND PLOTTING FEATURES

anterioration in the state of the second of the state of th

the grid, scalar, or extra points remaining after the constrains and matrix reductions have been applied. If a modal formulation is used, the output describes the motion of the <u>modal</u> coordinates and any user-supplied extra points. The NLLØAD output request provides the nonlinear displacement or velocity-dependent loads generated only for the transient analysis.

Note that grid, scalar, and extra points, as well as modal coordinates, may be intermingled in the output. The points with only one degree of freedom are printed in a special format. The "ID" = N number is printed only for the first point in a sequential set of up to six such points. The results for the second and succeeding points are printed on the same line with the first. If more than six points occur in a numerical sequence, the seventh "ID" will be printed so as to identify the second line of printout, and so on for every succeeding set of six such points.

In the general gridpoint output, the values given for T1, T2, and T3 correspond to the three translations, and R1, R2, and R3 correspond to the three rotations of that grid point. The orientation of these displacement components is defined by the output displacement coordinate system, which is either the NASTRAN "basic" rectangular system or a special coordinate system that is specified by the user for that grid point. If a rectangular system is used, the T1, T2, T3 and R1, R2, R3 directions correspond to the X,Y,Z(T1 = u_c (radial), T2 = u_{θ} (circumferential), and T3 = u_z (axial)). The rotations R1, R2, and R3 are given in units of radians and are defined as vectors with the same directions as the translations.

The formats of the element forces and stresses are not as uniform as the output formats for gridpoint data. Each element type has a particular set of output quantities which is printed in a self-explanatory format. These formats may vary with the type of analysis. Element force output, in general, is the self-equilibrating internal load system for that element in units of force (moment), or force (moment) per length. Element stress output, in general, consists of direct stresses at a particular point in the element. Both these sets of element output quantities are defined in the local element coordinate system. Section 2 of the User's Manual lists the output quantities available for each element. Examples of these printouts are given in Figures 15 and 16. The first example is a typical element stress output; the second is a more elaborate printout of stresses occurring in the harmonic analysis of an axisymmetric conical shell element. As with the grid point data printout for dynamic analyses, these values may be real or complex, and may be requested in either SØRT1 or SØRT2 format.

and the second second second second

i water

Trage ...

--}

= = \

1

Table 1. Output Request Codes for Combine and Reduce Operations.

	CØMBINE		MREDUCE
Code	Output	Code	Output
2	SØF table of contents	1	Current problem summary
3	CØNCT1 bulk data summary	2	Boundary set summary
4	CONCT bulk data summary	3	Summary of grid point ID numbers
6	GTRAN bulk data summary		The EOSS item for the structure
7	TRANS bulk data summary		being reduced
9	RELES bulk data summary	5	The EQSS item
11	Summary of automatically generated connections	6	The BGSS item for the new
12	Complete connectivity map of	7	The CSTM item > reduced
•-	final combined pseudostructure	8	The PLTS item pseudoscructure
13	The EQSS item	9	The LØDS item [/]
14	The BGSS item	10	Modal DØF set summary
15	The CSTM item { for the new psychotracture	11	Fixed set summary
16	The PLTS item	12	Summary of grid point ID numbers
17	The LØDS item ^J		in each fixed set
	REJUCE		CREDUCE
Code	Jutput	Code	Output
1	Current problem summary	1	Current problem summary
2	Boundary set summary	2	Boundary set summary
3	Summary of grid point ID numbers in each boundary set	3	Summary of grid point ID numbers in each boundary set
4	The EQSS item for the structure being reduced	4	The EQSS item for the structure being reduced
5	The EQSS item χ	5	The EQSS item \backslash
6	The BGSS item	6	The BGSS item for the new
7	The CSTM item (for the reduced	7	The CSTM item > reduced
8	The PLTS item	8	The PLTS item pseudostructure
9	The LØDS item [/]	9	The LØDS item 7
		10	Modal DØF set summary
		1 11	Fixed set summary
		12	Summary of grid point ID numbers in each fixed set

PRINTING AND PLOTTING FEATURES

dealersfell through

10 10 10

ŝ

and the the second second second

Table 2. Summary for Structural Plotting Options.

PLØTID	Optional to give plot identification, must precede @UTPUT(PLØT)
SET 1	Required and specifies sets of elements for plotting (UM 4.2 for options)
PLØTTER	Optional to specify plotter name and MDDEL names (UM 4.1) if other than SC 4020 (UM 4.2 for options)
PRØJECTIØN	Optional to specify type if other than @RTH@GRAPHIC (TM 13.)
AXES (VIEW)	Optional to device observer's coordinate system and to position the figure (UM 4.2)
MAXIMUM De formation	Required to specify scale for plotting deformed structure and must precede FIND card (UM 4.2)
SCALE	Required to specify scale of plotted object, may be replaced by FIND card (UM 4.2)
ØRIGIN	Required to define origin of plotted object relative to lower left-hand corner of paper, may be replaced by FIND card (UM 4.2)
VANTAGE PØINT	Required to specify location of observer with respect to structural model for either perspective or stereoscopic projections only, may be replaced by FIND card (UM 4.2, TM 13.)
PRØJECTIØN Pl ane Separatiøn	Required to specify R-direction separation of the observer and the projection plane for perspective or stereoscopic projections only, may be replaced by FIND card (UM 4.2, TM 13.)
gc ul a r Se paratign	Optional to specify S-coordinate separation of the two vantage points of other than 2.756 inches in the stereo- scopic projections only (UM 4.2)
CAMERA	Optional to specify type of film and/or paper and number of blank frames on microfilm plotters only (UM 4.2)



يجردونهم

Figure 1. Example of the NASTRAN Executive Control Deck Echo.

PRINTING AND PLOTTING FEATURES

ORIGINAL PAGE IS OF POOR QUALITY

0 1 U L ¥ U w ٥ u œ Э ⊢ 0 ∍ æ * S ø 5 5 Z < ۵

NAS

~

2 A GE

NASTRAN 11/30/76

NUVF WRER 24. 1978

1. 8. 6 . 6 .

SUBSTRUCTURE PHASE2 PASSVORD - MILSYN SOFRIN: FOC SOFPRIN: TOC AR:DUCE ARAIC VAME MA SOFPRIN: TOC MR:DUCE ARAIC VAME MA BOUVDAAY 5 HETHOD 9 DUTPUT 1,550,9,10 SOFPRIN: TOC METHOD 9 DUTPUT 2,7212 COMPOVEN MA MAME MO DUTPUT 1,550,9,10 COTPUT 2,50,9,10 COTPUT 1,500,9,10 COTPUT 1,500,9,00 COTPUT 1,500,9,000 Figure 2. Example of the NASTRAN Substructure Control Deck Echo.

10.3-6

ORIGINAL PAGE IS OF POOR QUALITY

ა

0

200

~

~

TRAILER FOR DATA BLUCK CASECC

1

B MESSAGE

Set CIAG

a far en a la serie de la s

ويروي وبريدود بالنوار المالك وتعتدي ترتهوه فترجعون



ŝ

È.

Ŀ

STATIC AMALYSIS OF A THREE-PANEL TRUSS Output request for all quantities

C I U ш TITLE • STATIC ANALYSIS OF A THREE-PANEL TRUSS SUBTITLE • DUTPUT REGUEST FOR ALL GUANTITTES LINE • 38 LCHO • BOTH SPC • 1 DUTPUT DISPLACEMENT • ALL SPC FORCE • ALL COPOPE • ALL COPOPE • ALL SPECASE 1 LABEL • 240-LB LUAD AT GRID POINT 4 LABEL • 240-LB LUAD AT GRID POINT 2 LABEL • 220-LB LUAD AT GRID POINT 2 LABEL • 120-LB LUAD AT GRID POINT 2 D F C K CONTROL CASE CARD COUNT

Figure 3. Example of the NASTRAN Case Control Deck Echo.

مويود والاستحاص دي

STATIC AMALYSIS OF A THREE-PAMEL IRUSS Dutput request fum all quantities

NOVEMBER 27, 1978 NASTRAN 11/30/75 PAGÉ

.

فكراد

1 1 1 1 </th <th></th> <th>Z </th> <th>•</th> <th>5</th> <th>£</th> <th>2</th> <th>بر</th> <th></th> <th>5</th> <th>-</th> <th><</th> <th>3</th> <th>u ~</th> <th>ى</th> <th>¥</th> <th>-</th> <th>ر د</th> <th>:</th> <th>2</th> <th></th> <th></th> <th></th> <th></th> <th></th>		Z 	•	5	£	2	بر		5	-	<	3	u ~	ى	¥	-	ر د	:	2					
\$ DEFINE GALD FOINT LOCATIONS \$ CALO 0.0 0.0 \$ SHOLO 240.0 0.0 \$ SHOLO 0.0 0.0 \$ SACO 0.0 0.0 \$ SACO 0.0 0.0 \$ SACO 1 1 \$ SACO 1 2 \$ SACO 1 0 \$ SACO 1 2 \$ SACO 1 2 \$ SACO 1 2 \$ SACO <t< td=""><td>. 1 2</td><td>•</td><td>m</td><td>:</td><td>*</td><td>•</td><td>•</td><td>\$</td><td>:</td><td></td><td>•</td><td>:</td><td></td><td>~</td><td>•</td><td>•</td><td>æ;</td><td></td><td>:</td><td>3</td><td>:</td><td>Ā</td><td>0</td><td>•</td></t<>	. 1 2	•	m	:	*	•	•	\$:		•	:		~	•	•	æ;		:	3	:	Ā	0	•
6#10 1 0.0 0.0 0.0 6#10 5 180.0 240 0.0 6#10 5 180.0 240 0.0 6#10 5 540.0 240 0.0 6#10 5 540.0 0.0 0.0 6#11 5 540.0 0.0 0.0 6#10 6 540.0 0.0 0.0 6#11 5 540.0 0.0 0.0 6#10 6 540.0 0.0 0.0 5 5 5 5 5 600 5 1 3 5 5 600 5 1 3 5 5 600 5 1 3 5 5 600 5 3 5 5 5 600 5 3 5 5 5 600 5 3 5 5 5	S DEFINE SRID	FOINT	3	IC A 1	LI ONS																			
GR10 2 180.0 0.0 0.0 GR10 5 180.0 240.0 0.0 GR10 5 540.0 0.0 0.0 GR01 1 1 2 4 5 5 1 3 5 5 CR00 1 2 4 5 5 5 CR00 7 3 3 5 <	GRID 1			0	••		0.0	_	0	•														
G410 3 160.0 240.0 0.0 G410 5 540.0 0.0 0.0 5 540.0 0.0 0.0 0.0 5 5 540.0 0.0 0.0 5 5 540.0 0.0 0.0 5 5 5 5 5 6401 1 2 4 5 6401 1 2 4 5 6401 1 2 4 5 6401 1 2 4 5 6401 1 2 4 5 5 6401 1 2 4 5 5 6401 1 2 4 5 5 6401 1 2 5 5 5 5 6401 1 2 5	GRID Z			-	60.08		0.0	_	C	0.0														
GR10 4 360.0 240.0 0.0 GR10 5 500.0 240.0 0.0 GR10 5 500.0 240.0 0.0 GR11 5 500.0 240.0 0.0 GR05ET 4:1 2 3456 GR05 1 2 3 3 GR01 1 2 3 3 GR01 1 3 3 3 GR01 1 3 3 3 CR00 3 3 3 3 CR00 3 3 3 3 CR00 7 3 3 5 CR00 7 3 3 5 CR00 7 3 3 5 CR00 7 3 5 5 CR00 7 5 5 5 CR00 7 7 5 5 CR00 7 7 7 7 CR00 3 6 5 <td>GK10 3</td> <td></td> <td></td> <td>~</td> <td>60.08</td> <td></td> <td>240</td> <td>•••</td> <td>9</td> <td>••</td> <td></td>	GK10 3			~	60.08		240	•••	9	••														
GR10 5 300.0 240.0 0.0 GADEFT 540.0 0.0 0.0 3456 5 SFECTFT CATD POINT CONNECTIVITIES 3456 3456 5 SFECTFT CATD POINT CONNECTIVITIES 3456 5 SFECTFT CATD POINT CONNECTIVITIES 3456 5 SFECTFT CATD POINT CONNECTIVITIES 3 5 CR00 2 3 6 CR00 2 3 7 SECTO 3 4 6 CR00 3 3 7 SECTO 3 4 6 CR00 3 3 5 7 SECTO 3 4 5 6 CR00 3 3 5 5 6 CR00 3 3 5 5 6 CR00 3 4 5 5 6 CR00 3 4 5 5 6 CR00 3 4 5 5 6 CR00 3 1 1 5 5 6 CR00 3 1 1 5 5 5 6 CR00 3 </td <td>GRID 4</td> <td></td> <td></td> <td>1</td> <td>160.0</td> <td></td> <td>0.0</td> <td>_</td> <td>0</td> <td>•••</td> <td></td>	GRID 4			1	160.0		0.0	_	0	•••														
GRID 6 540.0 0.0 0.0 3456 5 SFECTFY 6#1D 0141 1 2 3456 5 SFECTFY 6#1D 0141 1 2 3456 5 SFECTFY 6#1D 0141 1 2 3 5 5 CHOD 3 3 4 5<	GR10 5			61	160.0		240	•••	0															
G405ET 3456 5 \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	GRID 6			<i>u</i> 1	0.046		0.0	_	9	•••														
<pre>\$ PECIFY GAID POINT CONNECTIVITES \$ CAOD 2 2 1 3 \$ CAOD 2 2 2 1 3 \$ CAOD 4 1 2 4 \$ CAOD 4 1 2 2 4 \$ CAOD 7 1 3 4 \$ CAOD 7 1 1 0 \$ CAOD 7 1 10 \$ CAOD 7 0 0.0 -1.0 0.0 \$ CAOD 7 0 0.0 -1.0 0.0 \$ CAOD 1 10 \$ CAOD 1 10 \$ CAOD 1 10 \$ CAOD 7 0 \$ CAOD 7 0 \$ CAOD 7 0 \$ CAOD 1 10 \$ CAOD 7 0 \$ CAOD 1 10 \$ CAOD 1 1</pre>	GAD SE T															34.5	99							
C#00 1 1 1 2 C#00 2 2 1 3 C#00 4 1 2 4 C#00 7 3 4 5 C#00 7 3 4 5 C#00 7 3 4 5 C#00 7 3 4 5 C#00 9 1 3 6 C#00 9 1 2 2 C#00 9 1 0 C#00 9 0 C#0	SPECIFY GAID	NID4 0	5	NNO	HE CT I	V 1 T	1 E S																	
CK00 Z Z 1 3 CR00 3 3 Z 3 CR00 5 1 3 5 CR00 7 2 4 CR00 7 3 4 CR00 7 3 4 CR00 9 1 3 5 CR00 9 1 4 5 CR00 9 1 4 CR00 0.0 -10 0.0 1170 0.0 -11.0 0.0 1170 0.0 100 0.0 0	C+00 1	-					~																	
CR00 3 3 5 CR00 4 1 2 4 CR00 7 3 4 CR00 7 10 CR00 7 3 4 CR00 7 10 CR00 7 10 CR00 7 3 4 CR00 7 10 CR00 7 10 CR	CK00 2	2					~																	
CRUD 4 1 2 4 CRUD 5 1 3 5 CRUD 7 3 4 5 CRUD 7 2 120.0 0.0 -1.0 0.0 FURCE 1 2 2 FURCE 1 2 2 FURCE 1 1.0 7.07 FURCE 2 2 1700 0.0 -1.0 0.0 50 FINE ELEMENT PECFFICATIONS 50 FINE MATERIAL SPECIFICATIONS 50 CHUE MATERIAL SPECIFICATIONS 50 CHUE MATERIAL SPECIFICATIONS 50 CHUE MATERIAL SPECIFICATIONS 50 CHUE MUNDARY CONDITIONS 50 CHUE 1 100 0.0 -3 4.28 MATI 100 33.0.47 0.3 4.28 50 CHUE 1 12.0.4 50 CHUE 1 2 0.0 5 2 0.0 50 CHUE 1 12.0.4 50 CHUE 1 12.0	CR00 3						-																	
CROD 7 1 3 4 CROD 7 2 1100 0.0 -1.0 0.0 FURCE 1 4 CROE 2 2 2 FURCE 1 4 CROE 1 4 CROD 0.0 -1.0 0.0 -1.0 0.0 FURCE 1 4 CROE 2 2 2 CROE 1 100 0.0 0.0 -1.0 0.0 CROE 2 2 2 CROE 1 100 0.0 0.0 -1.0 0.0 CROE 2 2 2 CROE 1 100 0.0 0.0 -1.0 0.0 CROE 2 2.0 0.0 0.0 -1.0 0.0 CROE 2 2 2.0 0.0 0.0 0.0 -1.0 0.0 CROE 2 2 2.0 0.0 0.0 0.0 0.0 0.0 CROE 2 2 2.0 0.0 0.0 0.0 0.0 0.0 0.0 CROE 2 2 2.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	CRUD 4	-					3																	
CK00 5 3 3 5 CR00 7 3 4 5 CR00 7 3 4 5 CR00 9 1 4 2 5 CR00 9 1 4 2 5 CR00 9 1 4 2 6 FURE LOADING CUNDITIONS 240.0 0.0 -11.0 0.0 FURE 1 4 2 17000 0.0 -11.0 0.0 FURE 1 100 0.0 0.0 -11.0 0.0 FURE 2 2 2 1107 FURE 2 2 2 2 1107 FURE 2 2 2 2 0.0 0.0 -10.0 0.0 FURE 2 2 2 2 0.0 0.0 -10.0 0.0 FURE 2 2 2 2 1107 FURE 2 2 2 2 1107 FURE 2 2 2 2 0.0 0.0 -10.0 0.0 FURE 2 2 2 2 2 0.0 0.0 0.0 0.0 FURE 2 2 2 2 2 1107 FURE 2 2 2 2 1107 FURE 2 2 2 2 2 1107 FURE 2 2 2 2 2 1107 FURE 2 2 2 2 1107 FURE 2 2 2 2 1107 FURE 2 2 2 2 2 2 2 2 2 2 2 2 0.0 0.0 FURE 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	CROD >	-		(11)																				
CROD 7 3 4 5 CROD 7 3 4 5 CROD 9 1 4 5 5 5 5 5 5 6 7 0 9 1 4 6 7 0 0.0 -1.0 0.0 7 0 -1.0 0.0 7 0 -1.0 0.0 7 0 -1.0 0.0 7 0 0 0 0 0 0 0 7 0 11.77 7 0 11.77 7 0 0 0 0 0 0 7 0 0 0 0 0 0 0 7 0 0 0 0	CK00 0	•		m	-		\$																	
CRUD R 2 5 5 CRUD V 1 4 5 5 DEFINE LOADING CUNDITIONS FURCE 1 4 2200 0.0 -1.0 0.0 FORCE 2 2 120.0 0.0 -1.0 0.0 5 DEFINE ELEMENT PECPEKTES FORD 1 1.00 7.97 PROD 3 100 11.77 PROD 3 100 11.77 PROD 3 1.00 0.0 PROD 3 1.00 0.0 PR	CR00 7	m		4			\$																	
CK00 4 1 4 6 5 DEFINE LOADING CUNDITIONS FUNCE 1 4 240.0 0.0 -1.0 0.0 FORCE 2 2 2 120.0 0.0 -1.0 0.0 5 DEFINE ELEMENT PRUPEKTIES 5 DEFINE ELEMENT PRUPEKTIES 7 PROD 3 100 11.77 PROD 3 1.00 3.4.28 ATT 100 3.0.44 15.044 5 PROD 4 15.044 12.044 5 PROD 5 0.0 6 2 0.0 PARAM GRUNDARY CONDITIONS 5 PROD 1 1 12 0.0 6 2 0.0 PARAM GRUNDARY CONDITIONS 5 PROD 1 12 0.0 6 2 0.0 PARAM GRUNDARY CONDITIONS 5 PROD 1 12 0.0 6 2 0.0 PARAM GRUPT 0 PARAM FWES 1 FORD 2 1.00 0.0 PARAM FWES 1	CRUD A	2		'n			÷																	
* DEFINE LOADING CUNDITIONS * DEFINE LOADING CUNDITIONS FORCE 1 * SOFFINE LEMENT PROPERTIES * DEFINE ELEMENT PROPERTIES * DEFINE ELEMENT PROPERTIES * DEFINE ELEMENT PROPERTIES * DEFINE ELEMENT PROPERTIES * PROD * DEFINE ELEMENT PROPERTIES * PROUDE * DEFINE MOUNDARY CONDITIONS * PROUDE * DEFINE BOUNDARY CONDITIONS * PROUDE * DEFINE BOUNDARY CONDITIONS * PROUDE * DEFINE BOUNDARY CONDITIONS * PROUDARY CONDITIONS *	CK00 4	-1		4			¢																	
FURCE 1 4 240.0 0.0 -1.0 0.0 FORCE 2 2 1700 0.0 -1.0 0.0 5 DEFINE ELEMENT PHOPEKTES FROD 1 100 11.77 FROD 2 100 11.77 FROD 2 100 11.77 FROD 3 10001 11.77 FROD 3 10.0001 11.77 FROD 3 10.0000 11.77 FROD 1 100000 11.77 FROD 1 100000 11.77 FROD 1 100000 0.0 0.0 0 2 0.00 FROD 1 10000 0.00 0.0 0 2 0.00 FROD 1 10000 0.00 0.00 0 2 0.00 FROD 1 10000 0.0000 0.00 0.00 0.00000000000	S DEFINE LOADI	ING CUI	Ĩ	110	NS																			
FORCE 2 2 170.0 0.0 -1.0 0.0 5 DEFINE ELEMENT PROPERTES 7 DED 1 100 7.97 7 PROD 2 100 7.97 7 PROD 3 100 11.77 7 PROD 3 100 11.77 7 PROVICE MATERIAL SPECIFICATIONS 7 PROVICE MATERIAL SPECIFICATIONS 7 PROVICE MATERIAL SPECIFICATIONS 7 PROVICE MATERIAL SPECIFICATIONS 7 PROVIDE MATERIAL SP	FURCE 1	4					240	••	0	•		1	-	0		:0	~							
3 DEFINE ELEMENT PROPERTIES PROD 1 100 7.97 PROD 2 100 11.77 PROUTE MATERIAL SPECIFICATIONS PROVIDE MATERIAL SPECIFICATIONS PROVIDATIONS PROVIDE MATERIAL SPECIFICATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PROVIDATIONS PR	FORCE 2	2					120	•	0	•		1	-	0		3	~							
PROD 1 1.00 7.97 PROD 2 1.10 21.15 PROUDE 100 11.77 100 \$ PROVIDE ATT 100 31.047 \$ PROVIDE 100 30.047 0.3 \$ PROVIDE 15.044 0.3 4.28 \$ PREFINE 0.04 15.044 12.044 \$ PREFINE 0.04 15.044 0.3 \$ PREFINE 0.044 12.044 0.0 \$ PREFINE 0.0 6 2 0.0 \$ REQUEST 1 1 12 0.0 \$ REQUEST 1 1 12 0.0 \$ PARAM INES 1 0 \$ PARAM INES 1	S DEFINE ELEME	NT PRI	JPE	ƙ LI	ES																			
PRDD Z 1C3 Z1.16 PRDD 3 L00 11.77 PROVIDE MTL 100 11.77 ATI 100 30.047 0.3 4.28 ATI 100 1 2 0.0 5 4.28 AFFINE BOUNDARY CONDITIONS 5 2 0.0 5 2 0.0 5 5 6 0.0 5 5 0.0 5 7 0.0 5 2 0.0 0 2 0.0 0 2 0.0 5 5 5 1 2 1 2 2 0.0 2 0.0 2 0.0 5 2 0.0 5 2 0.0	PROD 1	100			.97																			
PROD 3 100 11.77 * PROVIDE MATK SPECIFICATIONS ************************************	PROD 2	100		~	1.10																			
\$ PROVIDE MATERIAL SPECIFICATIONS MATI 100 30.0+7 0.3 4.28 •MRGNS MAKUS 20.0+4 15.0+4 12.0+4 •ARGNS •MRGNS SPEFINE 800NDARY CONDITIONS 0.3 4.28 •MRGNS SPEFINE 800NDARY CONDITIONS 0.0 5 2 0.0 SPE 1 1 12 0.0 5 2 0.0 SPE 1 1 12 0.0 5 2 0.0 SPEMAH GROPNT 0 0170UT 5 2 0.0 5 PARAH GROPAT 0 0 12 0.0 5 2 0.0 PARAH GROPAT 0 0 0 5 0.0 5 5 0.0 5 5 0.0 5 5 0.0 5 5 0.0 5 5 0.0 5 5 0.0 5 5 0.0 5 5 0.0 5 5 5 5 5 5 5 5 <	PKOD 3	100		-	1.77																			
MATI 100 30.047 0.3 4.28 +MRGNS +MRGNS 20.044 15.044 0.3 4.28 +MRGNS 5 DEFINE BOUNDARY CONDITIONS SPC 1 1 12 5 REQUEST ADDITIONAL UUTPUT 5 REQUEST ADDITIONA	S PROVIDE MATE	RIAL	SPE	CIF	ICAT	NO3	5																	
-MRGNS 20.0+4 15.0+4 12.0+4 5 DEFINE BOUNDARY CONDITIONS 5 PC 1 1 12 5 REQUEST ADDITIONAL UUTPUT 5 REQUEST ADDITIONAL UUTPUT PARAM GRUPHY 0 PARAM INES 1 ENDDATA	MATL 100	30.0	1				0		4	2.	80										Ŧ	MRG	Ś	
S DEFINE BOUNDARY CONDITIONS SPC I L 12 0.0 6 2 0.0 F REQUEST ADDITIONAL UUTPUT PARAM GROPPNT 0 PARAM INES 1 ENDDATA	+MRGNS 20.0+4	15.0	**0	-	2.0+4																			
SPC I I I 12 0.0 6 2 0.0 5 REQUEST ADNITIONAL UUTPUT PARAM GRUPNT 0 PARAM [NES] ENDDATA	S DEFINE BOUND	DARY CI	NO	III	ONS																			
S REQUEST AUDITIONAL UUTPUT Pakam Gruppit o Pakam Ines 1 Enddata 22	SPC 1	-		٦	2	-	0.0	_	0	_		~				0.0	_							
PAKAM GROPNT O Pakam Intes 1 Enddata	S REQUEST ADDI	I I DNAI	С	UTP	10																			
PARAM INES 1 ENDDATA Calimata al	PAKAH GUUNT	0																						
	PARAM INCS	-																						
11.11.1 - 17.	ENDDATA																							
	C.16.M.T.= 44																							

PRINTING AND PLOTTING FEATURES

.

Figure 4. Example of the NASTRAN Input Data Deck Echo - Unsorted.

••• USER INFORMATION MESSAGE 207» BULK JATA RUT SUKTEDAKSJAT WILL PERFORMEN DECK.

TWRE LLEMENT BEAM OF LFNGTH 3.0 Each Elfment Hag Length # 1.0

PAGE

٠

•

:

• •••

•••

...

~

¢

....

• n,

.

: m

:-

~

c 80 I

د بد

۲

U A T

BULK ÷

3 7 R T E U

ORIGINAL CACE IS

OF POOR QUALITY

₹. ÷

1.-5 1.-5 1.-6

•••

• •

1.0

1.0

00000000

2020

10.5-9

223

1.-5

•

2. - 5 2.40 2. =h 2.-6

- 0.00

2. • • 50 • • 4. • 2.=6

<u>e</u>

2000.

<000° 0. . 1005 0. .0005

÷

. . .

•

ું

9/1/19	
NVALSEN	
11, 1974	
JANUARY	

CASE	CONTROL	DECK	OUTPUT	CONTROL

.

.

SURIED CAN'N CURINT

24

*** USEM FATAL MESSAGE 515, FURMAT ERMINR UN RULM DATA CARD FUNCE

2.-5

4. - h

å. - h . 333

5.

1. • 7 25.

MTHAUS

٦.

K. 1 - 1

Figure 5. Example of the NASTRAN Input Data Deck Echo - Sorted.

STATIC ANALYSIS OF A THREE-PANEL TRUSS Dutput Request fuk all quantitles

NUVEMBER 27, 1978 NASTRAN 11/30/76 PAGE

¢

1

LEVEL 2.0 NASTRAN DMAP COMPILER - SOURCE LISTING

OPTIONS IN EFFECT: GO ERR+2 NOLIST NUDECK NOREF

NUISCAR

- 1 BEGIN NO.1 STATIC ANALYSIS SFRIES P \$
- 2 FILE GPTP2=SAVE/EST1=SAVE S
- 3 FILE UG-APPEND/PGG-APPEND/UGV-APPEND/GM-SAVE/KNN+SAVE S
- 4 GP1 LEONINGEOM2./GPLNEGEXIN/GFDT.CSTM. 3GPDT.SIL/V.N.LUSEI/ V.N. NOGPDT/V.N.ALWAYS=-1 \$
- 5 SAVE LUSET \$
- 6 CHXPNT 6PL+EUEXIN+GPDT+CSTM+BGPDT+SIL 5
- 7 GP2 GEUM2, EQEXIN/ECT 5
- B CHKPNI ECT S

10.5-10

- 9 PARAML PCUA//C.N.PRES/C.N./C.N./C.N./V.N.NOPCOB S
- 10 PURGE PLTSETX, PLTPAR, GPSETS, ELSETS/NOPCOH \$
- IL CCND PLANOPCDB \$
- 12 PLTSET PCD4.E0EXIM.ECT/PLTSETX.PLTPAR.GPSETS.ELSLTS/V.N.NSIL/ V.N. JUMPPLOT=-1 5
- 13 SAVE NSIL, JUMPPLOT 5
- 14 PRTMSG PLTSETX// S
- 15 PARAM //ComoMPY/YowoPLTFLG/Comol/Comol S
- 16 PARAM //Com.MPY/W.N.PFILt/Com.O/Com.O S

Figure 6. Partial Listing of NASTRAN DMAP Statements for Rigid Format 1.

7

PRINTING AND PLOTTING FEATURES

 DCFAL=130022
 DECIMAL= 48145

 0
 10.340 SEM1 ENG
 2

 0
 10.350 FUC
 2

 0
 0.11.075 CORE LEFT
 28600

 0
 11.075 LINK BEGN
 NS02
 28600 N502
 DECIMAL
 42049

 DPEN NPTP
 42046

 DPEN NPTP
 42046

 DPEN NPTP
 42046

 DPEN SCRATCHL
 1

 CLOSE SCRATCHL
 1

 CLOSE SCRATCHL
 1

 CLOSE SCRATCHL
 1

 CLOSE SCRATCHL
 1

 DPEN TFFILE
 0

 DPEN SCRATCHL
 1

 DPEN SCRATCHL
 CLOSE XPTU 1 OPEN PUOL 2 CLOSE POUL 2 DECIMAL= 48145 . 0.000 ---- [INK END ----06CIMAL- 50580 0PEM POOL DECIMAL= 44008 z 0 EXECUTI CLUSE GEOMI VELIMAL CLOSE PUOL .005 XSFA SECUNDS NOGU-STATUS

 09.31.16.CGRE
 CCTAL=122101
 DECTM

 09.31.17
 DCTAL=12201
 DECTM

 09.31.19
 DCTAL=12207
 DECTM

 09.31.20
 DCTAL=12207
 DECTM

 09.31.21
 DCTAL=12207
 DECTM

 09.31.21
 DT
 DFEM

 09.31.21
 DT
 DT

 09.31.21
 DT
 DT

 2.405
 END
 IfP

 2.410
 XGPI
 DEC

 CCTAL=122101
 DEC
 DEC
 7.0 09.34.16. 09.34.16

> ORICI: OF IS

v. ...

Portion of a Dayfile Illustrating DIA6 13 (Core Length) and DIA6 15 (Open-Close). Figure 7.

10.5-12

Figure 8. Example of the Cutput from DIAG 19 - MPYAD Method Selection.

1

....

21 2AGE NASIRAN 11/30/70 0 z MQVENBEK 27, 1978 ••• USER INFORMATION MESSAGE 2119, SUBROUTINE CP4PRT - DIAG 22 SET DISP SETS VS. DUF FULLOWS. 27 5 0 0 3 * 30 ی 36 o • • . • • ٠ o < ø *~ 56 m 3 STAFIC AMALYSIS OF A TWREE-PANEL TAUSS Output reuvest fom all quantities 200 ٠ 1 1 ŧ . . T01ALS -----1 1 . . . 1 a EXT 100 TM3 q -----COFUM

Figure 9. Example of the Output Table Provided by DIAG 21.

10.5-13

CASE CONTROL DECK OUTPUT CONTROL

PRINTING AND PLOTTING FEATURES

PAGE NOVEH6EK 27. 1978 NASTRAN 11/30/78 **STATIC AMALYSIS DF A THREE-PANEL TRUSS Outout request fum all quantities**

1000

. Setterstere

22

.....

_

--1

	-10-	li	
	ţ	5 - 5 - 5	,
	÷	+ + ~ +	
	-1-		
CENENT SET			
SPC DISPLA	;		
	ł	11]	
	ł		
	-?-	111	
		1 11	

Figure 10. Example of the Output Provided by DIAG 22.

:

5 H U U 0 E C K SUBSTRUCTURE

N N S L & N N

4

PAGE

NASTRAN 11/30/74

NOVEMARE 24, 1978

ŝ,

	,						
	//+400+/04//1	\$ 0					
13621	LOSDEG 1						
50707	//DPY/+190	•/+206++10	:	•.•	•:•	•:•	
	• •/• •						
103em	CASECC 6EDH40	DVNAMICS, CST	1/USETP.EE	UR.EOST.D	1740/410	• 0101	•
	S. CLARCINS	No NOFI X / South	SE 3 PE / + RE/	AL+ 5			
520	LAC. SETAL						
1905	/K2 . H2 . P2	•	15. No DRV/	• ABASIC		1440/01	(• >
	• 2 A E C • 7 • 6 H A E •	/•K4"X• 5					
COND	LAT23 SKIPH						
E CUTV	K2 KFFK/NCF	1 2 3					
EOULV	108/X111 0K	1 x S					
C T NO	LPN13 LPN14						
SCEI	USETPAK2 .P2	•	KFFX, KFS	XexSSXeMF	s F X .	•	•
LABEL	Lents a						
CEAD	X F F X . EFF X .	. SEDR	USE TR. /LAI	RAK. PHLK.	MIR, DE	GR / + MU	DF < + /
	NEIGVS 5						
970	LANAPODE I GROOM	* // *					
E OUTV	JON/SING "AING	1 1					
CND	LBh23 MOF1X						
UNERE	USETR. PHTR. /P	1030/ oke/SIH	·S• S				
LABEL	LAP23 5						
MRED2	CASECCALAMARA	PHIS.EQST.US	TRAKZ AN	•	•	HQ. 24	
	IN, IN/ND	•	104, 14	13 150 No	DRY/+P	VEC+ \$	
LAFL	LAR33 S						
C CMD	FINIS, DRY S						
SOFUT	//DRY/+TGC	#/#SOFP#/0	•	*/*	•/•	::	/•
	• •/• •						
MRFDI	CASFCC. GEON4.	DYNAPICS.CST	1/USETR.EE	DR.EQST.C	MR/+83	ASIC +	

Figure 11. A Partial Listing of DMAP Alters Generated During Automated Multi-Stage Substructuring (Obtained from D1AG 23).

ALL NUMBER OF TRANSPORT

CASE CONTROL DECK OUTPUT CONTROL

A NATION OF A DESCRIPTION OF A DESCRIPTION

LE IS

OF +. . . .





Ŷ

A LOW THE THE REPORT OF THE REPORT OF THE REPORT OF



Figure 12. Example of a Grid Point Singularity Table.

t

PAGE NAVEMBER 20, 1978 NASTPAN 11/ 3/78

22

D15PLACEMENT APPENACM, PIGID FURMAT 11 FREQUENCY - 2.0000006+01

PIST4 FREOUE			ACH, PIGID F1 0006+01	11 1 4 44				SUBCASE I
					A C C E L F P A (MAGNITUDE/PH)	7 I 0 H Asfi	V E C T O R	
P014T	10.	Tyac	11	12	11	:	21	2
	-	IJ	0.0	0°0	0.0 0.0	0.0	0°0	3.0 0.0
	~	IJ	0°0	0°0 0°0	1.671510E+02 203.4311	0.0	1.637622E+02 23.4277	0.0
	•	9	0.0	0°0 0°0	6.417696F+02 203.4246	0.0	3.076057E+02 23.4179	0.0
	•	u	0°0	0°0 0°0	1,385247E+03 203.4186	0°0 0°0	4.379512E+02 23.4100	0.0
	•	U	0.0	0°0	2.367595F+03 203.4141	.0•0 .0•0	5.4]8274E+02 23.4061	0.0
	•	U	0°0 0°0	0°0 0°0	3.542418F+03 203.4115	0°0 0°0	6,355032E+02 23,4073	0.0
	~	e	0.0	0°0 0°0	4.994042E+03 203.4110	0.0	7.133312E+02 23.4126	0°0 0°0
	æ	u	0°0	0°0 0°0	6.383551E+03 203.4122	0.0	7.729722E+02 23.4198	0*0 0
	•	ى	0.0	0°0	7.971708E+03 203.4144	0.0	8.116790E+02 23.4263	0.0
4	0	y	0.0	0°0 0°0	4.6167516+03 203.4168	0.0	8.302117E+02 23.4301	0.0
		e	0.0	0.0	1.1293236+04 203.4189	0.0	8.347499E+02 23.4312	0.0 0.0

CASE CONTROL DECK OUTPUT CONTROL

a la contrativitation a subset

en der Mitter in eine property

Figure 13. Example of Complex Vector Output in SMT1 Format.

×.

•

· . ĸ.



· 1 ar

12.00

ιì

Prog. Mar. 1895 Anno 1999

12

		>
		►
		-
		U
		C
		-
		u
•		2
FURMAT		
0191e		
ADDORACH,	5	
DISPLACEMENT	• 01-1N104	

	•		VELNCITY	VECTOR	(SOLUTION SET)		
71 #6	TYPE	11	12	13	61	R 2	R 3
0.0	w	1.0000006+01					
1.0000006-01	u	5.21A204E+00					
2.001000E-01	2.	-3.335533E+00					
3.0000005-01	u	-8.98504 RE+00					
4.000000F-01	u,	-5.8201545+00					
5.0n00c0F-01	w	1.2779306+00					
6.000000f-01		P.254471F+00					
7.000006-01	w	8.062399E+00					
5.0000000 S	u	F.474503E-01					
9.00000E-01	LL.	-7.1046095+00					
1.000000F+00	LL.	-9.1371976+00					
1.020000E+00	u	514564E+00					
1.1000005+00	-	-2.73559E+00					
1.200006+00		6.206F27E+00					
1.30000F+00	-	9.532409F+00					
1.4000005+00	6 1.7	4.173374E+00					
1.500005+00	w	-4.9478635+00					
1.60000E+00	u.	-9.604847E+00					
1.70000nF+06	L.	-5.471220F+00					
1.A00000E+CO	w	3.6470296+00					
1.90000E+00	W.	9.4424085400					
2.0000005+00	u.	A.635385E+00					

Figure 14. Example of Solution Set Vector Output in SORT2 Format.

いるのである

And A water and

*

PRINTING AND PLOTTING FEATURES

.

MOVEMBER 20, 1978 NASTRAN 11/ 3/76 PAGE 37

4

IJISPLACEMENT APPPDACH. PIGID FORMAT 12

- CL-INSHI-ID -	2		0 7 8 N 1	FLEMENT	2 C E			
						TAMAR 2	SA-HIN	H.ST
	143	542	S A 3	SA4	AXIAL-JIKE SS		SB-MIN	H.SC
		282	583	284			0.0	
•		0.0	0.0	0.0			0.0	
0.0			0°0	0.0				
	•				•	c c	-3.1730436+03	
	2013230121 -	0.0	0.0	c•0	0.0		-4.7655976+03	-1.06+00
1.3PA000E-03	-3.1/50436403		0.0	0.0		0.0		
						a a	-1-6177466+04	
		•		0.0	0.0	0.0		-1.06+00
2.77600E-03	-1.6377466+04	0.0		0.0		0.0		
	-1.631670F+04	0.0						
				c c	0-0	. 0.0	-3.1621346+04	
	-1-147134F+04	0.0	0.0			0.0	-2.572238E+04	-1.0E+00
		0.0	0.0	0.0				
		1			•	c	-2.734461E+04	
•		•	с с	0.0	0.0			-1.0F+00
4.442000F-03	-2.734461E+04	0.0				0.0		
	->.2242655+04	0.0	0.0					
					•		-2.299755E+03	-
		<	0.0	0.0	0.0			1 -1.06+00
6.94000F-03	-7.7991955405			0.0		0.0		
	-2.8203755+03	0.0					,	. 01.00
				•	с с	7.873686E+04	0.0	-1.05 -00
	2.P73686E+04	0.0	0.0	0.0		2.596165E+04	0.0	
	3 504145F+04	0-0	0.0	0.0				
						6 0035385404	0.0	-1.06+00
			0	0.0	0.0			
•.716000E-03	5.893538E+04			0.0		5.2496792+04	0.0	
	5.749679E+04	0.0	•••	6 9				
				•	•	7.4405716+04	0.0	-1.00 + 00
	7.4015715+04	. 0.0	0.0	0.0		6 305614F+04	0.0	
			0.0	0.0				
	111111111111111111111111111111111111111						c c	-1.0E+00
			c c	0-0	0.0	-0-36436309.6		
1.2492005-02	5.425293E+04	0.0				4.7799885+04	0.0	
	4 .779988E+04	• 0•0	0.0					10 10 0
			I	0	· 0· 0	8.470672E+03	0.0	-4°46-01
	8.4706775+03	0.0	0.0	0.0		9.441112E+03	0.0	
7 - 3000 - 9E • T			0.0	0.0				
	0-37777 ** *						-4.152456+0	4
		•		0.0	0.0	0.0		4 -1 -0E+00
1.576800E-02	-4.4152456+04			0.0		0.0		
1	-3.7525716+04	6 0 0		•••				
						0.0	-8.6951846+0	
	-8-695184E+04	4 0.0	0.0	0.0		0.0	-7.496687E+0	+ -1.0E+00
	-7.495687F+04	4 0.0	0.0	0*0		•		
		•					-1.0373646+0	5
	-1.017364540	0.0	0.0	0.0	0.0			
		1						

10.5-19

Figure 15. Example of Element Stress Output in SMRT2 Format.

ł

CASE CONTROL DECK OUTPUT CONTROL

CONEAN LIPHLATS

JANUARY 31, 1974 4451844 9/ 1/79 PAGE

*

ELEMENT ID = 3 1 4 6 8 9 1	19 POJNT AN	16LE = 90.000	e etric c	1 M I C A L	ب بر ۲۰ ۲۰		I T S (CCUMEAX	-
TIME	F [968 D1914VCE	STHF 89E 9 NURMAL-V	JN ELEMENT CO NUMMAL=U	UPD SYSTEM 346.44-UV	PHING II	PAL BINESSES (Majin	ang ang an) (ng ang an)	14% [11/1 UNE 44
o •	0.0 5.490001E -01	0°0	00 ••	•••	4040°0	6 6 7	• •	•••
1.0000015-01	0,0 5,000006-01	-2,295768f -04 -2,301855£-04	-2.548452F-05 -2.547488E-05	-4,/74257E-05 -1,741014E-05	-77.452/ -85.161/	-1 . 4 46285E -05 -2 . 450319E -05		1 . 1 250995 - 00 1 . 0 55 / 4nE - 00
2.000001-03	0,0 5,010900£-01	-5.176778E-04 -5.241626E-04	-4,2/546/£-05 -4,015450E-05	-1,19606)E-04 4,151951E+06	1554.07- 1224.07-	-1 • 4 5 5 5 1 - 0 - - 0 • 3 4 5 0 5 4 5 - 0 5	-5.410440E-04	2,658827£-94 2,419391E-94
5 • 000001; = 03	8.0 5.0000vn£-01	-6,449272f-04 -6,735085£-04	-1。0828086-04 -1。178525-04	-1,5798126-08 1,1602745-64	-74 . 6024 /8 . 6667	-0.554524E-05 -9.040759E-05		5.12238*5-04 5.0107875-34
4.000001 - 33	0°0 5,090000£-U1	-5.496544.04 -6.0561/45-04	-2.9475076-04 -5.1408576-94	-1,5449256-08 2,65/588E-08	-66.6/91 58.5133	-2 , 59355 /5 +04 +1 , 5906446 +04		1,85000AL-04 5,20785AL-04

10.5-20

ORIGI

E IS LITY

Figure 16. Exam a of Stress Output for Axisymmetric Conical Shell Elements.

PRINTING AND PLOTTING FEATURES

INANYO SENT	C ANALY NSTRATI		STNC ONLET	40104 NO.	ATE0 2-3-0	1001	MAS	IES	2								-	2	2		i Ç			-			
SUGS TPUC TURE	2. 4UN	2. 1	HASE	Lo RF	•																						
		2	8			*	0	•	*	-	z	ې		ш 	•	4	-	_	-	U			2	~			
							w 0 w				***		6 > L	342	ICer	3 > w	0>=			-04				2 4 R I	n ≝ ⊨ 1	*	
SUBSTPUCTURE NG. NAME	3471	S	s	11	ទ	Ĩ	n İ	~	E	••	×	×	5		e	.			•					•	-		
1 ABASIC 2 BAASIC	e e		••	• •	• •	••	••	m m	-		•	m m	6 6														
							15	2E 0	Ĩ	EA	15 6	I VEN		ŽČ.	E S	10	le a	00	ONI	ICAI	TE S	DATA	15	5 Tů	RED	1 2	AVE -
unused space	585 5	•••	232		os. Cent.							لورا أحدا	Ű														
NIGHEST PLOCH	nsfo	•	2	•									Ric													OF	OF
											•									,						POOR	RIGINAL
											LITY	· /s														QUALITY	PAGE IS

...

a line in the

and the second second

à

.....

Figure 17. Example of an SMF Table of Contents After Two Phase 1 Runs.

 $\langle \rangle$

PRINTING AND PLOTTING FEATURES

TPU NASI	TRAND 25	C AVALY	SIS CN •	USING	AUTO-	ATED 2-3-3	400×	۲. د	I I I	\$15					C	4 4 4 A		54.0	0	C.	<	T R A	1 1 2	130	2	•	2 9 8		n D	
HODI	IL REUCE	I LUCO	NF.	MOTAL	RECON	۶۶۷,	5	3, 1	HASS	~	11. 1	*																		
			r	с я С	9 9 1	L U	a D	w	۵ 0	а. Э	4	-	ی 7	۴.	 	LL.	-	6) E	u L	C	L	Ū	z	1 6	z	S				
Sens	I AUC TUPE							0.00.0	8 () f ()	0 N F F	-000	• v	****	6 > U U	0 C > r	- a a F	1 C D V		ນດາງຮ	~ ~ ~ ~	مەرى			e I m M	-450	****	****	2		i
NQ.	NANE	TYPE	22	¥.	1	3	f	•	ļ		ļ		ļ	ļ																
-	ABASIC	•	0	0	00	00	m .		~ ~		~ ~		~ ~ ~	~ ~	m r	""	~ ~						en en	 4 4	m m					
~ ~	Ne AS IC	6 8 -	00	00		5 4 (~ ~ ~ ~			6 m e) (*) F	- en e	•	•							•							
• • •	73 7076 71076	r u f			N 10 41	n 0 3	r e J		• •• •		• •• • •	~ ~	. 		•	•								_	~					
		•	•	•	ı	,		•••	126	0	1751	- 15	61 4	t tig	z	しってい	e. S	1		ŝ	0'1I	ICAT	5	ATA	15	5 T OH	1 03	* 	34234	>
578-0	ED 5PACE	5 5 5 5		91754 14 14	+ 20K	FRI.																								

Figure 18. SAF Table of Contents at the End of Phase 2 Runs.

and the standard

10.5-22

5

62

CASE CONTROL DECK OUTPUT CONTROL

where it is the state of the second

. ..

. .

28 PAGE WIVE MYER 24, 1978 1457046 11/30/78 TBUSS JYNAMIC AMALYSIS USING AUTOMATED "GOAL SYNIHESIS Mastaan demoustraticn Prinilem ni. 2-3-3

3

j -

"DDAL REDUCE" COMJINE" MODAL RECOVERY. JUN 3. PHASE 2. PF 3

SURVAPY OF CURPENT PROBLEM

4.10

				•				
•	•			307	SAN SAN			
					°,			
4		FL.			30F			
¥		• •		5	246			
Ĕ.	• • • •	• •		51 2(+00			
	• • • •	••		F.	~			
DOU REAL		rse.		160	.			
	E	• •		i d			000	2
TIC TIC	• • Ī • • Z			AT	C 69			
UCT ULT		- <u>-</u>		JRE	×			
KTIF STR	SET AS SE TRAC	• • • • •		ъ и	ž G			5
195				THI	500	==		-
VEN ET				ac.	6 <	51.		-
500				. <u>e</u>	2			-
111			1		P C A			•
•						55	5.6	ŝ
				- <u>-</u>	1150	5 7 C		
ບຼີ		2		11.	ND ND	2 4 4 2 4 2 4 2 4 2 4 2 4 2 4 2 4 2 4 2	2 CL C	1
BAS		343		~	1060			2
•	~~~~~	• <u>Z</u>		5 Å	21 8	~		m
• • 68 14	• • •	• <u></u>	•	ATE	1	5 P	SPH SH	BOR
••••		AIA		LI S	2 × 5 2 × 6 3 × 6	NE T NE T		
N. FO	• • •			1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	500	,		
				12	ž			
		TURI		ST ST	E 7 E I			
001					He - O	,		
CAT CAT	176	1961		2				
1151		Ξ.Ϋ́		2010	ň 1 N -			
				u U				
1000				224	511			
		້		ī 7		•		
200		2 0 0 2 4 3 0 6		511	NOL C	5		
ANE		× × × × × × × × × × × × × × × × × × ×		ARR	TAT OA			
2 6 9		* *		C M I	к Ч			
					iii e			
				20 1	125			
				Ī	ī			

10.5-23

Figure 19. Example of the Summary of Current Problem Table for a Reduce Operation.

100

ODIAT Infermal Consolerat Edit and the substant Consolerat	10 701MT 10 101 104 110	INTERNAL Point to		EOSS ITE				NT RTRUS	1			
Figure 20. Example of the EQSS Item for a Reduced Pseudostructure. \$\$\$ 7 YMARIC AMALYSIS USING AUTOPATED FIDDAL SYNTHESIS AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 AL REDUCE, COMAINE, MODAL PECTOFER PUN 3, PHASE 2, RF 3 </th <th>119</th> <th>AE 1410</th> <th>СОМРОИЕМТ роб 1 1 1 1 1 1 1 1 1 1 1 1 1</th> <th></th> <th>R FOR SUBS' 64.10 POINT 10 105 105 105 111 111 111 117 117</th> <th>TRUCTURE RTRU Interal Point ID 6 9 12 12 13 13 21</th> <th>5.5 C C C C C C C C C C C C C C C C C C C</th> <th></th> <th>5 6810 POIN 10 105 106 112 112 115 115</th> <th>T INTEPNAL POINT 10 10 10 13 19 19 22 22</th> <th>50 0 4999999 0</th> <th>10 10</th>	119	AE 1410	СОМРОИЕМТ роб 1 1 1 1 1 1 1 1 1 1 1 1 1		R FOR SUBS' 64.10 POINT 10 105 105 105 111 111 111 117 117	TRUCTURE RTRU Interal Point ID 6 9 12 12 13 13 21	5.5 C C C C C C C C C C C C C C C C C C C		5 6810 POIN 10 105 106 112 112 115 115	T INTEPNAL POINT 10 10 10 13 19 19 22 22	50 0 4999999 0	10 10
USS DYWAMIC AWALYSIS USING AUTOPATED MIDAL SYMTHESIS STRAN DEPOUNSTVATICN PROALEF AUTOPATED MIDAL SYMTHESIS AL REDUCE, COMMINE, MIDAL PECOVERY, PUN 3, PHASE 2, RF 3 AL REDUCE, COMMINE, MIDAL PECOVERY, PUN 3, PHASE 2, RF 3 REMAL INTERNAL CONFUMENT SISTER - SCALAR INDEX LIST FOR SUBSTVUCTURE RTWUSS INTERNAL CONFUMENT SOSS ITEM - SCALAR INDEX LIST FOR SUBSTVUCTURE RTWUSS INTERNAL CONFUMENT SOSS ITEM - SCALAR INDEX LIST FOR SUBSTVUCTURE RTWUSS INTERNAL INTERNAL CONFUMENT SOSS ITEM - SCALAR INDEX LIST FOR SUBSTVUCTURE RTWUSS INTERNAL INTERNAL CONFUMENT SOSS ITEM - SCALAR INDEX LIST FOR SUBSTVUCTURE RTWUSS INTERNAL CONFUMENT SOSS ITEM - SCALAR INDEX LIST FOR SUBSTVUCTURE RTWUSS INTERNAL INTERNAL CONFUSENT SOSS ITEM - SCALAR INDEX LIST FOR SUBSTVUCTURE RTWUSS INTERNAL CONFUSENT SOSS ITEM - SCALAR INDEX LIST FOR SUBSTVUCTURE RTWUSS INTERNAL INTERNAL CONFUSENT SOSS ITEM - SCALAR INTERNAL INTERNAL INTERNAL INTERNAL CONFUSENT SOSS ITEM - SCALAR INTERNAL COMPONENT INTERNAL OF TO THE FORM TO SSILL OF SOSS ITEM - SCALAR INTERNAL INTERNAL TO			f igure	20. Exam	wple of the	EQSS Item fo	r a Reduced F	seudostru	cture.			
DAL REDUCE. COMAINE. MODAL PECOVERY. PUN 3. PHASE 2. RF 3 TERNAL INTERNAL COMPONENT TERNAL INTERNAL COMPONENT INTERNAL COMPONENT INTERNAL INT I I	USS DYMANI Stran Depoi	C AMALYSIS - KSTRATICN P	USING AUTOPATE Roques NO. 2-3-	0 M00AL 5	SYNTHESTS		A THE ROAD	4, 197e	NASTRAN 1	1/33/76	e e	63
FRVAL INTEGNAL COLAR INDEX LIST FOR SUBSTEUATION COMPONENT	DAL REDUCE	· COMAINE.	MJDAL PECOVEPY	· BUN 3.	PHASE c. R	F 3						
	7684AL 10 10 19 19 19 19 19 19	INTF #NAL 51L 10 7 113 16 113 16 119 27 25	CONFERENT 60 2005	- H C	- SCALAR IN Internal Foint 15 5 5 14 11 22 20 22 20 22	DEX LIST FOR Internal Sil ID 3 3 14 17 17 20 23	SUBSTEUCTUR FOMENT DOF 1 1 1 1 1 1 1 1 1 1 1 1	7 ¥ USS	INTERNAL Point to 3 6 12 12 15 15	INTERNAL 511 ID 9 12 15 15 24 24	100 100 11 11 11 11 11 11	1 2 3

Figure 21. Example of the EQSS Item - Scalar Index List for a Reduced Pseudostructure.

7

PRINTING AND PLOTTING FEATURES

-

10.5-24

ORIGINAL PAGE IS OF POOR QUALITY

TRUSS STYL	HIC ANA	TSUS P	N3 160 4	AUTOMATED 1 40. 2-3-3	* JADAL 5	VNTHEST	k a		8	÷1 1975	HASTRAN 1	1/30/76	P 4 6 E	•
MOAL LED	KE. CON	•] N] Ø	HODAL	REC CVERV.	PUR 3.	PHASE 21								
					865	S ITEN	FOK SU	STRUCTURE	219455	1				
				TEPNAL	CSTN 1	•		c U	4 1 1 0 4 0	1 E S				
				JENT ID	NG.			11			÷ 1			
							••		•• 30.1.1.10.0					
				~ •					.3020275+0	2 0.				
						> 0	.230	00CE+03	0.	•				
					1			1	э.	••				
				• •0	'	-	с.		0.	•				
				•	•	-	•		•	•				
				4	1	Ļ	•		•	•				
				•	•	,	•		••	•				
				10	•	-	•		•••	50				
				11	•	- -	•••			S				
				12	• •	-	• •							
						• -				0				
				1.5	•		•		•••	••				
				90	•		••		•					
				11	• •	-								
				•••		•			.0	•				
				20	•	-	••		•••	ō¢				
				23	•	-	•••							
				22	•	-	•		5					

CASE CONTROL DECK OUTPUT CONTROL

-

PAGE

HUVEH9ER 24, 1975 HASTRAN 11/30/76

Figure 22. Example of the BGSS Itam for a Reduced Pseudostructure.

PAGE PASTPAN 11/30/76 NrVe44co 24, 1978 MCDAL REPUCE, COMMINE, PODAL RECOVERY, JUN 3, PHASE 2, RF 3 TOUSS DYNAMIC AVALYSIS USING AUTCMATED AMDAL SYNTMESIS Mastram demonstratich Podrlem VC. 2-3-3

N C R B C R S LOAD SET IDENTIFICATION LODS ITEN FOR SURSTRUCTURE PTRUSS NO LOAD SETS FOR THIS COMPONENT. NC LOAD SETS FOP THIS COMPONENT. NG LCAD SETS FOR THIS COMPANENT. 689 980 NUMBER OF LEAD SETS ----CAMPONENT NAME ARASTC **JAASIC** SSNele 44 40 31.

Figure 23. Example of the LØOS Item for a Reduced Pseudostructure.

2

.

*. . . *)

PRINTING AND PLOTTING FEATURES

5. B. S.

65

۰.

NOAL TEDUCE, COMMINE,	PUID . YAJUC JECCY	2. PHASE 2. PF					
	M.0.41.034	20039 804 FOR 85000	1#15+435-03	Stiala Jalil'i	51		
	USASE CCDÉS APE D 1 3	- PIGIO 800 - POI - Included IN 40 - Sacluded Fack - Excludes Fack	NT 1946 SET Medal Set Medal Set	96 <u>2</u> 8485 9668465 96	MU-PARTEC	. 24 F I GN	
	3004		USAGE	u 1 aŭ			
	AL'FEE	CVCLES	COPE	61 1210a	511		
		3.5957736+00	-	161	40		
	~ ~	1.756351 .+01	1	201	ð		
		2.4492195+01	1	111	10		
	r	3,950602E+01	-	1.74	11		
	¢۲	6.109854E+01	1	135	12		
	•	+.92r614E+01	p=4	175	13		
	~	K.4453675+01	1	101	14		
	a	9, FF97536+C1		178	51		
	σ	I.117643E+02		179	15		
	91	1.2334945+02	-	110	17		
	11	1.2755534.02	-1	111	Γα		
	12	1.3035975+92	-1	211	19		
	6 1	1.3492176+02	-1	113	20		
	14	1.5350555+02	-	114	12		
		1.6261935+32	-1	115	22		
	15	1.403714E+92	•4	116	23		
	17	2.0070175+02		111	24		
	1	2.68550¢£+02	7	911	25		
	19	2.7398525+92	•				
	64	4.5877245+32	~1				

CASE CONTROL DECK OUTPUT CONTROL

\$\$

PAGE

4-VENACO 24, 1975 1421044 11/30/76

2

ίì,

and the second of

Figure 24. Example of the Modal DØF Summary for a Reduced Pseudostructure.

•••• •

TRUSS DYNAMIC AVALYSIS USING AUTCHATED "OFAL SYNTHESIS Vastram demonstration Prialem No. 2-1-3

MÜVEMMED 24, 1978 NASTPAN 11/30/78 PAGE

<u>ر</u>

52

NOOAL PERUCE, COMBINE, MODAL RECOVERY, JUN 3, PHASE 2, 4F 3

NTEPNAL Print No 1	INTERNAL 90F NO 1	DEGREES CF FFEEDCM 12	SUFFARY ++++++++++++++++++++++++++++++++++++	DE PSEUDOSTRUCTURE CONNECTIVITIES 400000000 P S E U O N S T R U C T U P E N A M E S 000000000000000000000000000000000
~	ſ	12	ABASIC	
F	ur.	12	A 45 I C	
•	7	•	MA 101	
\$	•	1	4.A 102	
Ð	σ	p i	M.K 103	
٢	10	1	MA 104	
£	11	1	78 105	
o	12	1	ra 106	
13	61	I	MA 107	
	1	12	AMASIC BBI	1510
-1	16	12	AAASIC 68	15 [C

Figure 25. Example of Grid Point Connectivity Output for a COMBINE Operation.

: ;

,

PRINTING AND PLOTTING FEATURES

Solute AND RECTVENT DUM 4. PHASE P. R. S. COMPONENT REASTC 1144 2.0000006-01 DI S P L A C E 4 E M T V F C T 0 P DI S P L A C E 4 E M T V F C T 0 P P. S. P. S. <th>TRUSS SYNAMIC ANALYSI Mastram demonstration</th> <th>S USING AUTOMA I PRJ9LEP NC. 2</th> <th>TED HODAL SYNTH</th> <th>HE 2 I 2</th> <th>Sens</th> <th>STRUCTURE</th> <th>89451C</th> <th></th>	TRUSS SYNAMIC ANALYSI Mastram demonstration	S USING AUTOMA I PRJ9LEP NC. 2	TED HODAL SYNTH	HE 2 I 2	Sens	STRUCTURE	89451C	
1146 2.6000006-01 n I S P L A C E 4 E N T V F C T D P 2 11.2460435-02 6.53341835-02 0.0 0.0 0.0 2 6 -1.2460435-02 0.0 0.0 0.0 0.0 0.0 2 6 -1.2460435-02 0.0 0.0 0.0 0.0 0.0 0.0 0.0 2 6 -1.2460435-02 0.01 0.0 <t< th=""><th>SOLVE AND RECOVERY, P</th><th>UN 4. PHASE 2.</th><th>RF 3</th><th></th><th>1460</th><th>PONENT</th><th>REASIC</th><th></th></t<>	SOLVE AND RECOVERY, P	UN 4. PHASE 2.	RF 3		1460	PONENT	REASIC	
1 1	T146 • 2.600006-01		r J d S I C	T N S N S O 1	γ ε ζ τ η ε			
1.2460437-02 6.5341835-02 6.5341855-02 6.5341855-02 <t< th=""><th></th><th>11</th><th>1</th><th>11</th><th>14</th><th>32</th><th></th><th></th></t<>		11	1	11	14	32		
1 2 4 2 0.0 0.0 1 2 4 3 0.0 0.0 0.0 1 4 3 0.0 0.0 0.0 0.0 1 4 3 0.0 0.0 0.0 0.0 0.0 1 4 3 0.0 0.0 0.0 0.0 0.0 0.0 1 5 1 5 7 7 7 0.0		1 . 2440435-02	A.634183F-02	0.0	0*0	0.0	0.0	
1.246043E-02 8.759338F-02 0.0 0.0 11 6 1.430501E-02 8.759338F-02 0.0 12 6 1.430501E-02 8.759338F-02 0.0 11 5 -4.131227E-15 9.89739F-02 0.0 12 6 -4.131227F-15 1.030199F-02 0.0 0.0 21 5 -1.430501F-02 8.75938F-02 0.0 0.0 0.0 21 5 0.0017F-15 1.030199F-01 0.0 0.0 0.0 0.0 21 5 1.5506017F-02 1.030199F-01 0.0 0.0 0.0 0.0 22 6 1.5506017F-02 1.030199F-01 0.0 0.0 0.0 0.0 23 6 1.3956667F-02 1.31677F-01 0.0 0.0 0.0 0.0 31 7 6 1.31677F-01 0.0 0.0 0.0 0.0 4 1 556667F-02 1.31977F-01 0.0 0.0 0.0 0.0 5 1.5556667F-02 0.0 0.0 0.0		2.4270035-15	6.535092F-02	0.0	0.0	0.0	0.0	
11 6 1.430501E-02 8.759338F-02 0.0 0.0 21 6 -1.430501E-02 8.759338F-02 0.0 0.0 21 6 -1.430501E-02 8.759338F-02 0.0 0.0 21 6 -1.430501E-02 8.759338F-02 0.0 0.0 0.0 21 1.526667E-02 1.039199F-01 0.0 0.0 0.0 0.0 21 5 -4.63050177F-15 1.110156F-01 0.0 0.0 0.0 22 6 -4.6305776-01 0.0 0.0 0.0 0.0 0.0 23 6 -4.6305776-02 1.01195776-01 0.0 0.0 0.0 0.0 23 6 -4.630576-02 1.01105776-01 0.0 0.0 0.0 0.0 31 7.55069576-02 1.3167776-01 0.0 0.0 0.0 0.0 41 5 -4.639576-02 1.3167776-01 0.0 0.0 0.0 5 -4.6395776-02 1.3167776-01 0.0 0.0 0.0 0.0 6 </td <td></td> <th></th> <td>6.4341P3F-02</td> <td>0.0</td> <td>0.0</td> <td>0.0</td> <td>0.0</td> <td></td>			6.4341P3F-02	0.0	0.0	0.0	0.0	
1 5 -4.1312276-15 9.6977395-02 0.0 0.0 21 6 -1.4335016-02 n.7593366-02 0.0 0.0 22 6 -1.4335016-02 1.0991996-01 0.0 0.0 22 6 -4.6335016-02 1.0991996-01 0.0 0.0 0.0 23 6 -1.5266626-02 1.0391996-01 0.0 0.0 0.0 0.0 21 6 -4.639576-02 1.319776-01 0.0 0.0 0.0 0.0 23 6 -4.6395716-15 1.319776-01 0.0 0.0 0.0 0.0 31 6 -1.55666762 1.317776-01 0.0 0.0 0.0 0.0 31 6 -1.55666762 1.317776-01 0.0 0.0 0.0 0.0 31 6 -1.556667662 1.317776-01 0.0 0.0 0.0 0.0 41 6 -1.5576967676 1.317776-01 0.0 0.0 0.0 0.0 42 6 -1.5370986766-02 1.3177776-01 0.0		1.4104016-02	R. 75933PF-02	0.0	0.0	0.0	c •0	
23 6 -1.53001E-02 A.759378F-02 0.0 0.0 21 5 -1.520602E-02 1.039199E-01 0.0 0.0 23 6 -1.520602F-02 1.039199E-01 0.0 0.0 23 6 -1.520602F-02 1.039199E-01 0.0 0.0 0.0 31 6 -1.520602F-02 1.011552F-01 0.0 0.0 0.0 33 6 -1.520602F-02 1.311875F-01 0.0 0.0 0.0 31 6 -1.550652F-02 1.311875F-01 0.0 0.0 0.0 31 6 -1.550652F-02 1.31977F-01 0.0 0.0 0.0 41 6 -1.5512511571271 1.31977F-01 0.0 0.0 0.0 42 6 -1.5576957E-02 1.53793919E-01 0.0 0.0 0.0 0.0			4.878739E-02		0.0	0.0	0.0	
21 6 -4.6601775-15 1.0391995-01 0.0 0.0 23 6 -4.6601775-15 1.1315425-01 0.0 0.0 23 6 -4.560627-02 1.0391995-01 0.0 0.0 31 6 -1.5266627-02 1.0391995-01 0.0 0.0 0.0 33 6 -1.5266627-02 1.31947575-01 0.0 0.0 0.0 33 6 -1.55666276-02 1.3194775-01 0.0 0.0 0.0 41 6 -1.55669276-02 1.3194775-01 0.0 0.0 0.0 42 6 -1.55669276-02 1.3194776-01 0.0 0.0 0.0 41 6 -1.55969276-02 1.3194776-01 0.0 0.0 0.0 42 -1.55969276-02 1.319778-01 0.0 0.0 0.0 0.0		1 - 1 - 1 - 0 - 0 - 0 - 0 - 0 - 0 - 0 -	A. 759338F-02	0.0	0.0	0.0	0.0	
23 6 -4.6601775-15 1.101542E-01 0.0 0.0 23 6 -1.526662E-02 1.000109E-01 0.0 0.0 31 6 -1.526662E-02 1.000109E-01 0.0 0.0 33 6 -1.556662E-02 1.31977E-01 0.0 0.0 0.0 33 6 -1.556662E-02 1.31977E-01 0.0 0.0 0.0 41 6 -1.556652E-02 1.31977E-01 0.0 0.0 0.0 42 6 -1.576652E-02 1.31977E-01 0.0 0.0 0.0 41 6 -1.576652E-02 1.531991E-01 0.0 0.0 0.0 42 5 -4.693382E-15 1.531991E-01 0.0 0.0 0.0			10-301050 1	0-0	0.0	0.0	¢•0	
23 6 -1.5266627-02 1.0791997-01 0.0 0.0 31 6 -1.5266627-02 1.317777-01 0.0 0.0 0.0 31 6 -4.8357316-15 1.3177777-01 0.0 0.0 0.0 31 6 -1.55664267-02 1.3177777-01 0.0 0.0 0.0 31 6 -1.55664267-02 1.3177777-01 0.0 0.0 0.0 41 6 -1.55664267-01 0.0 0.0 0.0 0.0 0.0 41 6 -1.57695777-01 0.0 0.0 0.0 0.0 0.0 42 5 -4.83579367-01 0.0 0.0 0.0 0.0 0.0 43 5 -4.8357876-01 0.0 0.0 0.0 0.0 0.0		4 . 4 6 0 1 7 7 5 - 1 5	1.1315626-01	0.0	0.0	0.0	0.0	
31 6 -4.036740E-02 1.31777E-01 0.0 0.0 0.0 32 6 -4.035731E-15 1.31777E-01 0.0 0.0 0.0 33 6 -1.556870E-02 1.31777E-01 0.0 0.0 0.0 33 6 -1.556870E-02 1.31777E-01 0.0 0.0 0.0 0.0 41 6 -1.55857E-02 1.537938E-01 0.0 0.0 0.0 0.0 42 5 -4.833382E-17 1.5337938E-01 0.0 0.0 0.0 0.0 43 5 -4.576657E-02 1.5337938E-01 0.0 0.0 0.0 0.0		-1-5266626-02	1.0391991	0.0	0.7	0.0	0.0	
32 6 -4.833731E-15 1.319476F-01 0.0 0.0 0.0 33 6 -1.5568846E-02 1.319477E-01 0.0 0.0 0.0 41 6 1.575692EE-02 1.532938E-01 0.0 0.0 0.0 42 5 -4.875657E-02 1.532938EE-01 0.0 0.0 0.0 43 5 -4.575657E-02 1.532938EE-01 0.0 0.0 0.0 43 5 -4.575657E-02 1.532938EE-01 0.0 0.0 0.0		1.9666406-02	1.3187576-01	0.0	0.0	0.0	C•0	
33 6 -1.566846E-02 1.31x757E-01 0.0 0.0 0.0 0.0 41 6 1.575652E-02 1.533938E-01 0.0 0.0 0.0 0.0 42 6 -1.576652E-02 1.5337938E-01 0.0 0.0 0.0 0.0 43 7 -1.576652E-02 1.5337938E-01 0.0 0.0 0.0 0.0	1 	-4.8357316-15	1.3198705-01	0.3	0.0	0.0	0.0	
41 6 1.575652E-02 1.532938E-01 0.0 0.0 0.0 42 6 -4.093382E-15 1.5331196E-01 0.0 0.0 0.0 0.0 43 7 -1.576652E-02 1.5337936E-01 0.0 0.0 0.0 0.0		-1.56684CE-02	1.3147576-01	0.0	0.0	0.0	0.0	
•2 6 -4.093382E-1f 1.533119E-01 0.0 0.0 0.0 •3 -1.576652E-02 1.532938E-01 0.0 0.0 0.0	1	1.5756526-02	1.5329386-01	0.0	0.0	0.0	0.0	
		-4. A933R2E-15	1.5311196-01	c.0	0.0	0.0	0.0	
		-1.5766525-02	1.5329386-01	c•0	0.0	0.0	C-0	

Figure 26. Example of Displacement Vector Output Requested Under a PRIMT Subcommand.

www.gurpe

CASE CONTROL DECK OUTPUT CONTROL

.

h

TOUSS DYAARIC AMALYSTS USING AUTOMATED AMOAL SYNTHESIS Mastbaa deadustration Poddlef NG. 2-3-4

NOVEMBER 74, 1972 MASTRAN 11/30/78 PAGE

163

ŝ

à

7

4

. .

÷

•

The second se

SOLVE AND RELEVERY, RUN 4, PHASE 2, RF 9

PPDAL COORDINATE ENERGIES FOR SUBSTPURTIPE MB

+ 3HE +	0-300006+0	1	י				
C1 89	TVFF	3004	F2EQUENCY	KINETIC	KE /TOTAL	POTENTIAL	PEITOTAL
101	14-H0DE	-1	1.1784496+01	1.1321316-02	4106.	2.9635785-01	E699.
102	3004-W1	•	5.715336E+01	7.5947245-03	1100"	1.7622106-04	.0036
103	IN-HODE		t.452575E+01	-6.1267666-27	COUC -	3.1657145-20	0000.
104	3-0	4	9.4355EF01	4 - 25 50965 - 44	10.0.	7.0943676-06	のかうい。
104	100M-M1	•	1.234045442	7.4202906-05	.0000	1.6835246-06	0000.
106	300x-x1	٩	1.4404545+02	3.2925435-06	000C.	5.1519325-08	.0000
107	14-POCE	•	1.5149406+02	4.221955E-30	0000.	6.413725E-30	.0000

DTAL E	INE OF FOR THE	S VECTOR		7.0903976+00	1.7000	2.985652E-01	1.0090

Figure 27. Example of Energy Output Requested Under a PRINT Subcommand.

PRINTING AND PLOTTING FEATURES

MUVEMAER 20, 1978 NASTRAN 11/ 3/78 PAGE

20

ì

-

ì

CUTPUT FPOR GPID POINT HEIGHT GENERATOR

PEFERENCE POINT .

c

: ٠ ٠ ٠ ٠ •••

 2.3120607.00
 -1.047315f-16
 0.
 1.256603f-15
 1.073122f-01
 -1.594735f-80

 7.004457f-17
 2.317069f+00
 0.
 -1.073122f-01
 -8.37688f-17
 2.332196f-80

 0.
 2.317069f+00
 0.
 -1.073122f-01
 -8.37688f-17
 2.332196f-80

 0.
 2.317069f+00
 1.594735f+00
 1.594735f+00
 -2.332196f+01
 0.

 1.009229f-19
 -1.073172f+01
 1.594735f+00
 6.741736f+01
 -1.010790f+01
 -1.0380476f+02

 1.0073127f+01
 4.199346f-16
 -2.332196f+01
 -1.610790f+01
 2.962A81f+02
 -7.379832f+00

 -1.994735f+00
 7.332196f+01
 0.
 -1.610790f+02
 -7.379832f+00
 2.377293f+00

 4.641393E+00 4.641393E+00 2-0.6. • I-C.6. 9.435776-16 6.8974306-01 1.0087056+01 -3.6230976-17 1.0087056+01 6.8974306-01 HO - RIGID BODY MASS MATRIX IN BASIC COMPDIMATE SYSTEM - TPANSFORMATION MATPIX FOR SCALAR MASS PARTITION *** ٠ ۰ 1.000000F+00 + ••• ٠ *** 1.150459f+01
 2.171737f-07
 2.171737f-02
 1.121044f+01
 2.194100f-02
 -2.037970f-01
 -2.194100f-02
 1.379537f+00 I(S) - INERTIAS PFLATIVE TO C.C. I(0) - PRINCIPLE INFETIAS • 0. 1.9000005+00 0. 2.3120696+00 2.3120696+00 2.3120696+00 1.0000006+00 SSA. • • : NASS ANTS SYSTEM (S) : : • ~ DIFFCTION -: :

> URIGINAL PAGE IS OF POOR QUALITY

CASE CONTROL DECK OUTPUT CONTROL

10.5-31

j,

Figure 28. Meight and Balance Output.

• •

-9.973675E-01

7.249831E-04

7.2479105-07 -2.0206765-02

• -7.188040F-03 • -4.997956F-03

9.471652F=01 7.250850E=02

• -2.0101026-02

•••

٠

2

٠

1.1208885+01

- TPANSFRRMATION MATRIX -- I(0) - 0T+I(5)+0

10+3065151.1

1.375392F+00

.

PRINTING AND PLOTTING FEATURES

. .

ŀ

ì

•

PAGE

MOVEMBER 20, 1978 NASTRAM 11/ 3/78

	VALUE	.3798E-11
79 67		12 - 3.6 13 - 7.6
3/76	1W104	• 1
NASTRAN 11/	VALUÉ	3.35276E-08 7.77596E-11
1978		
20°	[N]24	~ 4
NOV EN RE	VALUE	91038E-11 86265F-08
STRAN NO. 52	POINT	8 T2 -2+
OF UR NA	VALUE	27996E-00
רובף – אח נהגםS	7 M 104	7 F3 -7.
INERTLA REI	VALUE	7-72). 79965-12 52765-0 8
- 1638	r 2	1 (7 72 7.2 0 83 -3.3
EN CELL	10n ••	•
	•	e of Unit

Figure 29. Example of Residual Load Vector Output.

1994 - 2 - 4 - 1 - 144

AGVEM36 × 27, 1979 NASTMAN 11/30/74 PAGE

22

. e. 1

...

9E AM	
CELL	
2	
-	
ð	7
75.1S	1-11
ANAL	ND.
RANDON	PADBLEY
ANO	N.
ESPONSE	I WAL SHOT
	E 00
	STRAL
2	3

MATREN I	DNID) VING	1 NAME 101) 1	IS A REAL	NEULOJ C	NDH 06 X	RECTANG MAT	R]X.		
COLURN 1	RONS	1 THRU	30						
-6.793236-01 3.926036-32 3.498986+00	1.02379E-32 4.11330E+00 3.99304E-01	1.33448E+00 -2.09924E-01 6.36054č-32	-6.40077E-01 4.73194E-32 2.54215E+00	2.04439E-32 4.32499E+00 5.44560E-01	2.54715E+0U -2.92425E-07 6.62723E-32	-5.49590E-01 5.42055E-32 1.33648E+00	3.02270L-32 4.11331c+00 6.46077L-01	3.44898E+00 2.09528E-01 6.71147E-32	-3.74304E-01 5.47467c-32 6.74324c-01
COLUMN 2	RONS	1 THRU	30		****		8		
1.35748E+00 -1.0+202E-32 4.11331E+00	-2.74646E-33 -2.54218E+00 -4.19505E-01	-2.54213E+00 -1.09823E+00 -1.69344E-32	1.09423E+00 -1.2593E-32 4.11333E+00	-5.42608F-33 -3.3460%F-05 4.194655-01	-4.11327E+00 -1.35750E+00 +1.75922E-32	4.19497E-01 -1.43864E-32 2.54218E+00	-8.02264E-33 2.54214E+00 1.09425E+00	-4.11329E+00 -1.09825E+00 -1.78131E-32	-4.19476E-01 -1.54575E-32 1.35751E+00
COLUMN 3	ROWS	I THRU	30				+		
-2.028206+00 3.384726-29 1.338496+00	8.921136-30 -2.542166+00 -1.928886+00	3.49903£+00 1.64083£+00 5.50084£-29	-1.14215€+00 4.07454€-29 4.11326€+00	1.702915-29 -4.324976+03 -6.267296-01	4.11335E+0U -9.62900E-06 5.71435E-24	6.267316-01 4.573186-29 3.448936+00	2.60594E-29 -2.54214E+00 1.19212E+00	1.336J1t+03 -1.64062E+03 >.75611t-29	1.428425+66 5.126655-29 2.026145+00
COLUMN 4	ROWS	1 THRU	30						
2.67209E+00 -1.47479E-24 -2.94215E+00	-3.887126-25 4.113246+30 -2.161906+00	-4.113276+00 8.257406-01 -2.396836-24	8.25717E-01 -1.77754E-24 2.54219E+U0	-7.679646-25 -3.1474AE-05 -2.16179E+00	-2.54213E+CU 2.67211E+OU -2.48986E-24	-2.16178E+00 +2.03620E-24 4.11332E+00	-1.13546c-24 -4.11332£+00 8.25730£-01	2.54217E+00 8.23715E-01 -2.52113E-24	-2.16177£+00 -2.24435c-24 2.07213c+60
COLUMN 5	ROAS	I THRU							
3.24376E+00 -1.62672E-08 4.32496E+00	-4 .28736E-09 -3 .27140E-03 2.35985E-05	-4.325016+00 3.243746+00 -2.643746-09	-1.44555F-05 -1.96206E-08 -2.03443E-45	-8.47077E-09 -4.3249AE+00 3.24371E+00	1. 31207E-05 -2.91343E-05 -2.74636E-08	-3.24376E+UU -2.24546E+UU -4.32494E+00	-1.252436-08 3.293316-05 -9.013086-06	4.325006400 -3.243726400 -2.740446-08	7 2.340495-05 -2.475565-66 -3.243705+00

10.5-33

THE NUMBER OF NON-ZERO JORDS IN THE LUNGEST RECOPD -

30

THE DENSITY CF THIS MATRIX IS 100.00 PERCENT.

Figure 30. Matrix PHIA Printed by DMAP Module MATPRN.

CASE CONTROL DECK OUTPUT CONTROL

and the second second participants and the second
TRUSS JYWAMIC ANALYSI' USING AUTOMATED YUJAL SYNTHESI'S Wastran nemonstraticn proelem nd. 2-3-3	N NVE 485	24. 1978	P. ASTHAN 1	1/30/76	PAGE	4
MODAL BEDUCE. COMMINE, 409AL RECEVEPY, 20N 3, PHASE 2, PF 3						
EIGENVALUF ANALYSIS S		ICLVFNS M	E THOD)			
NUMEER OF EICENVALUES EXTPACTED	• • •	24				
MLMRER DE EICENVECTORS COMPUTED	• • • •	2				
MUPPER OF ELGENVALUE CONVERGENCE FA	· · Sabill	0				
NUMAER CF EIGFNVECTOR CJNVERGENCE F		o				
REASON FJR TERFINATION	• • • •	T				
LAPGEST DFF-DIAGOVAL MODAL 4ASS TF2	zM 0.					
	•	o				

•

Š

PRINTING AND PLOTTING FEATURES

Э

0

NUMMER OF DFF-DIAGONAL MCDAL MASS TEPMS FAILING CPITEPIQ4. • • •

MUDE PAIR. . .

* • • •

Figure 31. Example of the Eigenvalue Analysis Summary Table.

3

٠,

7. . .

TOUSS DYNAMIC ANALYSIS USING AUTOPATED MMDAL SYNTHESIS Mastran Cfmunstaaticn Ponglep nn. 2-3-3

99

PAGÊ

AASTRAN 11/30/76

d261 452 bisaniAuk

2.1

2

MODAL REDUCE, COMPLKF, MODAL RECCVEPY, PUN 3, PHASE 2, PF 3

Ξ
-
_
2
z
ę
-
•••
-
_
1
u
~

			PEAL EIGS	. N V A L U F S		
NC)E ND.	EXTRACTION Deger	EIGENVALUE	RAUIAN FPEQUENCY	CYCLIC Frequenty	GENFRALIZED Mass	GENERALIZED STIFFNESS
-	24	7.5018165+03	F. 6613035+01	1.3784896+01	3.01240CE-C1	2.2605976+03
~	"	1.0742145+05	3.2775216+32	5.2163345+0]	5.1536956-01	5.5361725+04
~	<i>د</i> ر	1.6437125+05	4.054272E+u2	6.432575F+01	3.7153266-01	5-4494421+34
*	12	3.705307E+05	6.9E7124E+02	9.6479FPE+01	4.151166F-C1	1.5414605+05
•	6 2	A.0124255+C5	7.753983E+C2	1.2340B5F+n2	4.2584541-01	2.5503646405
¢	19	P.6549415405	9.303221F+02	1.440545+07	3.4849465-6.1	3.149210F+15
~	19	9.05327CF+C5	5.51 E320E+02	1.514440E+02	1.3921246-01	1.2221036445
•	17	1.3530596+06	1.1637105+03	1.3513075+02	C_0	0.0
•	16	1.6127346+05	1 . 2699356 +03	2.0211455+07	0.0	0.0
10	15	1.836036E+96	1.354993£+03	2.156539F+02	0.0	
11	*1	2.027653E+06	1.423957c+03	7.25629FE+02	C • D	
12	13	2.0294946+06	1.424601E+03	2.2073235+02	0.00	
[]	12	7.434966E+06	1.57637ÄE+03	2.50H8A3F+02	0.0	0.0
1	11	2.751799F+06	1.656P55E+03	2.6401495+62	0.0	0.0
15	10	3.4588235+06	1.H59791F+03	2.9599505+72	C *3	0.0
1	o	3.5621775+C6	1.dA7373E+03	2.00344F+07	0.0	
17	8 0	3.621002E+06	1.902693E+03	3.02H54aF+02	0-0	
-	~	4.26?82CE+C6	2.0646605+03	3.2360785+72	0.0	0.0
2	£	4.466223F+06	2.1133446+03	3.3634015+52	0.0	0.0
20	s.	4.5117POE+C6	2.124095E+03	3.340602E+07	0.0	
21	4	4.771722E+06	2.144276+93	3.4766245+02	0.0	0.0
22	•	5.303790E+06	2.302996E+03	3.6653325402	0.0	0-0
2	~	5.612925E+06	7.369161E+03	3.770637E+02	C. 0	0.0
24	-	7.754982E+06	2.7847776+03	4 • 4 32113E + 02	c.0	0.0

CASE CONTROL DECK OUTPUT CONTROL

.

AND A CALLAR

Figure 32. Example of the Real Eigenvalue Output Table.

*****.

~ PAGE KASIMAN 11/30/78 0.00, AXES = +X++Y++Z+ SYMMETRIC NUVERREK 27, 1978 Figure 33. Plot Data Summary for One Plot. (INCHES) MESSAGES FROM THE PLOT MODULE ORTMOGWAPHIC PROJECTION Rotations (degres) — Gamma = 20.00, beta = 30.00, alpha = scale (dbject-to-plot size) = 8.351734e-02 X0 = -8.679378E-01, Y0 = 1.540936E+00 PPHENICAL MALL WITH PALSJUKE LUAUINC, NU MOMENTS UN BUUNDARY Mastran demonstration problem NG. 1-2-1 PAPER 512E = 7.5 X 7.5, PAPER TYPE = VELLUM THE FOLLOWING PLOTS ARE FOR AN SC-4020 PLUTTER AN END-DF-FILE MARK FOLLOWS THE LAST PLOT 0 A T A ŧ DATA -4 ENGINEERING OR LG IN PL011ER

10.5-36

PRINTING AND PLOTTING FEATURES

1.1.1.1

and the first second second stars of the second second second

د

.....

FIND SCALE'SET 2, DRIGIN 1000 FIND SCALE'URIGIN 1000, SET 1,VANT POINT,REGION 0.35,00.1, 0.9, 0.6 PITTLE - SECTION PLOTEL ELEMENTS (PERSPECTIVE PRUJECTION) PLOT SET 2, ORIGIN 1000, LARELS PLOT SET 2, DRIGIN 1000, SYNGLUS 9, SHAPE, PLOT SET 1, DRIGIN 1000, SYNGULS 9, SHAPE, SET 1, DRIGIN 1000 SYNGULS 9 SHAPE, SET 1, URIGIN 1000 SYNGULS 9 SHAPE TITLE • SPHERICAL SMELL WITH PRESSURE LOADING, MD MOMENTS ON BOUNDARY Subtitle • Mastram Dlmonstmation Problem ND. 1-2-1 Load • 1 VIEW 20.0, 30.0, 0.0 FIND SCALE DWIGIN 1 SET 1 PTITLE - UNDERLAMED SECTION TRIAZ ELEMENTS PLOT LABEL(BOTH), SYNBOLS 6 PLOT LABEL(BOTH), SYNBOLS 6 PTITLE - SECTION TRIAZ ELEMENTS WITH UNDERLAY PLOT STATIC DEFOMMATION 0.1, SET 1, ORIGIN 1,5MAPE, LABELS 0 I ں بد ¥ 0 9 0 PLOTID - MAŠTRAN DEMONSTRATION PROBLEM ND. 1-2-1 Output(PLOT) SPMERICAL SMELL WITM PRESSURE LOADING, NU MOMENTS ON BOUNDARY Mastram demonstration Problem ng. 1—2—1 **CONTROL** PLOTEL - EDGES AND CENTERLINE Set 2 • PLOTEL PERSPECTIVE PRUJECTION MAKINUM DEFORMATION 6.0 SET 1 - ELEMENTS TALA2 CASE MAKEMUN DEFORMATION 6.0 500 - 2 ALL ELENENTS 578655 - ALL DISP - ALL SPCF - ALL PLUTTER SC 001PU1 -. COUNT CARD . 2 1

Example of a Structural Plot Packet in the Case Control Deck. Figure 34.

CASE CONTROL DECK OUTPUT CONTROL

PAGÈ

MASTRAN 11/30/78

NGVENBEK 27, 1978

ì
PAGE NASIKAN 11/30/76 NGVENBER 27. 1978 **AEADELASTIC FLUTTEN ANALVSIS DF A FIFTEEN DEGREZ SWEPT WING** M**astran demonstration Problem ND. 10+2→1**

K VALUÉS .2001*) .167(0) .143(1) .125(2) .111(3) .130(4)

CASE CONTROL DECK ECHO

and the state of the

~

•

¢

.

TITLE • AERDELASTIC FLUTTER AMALYSIS DF A FIFTEEN DEGREÉ SWEPT WING Subtitle • Mastran demunstration pruntem nd. 10-2-1 Label • K values •20014) •167(0) •143(1) •125(2) •111(3) •10014) OUTPUTERYOUT) XIIILE = VELOCITY XIIILE = VELOCITY XIIILE = FREQUENCY (L) YBIILE = FREQUENCY (L) FCUARE = V-G AND V-F UAIA PUINTS TCUARELINESYMBOL--1 XYPAPERPLOT VG / 1(6,F) 2(6,F) 3(6,F) 4(6,F) 3(6,F) 6(6,F) BEGIM BULK LABEL - K ECMJ - 807H 596-1 NE THOD-10 C4c THOD-20 FAc THOD-20 CAND

Figure 35. Example of an XY Plot Packet in the Case Control Deck.

PRINTING AND PLOTTING FEATURES

Sec. Sec.

10.5-38

FRÉQUENCY FESPONSE AND RANDUM AMALYSIS OF A 10 CELL BEAM Mastran démunstration prublem no. 11-1-1

102

PAGE

NASTRAN 11/30/76

NOVEMBER 27. 1478

S U M M A R Y x y - 0 U T P U T

5.003758E+01 400T MEAN SQUARE VALUE - 3.742491E+02 Freduency de 2640 Crussings (n 2580) -Pouer-Spectral-density-function (f50f) displacement cuave of 9)

XY-PAIAS BETWEEN MMIM AMU XMAX WILL BE PRINTED XY-PAIAS WITMIM FRAME LIMITS WILL WE PLOTTED PLOTTER SPECIFIED IS SC 4020 PENSILE • 1

-THIS IS CUAVE I UF WHOLE FRAME

CUMVE TITLE = POWER SPECTRAL DENSITY OF POINT & DISPLACEMENT H-AKIS TITLE = T-AKIS TITLE = 5

THE FULLOWING INFURMATION IS FOR THE ABLVE DEFINED CURVE DNLY.

10.5-39

A 15 0F 4

* 0 L C 8 8 œ 0 4 ATA • • • • • • • • • Ż.

•

×,

.

۰,

4

Example of the XY Output Summary Table for One Curve.

Figure 36.

ł

ķ

.

.

1

с к к к 2141

4

ĸ × × ∍ \$ ي. د r 20

CASE CONTROL DECK OUTPUT CONTROL

FREQUENCY RESPONSE AND PANUOM ANALYSIS OF A 10 CELL BEAM NASTRAN DEMONSTRATION PROBLEW NO. 11-1-1

DISPLACENEN.

i,

						•										17																					
	CAPD NUMBER																																				
WHOLE FRAME	Y-VALUE	6.019458E-01	6.122720E-01	6.451568E-01	7.071608E-U1	4.1274176-01	9.93121690	1.3224506+00	2.0124936+00	3.9003166+00	1.366200£+01	2.7974226+04	9.381165£+00	2.077693E+00	5.122136E-01	4.0342956-01	2.2963756-01	1.4311785-01	9.5335666-02	6.690560E-02	4. P99345E-02	3.717110E-02	••	••	.0	•0•	•••	•	• •					•••	0,	•••	
COMPONENT = 5	X-VALUE	o.	5 .000000E + 00	1.0000006+01	1.5000005+01	2.000006+01	2.5000004.01	3.00000E+01	3.500000401	4.000006+01	4.5000006+01	10+30000004-c	5.500000E+01	6.000000E+01	••\$00000E+01	7.0000006+01	7.500000E+01	4.000000E+01	d.500006+01	9.000000E+01	9.500000E+01	1+0000006+02	1.0500005+02	1.1000006+02	1 • 1 50000E + 0 2	1 • 200000 E + 02	1 • 250000 E + 02	L.SUUUUEFUZ	1 400006403	1 - 4 500005 - 02	1.500006+02	1.5500006402	1.600000f+02	1.6500000000	1 . 700000E+02	1.750006+02	1 . 4000005 +0.2
URVE ID - 6	PAINT NURBER	-4	2	e	•	٩	٥	~	3	•	10	11	12	13	4 1	15	16	17	18	19	20	21	22	23	5	93	0 7	21	0	30		26	33	34	35	36	37

Figure 37. Example of a Table of XY Pairs Scheduled for Plotting.

٢

.

PRINTING AND PLOTTING FEATURES

an a series and a second second second second second second second second

103

PAGE

NASTRAN I ! . . . 78

NOVEMBER 27. 1978

;

10.5-40

SUBSTRUCTURE CONTROL DECK OUTPUT CONTROL

- Contraction - Mathematica - Landson - Landson - Landson - Mathematica - Mathematica - Mathematica - Landson -

10.6 SUBSTRUCTURE CONTROL DECK OUTPUT CONTROL

1

NASTRAN provides the automated substructuring user with facilities to print useful summaries of substructuring bookkeeping information, the status and contents of the substructuring data base file (SØF), descriptions of the results of Substructure Control Deck commands, as well as the normal engineering information of analysis results such as displacements, velocities, accelerations, forces of constraint, and applied loads.

Structural analysis results at the basic substructure level, including element forces and stresses, grid point force summaries, omitted degrees of freedom displacements, etc., are not printed using Substructure Control Deck commands. These data are requested using Case Control Deck output requests in Phase 3 substructuring runs. The automated substructuring system is completely described in Section 20, including the relationships between Case Control and Substructure Control Deck output requests. Summaries of the output requests available in the Substructure Control Deck are provided in this section.

There are three major print request commands and a plot request command included in the Substructure Control Deck. Each is described in the following sections and in detail in the User's Manual, Section 2.7.

10.6.1 SOFPRINT Command

The SØFPRINT command provides a means of determining the status of the Substructure Operating File (SØF) at any stage of an analysis. The command is often used to obtain the table of contents of the SØF which includes a list of all substructures, how the substructures are related, how much storage space is being used and remains available, and which substructure items exist. The form of the command is simply:

SØFPRINT TØC

Examples of S@FPRINT T@C printed output are shown in Figures 17 and 18.

The SØFPRINT command also may be used to request print of individual substructure items or matrices, such as a stiffness matrix or load.

All options for using the SØFPRINT command are described in the User's Manual, Section 2.7.

PRINTING AND PLOTTING FEATURES

10.6.2 ØUTPUT Subcommand

and the second of the second
The subcommand ØUTPUT is used with substructuring operation commands CØMBINE, REDUCE, MREDUCE, and CREDUCE. ØUTPUT requests specific information relating to the operation being performed. The command is used as shown below:

ØUTPUT 1,2,6,...,17

Each integer requests a particular item relating to the operation. A complete summary of the options available for ØUTPUT under each command is presented in Table 1. Use of ØUTPUT is illustrated in Figure 2, and examples of the resulting print are presented in Figures 19 through 25.

10.6.3 PRINT Subcommand

PRINT "name" is used as a subcommand of RECØVER or MRECØVER where "name" is any substructure. The PRINT subcommand is followed by case control type commands such as DISP, SPCF, ØLØAD. These commands are merged with any similar Case Control Deck commands to create a complete print request for the named substructure. For these cases the Substructure Control Deck commands override any identical Case Control Deck commands. Note that no output will be printed unless these specific requests (DISP, SPCF, ØLØAD) appear in either the Substructure Control or Case Control Decks.

ENERGY is a subcommand to PRINT in substructure control, but it is not available as a Case Control command. ENERGY is used to print relative modal coordinate contributions to analytical solutions. The ENERGY command only applies to substructures used in modal reduce (MREDUCE) operations.

For specific details of using PRINT and associated commands, refer to descriptions of the RECØVER and MRECØVER commands in section 2.7 of the User's Manual. Output generated by displacement and energy requests in substructuring is illustrated in Figures 26 and 27.

10.6.4 PLØT Command

Structural plots of undeformed structures are requested in Phase 2 substructuring runs using the command:

PLØT name

where name is any substructure (from a basic model to the completely assembled model). Any number of these commands may appear in the Substructure Control Deck. Undeformed structural plots will be

10.6-2

SUBSTRUCTURE CONTROL DECK OUTPUT CONTROL

en por contra de la

generated using Case Control Deck structural plot commands. The entire sequence of Case Control commands will be executed for each Substructure Control Deck command. Refer to Section 20 of the User's Guide and Section 4 of the User's Manual for more details on substructures and plotting.

territore Adv

ťι

MISCELLANEOUS PRINTOUT

10.7 MISCELLANEOUS PRINTOUT

Along with the standard output options, certain results are printed automatically or may be selectively requested independent of the Case Control options. Those quantities which are selected by the user are described below, followed by a description of the results which are output automatically.

10.7.1 Parameter-Controlled Output

Specialized output may be requested via the PARAM Bulk Data Card (User's Manual, Section 3.1). These options include:

<u>GRDPNT</u> - requests the weight and balance results (Figure 28). The output quantities for the user-specified grid points are:

- 1. Reference points.
- Rigid body mass matrix (MØ) relative to the reference point in the basic coordinate system.
- 3. Transformation matrix (S) from basic coordinate system to principal mass axes.
- 4. Principal masses and associated centers of gravity.
- 5. Inertia matrix I(S) about the center of gravity relative to the principal mass axes.
- 6. Inertia matrix I(Q) about the center of gravity relative to the principal inertia axes.
- 7. Transformation matrix (Q) such that $[I(Q)] = [Q]^T [I(S)] [Q]$.

The generality of this output is necessary due to the potential problems of the user's inputing directional masses and general inertia terms. The effects of scalar points are ignored, but all other mass effects are included.

<u>IRES</u> - requests the printout of the static analysis residual load vector (Figure 29). This will produce a printout of the error occurring at each solution degree of freedom for each static load vector. The form of the data is in general matrix output format printed by the DMAP module MATGPR (User's Manual, Section 5.3).

10.7.2 DMAP-Controlled Output

An Output Utility Module's (User's Manual, Section 5.3) printout is controlled by including DMAP modules MATPRN, MATGPR, and/or TABPT in the DMAP alters to a Rigid Format or in the user-

PRINTING AND PLOTTING FEATURES

specified DMAP program. The MATGPR module, which prints results with gridpoint references, and MATPRN module, which prints results using the internal numbering sequence, are highly recommended for finding difficult errors in a matrix formulation, but the volume of printout precludes their use in routine - debugging of large problems. Examples of the printout format are given in Figures 29 and 30. The TABPT module prints the internal NASTRAN table data specified in the DMAP instructions. The use of these modules requires some knowledge of the DMAP language (User's Manual, Section 5), and requires an understanding of the NASTRAN data block contents (Programmer's Manual, Section 2).

10.7.3 Automatic Output

Diagnostic printout which may be augmented with the Executive Control DIAG card (User's Guide, Section 1, and User's Manual, Section 2.2) is a convenient tool for the experienced NASTRAN user to analyze the execution of a problem. A description of all possible outputs is beyond the scope of this Guide. Also, an understanding of some of the available printout requires a knowledge of the NASTRAN program.

Some of this diagnostic output is automatic. This printout occurs in different locations, depending on the particular computer system being used. The standard output consists of the BEGIN and END times for every functional module and some of the major subroutines.

When a user requests a checkpoint be taken, the RESTART data is automatically printed. This data is written as it is created and is a duplicate of the checkpoint dictionary cards punched at the end of each run. If the restart deck is lost, the cards may be punched manually from this printout.

Every time an eigenvalue extraction routine is executed, the Eigenvalue Summary Tables are automatically printed. This occurs in normal modes, buckling, and complex eigenvalue analyses. The output shown in Figure 31 and 32 includes the eigenvalue routine diagnostics and the eigenvalues, fruquencies, generalized (modal) mass, and generalized (modal) stiffness for each solution root. Refer to the Rigid Format descriptions (User's Manual, Section 3) for a definition of the contents printed for each Rigid Format.

Each time a plot output request is made, a Plot Summary Table is automatically printed for both the structure (Figure 33) or plot (Figure 36) operations. Each plot produces a page of output containing data in the scale factors, origins, sets, labels, and/or curve maximums and minimums.

10.7-2

STRUCTURAL AND CONTOUR PLOTTING

10.8 STRUCTURAL AND CONTOUR PLOTTING

10.8.1 Structural Plotting

The structure plot control options, as summarized in Table 2, are described in detail in Section 4 of the User's Manual. Illustrative examples of their use are presented in Sections 18 and 19 of this Guide. The following describes the actual results and contains recommendations for general usage.

- 1. The undeformed plots are generated near the beginning of the NASTRAN execution. These plots provide visual verification of the geometry and element connectivities of a model. For large complex problems, it is recommended that several plots be made using different sets of elements, view angles, and labeling options in various combinations. The user may then select the best views of his structure for plotting the results of subsequent runs.
- 2. Use of the FIND option is recommended to automatically calculate the origin and scaling factors used for plotting each section of a structure. Figure 33 illustrates the plot data summary printed out for each plot requested. If the user specified his own data, it is possible that some points will not be included in the plot. This also occurs frequently when a different set of structural elements is plotted with the origin and scale of a previously plotted structure. The FIND card should be used before each PLØT command, unless the structure and view data are unchanged. Another problem occurs when a deformed structure is plotted and the MAXIMUM DEFØRMATION card is missing. Without it, the scale is chosen to place the extreme limits of the structure at the edge of the plot. Large deformations at these points, therefore, might not get plotted.
- 3. The use of symmetry perspective and stereo projections, and user-defined scales and origins is <u>not</u> recommended for the inexperienced user. These options are best approached in an experimental mode by using small test examples. Figure 34 shows part of a large structure plot packet in a Case Control Deck. It illustrates the use of many of the plotting options.
- 4. Quite frequently some of the plotter choices available in NASTRAN are not compatible with a certain installation hardware or may be inefficient for some plotters. Plot tape conversion programs are commonly available to remedy these situations. They require only an additional tape set-up and an extra job step in the job control deck.

10.8-1

10.8.2 Contour Plotting

والمراجع والمراجع والمراجع المراجع والمراجع فالمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع

Contour plots may be obtained by requesting them on PLØT cards and by including a separate CØNTØUR card in the structure plot packet. Nine types of stress and four types of displacement contour plots may be selected. The parameters on the CØNTØUR card are discussed in Section 4.2 of the User's Manual. In a normal NASTRAN run, these plots are available following the solution of the structure. In a substructuring run, they are available only in Phase 3 runs, where the basic structural relations between points and elements are recovered and the solution data for each substructure is known (User's Guide, Section 20).

10.9 XY PLOTTING AND PRINTING

<u>ار ا</u>

The XYPL@T options (User's Manual, Section 4.3) are used for a variety of purposes in dynamic analysis, especially to produce curve plots of selected variables versus time of frequency. Options are also available to print out selected components of the results versus time or frequency, or to print out the peak values only. Also, the printer itself may be used as a plotter when quick qualitative results are desired. The printed characters are placed in locations on the printout to represent one or more curves at a time. Figure 35 shows the XY request packet in a Case Control Deck. Figure 36 illustrates a typical XY Output Summary table, one of which is printed for each curve. When the XYPRINT command is used, a table of XY pairs is printed immediately following the summary table (Figure 37). This represents the data which will be plotted to fill the current request.

The command cards to produce XYPLØT printout or plots are described in Section 4.3 of the User's Manual. The following discussion is concerned with aiding the user in interpreting his output and avoiding problems.

- Quite frequently the user cannot anticipate the critical points and alements for which plots are desired. Plotting a number of points and elements may be inefficient and confusing. A common procedure is to checkpoint the solution run and printout much of the data. From these data, the critical points and elements and the plot scales may be selected, and a restart can be made for plotting only.
- 2. The automatic selection of scales in the XYPLØT option was designed for the general case and may be undesirable for a specific user problem. For instance, quantities which are to be compared should be plotted with the same scale with perhaps more than one curve plotted per page. But, the plotter automatically sizes each plot to reach the upper and lower limits of the plot region specified. The procedure suggested above should help avoid this problem.
- 3. The automatic labeling of the plots includes the point or element identification number and plot sequence number. This sequence number corresponds to the order in which the plots were created. A separate and unique TCURVE title for each frame should be input for every plot. Later, this title will be very useful when the user attempts to recall what he had intended to have shown in each plot.

10.9-1

PRINTING AND PLOTTING FEATURES

5

ě.

i.

4. The number of options available in the XYPLØT control package is usually overwhelming to the beginner. It is suggested that, at first, single plots to be placed on single frames and that most of the defaults be used. If a checkpoint has been taken, the options may be selectively and economically experimented with, using the restart feature of NASTRAN.

MATRIX TOPOLOGY

10.10 MATRIX TOPOLOGY

NASTRAN provides a special capability to print or plot the topology of any matrix. This capability requires the user to insert a DMAP alter packet with the SEEMAT instruction (User's Manual, Section 5.3) which identifies the matrix data block and selects the output options. Whether printed or plotted, the SEEMAT output will show the nonzero matrix elements positioned pictorially by row and column within the outlines of the matrix. SEEMAT identifies each degree of freedom which, for large matrices, will produce a large volume of output.

11.1 INTRODUCTION

This section is designed to orient a new user of NASTRAN to the basic requirements for operating NASTRAN on any of the standard computer systems listed in Table 1.

Section	Computer System
11.2	IBM 360/370 (#S) (PM 5.3)
11.3	UNIVAC 1108/1110 (Exec 8) (PM 5.4)
11.4	CDC 6000/CYBER (NDS,NDS/BE) (PM 5.5)

Table 1. Standard Computer Systems for NASTRAN

Though a vast majority of the NASTRAN subroutines are written in a subset of FURTRAN to be compilable with all three systems, certain utilities have been written in Assembly language to facilitate interfacing the NASTRAN Executive System with the resident operating system software.

NASTRAN consists of approximately 1600 subroutines of source code and is divided into 15 functional links (Programmer's Manual, Section 5). When the NASTRAN program is actually installed at the user's location, the local usage conventions must be considered in order to determine the most advantageous means of making NASTRAN accessible to the user. At certain installations, where NASTRAN is heavily used, it may be made available as a system resident program. Alternatively, the program may be stored off-line, but in load module form so that it can be readily loaded into the system when required. It should be noted that NASTRAN is designed to dynamically allocate core so that at execution time, all the core memory that can be made available will be used. Whenever practicable, the maximum core available should be used to achieve minimum run times. For further discussion of core requirements, see Section 14.

Each of the following sections will summarize the rudiments of the system control cards required to operate NASTRAN on each of the three computer systems. Examples are provided for many of the typical situations encountered, including usage of checkpoint tapes, user tapes, plot tapes, and substructure operating files. Not all conditions can be covered in this Guide. However, sufficient examples are given to indicate what questions to ask of the resident systems personnel in the event more complex arrangements are desired.

11.1-1

11.2 NASTRAN ON THE IBM SYSTEM 360/370 (DS)

11.2.1 Deck Organization Summary

-3

This section describes how the NASTRAN input deck (including control cards) must be set up for operation on the IBM 360/370 hardware (Programmer's Manual, Section 5.3). The overall organization of the deck is shown in the table below. Each item in the table is described in Section 11.2.2. Many examples of common deck set-ups are presented in Section 11.2.3. All system control cards have two slashes (//) punched in columns 1 and 2.

Table 1. Organization of the Input Deck

1	JØB card
2	NASTRAN instream procedure
3	EXEC card
4	Control cards for optional files
5	NASTRAN data deck
6	/*

11.2.2 Descriptions of the Input Deck

11.2.2.1 JØB Card

The JØB card identifies the job and furnishes parameters necessary for scheduling and accounting purposes. It is installation-dependent; therefore, check with cognizant personnel at your installation for the proper usage.

11.2.2.2 NASTRAN Instream Procedure

The NASTRAN instream procedure is a deck of about 100 control cards which are required to execute NASTRAN. These cards must be included in every run. However, some installations may have these cards permanently stored in the system (cataloged). In this case, the instream procedure deck may be left out of the input deck.

A listing of the NASTRAN instream procedures is given in Table 2, followed by descriptions of the numbered items.

11.2-1

C - 3

and the second
Table 2. Basic Job Control Language (JCL) for NASTRAN Runs.

1. //NASTRAN PROC NAME='XLJLR.NSTNLMOD.L1750.LOAD'. PUNITS=CYL+P1=1+P2=1+ 11 11 SUNITS=CYL+S1=3+S2=1+ 11 TUNITS=CYL+T1=3+T2=1+ 11 DENPLT=1 //*---------//*------//*---******************************** ---* PROCEDURE NASTRAN * //*---... //*---*********** //*------//*-----_____ 2. //NS EXEC PGM=NASTRAN, REGION=41úk, PARM=+CORE=(STAT, 64K)+ 3. //FT01F001 DD UNIT=SYSDA+SPACE=(CYL+(3+1))+ 11 DCB=(RECFM=FB+LRECL=80+8LKSIZE=7280+8UFNO=1) 4. //FT04F001 DD SYSOUT=A,DC8=(RECFM=VHA.LRECL=137.BLK512E=3429) 5. //FT05F001 DD DDNAME=SYSIN 6. //FT06F001 DD SYSOUT=A,DCB=(RECFM=VBA,LRECL=137,BLKSIZE=3429) 7. //FT07F001 DD SYSOUT=8 8. //FT14F001 DD DDNAME=PLT2 * FORTRAN PLOT REF 9. //INP1 DO OSN=NULLFILE EXTRA 10. //INP2 DD DSN=NULLFILE INPUT 11. //INP3 DD DSN=NULLFILE AND 12. //INP4 DD DSN=NULLFILE OUTPUT 13. //INP5 DD DSN=NULLFILE UNITS 14. //INP6 DD DSN=NULLFILE FOR 15. //INP7 DD DSN=NULLFILE MATRIX 16. //INP8 DD DSN=NULLFILE STORAGE 17. //INP9 DD DSN=NULLFILE AND 18. //INPT DD DSN=NULLFILE RETRIEVAL 11+ //* FOLLOWING FIVE CARDS DEFINE THE EXECUTIVE FILES. //* EXCEPT FOR THE HPTP AND POOL, THESE FILES ARE DEFINED TO BE NULL 1/* AND NO EXTERNAL USE CAN BE MADE OF THESE DATA SETS UNLESS PROC . //* OVERRIDE CARDS ARE SUPPLIED---SPECIFY UNIT, DSN, VOL-SER-NO. //* FURTHERHORE ANY OVERRIDES MUST APPEAR IN THE SAME ORDER AS ٠ //* THOSE BEING OVERRIDDEN. I.E., ALPHABETICAL. 11. 19. //NPTP DO UNIT=SYSDA, SPACE= (LPUNITS, (LP1, LP2)), DISP= (, PASS) 20. //OPTP OD OSN=NULLFILE+LABEL=(+NL)+DISP=OLD 21. //PLT2 DD DSN=NULLFILE 22. //POOL DD UNIT=SYSDA+SPACE=(LPUNITS+(LP1+LP2)) 23. //UMF DD DSN=XLJLR.UMF.BULK.L1750.DATA.DISP=SHR 110 //* **** EXAMPLE TAPE OVERRIDE CARDS FOR EXECUTIVE FILE DU'S **** **** QUESTIONS (?) MUST BE REPLACED WITH APPROPIATE NAME **** 11• 11+ //* //NS.NPTP DD UNIT=2400,VOL=SER=?,DSN=?,DISP=(NEW,KEEP), LABEL=(1,BLP) UNIT=2400,VOL=SER=?,DSN=?,DISP=OLD, 110 11 //+ //NS.OPTP DO 110 11 LABEL=(1+HLP)

3

۶

=

٦

Table 2. Basic Job Control Language (JCL) for NASTRAN Runs (Continued)

1

	//*							
	//*							
	//* *****	PR	INARY	SCRATCH	UATA SE	13		
	114							
24.	//PRIG1	DO	UNIT=3	YSDA.SP	ACETILPU	N1139(6)		
25.	//PR102	00	UNIT=5	YSDA.SP	ACE=(LPU	NI 130 (L		
26.	//PR103	DŪ	UNIT=5	SYSDA . SP	ACET(LPU	NIT31 (6)		
27.	//PR104	DD	UNITES	SYSDA+3P	ACENILPU	NI 130 (61		
28.	1/PH105	00	UNIT=1	SYSDA+SP	ACE ULPU	NI121(6)		
29.	//PR106	DD	UNITOS	SYSDANSP	ACESILPU	NITZICE		
30.	//PR107	ŬŌ	UNIT=	SYSDA+SP	ACE - LPU	WT12010	<u> </u>	
31.	//PK104	00	UNIT=	SYSDA	PACE	NIT30(6	P10676//	
32.	//PR109	DD	UNIT=	SYSDA+SP	PACESILPU	NI15+(6	P1:076//	
33.	//PR[19	0D	UNITE	SYSDA	PACER(LPU	NI 120 (D		
34.	//paili	00	UNIT=	SYSDA	PACE	NITSOCO	riteress.	
35.	//PR[12	υÖ	UNITE	SYSDA+SP	PACE= (LPU	NITS (6	-1. 0 -611	
36.	//PRILJ	00	UNIT=	SYSDA	PACE - ILPU	NI 150 (6		
37.	//PR114	00	UNITE	SYSDA	PACE	MIIS+(6		
38.	//PRI15	DD	UNIT=	SYSDA	PACESIEPU	MI131(0		
- 39.	//PRI10	00	UNIT=	SYSDA	PACE = (6PU	MITIC - 18		
40.	//PRI17	DD	UNIT	SYSDA		JN [13+ (6		
41.	//PRILM	UD	UNIT	SYSDAIS		141126°17	P).(P2))	•
4Z.	//PRILY	DD	UNIT.	SYSUAIS	PALE - (671)4113V(U		a b
43.	1164150	00	UNITE	STSDAIS	FALL = (6F)	JN 1 1 3 4 (0		
- 44.	//PR151	00	UNII	STSUAIS	FRUE - (8F) DACE - /1 81	JN 1 1 3 4 4 4		22
45.	//PRI22	00	UNITE	313UA13		INTTS./B	P1.LP2))	- Z
40.	//PRIZJ	00	UNIT	STSUAIS	PACE-ISPI	INTTS./A	D1.4P2))	•
47.	//PRI24	00		STSUATS		UNITE. (8	P1+6P2))	D
48.	//₽₩123	00		ST SURTS	PACFEILP	UNITSO	PILLPELL	E
47.	//₩₩160	00		SYSDAIS	PACES(LP	UNITS	PloLPd))	- F
- 3V. E 1	////////	00	LintTe	SYSDA	PACE=(LP	UNITS. (PLICPED	V
31.		00	UNIT	SYSDAS	PACEALLP	UNITS. (LPliLPd))	
56.	//00130	00	UNIT	SYSOA	PACE= ILP	UNITS+(LP1+LPd))	
5J. 84	//00131	00	UNIT	SYSUA	PACE - (LP	UNITS.(6P1+6P2}}	
- 54. E.E.	//00132	00	UNIT	SYSOAIS	PACE= (.P	UNITSO	LPl+LPX))	
	//•							
	//* ***	•• SE	CONDAR	RY SCHAT	CH DATA	SETS "		
	110							
56	//SECOL	UÐ	UNIT	SYSUAIS	5PACE= (65	UNITS . (651+652))	
57	//SECO2	DD	UNIT	SYSOA	PACE = (LS	UNITSO	621+62411	
- 58	//SECOJ	DD	UNIT	SYSOA	SPACE= (LS	UNITSO	F21+F2411	
59	//SEC04	DU	UNIT	SYSDA .	PACE = (L)	UNITSOL	621163C11	
60	. //SEC05	00	UNIT	SYSDA	SPACE = (LS	UNITSIC	031103CII	
61	. //SECON	DO	UNIT	SYSOA.	SPACE - (LS	UNITSIC	P314P3C11	
62	//SECOT	DÖ	UNIT	SYSDAN	SPACE = (L			
63	//SECON	00		-313041		UNITS . I	LS1+LS2))	
- 64	//SECUV	00			SPACE	UNITS. (L51+654))	
65	//36/10	00	UNIT	aSYSDA .	SPACE	UNITSI	151.654))	
6	//32641	00	UNIT	SYSDA	SPACE - ILS	UNITS I	651+65477	
67	//45014	00	UNIT	.SYSUA.	SPACE - IL	SUNITS . C	6211624))	
6	//5# (10	00	UNIT	-SYSOA.	SPACE-IL	SUN175+ (651+65d))	
61	//SEC15	00	UNIT	-SYSDA.	SPACE= (L	SUNITS I	621+624))	
1	//SNAPSHO	T DD	5750	UT=A.DC	u= (RECFM	• VUA + L AE	(CL=152+8L)	(214 6-044)
	//STEPLI	00	05N=	LNAME .D	ISP=SHR			
	//SYSUDU	₩ NÖ	5750	UT=A				
· · · · · · · · · · · · · · · · · · ·								

بالمناهب الا

مر منابعه، سالله Table 2. Basic Job Control Language (JCL) for NASTRAN Runs (Continued)

110 11+ ++++ TERTIARY SCRATCH DATA SETS ***** 11. UNIT=SYSUA.SPACE=(LTUNITS.(LT1.LT2)) 74. //TEROL DD UNIT=SYSDA, SPACE=(LTUNITS, (LT1+LT2)) 75. //TERO2 DD UNIT=SYSDA, SPACE=(LTUNITS+(LT1+LT2)) 76. DO UNIT=SYSDA+SPACE=(LTUNITS+(LT1+LTd)) DO UNIT=SYSDA+SPACE=(LTUNITS+(LT1+LTd)) DD UNIT=SYSDA+SPACE=(LPUNITS+(LP1+LPd)) //TEROJ //TEH04 77. 78. //XPTD ------//*----_____ -----END OF PROCEDURE ---//*---//*-----79. 11 PEND

٦

ż

NASTRAN ON THE IBM SYSTEM 360/370 (ØS)

and a start of the
Item

Description

1. The PRBC card defines the name of the instream procedure and sets default values for the symbolic parameters.

SYMBOLIC PARAMETER	DEFAULT VALUE	DESCRIPTION
R	400K	Defines region size.
Т	10	Defines CPU limit (minutes).
DENPLT	1	Defines the density of the plot tape (556 bpi)
NAME	XLJLR.NSTNLMØD. L1750.LØAD	Defines the data set name of the executable.
PUNITS	CYL	Defines the space allocation units for the primary files.
Pl	1	Defines the initial space allocation for the primary files.
P2	1	Defines the increment space allocation for the primary files.
SUNITS	CYL	Defines the space allocation units for the secondary files.
S1	3	Defines the initial space allocation forthe secondary files.
S2	1	Defines the increment space allocation for the secondary files.
TUNITS	CYL	Defines the space allocation units for the tertiary files.
TI	3	Defines the initial space allocation for tertiary files.
T2	1	Defines the increment space allocation for tertiary files.

- 2. The EXEC card defines the name of the program to be executed, NASTRAN, the region size in which NASTRAN is to be executed, and the default PARM.
- 3. The FT01F001 DD card defines the data set to be used as intermediate storage for NASTRAN input. BCD or EBCDIC card images are read and converted to BCD card images and written on Unit 1, then NASTRAN reads its input from Unit 1.
- 4. The FT04F001 DD card defines the data set that will contain the run log. The NASTRAN run log contains internal timing for NASTRAN and a trace of the modules that are executed in an execution. It is usually assigned to a printer (SYSBUT=A); however, it may be assigned to any device or set to dummy and thereby deleted as desired.
- 5. The FTOSFOO1 DD card is deferred by the use of the DDNAME=SYSIN parameter and will be discussed in the examples that follow.

- 6. The FT06F001 DD card defines the printed output from the NASTRAN run. This printed output is written in BCD. As with FT04F001, it can be modified at execution time by the user as desired.
- 7. The FT07F001 DD card defines the punched output from NASTRAN executions. It can be modified as desired.
- 8. The FT14F001 DD card is deferred by the use of DDNAME=PLT2 (see Item 23).

_)

5

Ē

- 9 thru 18. These DD cards define data sets to be used as user tapes through the use of INPUTT2 and @UTPUT2 routines. All have the parameter DSN=NULLFILE which restricts the operating system from allocating them unless they are supplied as override JCL.
- 19. The NPTP DD card is used to describe the new problem tape (checkpoint tape) to be used by NASTRAN. It is set up as a temporary file and should be overridden if it is to be saved for later use.
- 20. The @PTP card defines the old problem tape (previous checkpoint tape) for the NASTRAN run. It must exist prior to the NASTRAN run in which it is used, and in this procedure DSN=NULLFILE must be overriden before the @PTP could be used to retrieve data.
- 21. The PLT2 DD card defines the output data set containing the SC4020, CALCHMP or NASTPLT plotting data.
- 22. The POOL DD cards define the data set to be used as the Data Pool file. It is always present and refers to temporary scratch disk.
- 23. The UMF DD card defines the data set that contains the User Master File data to be input to NASTRAN. It must be specified if it is needed.
- 24 thru 55. These DD cards define the primary units to be used by NASTRAN as temporary working files. (Note, each DD card must reference only one unit.)
- 56 thru 70. These DD cards define the secondary units to be used by NASTRAN as temporary working files. (Note, each DD card must reference only one unit.)
- 71. The SNAPSHOT DD card defines the data set that contains the diagnostic dump if NASTRAN takes a user abort. It is assigned to the printer.
- 72. The STEPLIB card defines the data set name and the location of the executable.
- 73. The SYSUDUMP DD card defines the data set that contains the diagnostic dump if NASTRAN takes a system abort. It is assigned to a printer.

NASTRAN ON THE IBM SYSTEM 360/370 (95)

į.

- 74 thru 77. These DD cards define the tertiary units to be used by NASTRAN as temporary working files. (Note, each DD card must reference only one unit.)
- 78. The XPTD DD card defines the temporary data set that is to contain the checkpoint dictionary during CHECKP9INT runs in a GIN# compatible format (N#T BCD card images).
- 79. The PEND card signifies the end of an instream procedure. It should be removed before the JCL is placed on the PR@CLIB.

The following restriction should be noted: Plot tapes must be 7-track tapes when plotting for a Stromberg-Carlson plotter; any remaining tapes (problem tapes, UMF, PLT2 tape for the CALCOMP plotter, or the General Purpose Flotter, etc.) may be 7- or 9-track as the user desires.

The NASTRAN execution deck setup is presented as an instream procedure such that it may be verified before addition to the installation procedure library (PRØCLIB) to permit easy recall and reuse.

The procedure provided is intended for use on an IBM 360 computer operating under ØS. It should be reviewed for each installation prior to actual use. Among the items that may necessitate modification of the procedure are the following:

- Any modification to the standard IBM ØS operating system could make modification of the PRØC necessary. For example, all SYSØUT=A and SYSØUT=B data sets are provided with DCB information and space allocation. If a particular installation is using HASP (Houston Automatic Spooling System), it becomes necessary to remove the DCB parameters and the space allocation is no longer needed. Similar modifications will be necessary when running under ASP (Auxiliary Support Processor) or related systems.
- 2. The procedure as provided has temporary scratch file space sufficient for small to medium size problems. Medium to large problems will require the initial space allocation for primary and secondary files to be doubled (F1=2, S1=6), while very large problems often tax the resources of the computer and must be dealt with on an individual basis.
- 3. If the number of DD statements in the procedure is too great, it may be reduced (at some cost in performance) as follows with the most expendable data sets listed first:

- a. INPT, INP1, INP2, ---, INP9, BPTP, UNF, PLT2(and FT14F001). The data sets are not used by NASTRAN unless requested by the user through his data.
- b. TER01,TER02,---. Tertiary files may be eliminated if careful thought is given to the selection of the space allocation parameters Pl and Sl.
- c. The secondary files (SEC01,SEC02,---,SEC15) may be cut back to a single one (SEC01) as long as enough space is pre-allocated for the primary files (PRI01,PRI02,---) to run the problem. It is recommended that at least four or five secondary files be retained, however.

11.2.2.3 EXEC Card

The EXEC card actually causes NASTRAN to execute. The format is:

//name EXEC NASTRAN

anna an taobh aile an an taobh
where "name" is an alphanumeric name of up to 8 characters, the first of which must be alphabetic. "Name" is optional. If used, it should begin in column 3.

There must be at least one blank space preceding the word EXEC, and at least one blank space between EXEC and NASTRAN.

NOTE: For some unusual problems, it may be necessary to specify parameters on this card. These parameters are described in Section 11.2.2.2.

ORIGINAL PAGE IS OF POOR OUALITY

NASTRAN ON THE IBM SYSTEM 360/370 (ØS)

11.2.2.4 Control Cards for Optional Tape and Disk Files

No tapes are required to run NASTRAN. However, a variety of options are available to the user which require the definition of tapes and/or their disk file equivalents in the system control card deck. These additional files fall into two categories: 1) NASTRAN permanent files (see Table 3) which are referenced in the procedure deck (Section 11.2.2.2), and 2) Other FØRTRAN units not covered in the procedure deck. For additional information on how NASTRAN manages input/output in the IBM system, see Section 5.3.2 in the NASTRAN Programmer's Manual.

Tabl	e	3.	NASTRAN	Permanent	Files
------	---	----	---------	-----------	-------

File Name	Furpose
INP1 (1=1,2,,9,T)	User tapes referenced by DMAP utility modules INPUTT1 and ØUTPUT1 (UG 15, UM 5.3.2)
NPTP	New problem tape for checkpointing
ØPTP	Old problem tape for restarting
PLT2	Plot tape for SC4020, Calcomp, or NASTRAN General Purpose Plotter (assigned FØRTRAN Unit 14)
UMF	User's Master File

Note that the file names in Table 3 are in the same order (alphabetical) as they appear in the procedure deck (Section 11.2.2.2). In order to successfully override the procedure deck specifications, the defining cards must appear in the same order after the // EXEC NASTRAN card.

In addition, the user may wish to use any of the remaining FØRTRAN units not referenced in the procedure deck. These are needed when using the INPUTT2 and/or ØUTPUT2 DMAP utilities (User's Guide, Section 15; User's Manual, Section 5.3.2) and when running substructuring problems (User's Guide, Section 7; User's Manual, Section 2.7). The FØRTRAN units available for use are FT02F001, FT03F001, FT08F001-FT13F001, and FT15F001-FT23F001. Control cards defining these units must appear after the aforementioned set of procedure deck override cards.

Generally, NASTRAN permanent files (except for the plot tape PLT2) may be located on userassigned disk packs inst ' of on tape. If the user wishes to store any such file on disk, he must specify FILES=xxx (xxx=INP, ...,UNF) on the NASTRAN card (User's Guide, Section 6).

11.2.2.5 NASTRAN Data Deck

The NASTRAN data deck must be preceded by the following card:

//NS.SYSIN DD *

There must be at least one blank space on either side of "DD."

CARGO CONTRACTOR OF THE OWNER OF

11.2.2.6 /* Card

The data deck must be followed by one delimiter card with "/" punched in columns 1 and 2.

11.2.3 Examples

This section presents a number of examples which illustrate the deck of control cards needed to successfully execute NASTRAN for a variety of problems. These examples are annotated. However, because of the volume of the text, notations made for a given example are not repeated on subsequent examples. Therefore, if a card in some example is not understood, it will be necessary to refer to an earlier example where the card is annotated. The examples are organized as shown in Table 4. Table 5 identifies the key features of each example.

11.2.3.1 Example 1 - Standard

- 1. //NASTRAN JØB-----
- 2. Instream Procedure Deck
- 3. // EXEC NASTRAN
- 4. //NS.SYSIN DD *
- 5. NASTRAN Input Deck
- 6. /*

Notes: 1. See Section 11.2.2.1.

- 2. See Section 11.2.2.2.
- 3. See Section 11.2.2.3.
- 4. Identifies the following cards as NASTRAN input data.
- 5. See Section 11.2.2.5.
- 6. See Section 11.2.2.6.

and and the second second second and the second
2

Table 4. Organization of the Examples.

Example	Description
1-2	Non-restart runs
3-4	Restart runs
5-6	Runs using Substructure Operating Files
7-8	Runs performing data transmission to/from other programs
9-10	Runs using the User's Master File

Table 5. NASTRAN Key Features in Control Card Examples.

				Ex	ample	Number				
Feature	1	2	3	4	5	6	7	8	9	10
Plot		X	X							
Checkpoint			X							
Restart				X	x					
Substructuring					x	x				
Data Transmission							X	X		
User's Master File									X	X
INP1,,INP9,INPT							X			
NPTP			x		X					
₿ ₽ТР				X						
PLT2		X	x							
UNF									X	X
FTxxF001					X	X		X		

• • • • • • • • • • • • • • • • • •

11.2.3.2 Example 2 - Standard, with Plotting

1. // EXEC NASTRAN

2. //NS.PLT2 DD UNIT=2400,LABEL=(,NL),DSN=PL#T360,

and the second secon

- // DCB=(BLKSIZE=2400,RECFM=U,BUFNØ=1),
- // DISP=(NEW,KEEP)
- 3. //NS.SYSIN DD *
- Notes: 2. Defines the plot tape. The PLT2 DD card in the procedure deck is overriden. If the NASTRAN General Purpose Plotter is used, the DCB parameters should be as follows:

DCB=(BLKSIZE=3000,RECFM=F,8UFNØ=1)

11.2.3.3 Example 3 - Very Large Model, with Checkpoint and Plotting

1. // EXEC NASTRAN, R=600K, T=200, P1=5, P2=5, S1=10,

// S2=10,T1=10,T2=10

2. //NS.NPTP DD UNIT=2400,LABEL=(,NL),DSN=NPTP360,

// DISP=(NEW,KEEP)

- 3. //NS.PLT2 DD UNIT=2400,LABEL=(,NL),DSN=PLØT360,
 - // DCB=(BLKSIZE=2400,RECFM=U,BUFNØ=1),
 - // DISP=(NEW,KEEP)
- 4. //NS.SYSIN DD *
- Notes: 1. Procedure deck parameters are redefined to provide larger region size, longer execution time, and more scratch file space (Section 11.2.2.2).
 - 2. Overrides NPTP DD card. DCB parameters are internally generated.
 - 3. Overrides PLT2 DD card. Notice that NPTP and PLT2 cards are placed in same order as in procedure deck (Table 2).

11.2.3.4 Example 4 - Restart

- 1. // EXEC NASTRAN
- 2. //NS.@PTP DD UNIT=2400, V@L=SER=XXXXXX,

LABEL=(,NL),DSN=NPTP360,DISP=ØLD

3. //NS.SYSIN DD *

NASTRAN ON THE IBM SYSTEM 360/370 (\$S)

Notes: 2. Overrides @PTP DD card in procedure deck. NASTRAN buffer size must be same as in run which generated @PTP.xxxxxx is installation assigned.

- 11.2.3.5 Example 5 Substructuring, Initial SØF Creation, Checkpoint
 - 1. // EXEC NASTRAN

- 2. //NS.NPTP DD UNIT=2400,LABEL=(,NL),DSN=NPTP360
 - // DISP=(NEW, MEEP)
- 3. //NS.FT02F001 DD UNIT=2314,VØL=SER=XXXXXX,
 - // DSN=SØF1,SPACE=(TRK,(500)),
 - // DCB=BUFND=1,DISP=(NEW,KEEP)
- 4. //NS.SYSIN DD *
- 5. SØF(1)=FT02,,NEW
- Notes: 3. Allocates FØRTRAN Unit number 2 on a disk pack for use as a new SØF. xxxx::x is installation assigned. This card is placed after the NPTP override card.
 - 5. References FØRTRAN Unit number 2 in the Substructure Control Deck. Size of SØF is defined in SPACE field on JCL card.
- 11.2.3.6 Example 6 Substructuring, Multiple SØF's
 - 1. // EXEC NASTRAN
 - 2. //NS.FT02F001 DD UNIT=2314,VØL=SER=XXXXXX,
 - // DSN=SØF1,DISP=ØLD
 - 3. //NS.FT03F001 DD UNIT=3314,VØL=SER=XXXXXX,
 - // DSN=SØF2,SPACE=(TRK,(1000)),
 - // DCB=BUFNØ=1,DISP=(NEW,KEEP)
 - 4. //NS.SYSIN DD *
 - 5. SØF(1)=FT02
 - 6. S#F(2)=FT03
 - Notes: 2. JCL for SØF created in previous run. NASTRAN buffer size must be the same
 - 3. Allocates FØRTRAN Unit number 3 on a disk pack for use as extended SØF space.
 - 5. References FURTRAN Unit number 2 in the Substructure Control Deck.

 References FØRTRAN Unit number 3 in the Substructure Control Deck. New parameter is not needed for SØF extensions (in fact, must <u>not</u> appear).

11.2.3.7 Example 7 - User Tape Input

- 1. // EXEC NASTRAN
- 2. //NS.INPT DD UNIT=2400,V@L=SER=xxxxxx.LABEL=(,NL),
 - // DSN=INPT360,DISP=ØLD

- 3. //NS.SYSIN DD *
- Notes: 2. Overrides INPT DD card in procedure deck. File is used as input to INPUTT1 module (UM 5.3.2). Could also be an old tape attached for output via OUTPUT1 module (UM 5.3.2). Other file names that can be used are INPi(i=1.9).
- 11.2.3.8 Example 8 User Tape Output
 - 1. // EXEC NASTRAN
 - 2. //NS.FT19F001 DD UNIT=2400,VØL=SER=xxxxxx,LABEL=(,NL),
 - // DSN=USER360,DCB=(BLKSIZE=8190,RECFM=VS),
 - // DISP=(NEW,KEEP)
 - 3. //NS.SYSIN DD *
 - Notes: 2. Definition of additional FØRTRAN unit to be used by ØUTPUT2 module (UM 5.3.2). The number 19 will appear in the appropriate location in the ØUTPUT2 DMAP statement.
- 11.2.3.9 Example 9 Demo Driver Decks and UMF Tape
 - 1. // EXEC NASTRAN
 - 2. //NS.UMF DD UNIT=2400,VØL=SER=xxxxxx,LABEL=(,BLP),
 - // DSN=UMF,DCB=DEN=2,DISP=@LD
 - 3. //NS.SYSIN DD UNIT=2314,VØL=SER=XXXXXX,
 - // DSN=XLJLR.UMF.DRIVER.L1750.DATA(D10120),
 - // DCB=BUFN#=1,DISP=SHR
 - /*

Notes: 2. Overrides the UMF DD card to reference the UMF tape.

 Overrides the SYSIN DD card to reference the Driver Deck file. Demo problem D10120 (member name) is used as an example. Notice that no instream bulk data cards are required.

Sand and the second
11.2.3.10 Example 10 - UNF File on Disk

-}

Ì.

- 1. // EXEC NASTRAN
- 2. //NS.UNF DD UNIT=2314, VIDL=SER=XXXXXX.
 - // DSN=XLJLR.UNF.BULK.L1750.DATA,DISP=SHR
- 3. //NS.SYSIN DO *
- 4. NASTRAN FILES-UNF
- 5. {NASTRAN Executive Control and Case Control decks.}
- 6. BEGIN BULK
- 7. ENDOATA
- 8. /*

Notes: 2. Overrides the UNF DD card to reference UNF disk file.

- 4. Required to signal to NASTRAN that UMF resides on disk.
- 5 thru 7. Executive Control deck, Case Control deck, BEGIN BULK card, and ENDDATA card are required, but no Bulk Data deck is required. Reference must be made to UMF deck in Executive Control.

11.3 NASTRAN ON THE UNIVAC 1108/1110 (EXEC 8)

11.3.1 Deck Organization Summary

This section describes how the NASTRAN input deck (including control cards) must be set up for operation on the UNIYAC hardware (Programmer's Manual, Section 5.4). The overall organization of the deck is shown in the table below. Each item in the table is described in Section 11.3.2 in the following text. Section 11.3.3 explains the procedure to catalog NASTRAN. Many examples of common deck set-ups are presented in Section 11.3.4.

Table 1. Organization of the Input Deck

ו	RUN card
2	QUAL card
3	Control cards for tapes
4	Control cards for heading and Link 1
5	NASTRAN data deck
6	Control card for Links 2-99
7	FIN card

Throughout this section, the NASTRAN program is assumed to be cataloged on FASTRAND. If NASTRAN is stored on tape instead, it will be necessary to catalog it. Section 11.3.3 explains this procedure.

Many control cards are exemplified in this section. In all cases, they begin with the symbol "@". This symbol (a 7/8 multi-punch) must be punched in column one of the control card. An alphanumeric string of characters always immediately follows this symbol, beginning in card column two.

11.3.2 Descriptions of the Input Deck

11.3.2.1 RUN Card

The RUN card identifies the run and furnishes parameters necessary for scheduling and accounting purposes. It is installation-dependent; therefore, check with cognizant personnel at your installation for the proper usage.

11.3-1

11.3.2.2 QUAL Card

1

=

À

The QUAL card is used to qualify for NASTRAN the file names which are referenced downstream in the input deck. The format is:

QUAL qualifier

Replace "qualifier" with the proper alphanumeric string in use at your installation. This string should be preceded by at least one blank space.

11.3.2.3 Control Cards for Tapes

and succession in the second states of the second states and

No tapes are required to run NASTRAN. However, if the user wishes to checkpoint, restart, plot or have access to user tapes, etc., tapes are needed. The tape units which may be used have been assigned by the system utility routine NTAB\$ (Programmer's Manual, Section 5.4.2) as follows:

FØRTRAN Unit No.(s)	External Name
1-7	Standard units assigned for operations
8	<pre>BPTP - Old Problem Tape used for restart</pre>
9	NPTP - New Problem Tape used for checkpointing
10	UMF - User Master File used for input
11	NUMF - New User Master File used for output
13	PLT2 - Plot tape for SC4020, CALCOMP, or NASTPLT Plots
14	INPT - User tapes for input and/or output
15-23	INPi (i=l-9) - Available for user tapes
24-39	24-39 - Available as UT1-UT5 user tapes

These units will be dynamically assigned by NASTRAN for internal use unless they have been previously assigned by user-supplied assign cards. The user supplied tape control cards must reference the external name as shown in the following examples:

MASG,T PLT2,T,tapeno or MASG,T 25,T,tapeno Note that "tapeno" must be replaced with the reel number of the tape to be used; and one or more blank spaces must follow "MASG,T."

When executing on the UNIVAC 1108/1110, the external names INPT and INP1 can be assigned as user tapes and referenced via the companion DNAP utilities (User's Manual, Section 5.3.2) INPUTT:

NASTRAN ON THE UNIVAC 1108/1110 (EXEC 8)

and the second
and @UTPUT1. These same FØRTRAN unit numbers (14-23) plus units 24-29 may be referenced directly as user tapes via the companion DMAP utilities (User's Manual, Section 5.3.2) INPUTT2 and @UTPUT2. The user must, however, take care that the FØRTRAN unit number referenced by INPUTT1 and/or @UTPUT not also be referenced by INPUTT2 and/or @UTPUT2 in the same NASTRAN run.

11.3.2.4 Link 1 Card

Two cards are required to suppress the system heading on each page of output and to initiate the execution of Link].

@HDG,N

@XQT *NASTRAN.LINK1

11.3.2.5 The NASTRAN Data Deck

This is the standard NASTRAN data input consisting of the Executive Control Deck, the Substructure Control Deck (optional), the Case Control Deck, and the Bulk Data Deck. (See Sections 6, 7, 8, and 9 of this guide for details.)

11.3.2.6 Other Links

A control card adds about 1400 link execution cards to the run stream from a NASTRAN program file. These cards are required to execute Links 2 through 15 and 99 of NASTRAN. The format is:

CADD *NASTRAN. NASTRAN

One or more blank spaces must follow "@ADD". Omitting this card will cause NASTRAN to exit after executing Link 1 and serve as an economical input data check.

11.3.2.7 The FIN Card

The FIN card indicates the end of the input deck. The format is:

ØFIN

11.3.3 Cataloging the NASTRAN Program

If NASTRAN is stored on tape at your installation, it will be necessary to copy it to NASTRAN before it can be executed. To perform this task, the following cards should <u>replace</u> the QUAL card (item 2) in the input deck:

ODELETE, C NASTRAN ODELETE, C NASTRAN. NASTRAN OASG.UPR NASTRAN, F//PØS/20

11.3-3

@ASG,T	ABSTAP, T, XXXXXR*
OCOPY,G	ABSTAP. NASTRAN.
ØFREE	ABSTAP.

*Replace "xxxxx" with the serial number of the tape.

11.3.4 Examples

This section presents a number of examples which illustrate the deck of control cards needed to successfully execute NASTRAN for a variety of problems. These examples are annotated. However, because of the volume of the text, notations made for a given example are not repeated on subsequent examples. Therefore, if a card in some example is not understood, it will be necessary to refer to an earlier example where the card is annotated. The examples are organized as shown in Table 2. Table 3 identifies the key features of each example.

11.3.4.1 Example 1 - Options = none

1	erun	
2	@QUAL qualifier	
3	@HDG,N	
4	@XQT *NASTRAN.LINK1	
	(NASTRAN DATA DECK)	
5	@ADD *NASTRAN.NASTRA	N
6	ØFIN	

Notes: 1. Installation-dependent RUN card.

2. Replace "qualifier" with the proper alphanumeric string in use at your installation.

3. Suppress system heading on output pages.

- 4. Add control card to initiate execution.
- 5. Add control cards needed for NASTRAN execution.
- 6. Terminates run.

NASTRAN ON THE UNIVAC 1108/1110 (EXEC 8)

_.` =:`

Ē

.

Table 2. Organization of the Examples.

Example	Description				
1-4	Non-restart runs				
5-8	Restart runs				
9-14	Runs using the User's Master File				
15-17	Runs performing data transmission to/from other programs				
18-19	Substructuring runs				

Table 3. NASTRAN Key Features in Control Card Examples.

Feature	Example Number																		
	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19
Restart					X	X	X	X											
Checkpoint			x	X			x	x					x	x					
Plot		x		x		X		x				x		x					
User's Master File									X	x	x	x	x	x					
Data Transmission															x	X	X		
Substructuring									į									x	x
User Tapes															X	x	x	x	X
NPTP			x	X			x	X					X	x					
NUMF									x	x				ł					
ØPTP					x	x	x	x											
PLT2		x		X		X		X				x		x					
UMF										x	X	x	X	x					

3

11.3.4.2 Example 2 - Options = plot

والمعادية فللمناجع

į,

- 1 ORUN ---
- 2 QUAL qualifier
- 3 MSG,T PLT2,T,XXXXXW

- 4 PXQT *NASTRAN.LINK1
- 5 (NASTRAN Data Deck)
- 6 MADD *NASTRAN. NASTRAN
- 7 OFIN
- Notes: 3. This card defines the plot tape. Replace "xxxxx" with the reel number of the tape.

11.3.4.3 Example 3 - Options = checkpoint

- 1 PRUN - - -2 QUAL qualifier 3 MSG,T NPTP,T,XXXXXW 4 **EXQT** MASTRAN, LINKI 5 (NASTRAN Data Deck) 6 *NASTRAN, NASTRAN
- 7 🗣 FIN

.

Notes: 3. This card defines the New Problem Tape (or checkpoint tape). Replace "xxxxx" with the real number of the tape.

- 11.3.4.4 Example 4 Options = checkpoint, plot
 - 1 PRUN - - -2 PQUAL qualifier 3 MASG,T NPTP,T,XXXXW 4 MSG,T PLT2,T,yyyyyW 5 TOXO *NASTRAN.LINK1 6 (NASTRAN Data Deck) 7 MDD *NASTRAN.NASTRAN **OFIN**

and a general second

and the second state of the se

11.3.4.5 Example 5 - Options = restart

1	ØRUN	
2	QUAL	qualifier
3	@ASG,T	ØPTP,T,xxxxR
4	ex qt	*NASTRAN.LINK1
5	(NASTRAN I	Data Deck)
6	@ADD	*NASTRAN.NASTRAN
7	@ FIN	

والهمود بوطارحاته المحافج وجرحا حادا الم

Notes: 3. This card defines the Old Problem Tape (or restart tape). Replace "xxxxx" with the reel number of the tape.

11.3.4.6 Example 6 - Options = restart, plot

1	ØRUN	
2	@ QUAL	qualifier
3	@ASG,T	ØPTP,T,xxxxxR
4	@ASG,T	PLT2,T,yyyyyW
5	€XQT	*NASTRAN.LINK1
6	(NASTRAN I	Data Deck)
7	@ADD	*NASTRAN.NASTRAN
8	ØFIN	

11.3.4.7 Example 7 - Options = restart, checkpoint

1	ØRUN	
2	QUAL	qualifier
3	@ASG,T	NPTP,T,xxxxXW
4	@ASG,T	ØPTP,T,yyyyyR
5	ex qt	*NASTRAN.LINK1
6	(NASTRA*. Dat	a Deck)
7	PADD	*NASTRAN.NASTRAN
8	ØFIN	
1	ØRUN	
---	-------------	-----------------------
2	QUAL	qualifier
3	ØASG,T	NPTP,T,xxxxXW
4	ØASG,T	Ø PTP,T,yyyyyR
5	@ASG,T	PLT2,T,zzzzW
6	@XQT	*NASTRAN.LINK1
7	(NASTRAN D	Data Deck)
8	0AD	*NASTRAN. NA STRAN
9	QEIN	

11.3.4.8 Example 8 - Options = restart, checkpoint, plot

AN CHARGE HE WAS

11.3.4.9 Example 9 - Options = create a UMF

1	QRUN	
2	@QUAL	qualifier
3	@ASG,T	NUMF,T,xxxxXW
4	@XQT	*NASTRAN.LINK1
5	(NASTRAN I	Data Deck)
6	@ADD	*NASTRAN.NASTRAN
7	@FIN	

Notes: 3. This card identifies the tape for the New User's Master File. Replace "xxxxxx" with the serial number of the tape.

11.3.4.10 Example 10 - Options = edit the old UMF and create a new UMF

1 **@RUN** - -2 **QUAL** qualifier 3 @ASG,T NUMF, T, XXXXXW 4 @ASG,T UMF,T,yyyyyR 5 exq1 *NASTRAN.LINK1 6 (NASTRAN Data Deck) 7 **QADD** *NASTRAN. NASTRAN 8 ØFIN

Notes: 3. This card identifies the tape for the User's Master File. Replace "xxxxx" with the serial number of the tape.

11.3-8

1

11.3.4.11 Example 11 - Options = UMF

1	@RUN	
2	@QUAL	qualifier
3	@ASG,T	UMF,T,xxxxxR
4	@XQT	*NASTRAN.LINKI
5	(NASTRAN	Data Deck)
6	QADD	*NASTRAN.NASTRAN
7	ØFIN	

11.3.4.12 Example 12 - Options = UMF, plot

1	@RUN	
2	@QUAL	qualifier
3	@ASG,T	PLT2,T,XXXXXW
4	@ASG,T	UMF,T,xxxxxR
5	@XQT	*NASTRAN.LINKI
6	(NASTRAN	Data Deck)
7	QADD	*NASTRAN.NASTRAN
8	ØFIN	

11.3.4.13 Example 13 - Options = UMF, checkpoint

	0.0111	
1	@KUN	
2	QUAL	qualifier
3	@ASG,T	NPTP,T,×××××W
4	@ASG,T	UMF,T,XXXXR
5	@XQT	*NASTRAN.LINK
6	(NASTRAN I	Data Deck)
7	@ADD	*NASTRAN. NASTRAN
8	@F IN	

11.3-9

.

11.3.4.14 - Example 14 - Options = UMF, checkpoint, plot

1	ØRUN	
2	@QUAL	qualifier
3	@ASG,T	NPTP,T,xxxxW
4	@ASG,T	PLT2 ,T ,yyyyyW
5	@ASG,T	UMF,T,zzzzR
6	ØXQT	*NASTRAN.LINK1
7	(NASTRAN D	ata Deck)
8	@ADD	*NASTRAN.NASTRAN
9	ØFIN	

11.3.4.15 Example 15 - Options = user tape input

.

1	ORUN	
2	QUAL	qualifier
3	@ASG,T	INPT,T,xxxxxR
4	@XQT	MASTRAN.LINK1
5	(NASTRAN Data	Deck)
6	@ADD 1	NASTRAN. NASTRAN
7	ØFIN	

Notes: 3. This card identifies the user's input tape. Replace "xxxxx" with the serial number of the tape. More than one user input tape may be used in a given job. Additional tape names are INPi, i=1,9.

11.3.4.16 Example 16 - Options = user tape output

1	ØRUN	
2	equal	qualifier
3	@ASG,T	INP1,T,XXXXXW
4	€XQT	*NASTRAN.LINK1
5	(NASTRAN D	ata Deck)
6	GADD	*NASTRAN.NASTRAN
7	ØFIN	

11.3-10

A MALE IN THE MALE AND A MALE AND A MELE AND A

NASTRAN ON THE UNIVAC 1108/1110 (EXEC 8)

an in a ha and the second s

Notes: 3. This card identifies the user's output tape. Replace "xxxxxx" with the serial number of the tope. More than one user output tape may be used in a given job. Additional to names are INP1, j=1,9.

11.3.4.17 Example 17 - Options = user tape input and putput

1 **PRUN** - - -2 **QUAL** qualifier 3 **ØASG,T** INP3,T,XXXXXW 4 **MASG,T** INP7,T,yyyyyR 5 **exqt** *NASTRAN.LINKI 6 (NASTRAN Data Deck) 7 **QADD** *NASTRAN.NASTRAN 8 ØFIN

11.3.4.18 Example 18 - Options = new SØF on disk

ו	@RUN	
2	QUAL	qualifier
3	@ASG,U	INP1,F///1000
4	€XQT	*NASTRAN.LINK1
5	NASTRAN	FILES=INPI
	•	(other NASTRAN data cards)
6	SØF(1) = 1	NP1,400,NEW
	•	(other NASTRAN data cards)
7	QADD	*NASTRAN.NASTRAN
8	Q FIN	

Notes: 3. The card identifies the file to be used as the S@F. Allowable names for the S@F files are INPT and INP1, i=1,9.

5. The specification required because INP1 resides on disk.

6. SØF declaration in the Substructure Control Deck.

11.3-11

SYSTEM CONTROL CARD DECKS

11.3.4.19 Example 19 - Options = old SØF on tape ·

e

1	erun	
2	equal	qualifier
3	@ASG,T	INPT,T,XXXXR
4	exqt	*NASTRAN.LINKI
	•	(NASTRAN data cards)
5	SØF(1)=INPT	,250
	•	(other NASTRAN data cards)
6	ØÅDD	*NASTRAN.NASTRAN
7	@FIN	

Notes: 3. This card identifies the user's SØF. Replace "xxxxx" with the serial number of the tape. Additional SØF's (up to 10) are similarly defined.

NASTRAN ON THE CDC 6000/CYBER (NBS AND NBS/BE)

11.4 NASTRAN ON THE CDC 6000/CYBER (NØS AND NØS/BE)

11.4.1 Organization of the Input Deck

This section describes the organization of the NASTRAN input deck. Table 1 lists the items appearing in the overall deck setup.

1	JØB description cards
2	Control cards to load NASTRAN
3	Optional tape setup cards
4	Execution procedure cards
5	NASTRAN data deck
6	EØR and EØF cards

Table 1. Organization of the Input Deck

Each item in the table is described in Section 11.4.2. Several examples of common deck setups are presented in Section 11.4.3. For more information concerning NASTRAN on the CPC 6000/CYBER, see Section 5.5 in the NASTRAN Programmer's Manual.

11.4.2 Descriptions of the Input Deck

11.4.2.1 JØB Description Cards

These cards vary according to installation and whether or not a terminal is being used. A key card is the JØB card. It contains limits on core size, CPU time, IØ time, tapes, and priorities.

11.4.2.2 Control Cards to Load NASTRAN

NASTRAN operates as 15 separate programs, or links, under the NØS and NØS/BE operating systems. These 15 files may be stored on tape, on the system permanent files, or on private disk packs. These options may vary considerably from site to site.

If NASTRAN exists on system or private disk files, an ATTACH card is used for each link. If it is stored on tape, the links must be copied separately onto random access devices.

11.4-1

SYSTEN CONTROL CARD DECKS

Various methods are used to accomplish the latter, depending on the installation. Consult the systems programmer or NASTRAN representative for assistance in this case.

If the user has a foreknowledge of which NASTRAN links are actually needed for a given execution, he can save time and expense by attaching or copying from tape only the required links.

11.4.2.3 Optional Tape Setup Cards

5

Ρì

No tapes are required to run NASTRAN. However, if the user wishes to checkpoint, restart, plot, run substructuring, or have access to user tapes, etc., tapes, or their disk equivalents, are required. The user should refer to the Programmer's Manual, Section 5.5.2 and 5.5.4 as well as his local systems personnel for additional detail concerning the procedures for assigning tapes for use by NASTRAN. The tapes which are available to the user are as follows:

File Name	Purpose		
INPi (1-T,1,2,,9)	User tapes referenced by DMAP utility modules INPUTT1 and OUTPUT1 (UG 15, UM 5.3.2)		
UTi (i=1,,5)	User tapes referenced by DMAP utility modules INPUTT2 and OUTPUT2 (UG 15, UM 5.3.2)		
NPTP	New Problem Tape for checkpointing		
gp tp	Old Problem Tape for restarting		
UNF	User Master File		
PLT2	Plot tape for NASTRAN General Purpose Plotter, SC4020, Calcomp, or DD80 plots		

The user must reference these file names in his job control deck if they are to be used on a NASTRAN run. The user tapes INPi and UTi are accessed via the DMAP parameter P2 (UM 5.3.2) corresponding to the specified file name.

Substructure runs require (1 to 10) Substructure Operating Files. These also must be defined by control cards unless they are being used only for scratch purposes within the run. The logical file names must be 4 letters in length, and must match the names on the SØF cards in the Substructure Control Deck (User's Guide, Section 7).

Typical formats of the tape/disk control cards are:

REQUEST, @PTP, HI.4363, RØL REQUEST, PLT2, HI.(SAVE, RING, NL))-tape ATTACH(S@F1, @LDS@F, ID=xxx)}-disk

11.4-2

NASTRAN ON THE CDC 6000/CYBER (NOS AND NOS/BE)

and the second
These requests also vary with the local operating system.

11.4.2.4 Execution Procedure Cards

Due to differences in link switching, the execution procedure differs between the NØS and NØS/BE operating systems. NØS, at most, requires only RFL, REDUCE, and execution command cards, as follows:

RFL(220000) REDUCE(-) LINK1,INPUT.

Under N#S/BE, the deck setup reads in a CYBER Control Language procedure file, and transfers control to it (see Section 11.4.3).

11.4.2.5 NASTRAN Data Deck

This is the standard NASTRAN data input consisting of the Executive Control Deck (User's Guide, Section 6), the Substructure Control Deck (User's Guide, Section 7), the Case Control Deck (User's Guide, Section 8), and the Bulk Data Deck (User's Guide, Section 9).

11.4.2.6 EDR and EDF Cards

An EBR card (7-8-9 multi-punch in column 1) separates all input records from each other and from the system control cards. An EBF card (6-7-8-9 multi-punch in column 1) terminates the entire deck.

11.4.3 Examples

The following examples show control card setups for using NASTRAN under the N@Sand N@S/ BE operating systems. Accounting, tape results and plot library control cards may vary at different installations. The first nine deck setups illustrate usage on N@S. Example 10 illustrates the additional task that needs to be performed when executing under N@S/BE. Table 2 identifies the key features of each example.

Fosturo		Example Number								
reature	1	2	3	4	5	6	7	8	9	10
Plot		x								
Checkpoint		x		x				x		
Restart			x	x				x		
Substructuring					x	x				
User's Master File							X	X		
Data Transformation									x	
NØS/BE										x

Table 2. Key Features of Examples.

11.4.3.1 Example 1 - Simple Load and Execute (NØS)

l .

ì

Ĩ

1.	NASTRAN, T500.
	USER
	CHARGE
	DELIVER
	A TO NASTRAN EDAM TARE TO DISC
2	PERHEST/TAPE NT D+DE I D-KHINASTAI
2	DEWIND TADE
J.	CODVDE TADE I TAVA
5	CADVDE TADE I TNV2
۶. د	CODVDE TADE LINK2.
7	CAPTER, MELLINKS.
6	CODVDE TADE LINK4.
0.	CONDE TADE LINKS.
10	CONDE TADE I INFT
10.	CONDE TADE A TANKA
11.	CONDE TADE LINKS.
12.	CONDE TARE, LINKY.
13.	COPTOF, TAPE, LINKIO.
14.	COPTER, TAPE, LINKIT.
12.	CWPTBF, TAPE, LINKI2.
10.	CUPTER, TAPE, LINKIS.
17.	COPYBF, TAPE, LINK14.
18.	COPYBE, TAPE, LINK15.
19.	RETURN, TAPE.
	*EXECUTE NASTRAN
20.	RFL (220000)
21.	REDUCE (-)
22.	LINKI, INPUT.
23.	/EØR
24.	(NASTRAN INPUT DATA)
25.	/EØF

Notes: 1. Job cards, installation dependent.

- 2. Tape containing 15 NASTRAN link files is attached.
- 3. Make sure tape is rewound.

E.

NASTRAN ON THE CDC 6000/CYBER (N#S AND N#S/BE)

and the second second second

4-18. Copy tape to 15 individual temporary disk files.

- 19. Free tape drive for other jobs.
- 20. Set maximum field length for NASTRAN execution.
- 21. Allows NASTRAN to dynamically alter field length.
- 22. Execute NASTRAN link 1. Other links will be dynamically executed.
- 23. End-of-record (7-8-9 multi-punch in column 1). Separates system control cards from NASTRAN input deck.
- 24. NASTRAN input data, consisting of the Executive Control Deck (UG 6), the Substructure Control Deck (UG 7), the Case Control Deck (UG 9), and the Bulk Data Deck (UG 9).
- 25. End-of-file (6-7-8-9 multi-punch in column 1). Job Terminator.
- 11.4.3.2 Example 2 Checkpoint and Plot

NASTRAN, T500. 1. USER, -----CHARGE, ------DELIVER. -----*--DUMP NASTRAN FROM TAPE TO DISC 2. REQUEST (TAPE, NT, D=PE, LB=KU)NASTO1 REWIND, TAPE. COPYBF, TAPE, LINK1. 3. 4. 5. COPYBF, TAPE, LINK2. COPYBF, TAPE, LINK3. COPYBF, TAPE, LINK4. COPYBF, TAPE, LINK5. 6. 7. 8. 9. COPYBE, TAPE, LINK6. COPYBF, TAPE, LINK7. COPYBF, TAPE, LINK8. COPYBF, TAPE, LINK9. 10. 11. 12. 13. COPYBE, TAPE, LINK10. CBPYBF, TAPE, LINK11. 14. COPYBE, TAPE, LINK12. COPYBE, TAPE, LINK13. 15. 16. 17. COPYBF, TAPE, LINK14. 18. COPYBE, TAPE, LINK15. RETURN, TAPE. *--TAPE SETUP REQUEST (NPTP, NT, D=PE)SAVE 19. 20. 21. REQUEST (PLT2, MT, HI) SAVE 22. RFL(220000) 23. REDUCE(-) 24. LINK1, INPUT. 25 /EØR (NASTRAN INPUT DATA) 26. 27. /EØF

Notes: 20. Attach NPTP for checkpointing.

21. Attach 7-track PLT2 for plotting using the SC4020, Calcomp, DD80, or NASTRAN plotters.

11.4.3.3	Example	3	-	Static	Analysis	Restart

-١

1.	NASTRAN, T500.
	USER
	CHARGE
•	ATTACH NASTRAN FILES NEEDED FOR STATICS
۷.	ATTACH(LINK1, NASTLINK1, ID=LRC)
3.	ATTACH (LINK2, NASTLINK2, ID=LRC)
4.	ATTACH(LINK4, NASTLINK4, ID=1 PC)
5.	ATTACH(LINKS NASTLINKS TOPLAC)
6.	ATTACH(I INKA NASTI INKA ID-LOC)
7.	ATTACH(I INK12 NASTI INK12 ID-LOC)
Ŕ	ATTACHI I NHY 2 HACTI THEY 2 TO LOOK
ă	ATTACH (LINKIS, MASTLINKIS, IU=LRC)
3.	TADE CETTIN
10	
10.	KEQUEST (WPTP, NT, D=PE)VSN=XXXXXX
11	RFL(220000)
12.	REDUCE(-)
13.	LINKI, INPUT.
14.	/EGR
15.	(NASTRAN INDUT DATA)
16	/FOF
	/ .

Notes: 2-9. Links needed for static analysis attached. Reside on disk.

10. Attach @PTP for restart. xxxxxx is installation assigned.

1.	NASTRAN, T500.
	USER,
	CHARGE,
	DELIVER
	*DUMP NASTRAN FROM TAPE TO DISC
2.	REQUEST (TAPE NT. D. PF. I. R. KII) NASTOL
3.	REWIND, TAPE.
4.	COPYBE, TAPE LINKI
5.	COPYBE, TAPE LINK2
6.	COPYBE, TAPE, LINK3.
7.	COPYBE, TAPE LINK4.
8.	CPPYBF, TAPE, LINKS.
9.	COPYBE, TAPE, LINKG,
10.	CPPYBF, TAPE, LINK7.
11.	CPPYBF, TAPE, LINKB.
12.	COPYBE, TAPE, LINK9.
13.	COPYBF, TAPE, LINK10.
14.	C@PYBF,TAPE,LINK11.
15.	CPPYBF, TAPE, LINK12.
18.	COPYBE, TAPE, LINKI 3.
17.	COPYBE, TAPE, LINKI4.
18.	COPYBE, TAPE, LINK15.
19.	RETURN, TAPE.
	*TAPE/DISK SETUP
20.	REQUEST (@PTP, NT, D=PE) VSN=XXXXXX
21.	REQUEST (NPTP, *PF)
22.	RFL(220000)
23.	REDUCE(-)
24.	LINKI, INPUT.
	*CATALOG NEW FILE

.

NASTRAN ON THE CDC 6000/CYBER (NOS AND NOS/BE)

The second second as a second s

CATAL#G(NPTP, NEWPTAP, ID=LRC, RP=30) 25.

26. /EØR

- 27. NASTRAN FILES - NPTP
- (NASTRAN INPUT DATA) 28.
- (REMAINING NASTRAN INPUT) 29.
- /EØF

Notes: 21. Request NPTP as a permanent file to be subsequently cataloged.

- 25. Catalog NPTP.
- 27. NASTRAN card must have FILES=NPTP, as NPTP will reside on disk.

11.4.3.5 Example 5 - Substructuring, Initiate SØF

1. NASTRAN, T500. USER, -----CHARGE, -----DELIVER. -----*--ATTACH NASTRAN 2. ATTACH(LINK1, NASTLINK1, ID=LRC) ATTACH(LINK2,NASTLINK2,ID=LRC) ATTACH(LINK3,NASTLINK3,ID=LRC) 3. 4. ATTACH(LINK4, NASTLINK4, ID=LRC) ATTACH(LINK5, NASTLINK5, ID=LRC) 5. 6. ATTACH(LINK6,NASTLINK6,ID=LRC) ATTACH(LINK7,NASTLINK7,ID=LRC) 7. 8. ATTACH(LINK8,NASTLINK8,ID=LRC) ATTACH(LINK9,NASTLINK9,ID=LRC) 9. 10. ATTACH(LINK9, ASTLINK9, ID-LRC) ATTACH(LINK10, NASTLINK10, ID-LRC) ATTACH(LINK11, NASTLINK11, ID-LRC) ATTACH(LINK12, NASTLINK12, ID-LRC) ATTACH(LINK13, NASTLINK13, ID-LRC) ATTACH(LINK14, NASTLINK14, ID-LRC) 11. 12. 13. 14. 15. ATTACH(LINK15,NASTLINK15,ID=LRC) *--SØF SETUP REQUEST(SØF1,*PF) 16. 17. *--EXECUTE NASTRAN 18. RFL(220000) 19. REDUCE(-) 20. LINK1, ÍNÞUT. *--SØF CATALØG 21. CATAL@G(S@F1,NASTS@F1,ID=LRC,RP=30) 22. /EPR 23. (NASTRAN INPUT) 24. SOF(1)=SOF1,500,NEW

- 25. (REMAINING NASTRAN INPUT)
- 26. /EØF

Notes: 2-16. Nastran resides on permanent files.

- 17. Request SØF1 for future catalog.
- 21. Catalog SØF1.
- 24. SØF card in Substructure Control Deck must reference logical file name in Job Control Deck (UG 7). NEW must appear on card.

11.4.3.6 Example 6 - Substructuring, Extend S#F

and the second
- 1. NASTRAN, T500. USER, -----CHARGE, -----DELIVER. -----*--DUMP NASTRAN FROM TAPE TO DISC REQUEST(TAPE,NT,D=PE,LB=KU)NASTO1 2. REQUEST(TAPE,NT,D= REWIND,TAPE. COPYBF,TAPE,LINK1. COPYBF,TAPE,LINK2. COPYBF,TAPE,LINK3. COPYBF,TAPE,LINK4. COPYBF,TAPE,LINK6. COPYBF,TAPE,LINK7. COPYBF,TAPE,LINK8. COPYBF,TAPE,LINK8. 3. 4. 5. 6. 7. 8. 9. 10. 11. COPYBF, TAPE, LINK9. COPYBF, TAPE, LINK10. COPYBF, TAPE, LINK10. COPYBF, TAPE, LINK12. 12. 13. 14. 15. 16. COPYBE, TAPE, LINKI3. 17. COPYBE, TAPE, LINK14. COPYBE, TAPE, LINKIS. RETURN, TAPE. 18. 19. *--SØF SETUP 20. ATTACH(SØF1,NASTSØF1,ID=LRC) 21. RFL (220000) 22. REDUCE(-) 23. LINKI, INPUT. *--SØF EXTEND 24. EXTEND(SØF1) /EDR (NASTRAN INPUT) 25. 26. 27. SØF(1)=SØF1,500 28 (REMAINING NASTRAN INPUT)
- 29. /EØF

Notes: 20. Attach permanent file containing old SØF.

- 24. Extend SØF to save newly written data.
- 27. SØF declaration in Substructure Control Deck.

11.4.3.7 Example 7 - UMF Run

- NASTRAN, T500. 1. USER, ------CHARGE, -----DELIVER. -----*-- DUMP NASTRAN FROM TAPE TO DISC Ż. REQUEST (TAPE, NT, D=PE, LB=KU) NASTO1 REWIND, TAPE. 3. 4. COPYBE, TAPE, LINK1. COPYBF, TAPE, LINK2. COPYBF, TAPE, LINK3. COPYBF, TAPE, LINK4. COPYBF, TAPE, LINK5. 5. 6. 7. 8. 9.
- CPPYBF, TAPE, LINK6. 10. COPYBE, TAPE, LINK7.

11.4-8

7-1 in

NASTRAN ON THE CDC 6000/CYBER (NOS AND NOS/BE)

Contra Mary and

Law Contractor

•••°,

*

COPYBF, TAPE, LINK8. 11. 12. COPYBF, TAPE, LINK9. COPYBF, TAPE, LINK10. COPYBF, TAPE, LINK11. COPYBF, TAPE, LINK12. 13. 14. 15. 16. CØPYBF, TAPE, LINK13. COPYBF, TAPE, LINK14. 17. COPYBF, TAPE, LINK15. 18. 19. RETURN, TAPE *--SET UP UMF TAPE 20. REQUEST(UMF, NT, D=PE)VSN=xxxxxx *--EXECUTE NASTRAN 21. RFL(220000) 22. REDUCE (-) 23. LINK1, INPUT. 24. /EØR 25. (NASTRAN INPUT DATA) /EØĖ 26.

ì

Notes: 20. UMF tape set up. Reference is made to it in the Executive Control Deck of the NASTRAN input (UG 6).

11.4.3.8 Example 8 - Executive Demonstration Problems 1-1-1 and 1-1-1A.

These problems test Statistics Rigid Format #1 and the restart code, and use the User Master File to obtain bulk data. Note that the PUNCH file, which contains restart table data, must be PACKed to eliminate record marks (7-8-9 cards) within the file.

NA' ,AN ,T500. 1. U' .<, -----C +RGE, -----REQUEST (TAPE, NT, D=PE, LB=KU) NASTO1 2. REWIND, TAPE 3 COPYBF, TAPE, LINK1. COPYBF, TAPE, LINK2. 4. 5. CØPYBF, TAPE, LINK3. 6. COPYBE, TAPE, LINK4. 7. COPYBF, TAPE, LINKS COPYBF, TAPE, LINK6 8. 9. COPYBE, TAPE, LINK7 10. COPYBF, TAPE, LINK8 11. 12. COPYBF, TAPE, LINK9 COPYBF, TAPE, LINK10 COPYBF, TAPE, LINK11 13. 14. 15. COPYBE, TAPE, LINK12 COPYBF, TAPE, LINK13 16. COPYBF, TAPE, LINK14 COPYBF, TAPE, LINK15 17. 18. RETURN, TAPE . 19. *--GET THE UMF AND DRIVER DECKS REQUEST(TAPE,NT,D=PE,LB=KU)NAST03 20. COPYPE, TAPE, UMF. 21. 22. COPYBE, TAPE, OLDPL. REWIND, UMF 23. 24. RETURN, TAPE. *-- UPDATE A DRIVER DECK AND EXECUTE 25. UPDATE,Q,C=NASDATA.

SYSTEM CONTROL CARD DECKS

REDUCE(-)
LINKI NASDATA.
REDUCE
+PROCESS FILES FOR RESTART
REWIND, NPTP, PUNCH, UMF.
PACK, PUNCH.
RETURN. OOL.
RENAME OPTP=NPTP.
* UPDATE SECOND DRIVER DECK AND EXECUTE
UPDATE, Q, C=NASDATA.
RFL (220000)
REDUCE(-)
LINKI NASDATA.
/EGR
*C DDC1011
/EØR
*C DDC1011A
*I DDC1011A.3
*READ PUNCH
/EQF

Notes: 20. Attach tape containing UMF and UMF Driver Decks.

Contention - Vier de las

MARK THE SHOULD BE AND A DESCRIPTION

27.

- 21. Copy UMF to permanent file.
- 22. Copy UMF Driver Decks to ØLDPL for CDC UPDATE utility.
- 25. Obtain driver deck for input into NASTRAN.
- 28. NASTRAN execution. Input found on file NASDATA, generated by update.
- 29. Turn over control of field length to system.
- 31. Pack punch file containing checkpoint dictionary.
- 32. Scratch previous NASTRAN P00L file.
- 33. Old NPTP is now new ØPTP.
- 34. Obtain follow-up driver deck for input into NASTRAN.
- 37. Execute NASTRAN a second time.
- 39. UPDATE command to place driver deck DDC1011 onto UPDATE compile file.
- 42-43. Checkpoint dicitionary residing on file PUNCH is inserted into driver deck DDC1011A for restart purposes.

11.4.3.9 Example 9 - Data Transmission

NASTRAN, T500. 1. USER, -----CHARGE, -----DELIVER. -----*--DUMP NASTRAN FROM TAPE TO DISC REQUEST(TAPE,NT,D=PE,LB=KU)NAST-1 2. REWIND, TAPE. 3.

- COPYBF, TAPE, LINK1. COPYBF, TAPE, LINK2. 4.
- 5.

11.4-10

٠

1. "这一些一些种情况起来了我们的情况。"他的话题是一些一个人

6.	CØPYBF, TAPE, LINK3.
7.	COPYBE, TAPE, LINK4.
8.	COPYBE TAPE LINKS.
9.	COPYBE TAPE LINKG.
10.	COPYBE TAPE I INK7.
ii.	COPYBE, TAPE, LINKS.
12	COPYRE TAPE I INKO
13	COPYRE TAPE I INKIO
14	CODVRF TADE I INVII
15	CAPYRE TADE I INV12
16	CADVRE TADE I INVIS
17	CADVRE TADE LINKIA
18	CODVRE TADE I INVIS
10.	DETION TADE
13.	A HEED TADE CETHD
20	TUSER TAPE SETUP
20.	
21.	REQUEST (UIS, NI, U=PE) SAVE
~~	TEXECUTE NASTRAN
22.	RFL (220000)
23.	REDUCE(-)
24.	LINKI, INPUT.
25.	/EØR
26.	(NASTRAN INPUT DATA)
27.	/EØF
Notes	s: 20. Requests user tape to be read by DMAP utility INPUTT? (UM 5.3.2)

^{21.} Requests user tape to be written on by DMAP utility OUTPUT2 (UM 5.3.2). FØRTRAN unit 15 is implied.

11.4.3.10 Example 10 - Execution under NØS/BE

1.	NASTRAN, T500.
	USER
	CHARGE
2	REQUEST (TAPE .NT. D=PE. LB=KU, E=SI)NASTO1
3.	REWIND TAPE.
Δ.	CREVER TAPE I INKI
5	COPYRE TADE 1 INK2
<u> </u>	CODVDE TADE I INVO
7	CADVAC TADE I TNVA
<u>6</u>	CODVOC TADE I INVE
o. 0	CADVDE TADE I TAVE
3.	CONVOL TADE I TANZ
U .	CUPTOF, TAPE, LINK/.
1.	CWPTBF, TAPE, LINKS.
Ζ.	CUPTER, TAPE, LINKS.
3.	COPYBF, TAPE, LINKIO.
4.	COPYBF, TAPE, LINKII.
5.	CDPYBF, TAPE, LINK12.
6.	COPYBF, TAPE, LINK13.
7.	COPYBF, TAPE, LINK 14.
8.	COPYBF, TAPE, LINK15.
9.	RETURN, TAPE.
	*READ AND EXECUTE NASTRAN PROC.
20.	C9PYCR, INPUT, NASTRAN
21.	BEGIN, NASTRAN, NASTRAN.
2.	/EØR
23.	.PROC.NASTRAN.RFL=200000.INPUT=.OUTPUT=.PUNCH=.

#RFL,RFL. LINK1,INPUT, #UTPUT,PUNCH. 24. 25.

11.4-11

SYSTEM CONTROL CARD DECKS

26. WHILE,RIG.EQ.O,REPEAT. 27. BEGIN,NEXT,TAPE99. 28. ENDW(REPEAT) 29. REVERT. 30 EXIT. 31. REVERT(ABØRT) 32. /EØR 33. NASTRAN SYSTEM(76)=1 34. (NASTRAN INPUT DATA) 35. /EØF

اليهاد والاجتماع بالعقر وتجاد كالعبة القارب الأراج الأراد

Ì

Notes: 20. CYBER Control Language Procedure File (CCL Proc) is copied to file NASTRAN.

21. Transfer control to file NASTRAN.

23-31. CCL Proc.

33. SYSTEM(76) must be set equal to 1 on the NASTRAN card to inform NASTRAN of NØS/BE operating system.

12. NASTRAN INFORMATION SOURCES

Contraction Contract Contract of the

12.1 NASTRAN NEWSLETTER

Periodically, the NASTRAN Newsletter is published to provide NASTRAN users with timely information on the development of new capability, Software Problem Reports (SPR), Documentation Error Reports (DER), and user's experiences. The Newsletter is free and may be subscribed to by writing COSMIC at the following address:

> Computer Software Management and Information Center University of Georgia Suite 112, Barrow Hall Athens, Georgia 30603 Telephone: (404) 542-3265

A blank form to be used in subscribing is shown in Figure 1. The information requested on this form is used by COSMIC to help them provide better service to all NASTRAN users.

NASTRAN INFORMATION SOURCES

¥

NO CHANGE **OB AREA**

PREFIX

INITIALS

LAST NAME

Ĩ

mille &





١,

NASTRAN HEWSLETTER

with the second second second second





.....

.

12.1-3



Figure 1. Application to receive NASTRAN Newsletter (cont'd).

?

a company of the proceeding of the

,

12.1-4



.

Figure 1. Application to receive NASTRAN Newsletter (cont'd).

and the second second second second

an an Ville a strategical

NASTRAN NEWSLETTER

. . . . **.**

SOFTWARE PROBLEM REPORT (SPR) LOG

12.2 SOFTWARE PROBLEM REPORT (SPR) LOG

Ē

Another service called the Software Problem Report (SPR) Log is also provided by COSMIC as part of their NASTRAN maintenance service. This log is available to all interested users by writing to COSMIC. The SPR Log is essentially a working document whose primary value lies in its currency. For this reason, it is continuously updated and published at frequent intervals.

The user is encouraged to submit SPRs to COSMIC as a service to all other users of NASTRAN. The recommended procedures for doing so are presented in Section 12.4. The Construction of the second second second

12.3 NASTRAN USER'S COLLOQUIUM

More formal reports on application and development of NASTRAN have been published by NASA following each NASTRAN User's Colloquium held at various NASA centers around the country. The Colloquium proceedings now available are entitled:

"NASTRAN: USER'S EXPERIENCES"

- 1. NASA TM X-2378, September, 1971
- 2. NASA TM X-2637, September, 1972
- 3. NASA TM X-2893, September, 1973
- 4. NASA TM X-3278, September, 1975
- 5. NASA TM X-3428, October, 1976

"SIXTH USER'S COLLOQUIUM"

6. NASA CP-2018, October, 1977

"SEVENTH USER'S COLLOQUIUM"

7. NASA CP-2062, October 1978

These proceedings may be obtained at modest cost from:

National Technical Information Service Springfield, Virginia 22151

SPR SUBMITTAL

and the second
12.4 SPR SUBMITTAL

It is a goal that every NASTRAN run terminate with either correct results or a user-oriented message describing the difficulties encountered. Progress towards this goal is made by means of user-submitted materials describing software problems. Figure 1 shows the standard SPR submittal form.

The following guidelines are recommended so that each Software Problem Report (SPR) can be efficiently handled.

- If you suspect an error, first try to determine if the error is already known. An SPR log is available through COSMIC which lists known errors. This log is indexed by module and often contains a suggested user avoidance procedure.
- If the malfunction involves a computer-dependent termination, such as system abends on the IBM 360 or error modes on the CDC and UNIVAC computers, consult first with your own programming operations staff for possible job set-up errors.
- 3. Having decided that the problem is caused by an unknown malfunction in NASTRAN, and if you have programmers available, attempt to determine the cause by consulting them and the listings. The best error submittal is one which contains a run which exemplifies a stated error and contains a suggested program correction along with the statement, "We tried this fix and it worked," or even "We tried this and it didn't work."
- 4. Having exhausted your own resources, consider sending the run to COSMIC for further investigation. If you consider this approach, be sure that the run has as much diagnostic data as possible. In particular, be sure that DIAG's 1, 2, 3, 8, 13, 15, 16 (for inverse power problems), and 19 were on and that a dump was taken. Be sure to include an echo of all the input data and the Run Log (the day file (CDC), Log (UNIVAC), or FTO4 (IBM)). If this data is not available, please rerun the job to obtain it. You should also take the time to describe the circumstances surrounding the run, i.e., "I changed this one card from when it ran before," etc. A complete NASTRAN data deck in card form should also be submitted, because later testing and other reruns may need

12.4-1

NASTRAN INFORMATION SOURCES

to be made. If plots have been made, send copies of them. If user tapes are involved, dump them and send the dump. Be sure to give your name, address, phone number, and as precise a description of the difficulty as you can.

If you take the time to follow these guidelines, COSMIC will be able to respond to SPRs more quickly and more economically. The processing of SPRs can be delayed for extended periods when insufficient information is submitted to permit prompt identification and correction of the reported error.

Ê,

SPR SUBMITTAL

j

.

e.

	Date:
)riginator:	NSMO_Use
Organization:	
Address:	Priority :
	Date Rec'd. :
	Date Assigned:
Phone No.:	
M. A	Level:
Materials Submitted:	Computer:
() Output: Runs	Rigid Format:, 🗋 Disp 🗍 Heat 🗋 Aero
() Deck	or DMAP Alters
() Plots	Error Message:
() Letter	Module:
() Dump	Subroutine(s):
() Traceback	
() Fix:	
() Program Listing	
() Link Map Listing	Avoidance (if known):
() Other:	
	Estimate correction effort (if known):
Description:	
[NSMO Use
Level Fixed :	
Test Problem :	
Vanddiad by News	
verified by NSAU :	

Figure 1. NASTRAN Software Problem Report (SPR).

12.4-3

and the an and

12.5 DER SUBMITTAL

It is also a goal of COSMIC to supply the user with complete and accurate documentation to facilitate the application and development of NASTRAN. Even though COSMIC carefully edits all documentation, errors of commission and errors of omission do occur. COSMIC recognizes, however, that it is the user who ultimately puts the documentation to its most severe test. Therefore, COSMIC welcomes all user reports of documentation errors. Valid errors will be reported to other users via the NASTRAN Newsletter.

A sample DER form is shown in Figure 1. The user is encouraged to submit this form to COSMIC along with a copy of the documentation page with annotations to indicate the error. If appropriate, also provide the specific text which would serve to correct that error. Some types of documentation errors may indicate the presence of software problems as well. If this is known, the user should consider submitting a supporting SPR following the procedures suggested in Section 12.4.

NASTRAN INFORMATION SOURCES

 \mathbb{R}^{n}

ē.

Ì

DER		SPR		NE	W CAPABILI	ITY				
			NASTRAN I	DOCUMENTATI	ON ERROR I	REPORT (DE	R)			
Origina	tor:					Date:		······································		
Organiza	ation	:			Phone No.:					
		-								
Kanua 1		Theoret	Ical				— ¬			
		User's					_]	Pa-ua		
		Program	mer's				- }	Numbers		
		Demonst	ration _				_)			
Descrip	tion	of the Er	ror and l	Correction	<u>s</u>					
(А сору	of t	he page(s) in erro	or with ma	rked corr	ections s	hould be	attached.)		
				<u></u>						
•										
••••								. <u></u>		
. <u></u>										
3										
Comment	5									
						<u></u>				
•										
••••••		<u></u>								
Edit Appro	or val /	NSMO Approva]	MPSL Entry	Change Made	Editor Check	NSMO Verif.	Editor Verif.	MPSL Entry		
Dat	e	Date	Date	Date	Date	Date	Date	Date		

Figure 1. NASTRAN Documentation Error Report (DER).

____·--

12.5-2

13. OPERATING PRINCIPLES OF NASTRAN

يري يورد در الارتجام معالية الارتقال الارتقال الارتقاد المناد الريد الراد المعادية ال

13.1 INTRODUCTION

To model a structure in NASTRAN requires a knowledge of the input and control options but little understanding of how the program itself works. However, a fundamental knowledge of the basic operating principals of the program will assist the user in fully using NASTRAN's capabilities. With the "black box" stigma of the program removed, the NASTRAN user is free to use many of the special purpose diagnostics and efficiency controls available to him. Understanding the operating principles also aids understanding processing failures, interpreting diagnostic output, and planning unusual analyses.

Presented in this section is an overview of NASTRAN as a computer software system. The overview includes a description of the major subdivisions of NASTRAN: the executive system, the functional modules, the input/output utilities, the matrix algebra utilities, and the data base file structures. Included here is a summary of standards and conventions used in NASTRAN code development to implement the theory of NASTRAN applications. The conventions described include the engineering notation used, the nested vector concept, the open core concept for data storage, formats used for matrices and tables, and fundamental concepts of the NASTRAN input/output (IØ) system. This section concludes with a discussion of the theory behind NASTRAN restart procedures.

In-depth treatment of the topics presented in this section are found in Sections 1 of the Theoretical and the Programmer's Manuals. Specific references to other material are also provided.

PROGRAM DESIGN OVERVIEW

13.2 PROGRAM DESIGN OVERVIEW

1

An understanding of the NASTRAN program design can be facilitated by a knowledge of some of the design criteria which are related to computer science rather than structural analysis. Chief among these criteria are:

- 1. Simplicity of problem input deck preparation.
- 2. Minimization of chances for human error in problem preparation.

- 3. Minimization of need for manual intervention during program execution.
- 4. Capability for step by step problem solution, without penalty of repeated problem set up.
- 5. Capability for problem restart following unplanned interruptions or problem preparation error.
- 6. Minimization of system overhead in the three vital areas by:
 - a. Not diverting core storage from functional use in problem solution.
 - b. Not diverting auxiliary storage units from functional to system usage.
 - c. Not using system housekeeping time for performing executive functions that do not directly further problem solution.
- 7. Ease of program modification and extension to new functional capability.
- 8. Ease of program extension to new computer configurations and operating systems.
- 9. Maintain ability to operate efficiently under a wide set of computer configurations.

The design goals for NASTRAN are achieved through modular separation of program activities. The major functional subdivisions of NASTRAN are presented in Figure 1. They are:

- The Executive System It provides for overall control of the job. It allocates files, supervises the input process, generates the execution sequence. It acts as an interface between the program and the computer operating system.
- System data and module parameters These data are stored in small, core-held common blocks. These tables serve to define the status of the system and provide communication between the system and modules.

13.2-1

OPERATING PRINCIPLES OF NASTRAN

- 3. Functional Modules These are self-contained blocks of code designed to perform selected analytical functions such as element matrix generation or assembly with a minimum number of interface data blocks. To provide simplicity and convenience for updating and maintaining the system, the following restrictions are applied to functional modules:
 - a. Modules may interface with other modules only through auxiliary storage files, as opposed to passing information between each other while in core.
 - b. Since the availability and allocation of auxiliary files for module execution interact with the execution of other modules, no module can directly specify or allocate physical files. The auxiliary storage allocation is reserved as an Executive function.
 - c. Modules operate as independent subprograms, and may not call, or be called by, other modules. They may be entered only from the Executive System.
- 4. GINØ (General Input/Output utilities) A comprehensive set of utility subroutines developed for manipulating data blocks. GINØ routines are always used to read or write data blocks for internal NASTRAN operations.
- 5. Data Block A collection of unformatted tables and/or matrices used for writing to and reading from auxilliary storage. The contents and formats are defined by the routines creating them. Every NASTRAN file is made up of data blocks except for usersupplied files defined via the job control, all files are allocated temporarily.
- Matrix Utilities A set of mathematical subroutines provided for matrix decomposition, matrix multiply, matrix partitioning, eigenvalue analysis, etc.

These major subdivisions of NASTRAN and their relationships to each other are described in more detail in the following three sections.

PROGRAM DESIGN OVERVIEW

i nanan ni ni ba

.

ļ

كترأد للم

È,

Ň



Figure 1. NASTRAN Program Subdivision.

13.2-3

THE EXECUTIVE SYSTEM

and the second s

13.3 THE EXECUTIVE SYSTEM

The essential functions of the Executive System are:

- Establish and control the sequence of module executions according to options specified by the user.
- 2. Establish, protect, and communicate values of parameters for each module.
- Allocate system files to all data blocks generated during program execution. A file is "allocated" to a data block, and a data block is "assigned" to a file.
- 4. Maintain a full restart capability for restoring a program execution after either a scheduled or unscheduled interruption.

The Executive System is open-ended in the sense that it can accommodate an essentially unlimited number of functional modules, files, and parameters. Modifications of the Executive System to accommodate changes or additions to functional modules are performed with changes to control data tables.

The Executive System controls the execution of NASTRAN in two phases: (1) the preface, in which a basic setup is performed; and (2) the execution of the DMAP program in which the modules are controlled. The key executive subroutines performing these steps are:

- XCSA (Control Section Analysis) Processes the Executive Control Data Deck and loads the DMAP code from the Rigid Format Tables.
- GNFIAT (Generate FIAT) Allocates the NASTRAN working files and assigns initial data block names to the File Allocation Table (FIAT).
- 3. XGPI (General Problem Initialization) Processes the Direct Matrix Abstraction Program (DMAP) much like a compiler. Restart logic is also performed to identify the modules to be re-executed. The output is the Operational Sequence Control Array (ØSCAR), which provides the information for executing the modules.
- 4. XSFA (Segment File Allocator) Manages the data block to physical file relationships throughout the NASTRAN run. It reallocates files when new data blocks must be allocated and previously used data may be discarded to free the file to the system.

OPERATING PRINCIPLES OF NASTRAN

5. XSEMi (i = 1,2...M) (Executive Sequence Monitor, Link i) - The link-driven routines which control the calls to the functional modules. If a module resides in the ith link, the XSEMi routine requests the file and parameter allocations and directly calls the module. If the module resides on another link, the XSEMi routine returns to the overall control for loading another XSEMi routine. Each NASTRAN link is a major overlay segment of the program.

The general programming philosophy behind the Executive System routines is that reliability and efficiency are paramount concerns. It was also anticipated that few changes were to occur in the Executive System as the NASTRAN system grew, thus, few general rules were imposed on the Executive System code, resulting in a sophisticated, but difficult to modify, system.

In addition to the above specialized functions, the Executive System controls the input data card processing, and provides various other functions such as defining open core and clock time. For more information on the topics above, consult the NASTRAN Programmer's Manual Sections 1.2, 2.4, 2.2, and 4.1 through 4.20.

FUNCTIONAL MODULES

and the second of the second

13.4 FUNCTIONAL MODULES

A NASTRAN module consists of one or more subroutines which perform a single computational process. The size of a module may range from less than ten lines of code to many thousands of executable statements. Functional modules serve to manage or control a series of operations, which may be executed by other modules or matrix utilities, to accomplish an analytical objective. In general, functional modules are executed in the sequence specified by the NASTRAN Rigid Formats or by user DMAP instructions.

13.4.1 Module Types

Four module types occur in NASTRAN, and each type is limited to a specific range of functions. These are:

- Preface Modules Executes under direct control of the Executive System during the problem initialization. Since these modules are not used by DMAP and their data block names are fixed, many of the normal NASTRAN coding restrictions do not apply. For example, preface modules may write on their own input files or on the reserved NASTRAN system files.
- Executive Operation Modules Perform logical DMAP operations such as JUMP, L@@P, and FILE declaration. Except for the checkpoint module, none of these modules perform any major calculations or I/@ operations.
- Utility Modules Designed for general use with simple inputs and outputs. Utility functions range from performing print requests to complex matrix operations.
- 4. Analysis Oriented Modules Provided for specific steps in the analysis of structures which are not easily defined with matrix algebra. Examples are the element table assembler (TA1) and the stress recovery routine (SDR2). Other modules are provided for more efficient processing of matrices than available with Utility Module DMAP steps. An example is module SSG2 which reduces the static load vectors using a series of matrix operations.

13.4.2 Module Data Interfaces

Communication between modules is performed using data block files and parameters. The number
OPERATING PRINCIPLES OF NASTRAN

of input files, output files, and parameters for every DMAP module are pre-defined in the block data program XMPLBD.

and a strategy of the state of the strategy of

Input data block files are specified as file number 101, 102, 103 ... 100+N according to the DMAP calling sequence. These numbers are used in the calls to the GINØ utility subroutines for all file operations. Similarly, the output files are defined as 201, 202, 203 ... 200+n and the internal scratch files are labled 301...300+N. Several of the GINØ system error messages reference these numbers. The user must refer to the Rigid Format description in the User's Manual to determine the actual data block in question.

Parameters are stored in the blank common block (C@MM@N /BLANK/ list). The types may be integer, real, complex, or BCD characters. Floating point numbers may be single or double precision. The type of each parameter is fixed in the module code and in the system tables. Modules may use the parameters as either input or output data.

13.4.3 Module Overlays and Open Core

Each module in NASTRAN generally is assigned separate overlay segments independent of the other modules. During execution only the module code and the required utility subroutines are loaded into core. Furthermore, a common block assigned by the module is loaded below the current code for use in an "open-core" storage space. Open core exists from this point to the maximum core space requested for the execution. All matrix and table data required for a given module operation are brought into open core for temporary storage. Results are moved from open core to data block files for access by other modules.

A typical NASTRAN overlay "tree" is illustrated in Figure 13.2. Shown are examples of modules which are divided into separate parts, consisting of a small "driver" routine which calls the major sections of code located near the bottom of core. This scheme allows the large and complex matrix utility routines to be used by more than one module in a link, yet still allows the largest possible open core space for each module.

Each module in NASTRAN performs it's own open core allocations. The executive subroutine KØRSZ (core size) is used to determine the available length of core relative to a particular data location. The module allocates space for GINØ Input/Output buffers, along with all core-held variable length arrays. The modules are also responsible for their own spill processing. This occurs when a particular set of arrays do not fit in the available open core and must be processed a piece at a time using scratch files for the extra data.

13.4-2

FUNCTIONAL MODULES

an an ing takan pang ang pang barang a

13.4.4 Module Restrictions

A number of general rules have been followed in the development of NASTRAN modules which provide efficiency, reliability, and east of maintenance. Understanding these rules may aid in the use of the program. The general module design rules are:

- Only GINØ_utility routines (e.g., @PEN, CLØSE, READ, WRITE, PACK) are used to access data blocks and scratch files.
- 2. Only substructure utilities (e.g., SØF@PEN) are used to access the automated substructure operation file (SØF).
- 3. No data is written on input data blocks. (Note, however, that a module may read output data blocks.)
- 4. Values of system parameters and arrays are not changed.
- 5. Utility routines are used whenever possible to avoid unnecessary code duplication. An example is the use of the MESAGE routine which stores common error message print formats on a separate overlay.

For additional module coding restrictions and conventions, refer to Sections 1.3 through 1.5 of the NASTRAN Programmer's Manual. For discussions of the utility routines refer to Sections 1.7, 3.4, and 3.5 of the Programmer's Manual. For individual module descriptions refer to Section 4 of the Programmer's Manual. OPERATING PRINCIPLES OF NASTRAN

,

Ł

÷.



Figure 2. Illustration of NASTRAN Overlay System

•••••

THE NASTRAN I/ SYSTEM

about the set of more a

13.5 THE NASTRAN I/Ø SYSTEM

NASTRAN is primarily a file-oriented program, using mass storage devices for nearly all major data transfers. The four types of Input/Output systems provided in NASTRAN are the external user interfaces, the internal data blocks, the checkpoint/restart file, and the substructure data base. These are described individually below.

13.5.1 External User Interfaces

Aside from the conventional input data file and output print files, NASTRAN may be used to read or write data on other user-supplied external files. These are summarized below;

- Plot files are created for both structure plots (plots of the model geometry) and dynamic response curves (XY output plots). The format is dependent on the computer hardware and plotter hardware.
- The punch file is used for several purposes. The restart dictionary data are automatically generated and stored on the punch file. The DMAP program may be punched if DIAG 17 is requested. All printout data may also be punched for post-processing by other programs.
- 3. User Files are machine-dependent file names that are requested by the user in his system job control cards. Modules using these files must be defined in a DMAP program. Modules INPUTT1 and @UTPUT1 read and write data blocks in NASTRAN format. INPUTT2 and @UTPUT2 read and write data blocks which are converted to F@RTRAN format on the user files.
- 4. Substructure Operating Files (Spr) are direct access device files defined by the user for storing substructure data. A data base management system automatically provides the required NASTRAN I/Ø interface.

For more information on the external I/Ø interfaces of NASTRAN, see Section 5.0 of the NASTRAN Programmer's Manual for system interface descriptions versus computer type.

13.5.2 NASTRAN Data Blocks

Data blocks in NASTRAN are treated as separate, self-contained files used for temporary data storage. The Executive System allocates the storage devices while the modules perform the calculations to generate data for storage in the data blocks.

OPERATING PRINCIPLES OF NASTRAN

(1 + 1) = (1 + 1) + (1 +

Each data block may be subdivided into an unlimited number of records. Each record may contain an unlimited number of words or may be completely null. In rare cases a data block may also contain several logical files, but this practice is discouraged. Detailed information on the data blocks used in the NASTRAN Rigid Formats is contained in Section 2.3 of the Programmer's Manual.

The most general form of data block is the table. Any format of data may be used with mixed integers, floating point number, and BCD characters. No data conversions are performed on tables.

A standardized packed matrix format is used for all NASTRAN data block matrices. Only nonzero terms and their packed identifiers are stored on the file. A set of matrix utility subroutimes are used to read and write matrix terms. All NASTRAN matrices are stored by columns, i.e., each column is a separate logical record.

Ali mASTRAN data block names are stored in core-held tables which contain 6 words of so-called "trailer" information. The trailer data is used to identify the characteristics of files without actually performing any I/Ø operations. The trailers for matrix data blocks are particularly use-ful in that they contain the dimensions and density of the matrix. The input executive control card DIAG 8 will produce a printout of each trailer as the data block is created. The contents for matrices are:

WORD	DEFINITION
1	Internal Data Block Name
2	Number of Columns
3	Number of Rows
4	Type of Matrix (square, rectangular, etc.)
5	Type of Matrix Terms (single, double, etc.)
6	Maximum Number of Non-Zero Words in a Column
7	Matrix Density x 10 ³

For definitions of the types of matrices see the DMI (Direct Matrix Input) bulk data card description in Section 2.4 of the User's Manual.

13.5.3 Checkpoint Files

The NASTRAN checkpoint file is defined by the user on his machine-dependent job control input. A single operating system file is required, which may be on any of the available system permanent 2

THE NASTRAN 1/9 SYSTEM

bulk storage devices, including multi-reel tapes. If a checkpoint is requested, the contents of the NPTP (New Problem TaPe) will include the current case control data, the sorted bulk data, all NASTRAN data blocks written via the DMAP request CHKPNT, and the parameter values at each checkpoint stop. Associated with the NPTP file is a checkpoint dictionary which is normally written on the punch file in card images. The restart dictionary contains descriptions of the data blocks and the XVPS (Executive Variable Parameter Save Table) stored on the NPTP file.

On restarts, the previously generated checkpoint file is assigned as the **PPTP** (**pld Problem TaPe**) file. The corresponding checkpoint dictionary is included in the Executive Control Deck and the program automatically proceeds with the restart process.

13.5.4 Substructure Operating Files

Ľ, j

For automated multi-stage substructure analysis, the Substructure Operating File (SØF) is a single logical file used to store all data necessary for a complete multi-stage substructuring analysis. The SØF may actually reside on one to ten physical files. However, these physical files are chained together to form a single logical file for use in the analysis of larger problems.

Each physical file comprising the SØF must be a direct access disk file. These files are not processed with NASTRAN GINØ operations, but treated as external user files. In a substructure analysis, NASTRAN stores data on the SØF which must be saved from run to run. Therefore, it is the user's responsibility to maintain the physical disk files comprising the SØF from one execution to the next. A complete description of how to use and maintain the SØF is presented in Section 20.12 of the User's Guide and Section 2.7 of the User's Nanual.

13.5-3

NASTRAN CHECKPOINT/RESTART LOGIC

الداني الحالية من من ترجيع والمنظم والمنظمة المنظمة المنظمة المنظمة والمن والمراجع من المنظمة والرواي والمنت

13.6 NASTRAN CHECKPOINT/RESTART LOGIC

NASTRAN has a sophisticated checkpoint and restart capability. This capability allows the user to efficiently recover after either a scheduled or unscheduled interruption. Some situations applicable to the use of checkpoint/restart are listed below:

- 1. A run may abort due to insufficient time, a user data error, or insufficient core.
- 2. The user may request termination of his run at selected steps during the analysis for the purpose of examining intermediate results prior to continuing the run.
- 3. The user may wish to rerun only a portion of the analysis with limited data changes.

The overall design philosophy for restart is twofold. A restart selectively executes only the modules necessary to accomplish a user-input data change. The user is able to change any part of his problem including model data changes, additional cases, or more output requests. At the same time restarts are automatic as far as user interference is concerned. The user need only check-point his orfginal run and submit changes to the original run on subsequent runs.

13.6.1 Restart Types

In NASTRAN there are four general types of restarts. Unmodified Restart (UMR), Pseudo Modified Restart (PMR), Modified Restart (MR), and Rigid Format Switch (RFS). In Section 3 of the User's Manual, UMR's and PMR's are described together as Unmodified Restarts. These classifications are for descriptive and internal purposes. The user need not know anything about which type is which. The basic characteristics of each type will be described below.

An Unmodified Restart results when the user simply resubmits a problem with no data changes. It is used to continue a solution from the point of interruption. Presumably the problem aborted previously due to time expiring, or the user-requested early termination for the purpose of examining intermediate results prior to completing the solution. The restart dictionary (created while checkpointing) is processed, and the solution is started again at the last re-entry point (after the last successful checkpoint).

A Pseudo Modified Restart occurs when the user requests additional output from the program which simply requires the re-execution of an output module such as the Structure Plotter, the Grid Point Weight Generator, or the Stress Data Recovery module. The numerical solution is

13.6-1

OPERATING PRINCIPLES OF NASTRAN

not effected by these modules; only new output is generated. Note that a PMR is the common case since printer output, plotter output, etc. is usually requested. A true UMR is rare. On a PMR, output modules are re-executed to display requested output, and then the problem is continued at the re-entry point.

A true Modified Restart occurs when some numerically significant data change. The modules which process these data must be re-executed. These modules are re-executed to regenerate their output data blocks based on the new data and the problem is continued at the re-entry point.

A Rigid Format Switch (RFS) is a special form of Modified Restart in which a problem changes from one solution type to another. One example would be: a user has solved for the static solution on Rigid Format 1 and now wants to find the normal modes by using Rigid Format 3. RFs may or may not require data changes. The key difference here is that the re-entry point cannot be used to determine the proper place to restart. For a RFS, NASTRAN examines the final modules on the new Rigid Format and lets the File Name Table chain the solution back to the proper restart point.

13.6.2 Restart Decision Process

an and the second of the second s

To understand, in general, how the above types of restart are implemented, it is necessary to consider the Module Execution Decision Table (MEDT), which is associated with each Rigid Format. The Module Execution Decision Table has one entry for every DMAP instruction in the Rigid Format. A Card Name Table associates bits of the MEDT with selected bulk data card names, case control selections, and parameter names. The File Name Table associates bits of the MEDT with selected bits of the MEDT with selected bits of the MEDT with selected data block names. The following example illustrates the use of these tables in determining the effects of changing a bulk data card on a Modified Restart.

Suppose the FØRCE bulk data card is to be changed when executing Rigid Format 1. DMAP modules BEGIN, FILE, FILE, GP3, SAVE, PARAM, PURGE, CHKPNT, SSG1, CHKPNT, EQUIV, etc., will be executed since the bit associated with loads is set for each of those modules.

Each part of the NASTRAN Preface contributes to processing the information for a restart. The XCSA module extracts and stores the Card Name Table, the File Name Table, the Module Execution Decision Table, the DMAP sequence, and the Rigid Format Switch bit if any. These are written in the XCSA Executive Control Table on the New Problem Tape for later use by module XGPI. IFP1 compares the current CASECC (case control) data block with the one submitted on the previous run (a copy of CASECC is stored on the Old Problem Tape for this purpose).

13.6-2

NASTRAN CHECKPOINT/RESTART LOGIC

and set of the state of the set

Three types of changes are noted by IFP1:

- Changes in data set selection such as Load set, SPC set, etc.;
- 2. The occurrence of output requests for printer, plotter, punch;
- 3. Changes in the looping structure of the problem.

XSORT analyzes bulk data card changes in a similar manner; setting the proper bits in the tables. XGPI then analyzes this information to decide which modules are to be executed. Input data blocks are then checked for RFS restarts to find if they are to be generated or available on the checkpoint tape. If the input data blocks are missing, the modules which generate them are flagged and the process continues until all module data requirements are satisfied.

13.6.3 Restart Rules

-

Some of the basic program rules followed in the restart decision process are:

- 1. The first choice of re-entry points is always the last entry in the checkpoint dictionary.
- If a data block occurs more than once in the dictionary, the last entry before the reentry point is always used.
- 3. On DMAP looping problems the program will restart either at the last entry for the last loop or at the first, depending on the changes. Restarts in the middle of a set of loops must be controlled by manually editing the dictionary.
- 4. User-supplied DMAP ALTER code is always executed. ALTER code defined in the checkpoint run is not saved and must be included in the restart if required.

NASTRAN PROGRAMMING CONVENTIONS

water a transform and the second second second second

13.7 NASTRAN PROGRAMMING CONVENTIONS

NASTRAN development encompassed a comprehensive set of rules for terminology, messages, and user controls. These have included the use of consistent mneumonics for the names of data program variables, and module/subroutines. These conventions have been useful both to the programmer and the engineer. A unique example, described below, is the use of vested degree of freedom sets for naming vectors, matrices, and their partitions. This nomenclature system is followed from the basic theoretical development, through the user data inputs, to the actual code. The reader is directed to Section 7.1 of the User's Manual for a comprehensive dictionary of terms which frequently occur in the use of NASTRAN.

This section describes some of the conventions particularly useful when running NASTRAN. Some of the particular diagnostics and notation systems are included along with error message rules and some methods for overriding the system.

13.7.1 Diagnostic Outputs

The executive control card, DIAG, is particularly useful in diagnosing problems in NASTRAN. The format is:

DIAG n₁, n₂, n₃...

where each integer n produces a specific type of diagnostic output. Some sample diagnostics are given in the list below:

<u>n</u>	Type of Output
8	Prints matrix and table trailers when they are generated
13	Prints open core length
15	Trace open/close operations on data blocks
21 ,22	Print degree-of-freedom definitions for all points and all types of constraint sets
23	Print Automatically-generated DMAP ALTER data for substructure problems

The diagnostics referenced above are most commonly used and greatly assist the user in debugging NASTRAN problems. These and all other diagnostics are described in detail in Section 10.2 of the User's Guide.

13.7-1

13.7.2 Matrix/Vector Degree of Freedom Sets

In NASTRAN, vectors and matrices share the same format. Each row or column corresponds to a single degree of freedom defined as a generalized displacement at a grid, scalar, or extra point. Each degree of freedom, u, belongs to one or more sets (e.g., u_g and u_m) as illustrated in the diagram below:



where the sets on the left-hand side are selected by the user as constraints either explicitly on SPC, NPC data or implicitly as subroutine degrees of freedom and left-hand displacement sets are defined in the list below:

 u_m coordinates are eliminated as independent degrees of freedom by <u>multi-point</u> constraints

us coordinates are eliminated by single point constraints

NASTRAN PROGRAMMING CONVENTIONS

يو اين د د **چ**ر دهندها خر کړ

u_	coordinates	are	omitted	by	structural	matrix	partitioning	
----	-------------	-----	---------	----	------------	--------	--------------	--

- u, are coordinates to which determinate reactions are applied in static analysis
- ug are the remaining structural coordinates used in static analysis (points left over)
- ue are extra degrees of freedom introduced in dynamic analysis to describe control systems, etc.

In static analysis, we are concerned only with the grid point set, u_g , and its subsets. The application of constraints and partitioning to the stiffness matrix involves, essentially, the elimination of u_m , u_s , u_o , and u_g to form a stiffness matrix, k_{gg} , with coordinates defined by u_g .

The physical significances of these computations are explained in Section 3.5 of the Theoretical Manual. The concept of nested vector sets is extremely important in the theoretical development of NASTRAN. The reader may, in fact, find it useful at some point to memorize the relations defined above among the displacement sets.

Load vectors are identified with the same notation. Rectangular matrices are, whenever necessary to clarify the meaning of the symbol, distinguished by double subscripts referring to the vector sets associated with the rows and columns of the array.

In the NASTRAN DMAP, a stational set of names have been given to the various types of matrices. If x corresponds to the degree-of-freedom set or size, the following list explains the names:

- UxV Matrix of displacement vectors
- Px Matrix of load vectors
- Kxx Stiffness matrix
- Mxx Mass matrix
- Bxx Damping matrix
- Gx Transformation matrix for dependent coordinates, x
- Dx Rigid body transformation
- Qx Forces of constraint
- PHIx Eigenvector matrix
- Uxx Upper triangular u imposition factor
- Lxx Lower triangLlam decomposition factor

For specific information on any data block, refer to Section 2 of the Programmer's Manual.

OPERATING PRINCIPLES OF NASTRAN

In some cases, additional letters are added to the basic data blank name to denote a special form of the matrix data. The following list defines these key letters:

Ø - Output data block

D - Differential stiffness

L and NL - linear and non-linear

1 or 2 - Denotes sort on output data.

13.7.3 Error Messages

NASTRAN provides a comprehensive system of error checking to facilitate recovery from error conditions. Section 6.1 of the User's Manual contains a list of all user and system message along with explanations when required. Some useful general comments on the error message system are listed below:

- 1. Warning Messages Are issued when a logical discrepancy is encountered, which may be extraneous to the actual problem, and the program may proceed.
- User Fatal Messages Are always related to input data errors or serious structure modeling problems when no alternative path is possible. Insufficient time limits or core size may also cause these messages.
- 3. System Messages Are intended to occur only when the program code is in error. However, these may occur due to user DMAP changes or after user warning messages have occurred, and the system cannot recover from the bad data.
- 4. Potentially Fatal Messages Occur in the DMAP compilation and in substructure problems. These are serious, but not necessarily fatal, errors. Using DMAP, the user may specify these errors to be fatal with the XDMAP control card.

When the cause of a fatal error condition cannot be immediately determined, the user has several recourses. The user should survey the run output for automatically generated informative, warning, or other fatal messages which may help identify the source of the problem. The user should determent from the computer system "day file" which NASTRAN module was in execution at the time of the error. Then he should determine which calculations were being performed on what data. The user will find details of Rigid Format calculation steps in Section 3 of the User's Manual.

HASTRAN PROGRAMMING CONVENTIONS

and the second states a second state of the

白いましたろう

If the procedures outlined above do not uncover the source of the problem, the job should be re-run with requests for appropriate additional diagnostic output. This additional diagnostic output should be requested using the DIAG card in the Executive Control deck. Also, DMAP Afters could be included to obtain table and matrix printout of key information which would help identify the problem.

If rerunning the job and evaluation of the new data obtained do not identify the source of the problem, then the error should be reported to COSMIC. Refer to Section 12 of the User's Guide for those procedures.

14. ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

a manager and a star of the second and and

14.1 INTRODUCTION

An important aspect to using any finite element program is the task of estimating costs and resource requirements to determine feasibility of processing a particular structural model. Because small, simple models usually do not tax the capacity of the computer hardware and they usually do execute economically, the user may be unjustifiably tempted to linearly extrapolate the costs for processing much larger models. Depending on model size and configuration, the user may encounter serious problems with excessive execution times, job turnaround, file space, and overwhelming amounts of printout. The actual costs and resource requirements for a NASTRAN analysis are complex functions of matrix size, density, bandwidth, and available computer resources, and may increase dramatically with problem size and complexity.

Methods used for estimating cost and resource requirements for a NASTRAN analysis are presented in this section. Because detailed explanations of the algorithms and timing equations are given in the Theoretical and Programmer's Manuals, this section is oriented toward presenting more practical working procedures. Included is a summary of critical operations performed in each Rigid Format and timing parameters for estimating these critical steps. Simplified equations are included for estimating CPU time versus problem size and available computer core. Also, methods are presented for estimating other resource requirements, such as file space and tape requirements.

1

METHODS FOR PREDICTING NASTRAN COST FACTORS

14.2 METHODS FOR PREDICTING NASTRAN COST FACTORS

Several methods are available to the user for predicting cost requirements of a particular structural model executing in a particular type of analysis. Estimates may be based upon experience, tests, and/or detailed examination of the computerized operations. The different methods are further explained in the following paragraphs.

14.2.1 Past NASTRAN Experience

It should be noted that the NASTRAN program is designed to treat large models. However, the sizes of the resulting matrices may exceed the capacity of the more efficient in-core processing algorithms. Therefore, the NASTRAN matrix processing modules also include a necessary "spill" logic to provide for those conditions in which the required segments of matrix data cannot be held in core all at once and must be spilled to periphery storage devices. Thus, whenever this "spill" logic is invoked, there is usually a quantum jump in processing cost due to the extra data transfers to and from periphery storage. Estimating computer run times when spill will occur is a complex process. For this reason, using previous NASTRAN experience, described next, is highly recommended.

If previous NASTRAN models = me similar to the problem in question, examination of the old runs will provide valuable insight into the costs of the different aspects of the analysis. A NASTRAN run log is recommended for cataloging pertinent data and costs of previous jobs. This log should include typical parameters such as number and type of elements, matrix sizes and densities, and matrix bandwidths, along with the corresponding execution times of key operations such as EMG, MPYAD, DECOMP, SDCOMP, FBS, and READ, etc. The start and stop time for each of these modules are given in the computer output, called the "Day File." The matrix size parameters may be obtained by including a DIAG 8 card in the Executive Control Deck. However, as mentioned previously, <u>extrapolating</u> run times is dangerous because the timing functions are complex. The run costs of some operations are proportional to the square of some, and to the cube of other size parameters. Also, when the size of the problem requires "spilling," i.e., becomes larger the can be conveniently handled in core, the cost of the additional multiple-pass operations may be several factors larger than those for the in-core run time.

14.2.2 Testing for Cost with NASTRAN

In several critical matrix operations such as multiplication, decomposition, and forwardbackward substitution, NASTRAN automatically estimates the CPU time before actually executing

14.2-1

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

the step. If the time limit given on the Executive Control card "TIME" will be exceeded, the program exits with a message containing that time estimate. The user may obtain the NASTRANcalculated estimate at a small cost by submitting his job with a deliberately low limit on the TIME card. A similar procedure may be used in dynamic analysis when many time steps or frequencies are expected and the cost is known to be proportional to the number of solutions. The user may submit the job for only a small number of steps in order to obtain the actual run time per step and then extrapolating the solution time for the larger job. Note that this procedure will only approximate the cost for eigenvalue extractions using the inverse power method. It would not be valid for the Givens-Householder and Hessenberg methods of eigenvalue extraction.

14.2.3 Evaluation of the Computerized Algorithms

⊇.

ì

When the operational steps within the program and the size and form of the data to be processed are known, the resources required to process these data may be obtained by an analysis of the code algorithms. Fortunately, many of the major operations in NASTRAN have been analyzed for the purpose of automatically performing detailed internal estimates of the central processing time (CPU) in order to protect the user. These equations involve the size and properties of the input matrix data for each step, the available computer core, and the particular timing parameters for each computer. These equations, which are described in the Programmer's Manual, also may be used by the analyst to estimate the required resources and determine feasibility prior to the embarking on the lengthy tasks of preparing and checking the finite element model, its constraints, and the solution processing requirements. The following sections describe how to proceed with these estimating tasks.

REQUIREMENTS FOR THE RIGID FORMATS

a contractor

14.3 REQUIREMENTS FOR THE RIGID FORMATS

Each type of solution in NASTRAN, or Rigid Format, requires its own characteristic mixture of computer resources. Summarizing all of the calculation steps for each of the Rigid Formats is beyond the scope of this guide. For the interested user, the detailed steps taken for developing and processing of all matrix equations are presented in Section 3 of the User's Manual for every Rigid Format. Some of the specialized operations such as PLA3, PLA4 (Plastic Analysis) are not covered below and would require separate and detailed analysis of the program code.

For most purposes, the more costly operations in the Rigid Formats may be isolated to a few modules. These critical modules are named and the operations they perform are summarized in Table 1. The major calculation modules and subroutines used in each of the Rigid Formats and in the substructure operations are listed in Tables 2 and 3. In the analyses which could involve boundary set changes, nonlinear iterations, or time step changes, groups of operations would be repeated by looping through selected segments of these Rigid Formats. These groups of operations are denoted by underlining of the module names involved in that looping. Available procedures for predicting the running times for most of the operations are given in the subsequent sections.

The matrix size coefficients referenced in Tables 2 and 3 are:

- L: Static matrix size after all constraints and omitted degrees of freedom have been removed
- R: Rigid body support degrees of freedom
- D: Dynamic matrix size = L + extra points
- H: Modal dynamic size = no. of modes + extra points
- NL: Number of loads
- NF: Number of solution frequencies
- A_i: Substructure matrix size, input to operation
- C: Contributing substructure matrix size
- B: Boundary size for output reduced matrices

14.3.1 Static Condensation and Guyan Reduction

Matrix static condensation and the associated Guyan reduction procedure are provided in all NASTRAN Rigid Formats as a means of reducing the order of the solution matrices. This feature is optional and is selected by use of ØMIT or ASET data cards. Reasons for using this feature are presented in the User's Manual, Section 1.4.

The most costly processing steps for static condensation and Guyan reduction are described below. Equations and numerical coefficients to be used for computing CPU running times and core requirements are presented in later sections.

Three major steps are involved in static condensation, and a fourth step is used for Guyan reduction. These steps are:

1) SDC@MP (o x o) (Decompose K_{oo})

$$[K_{oo}] = [L_{oo}][U_{oo}]$$
(1)

where the "o" degrees of freedom are either those unconstrained degrees of freedom specified on the ØMIT data cards, or they are those <u>not</u> listed on ASET data cards.

2) FBS (o x a) (Solve for matrix G_0)

$$[L_{oo}][U_{oo}][G_{o}] = -[K_{oa}]$$
⁽²⁾

where the "a" degrees of freedom are either those unconstrained degrees of freedom specified on ASET data cards, or they are those associated with points.

3) MPYAD (a x o x a) (Solve for reduced stiffness matrix, K_{aa})

$$[\mathsf{K}_{aa}] = [\bar{\mathsf{K}}_{aa}] + [\mathsf{K}_{oa}]^{\mathsf{T}}[\mathsf{G}_{o}]$$
(3)

4) For other matrices (M, B, etc.), the steps for each type are (using the mass matrix as an example):

$$\mathsf{MPYAD} (o \times o \times a), \ \mathsf{M}_{1} = [\mathsf{M}_{oo}][\mathsf{G}_{o}] + [\mathsf{M}_{oa}]$$
(4)

$$\mathsf{MPYAD} (\mathbf{a} \times \mathbf{o} \times \mathbf{a}), \ \mathsf{M}_2 = [\mathsf{M}_{oa}]^T [\mathsf{G}_0] + [\breve{\mathsf{M}}_{aa}] \tag{5}$$

$$\mathsf{MPYAD} (\mathbf{a} \times \mathbf{o} \times \mathbf{a}), \mathsf{M}_{\mathbf{a}\mathbf{a}} = [\mathsf{M}_{2}] + [\mathsf{G}_{0}]^{\mathsf{T}}[\mathsf{M}_{1}]$$
(6)

14.3-2

14.3.2 Element Matrix Generation

The cost of generating matrix partitions of stiffness, mass, and damping coefficients for the finite elements in NASTRAN is associated primarily with CPU activity and is not significantly affected by core size of I/Ø transfers. Each of these element matrix partitions is generated in core-held arrays using individually designed algorithms and procedures appropriate to each element type. Because of the differences in algorithms and procedures, the cost of generating matrices for each element depends upon the element type, its configuration, and its properties. For instance, the cost of matrix calculation for the BAR element will increase when pin joints, grid point offsets, and local coordinate systems at the connected points are selected by the user.

Typical execution costs of NASTRAN for the most used element types are shown in Table 4 for various computer models. Note that the costs for some major element options are shown. For different models of computers from the same manufacturer, the costs will be nearly proportional to the outer loop factor, M, described later in Section 14.4. However, note that the element execution times are not proportional to M when comparing costs between manufacturers. This effect is caused by the differences in compilers and utility subroutine efficiencies.

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

14.3.3 Determination of Open Core Size

The actual working core parameter, W, is dependent on the core allocated by the user, the space used by the necessary portion of NASTRAN, and the type of matrix data. In the sections which follow, W is defined as the number of matrix terms which may be placed in working core. If C is the number of core words available, the number of terms is:

$$W = C/p \tag{7}$$

where:

p = {
 1, real single precision
 2, real double precision or
 complex single precision
 4, complex double precision

For IBM and UNIVAC machines, double-precision arithmetic is normally required and is the default. For CDC hardware, only single precision is supported. For all machines, complex arithmetic is used in the solution and data recovery sections of Rigid Formats 7, 8, 10, and 11, and in the aeroelastic Rigid Formats only.

The exact number of words in open core may be obtained by including a DIAG 13 card in the Executive Control Deck. The requested printout will be included in the "day file" where the total space for open core available to the user is defined for each module. However, each module requires from two to six "buffers" for storing active input/output file data. The number of buffers depends upon the module, the input and output options, and upon the spiil activity in large-order problems. The buffer size times the number of buffers required must be subtracted from the available core size to obtain the factor, C, which is the number of words of working core for storage of matrix data. The user must refer to the Programmer's Manual for explicit buffer requirements for a particular NASTRAN module, e.g., SDC@MP.

REQUIREMENTS FOR THE RIGID FORMATS

Table 1. Definition of Critical Operations.

ì

APD, AMG, FAI	-	Aeroelastic data processing modules
COMPLEX MODES	-	Extract general eigenvalues, eigenvectors
DDR1	-	Dynamic Data Recovery - Recovers grid point displacements, etc.
DECOMP	-	General decomposition for unsymmetric matrices
DSMG	-	Differential Stiffness Matrix Generator, similar to EMG
EMG	-	Element Matrix Generation
FBS	-	Forward/Backward Substitution
MPYAD	-	Matrix Multiply - Add
MODES	-	Extract Normal Modes (READ module)
PLA3, PLA4	-	Piecewise Linear calculation modules
RMG	-	Radiation Matrix Generator
SDC	-	Symmetric Matrix Decomposition
SSGHT	-	Heat transfer statics nonlinear solution generator

14.3-5

610	COLOR LACTOR	(NXM) - DENOTES MATRIX SIZE A.B DENOTES LOOP	MAJOR OPTIONS
Э. Р	LUUP FALIURS		
		EMG. SDCOMP(LxL), FBS (LXNL)	GMIT - K
		FMC SUCOND (FXT), FBS (FXK), FBS(FXNF)	GMIT - K,M
			G MIT - K,M
		ETTA, FRANCIS (1777)	BMIT - not recommended
	Load Factors	EMG, SOCOMP(LXL), FBS (LXI), USHE	
		EMG, SDC0MPP (LxL), FBS (Lx1), M0DES	GMIT - not recommended
		FMC. SDCOMP (LxL), FBS (Lx1), PLA3, PLA4	BMIT - not recommended
_	road steps	CAMPIEX MADES (DXD)	GMIT - K,M,B,K4
		CONTRACT (DAME) (DAME)	G MIT - K,M,B,K4
	Frequencies	UE LOW (UXU-UDIT LEALS ' US)	0NT - X M 8 K4
-	Time Step Sizes. Time Steps	DECOMP (DxD), MPYAD (Dx1), FBS (Dx1)	
		MODES, COMPLEX MODES (HXH)	0HIT - K,M
		MØDES, MPYAD (H×D×NE), DDR1	GMIT - K,M DMI - Same as RFB (Hsize
		MØDES, MPYAD (H×D×NT), DDR1	GMIT - K,M DAI - Same as RF9 (Hsize

Table 2. Critical Operations Versus Rigid Format.

- `

i≝ --- ì

÷,

NOTE: Underlined groups involve looping.

1

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

14.3-6

MAJOR OPTIONS	GHIT - K,KD,M	L) ØMIT - K	ØMIT - K,M					G MIT = K,B,R	
CRITICAL CALCULATIONS (NXM) - DENOTES MATRIX SIZE A,B - DENOTES LOOP	EMG,SDCØMP (LXL), DSMG, MØDES	EMG, MPYAD (2LxLxL), MPYAD (2LxLx2L), DECØMP(2Lx2	EMG, MPYAD (2LxLxL), MPYAD (2LxLx2L), MØDES	EMG, APD, EIGEN, AMG, <u>FAI, CØMPLEX, EIGEN</u>	EMG, APD, EIGEN, AMG, FAI, COMPLEX, EIGEN	SAME AS RF1	EMG, RMG, DECØMP(1xL), SSGHT	RF9 + RMG	
LOOP FACTORS		Cyclic Indices		AERO Factors					
RIGID Format no.	13	14	15	10(AER#)	11(AERØ)	1 (HEAT)	3(HEAT)	9(HEAT)	

Table 2. Critical Operations Versus Rigid Format (Cont'd).

Ì

REQUIREMENTS FOR THE RIGID FORMATS

Constraint and the second s

.,

معصد معربها بالاستريدا

A Life of Life

Table 3. Critical Substructure Operations.

.

ì

PHASE 1	Same as selected Rigid Format without solution step	'RSAVE' avoids later DEC OMP
PHASE 2 COMBINE C = 2H ^T A ₄ H ₄	MPY3 (C × A _i × A _i × C), i = 1,2,,N components for <u>each</u> matrix type)	'AUTØ' combine requires geometry search
REDUCE A + B (Ø * A - B)	K: DECØMP (Ø × Ø), FBS (Ø × B), MPYAD (B × Ø × B) M, B, or K4: MPYAD (A × A × B), MPYAD (B × A × B)	User selects matrices to be processed. B points
NREDUCE	Same as REDUCE plus mode extraction	8 points include boundary
Sørve	(See corresponding Rigid Formæt, g = substructure size)	wor yrus user-serected modal displacement dof
RECØYER	For Combined S.S.: MPYAD (A _o x C ₁ x NL) For Reduced S.S.: MPYAD (A x B x NL) DECOMP (Ø x Ø), FBS (Ø x NL)	
PHASE 3	Same as Phase] except restart is possible to avoid matrix reformulation costs.	

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

٠

÷ •

ana ya **M**ana ana a

14.3-8

ar 1

Ì.

ESTIMATING CPU TIME REQUIREMENTS

14.4 ESTIMATING CPU TIME REQUIREMENTS

In the following sections, numerical methods are given for estimating the central processing unit (CPU) time for the critical NASTRAN calculations. It must be noted that the given equations are based on average problem characteristics and have been determined from simplified code algorithms. Although the coefficients to be used in the equations have been determined empirically from actual hardware tests, their range of accuracy is limited by the range of the tests and the approximate nature of the equations.

The basic matrix timing coefficients for various computer hardware are given in Table 1. The three basic operations listed are:

Mg - Tight Loop Multiply-Add

These values define the time for real arithmetic performed primarily in computer index registers.

M - Loose Loop Multiply-Add

These values define the time for real arithmetic performed by FØRTRAN code with core data transfers.

P - Pack/Unpack for Matrix I/Ø

These values define the average CPU time to transfer one matrix term with row and column indices to or from a location in the file buffer.

Timing data given in Table 1 correspond to the default computer arithmetic precision selected by NASTRAN. All CDC system times are for single-precision operations. The timing data for the other machines correspond to double precision, the normal default. If a different precision is used in a key arithmetic operation, the coefficients should be adjusted accordingly.

The standard nomenclature used in NASTRAN timing calculations is given in Table 2. These terms are used in the basic timing procedures and equations which follow.

14.4.1 Matrix Multiplication and Addition - MPYAD

Four separate methods are used by NASTRAN for multiplication of matrix data blocks. Three methods are used by the MPYAD module for computing the product of two matrices and adding a

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

third (A·B+C). Additionally, the MPY3 module is provided for the specialized "triple multiply" operation (H^TAH), frequently encountered in substructure analysis.

In the MPYAD module, the estimated running time for each of the three alternatives is calculated and the fastest method is selected by the program. The choice taken will depend on the relative sizes and densities of the input matrices and the amount of computer core available. Figure 1 provides approximate equations for the three MPYAD methods. Note that it is the time to multiply which predominates. The time to add is negligeable by comparison. Large-order matrices with ranges in density of from 10 percent to 50 percent are assumed. For the more detailed equations, consult Section 3.5.12 of the NASTRAN Programmer's Manual.

14.4.2 Symmetric Triangular Decomposition (SDCØMP)

A sparse matrix Gaussian elimination method is used in NASTRAN for triangular decomposition of symmetric matrices. This method does not require to so-called "banded matrices" to realize efficient processing. In effect, the method uses a variable length band, processing only the "active columns," in which non-zero terms occur. Full details on the computational algorithms are given in Section 4.7 of the Programmer's Manual. Also note that the rarely used "passive columns" effects are deleted in the method shown below.

Figure 2 illustrates the method used for calculating the "active column" size for each row of a matrix. The characters "X" signify non-zero, off-diagonal partitions of the input matrix. Typically, each X may represent six degrees of freedom or one grid point in the model. During the decomposition, the non-zero terms create additional terms in the rows below. The resulting region defined by the dashed lines will be completely populated in the resultant triangular factor. The number of active columns at each row is measured by counting the columns inside the dashed regions for that row. The algorithm described below may be used to calculate the CPU time for the in-core process. This is followed by a method for estimating the out-of-core execution time which involves "spill logic."

Symmetric Decomposition in Core

The steps used for calculating execution time are:

1. Calculate or estimate the active column factor Ci for each row i.

ESTIMATING CPU TIME REQUIREMENTS

and a second
2. Calculate the maximum value CMAX, the average value C_{avg} , and the RMS value CBAR(\overline{C}), where:

$$\bar{c}^2 = \frac{1}{N} \sum_{i} c_i^2$$
(1)

$$C_{avg} = \frac{1}{N} \Sigma C_{i}$$
 (2)

and N is the order of the matrix.

3. Determine if matrix spill logic is required. No spill occurs when the following condition is met:

$$W > \frac{1}{2} C_{max} (C_{max} + 3)$$
 (3)

4. If no spill occurs, the timing equation is:

$$T = \frac{1}{2} M_{\chi} \tilde{C}^2 N + P C_{avg} N$$
 (4)

Note that the second term may usually be ignored.

Note that the example in Figure 2 shows a poorly banded matrix having a significant number of widely spaced coupling terms. However, examining the number of active columns listed down the right side of the figure, we observe that the actual value of \overline{C} is only about 7D, where D is the number of active columns per partition X. By using the active column technique of decomposition, the running time would be comparable to a banded matrix with B = 7D.

14.4.3 Symmetric Decomposition with Spill

When the matrix decomposition process requires spill logic to compensate for limited core storage, scratch files are used for temporary out-of-core storage of results. The non-zero matrix terms shown in Figure 3 are subdivided into core-sized partitions, or "spill groups," which are spilled to and read back into core from these scratch files as necessary to finish the processing. Note that once a group is spilled, each row of that group is then treated separatel and, as such, is read back in only if it is required for calculations involved in processing a subsequent group when it occupies core.

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

The actual spill logic in the NASTRAN symmetric decomposition routines is very complex, and no single equation is adequate for estimating the resource requirements. A recommended procedure for large-order problems is to submit the job with a deliverately low estimate on the NASTRAN "TIME" card of the Executive Control Deck. The SDCØMP routine will perform the time and core requirement calculations, print the pertinent decomposition parameters, and exit due to the low user-specified time limit. However, this recommendation requires full development of the solution matrices, which may be expensive. Therefore, an approximate method is given below which will aid in preliminary estimation.

1. Horizontally subdivide the non-zero partitions of the matrix into spill groups which would fit in available core. Figure 3 illustrates the method using a banded matrix for simplicity. The number of rows for each spill group, S_i , is chosen such that:

$$\frac{S_{i}}{2} (2C_{i} - S_{i}) + 2C_{i} < W$$
 (5)

where W is the total number of terms which may fit into core. The shaded area in Figure 3 corresponds to the available core space. Incidentally, note that S_i is less than C_i if spill logic is required.

- 2. For each <u>now</u> in a spill group, determine the number of groups below that spill group which are affected by that row. This determines the number of passes, N_{pi}, required for reading that row in from storage. Note that each row in a spill group may have to be treated separately. In the example, the first set of short rows in Group 1 will be processed with two passes, once when processing Group 2 and once when processing Group 3 The long rows forming the second part of Group 1 will require three passes since it affects three groups below, Groups 2, 3, and 4.
- 3. The additional CPU time required for the spill processing is approximately the following value:

$$T_{s} = \sum_{i}^{N} 2^{p} C_{i} N_{pi}$$
 (6)

where P is the average coefficient of the pack and unpack time, C_i is the number of active columns for each row, and N_{pi} is the number of spill passes required to be performed on row i. Add T_s to the estimated time manually required for decomposition without spill logic.

ESTIMATING CPU TIME REQUIREMENTS

الهابية العائمية بالرمع فالالا

14.4.4 Forward-Backward Substitution - FBS, GFBS

Forward-Backward Substitution is the second step of the general solution of simultaneous equations. It follows the decomposition of a matrix described above. The FBS module performs these operations for decomposed symmetric matrices, whereas the GFBS (General FBS) module is used when the decomposed matrix is nonsymmetric. Each is discussed separately below. The basic matrix equation of both the backward and the forward substitution steps is:

$$[U_{nn}][X_{np}] = [Y_{np}]$$
(7)

Support of the local distance of the local d

where [U] is the upper triangular matrix resulting from decomposition, [Y] is given, and [X] is to be determined as the matrix of P solution vectors, each n rows long.

For the <u>symmetric code</u>, the parameter C_{avg} for the average number of active columns is obtained from the same estimate used for symmetric decomposition. The FBS time is:

$$T_{FBS} = 2npM_{g}C_{avg} + P(2nC_{avg} + \frac{np}{2})K_{f}$$
(8)

where $K_f \sim np/W$. K_f is the integer number of passes, where one "pass" is required for each core-held partition of the matrix Y.

For the <u>unsymmetric code</u>, the bandwidth B, the column size C of the upper triangular factors may be different from those of the lower factor (\tilde{B} and \tilde{C}). Combining the two passes (the forward and backward substitution steps), we obtain the following equation:

$$T_{FBS} = npM_{g}(B + \bar{B} + C + \bar{C}) + nP(B + \bar{B} + \bar{C} + C + \frac{P}{2})K_{f}$$
 (9)

where $K_f \sim np/W$. K_f is the integer number of passes where one pass is required for each core-held partition of matrix Y.

14.4.5 Mode Extraction (Module READ)

Four alternate methods of extracting normal modes are available in NASTRAN. Although the inputs and outputs from the calculation are nearly identical, the internal operations and the time and core requirements differ considerably. The methods of estimating the requirements for each of the four methods are described separately below:

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

and a second
- 1) <u>Givens Householder</u> (GIV) with QR Transforms This process requires a Cholesky decomposition of the mass (or differential stiffness) matrix. In most cases, the ØMIT (or ASET) procedure must be used to remove massless components and may be used to reduce the matrix to core-held size, in which case the execution times for computing the manual modes of the resulting matrices are relatively short. However, note that the execution times to obtain these final reduced size matrices may be quite significant. When the size of the input dimension, N, becomes too large for the final matrices to fit in core (i.e., (N(N+1)/2 > W)), the execution time requirements increase significantly due to spill. The following calculation steps are performed for Givens eigenvalue extraction:
 - a) Use SDCØMP on the mass matrix (order = N), perform an FBS to invert the triangular factor, and use MPYAD to pre- and post-multiply the stiffness matrix to obtain the dynamic matrix. Estimating equations for these steps have been given earlier (refer to Theoretical Manual, Section 9.2).
 - b) Tridiagonalize the dynamic matrix with a CPU time for execution of approximately

$$T_{TD} = 4/3 \text{ M} \cdot \text{N}^3 + 1/2 \text{ PNWK}, \qquad (10)$$

where

$$K = \{ (1/2 \ \frac{N(N+1)}{M} - 1) \}, \ (K \ge 0)$$
 (11)

and the term in brackets is rounded up.

c) Inverse transform the resulting N_v eigenvectors to physical coordinates. This requires a CPU time (T_v) of approximately

$$T_{\rm U} = 3MN^3 N_{\rm U} \tag{12}$$

2) <u>Inverse Power with Shifts (INV)</u> - In the inverse power method, the normal modes (eigenvectors) are obtained indirectly through an iteration process. The matrices are decomposed into triangular factors at "shift points," each of which may be used to extract several eigenvectors. Each eigenvector requires an approximate average of six FBS steps and six matrix multiplies with a single column right-hand matrix. For estimating the running time:

ESTIMATING CPU TIME REQUIREMENTS

enter area a

- a) Calculate the SDCØMP time, T_D , for decomposing the sum of the stiffness and mass matrix. Calculate the FBS and MPYAD time, T_{FM} , for a single column. See the other parts of this section for the appropriate estimating algorithms.
- b) The total time estimate, T, would then be approximated by the equation:

$$T + T_{D}(N_{V}/3) + 6T_{FM} \cdot N_{V}$$
 (13)

where N_v is the expected number of eigenvectors to be extracted. Note that these estimates are obtained from experience with well-behaved systems. When closely grouped frequencies occur or a poor initial estimate of a region is chosen, more iterations may be required and/or extra shirt points may have to be considered. All of these will require an increase in the above estimate.

- FEER Method No accurate methods are currently available for estimating the running time for this method. However, tests indicate it is faster than the inverse power method.
- 4) <u>Determinant Method (DET)</u> This method iterates to an eigenvalue by fitting curves through the matrix determinant at trial eigenvalues. Because of its inherent inefficiencies, the DET option is <u>not recommended</u> for general use. The only reason for its existence is for checking results from the other methods.

Computers.
Typical
for
Parameters
Timing
-
able

かい いいやつや 発

HATTATE CONTRACTOR AND A MARCEL & LAND

MAKE			BM 360/				Ξ	BM 370,			ğ	ω.		CYBER		NINU	/AC
MODEL	20	65,7	75	85	91,5	155	158	165	168	195	6400	6600	173	175	176	1108	1110
CONFIG.	m	4	S	ور	2	=	12	10	13	6	10	4	16	18	19	m	20
	Ę	x	1	°	, o	25	ه ۲	8	1.7	0.3	15	L.	8	2.7	1.2	14	
ີ ສ	550	3 8	2 8	u u	2.5	09	14	2.5	2.2	2.0	38	01	12	4.3	1.9	35	
م	20	160	100	50	36	145	60	30	20	36	110	53	65	27	7	60	•

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

14.4-8

ESTIMATING CPU TIME REQUIREMENTS

a ar en an marger parte a part

۰.

Ļ

and the same the desire

Table 2. Nomenclature for Timing Equations.

T	-	CPU time
N	-	Order of square matrix
Ē	-	RMS average of number of active columns
Cavo	-	Simple average of number of active columns
W	-	Effective core space (number of open core cards/number of words per matrix terms)
Me	-	Tight loop multiply time coefficient
M	-	Loose loop multiply time coefficient
Ρ	-	Pack/Unpack coefficient for matrix I/#
8,Ē	-	Bandwidths for unsymmetric decomposition
n,m,p	-	Natrix dimensions for MPYAD
ĸ	-	Number of passes for spill processing
ρ _Α ,ρ _Β	-	Density of matrices A and B

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

.....

120 1 - AST. - D. S. B. B. S. B.

$$\frac{MATRIX PROBLEM}{(n \times n)} \times [B(n \times p)] = [C]$$
or
$$[A(n \times m)]^{T} \times [B(n \times p)] = [C]$$

METHOD 1 (Holds unpacked parts of band C in core)

$$T = M_{\ell} \rho_A mnp + P \rho_A mnk_{\uparrow}$$

where

Ë١

$$K_1 \geq \frac{(m+n)p}{W}$$

 ${\rm K}_{\mbox{\scriptsize i}}$ is the integer number of passes in reading the A matrix.

METHOD 2 (Holds packed parts of A in core)

$$T = (M + M_{\ell}) \rho_{A} mnp + P(np + mp) K_{2}$$

where

$$K_2 \ge \frac{2\rho_A mn}{W}$$

 K_2 is the integer number of passes in reading the B and C matrices.

METHOD 3 (Sparse B Matrix, Transpose Case Only)

$$T = M_{\ell} \rho_{B} mnp + P \rho_{B} (np + mp) K_{2}$$

Figure 1. Approximate Timing Equations for MPYAD.

14.4-10

1.1

1.00

Sugar an

ţ

a an a con contra



Figure 2. Region of Matrix Affected by Symmetric Decomposition.
a thur have



Figure 3. Subdividing Symmetric Upper Triangular Matrix Partition into "Spill Groups."

ESTIMATING OTHER HARDWARE REQUIREMENTS

14.5 ESTIMATING OTHER HARDWARE REQUIREMENTS

In addition to the CPU time requirements previously derived, many other cost factors are involved in estimating the overall expense of a NASTRAN analysis. These factors range from the hardware costs of printing and plotting the results to the very significant costs of the engineering effort necessary to model the structure. Estimating the labor costs of performing an analysis is a major subject far beyond the scope of this guide. This section, therefore, deals only with other hardware factors which include I/Ø charges, tape mounts, and permanent file requirements. Because many of these items are highly dependent on the operating systems or installation facilities, additional information may be required by the user, which must be obtained from his local computing facility.

14.5.1 Input/Output Data File Factors

In NASTRAN, most of the intermediate data storage is placed on scratch, bulk storage files which may be repeatedly accessed during the job execution. On most modern hardware, the actual cost for this process is billed in addition to the CPU costs, and defined as the local "I/O" cost.

Several different methods are used by the computer operating systems to charge the user for this I/p activity. Some of these are:

- 1) Charges for the total number of words or bits of data transferred through the I/\emptyset hardware.
- 2) Charges for the total file space requested and/or used by the program.
- 3) Charges for the number of calls to the peripheral processor (PP) routines which control the I/# operations.
- 4) Charge for the total actual clock time, which, on some systems may be the primary cost factor.

Many operating systems use a combination of the above factors in calculating system resource units or equivalent cost units for I/\emptyset activities. In NASTRAN, all of these activities tend to increase directly with model size, and their costs tend to be proportional to CPU costs. However, unlike the CPU charges, the user may modify the I/\emptyset costs by changing NASTRAN parameters or procedures. Some of his options are:

14.5-1

ESTIMATING RESOURCE REQUIREMENTS FOR NASTRAN

- 1) If the system charges are primarily for I/Ø routine calls as in (3) above, the buffer size parameter, BUFSIZE, defined on the NASTRAN card, may be increased to increase the amount of data transferred in each I/Ø call, which reduces the number of calls required See Section 2.1 of the User's Manual for details. However, note that the increased buffer space will correspondingly decrease the core space available for computation which, in turn, could affect the CPU cost.
- 2) If the system charges are primarily for the amount of data storage, as in (1) above, and the problem size is small, the BUFSIZE parameter may be decreased. This will reduce the space taken up by nearly empty blocks of data on the file and will give the program more useable core space.
- 3) On IBM hardware, the size and number of working files is controlled by the Job Control Deck. Although the standard NASTRAN setup is designed for efficiency in a normal analysis, the number and sizes of these files can be expanded or contracted for unusual cases. On other hardware, the number and size of the files are controlled by the program.
- 4) With very large structural matrices, the checkpoint file may become very large. In many cases, much of the data stored on this file will <u>not</u> be used in scheduled restarts. A suggested procedure is to run a small pilot model through the entire procedure and note which checkpoint files are necessary for the restart. The DMAP ALTER feature may be used to eliminate the unnecessary checkpoint instructions to reduce the size of the checkpoint file.

It should be noted that all of the above procedures are recommended only for the experienced NASTRAN user. Some experimentation with pilot runs may be necessary to check out and validate the particular combination of operating system changes and NASTRAN analysis modifications selected by the user.

14.5.2 Tape and Permanent File Requirements

The NASTRAN user may employ several options to archive data on bulk storage devices for subsequent jobs. These include using the checkpoint procedures or the *BUTPUTi* modules which write user tapes. Since these user tapes generally contain only single matrices or block data, their size requirements are not crucial. However, some care may be necessary in the use of the

14.5-2

ESTIMATING OTHER HARDWARE REQUIREMENTS

checkpoint procedures which are initiated with the Executive Control Deck CHKPØINT command. The **@UTPUTi modules are called into service with DMAP alters input also via the Executive Control Deck.**

The NASTRAN checkpoint/restart procedures may employ system permanent files, private disk packs, or physical tapes for the NPTP (New Problem Tape). The NPTP file is a single file which could easily exceed the space available on a single tape or disk pack. The user is cautioned to check with his computer center operations to obtain procedures for specifying multi-unit files for checkpoint runs on any large-order structural problems.

14.5.3 Substructure Operating File (SØF) Requirements

<-4

In the automated multi-stage substructure system (AMSS), a data base file system is used to store all data required for solving problems with substructuring. The Substructure Operating file (SØF) is a single random access data base which may reside on from one to ten system files. New files may be added to store the SØF as the data base grows, but the maximum size of each file must be specified on the run in which the file is first used. File size resources for each new substructure that is to be added to the data base are estimated based on the expected size of the corresponding NASTRAN data block to be stored. Each substructure will require storage of basic structural data including the stiffness matrix, with options for the mass, load, damping, and solution matrices. In addition, transformation matrix (HØRG) is required for each substructure which contributes to a higher level combination structure or to a reduced structure. The system automatically reports the space available on the SØF.

Several options are given in the system to maintain and control SØF file space. At each stage of the process, the space used currently and the unused space available are printed automatically. Additional files may be attached at any stage to provide more space for storage of the SØF. Finally, unused substructure data or data no longer required for subsequent operations may be deleted or edited from the files by the user in order to make that space available for future operations.

14.5-3

15. DIRECT MATRIX ABSTRACTION PROGRAM (DMAP)

15.1 DWAP OVERVIEW

DNAP is the "programming language" of NASTRAN which controls the execution logic and sequence of the NASTRAN program operations. Each DMAP instruction causes a block (module) of NASTRAN code to be executed, and each Rigid Format is simply a sequence of DMAP instructions permanently stored in NASTRAN. The user may code and execute a DMAP sequence in place of a Rigid Format. He may also make changes to the Rigid Formats through the use of ALTER and DMAP data. In either case, these instructions are always included in the Executive Control Deck (UG 6).

The format and detailed usage of the user-oriented DMAP instructions are described in Section 5 of the User's Manual. NASTRAN modules are described in Section 4 of the Programmer's Manual. The types of tasks which can be performed by NASTRAN using DMAP instructions fall into the following categories.

- 1. Executive, e.g., jump, checkpoint, equivalence files, exit (UM 5.3.4).
- 2. Utility, e.g., print or copy files, manipulate parameters (UM 5.3.2).
- 3. Matrix Arithmetic, e.g., add, multiply, transpose matrices (UM 5.3.1).
- 4. Structural, e.g., assemble tables, calculate element data (PM 4).

The first three categories are intended for general user applications. However, the structura oriented modules require specialized input tables and are intended for use in the Rigid Formats. In addition, several dummy modules have been implemented in NASTRAN for which the user may implement his own code (PM 6.7 and 6.12).

The next section presents a guide to the rules governing the use of DMAP instructions in NASTRAN. This is followed by a description with illustrative examples using each of the four categories of operations defined above.

15.1-1

FUNDAMENTAL DMAP RULES

15.2 FUNDAMENTAL DMAP RULES

Ēγ

The DMAP code may be viewed as a program in which the variables are data blocks residing^{*} on I/# devices and simple parameters which remain in core. The DMAP module instructions are equivalent to subroutine calls in which the user selects the input and output variables and the subroutine performs a predefined operation. The general format for a module instruction is:

MØDULE IN1, IN2, ... /ØUT1, ØUT2, ... /param 1/param 2/\$

where MDDULE is the name of the operation, IN1, IN2, etc., are user-selected names of input data blocks, ØUT1, ØUT2, etc., are names of output data blocks, and param 1, param 2, are fields used for defining the parameters. As in a FØRTRAN subroutine, the number of variables and their order in a module instruction are important.

The input and output data block names are chosen arbitrarily; their position in the instruction provides the instrucțions to the module. Guidelines for data block specification are:

- Blank fields or two successive commas indicate a "purged" data block. The module specifications indicate where purged data blocks are legal.
- Data blocks must reside on distinct files, i.e., the names are distinct and not equivalenced.
- 3. Input data blocks should be logically available as output from previous modules, as outputs from the preface, or as checkpointed data blocks on a restart run.

The parameter fields in the DMAP instructions may take several forms. Examples are listed below:

/(blank)/ or dropped from end - The module's default value is used.

/VAR/ or /V, N,VAR/ - VAR is a variable input parameter name.

/23.2/ or /C.N.23.2/ - An input constant, 23.2 is specified.

/VAR=16 or /V,N,VAR=16/ - The variable VAR is given an input value if not previously set. /V,S,LUSET/ - The output parameter LUSET is saved.

The types of constants used in the parameter fields are specified in the module descriptions and must be consistent. The types of constants used are integers, real numbers, complex numbers, and BCD flags.

EXECUTIVE OPERATIONS

And the second
15.3 EXECUTIVE OPERATIONS

The formats for the Executive instructions are less restrictive than the functional modules. The input and output data block fields may be missing and parameters are specified with only the variable name. Executive Operations instructions or statements are divided into general categories as follows:

- Declarative instructions FILE, BEGIN, and LABEL which aid the DMAP compiler and the file allocator.
- Instructions CHKPNT, EQUIV, PURGE and SAVE which aid The NASTRAN executive system in allocating files, interfacing between functional modules and in restarting a problem.
- 3. Control instructions REPT, JUMP, CØND, EXIT, and END which control the order in which DMAP instructions are executed.

The rules associated with the Executive Operations instructions are distinct for each instruction and are discussed individually in Section 5.7 of the User's Manual.

Some examples of executive module formats are:

BEGIN - Define start of DMAP
LABEL LBL1 - Provide a label for jumps, etc.
CGND LBL2, PARAM - Conditional jump to label LBL2
EQUIV DB1, DB2/PARAM - Equivalence two data blocks
CHKPNT DB1, DB2, DB3 - Write data blocks on checkpoint file
EXIT - Stop the program

The example that follows is most useful for those situations which call for controlling the partial execution of a Rigid Format. The subsequent sections present additional examples of execution operations.

DIRECT MATRIX ABSTRACTION PROGRAM (DHAP)

Example 1 - Partial Execution of Rigid Format 1

Alter Rigid Format 1 (Static Analysis) to stop execution after plotting the undeformed structure.

Card No.		
1	ID	EXAMPLE1, MODES
2	APP	DISP
3	SØL	1,0
4	TIME	2
5	ALTER	17)
6	EXIT	s }
7	ENDAL TER	۲
8	CEND	
9	TITLE =	DMAP EXAMPLE 1
10	PLØTID =	MØDE1 DESIGN CHECK
11	ØUTPUT (PLØT)
12	SET 1 AL	L
13	FIND SET	1
14	PLØT	

and a state of the second state

Ξ.

..... **`**

. ---

> 15 BEGIN BULK

> > BULK DATA FIELD

		2	3	4	5	6	7	8	9	10
16 17 18 19 20 21 22	CØNRØD CØNRØD GRID GRID GRID MATI ENDDATA	1 2 1 2 3 1	1 2 3.+7	2 3 0.0 1.0 2.0	1 1 0.0 1.0 0.0	1.0 1.0 0.0 0.0 0.0		123456 3456 123456		

NOTES ON CARDS:

2. Use displacement approach.

3. Use Rigid Format 1.

5. Insert the immediately following DMAP instruction(s) after Line 17 of the Rigid Format identified by the SØL card (UM 2.3.1).

NASTRAN will execute the Rigid Format up to the point where it encounters "EXIT" where it will terminate.
 End of Rigid Format alters.

15.3-2

UTILITY OPERATIONS

15.4 UTILITY OPERATIONS

The Utility Operation instruction (User's Manual 5.3.2) can be divided into general categories as follows:

- 1. Read and write of user tapes (e.g., INPUT, INPUTTi, **g**UTPUTi)
- 2. Matrix and table printout (e.g., MATGPR, MATPRN, MATPRT, TABPRT, TABPT)
- 3. Manipulate parameters (e.g., PARAM, PRTPARM, SETVAL)
- 4. Display matrix topology (e.g., SEEMAT)
- 5. Generate partitioning vectors (e.g., VEC)

These utilities are particularly helpful for debugging a structural model as well as for checking out new routines developed by a user. Also, when external programs are to be used for data preparation for input to NASTRAN, or for post processing the results obtained from NASTRAN, these utilities facilitate the transmission of the data (User's Guide, Section 9). Examples of these utilities are presented below and in subsequent sections.

Example 2 - Printout of Table Data

Alter Rigid Format 1 (Static Analysis) to print the grid point coordinates in the basic coordinate system.

Card No. 1 ID EXAMPLE2, BGPDT 2 APP DISP 3 SOL 1,0 4 TIME 3

- 5 ALTER 5
- 6 TABPT BGPDT,, ,//\$
- 7 ENDALTER
- 8 CEND
- 9 TITLE = DMAP EXAMPLE 2
- 10 BEGIN BULK

BULK DATA FIELD

	<u> </u>	2	3	4	5	6	7	8	Q	10
11	CONROD	1	1	2	h	1.	[[10
12	CONROD	2	2	3	h	1.				
13	GRID	1		.0	.0	.0		123456	1	
14	GRID	2		h.	1.	o.		3456	ľ	
15	GRID	3		2.	0.	ο.		123456		:
16	HATI	1	3.E+7							l
17	ENDDATA			1						

- NOTES ON CARDS: 2. Use displacement approach.

 - Use displacement approach.
 Use Rigid Format 1.
 Insert the immediately following DMAP instruction(s) after Line 5 of the Rigid Format identified by the SØL card.
 Print the Basic Grid Point Data Table (PM 2.3.3.5).

 - 7. End of Rigid Format alters.

UTILITY OPERATIONS

Example 3 - Write User Tape and Printout Intermediate Matrix

The following illustrates the writing of a user tape via FØRTRAN for use in subsequent processing outside NASTRAN. The matrix data represents the mode shapes for the reduced set (u_a) of dynamic degrees of freedom (Theoretical Manual, Section 9.3.1) computed with Rigid Format 3 (User's Manual 3.4.1).

Card No	
۱	ID EXAMPLE3,TEST2
2	APP DISP
3	TIME 10
4	SØL 3,1
5	ALTER 91
6	B UTPUT2 PHIA,,,,//-1/11/C,N,TEST2 \$
7	OUTPUT2, ,,,,//-9/11 \$
8	MATGPR GPL,USET,SIL,PHIA//C,N,FE/C,N,A \$
9	ENDALTER
10	CEND
11	TITLE = BØX3 FRAME
12	SPC = 12
13	METHØD = 3
14	DISP = ALL
15	PLØTID = MØDE SHAPES WITHØUT EQUIPMENT
16	ØUTPUT(PLØT)
17	SET) = ALL
18	VIEW 34., 30., 0.
19	MAXIMUM DEFØRMATIØN 12.0
20	FIND SCALE
21	FLØT MØDAL DEFØRMATIONS 0, SET 1 SYMBØLS 2, SHAPE
22	BEGIN BUIK
23	ENDRATA

DIRECT MATRIX ABSTRACTION PROGRAM (DMAP)

and a set of the State of the Constitutions at the set of the set

- NOTES ON CARDS: 4. Requests execution of Normal Modes Analysis.
 - 5. Insert following alter package following DMAP instruction 91, **Rigid Format 3.**
 - 6. Requests output via FØRTRAN of the mode shapes for displacement set u_a
 - to user tape on FØRTRAN Unit 11 after rewinding and writing the NASTRAN label TEST2. (See UG 10 for FØRTRAN unit assignments.) 7. Writes final end of file (EØF) on the tape. Note that the first comma is required as a delimiter when the first field is blank.
 - 8. Requests printout of the same mode shape matrix for all modes generated (Number of modes is FE, Degrees of Freedom Set is A).
 - 9. Concludes the alter package.

 - 10. Concludes the Executive Control Deck. 11-21. Case Control deck which specifies the constraint set, selects from Bulk Data the method of eigenvalues extraction, requests printing of the mode shapes for all grid points and directs the structural plotter package to plot all mode shapes underlayed by the undeformed plot of the model using all elements.
 - 22-23. Contain the Bulk Data Deck for the model which would include either ØMIT or ASET data to reduce problem size via Guyan reduction to the desired displacement set u_a.

15.5 MATRIX OPERATIONS

The Matrix Operation instructions (User's Manual 5.3.1) include the basic functions of adding, multiplying, partitioning, merging, transposing and solving matrix equations with single and double precision, real or complex data. These DMAP instructions may be used to process any matrix data generated by NASTRAN or input to NASTRAN from outside sources via the user tape facilities.

manunanter an interior an at the state of th

Examples 4 - Processing with User Supplied Matrices

The following example illustrates a complete DMAP sequence for reading two matrices from user tape, adding them together and printing the results.

Card No.

1	ID	EXAMPLE4,ADD
2	APP	DMAP
3	TIME	5
4	BEGIN	\$
5	INPUTT2	/MTXA,MTXB,,,/-1/15/*B9U* \$
6	ADD	MTXA,MTXB/MTXSUM/ \$
7	MATPRN	MTXA,MTXB,MTXSUM,,// \$
8	END	\$
9	CEND	
10	BEGIN BU	LK
11	ENDDATA	

NOTES ON CARDS: 2. Defines DMAP approach.

- Initiates input of user specified DMAP sequence.
 Reads in two matrices (MTXA and MTXB) from user tape on FØRTRAN Unit 15 which were written via FØRTRAN by a program external to NASTRAN. (Note, this tape could also be written with NASTRAN using ØUTPUT2 (UG 7.3.2.2) NASTRAN tape label B9U is used for positive identification.
- Add the two matrices to form MTXSHM.
 Printout all three matrices
- Printout all three matrices.
- 8. Terminates DMAP sequence.
- 9. Terminates Executive Control Deck.
- 10-11. Case Control Deck is empty except for initiation of the Bulk Data Deck which should contain at least one Bulk Data Card (e.g., DMI card with null data and the ENDDATA card required to terminate the NASTRAN input stream.

STRUCTURAL OPERATIONS

15.6 STRUCTURAL OPERATIONS

The Structural Operation instructions (Programmer's Manual, Section 4) include all the basic functions for solving structural problems. These include processing of basic input data, generation and assembly of element and user specified stiffness, mass, damping and load matrices, execution of static and dynamic analyses and data recovery. The modules called by these DMAP instructions may in turn call the utility matrix operation subroutines as required to process the data.

The best examples of use of the Structural Operation DMAP instructions are the Rigid Formets themselves. A simple application is illustrated below. For more complex applications, see the Substructuring sample input in this Guide, Section 20. Additional examples are given in the User's Manual, Section 5.4.

Example 5 - Replace Program Generated Data with User Data

This example illustrates an alter to Rigid Format 3 (Normal Mode Analysis) so that a mass matrix input on DMIG Bulk Data cards is used in place of the NASTRAN generated mass matrix.

DIR / MATRIX ABSTRACTION PROGRAM (DMAP)

Card No.		
1	ID	EXAMPLE5, DMIG
2	APP	DISP
3	TIME	2
4	SOL	3,0
5	ALTER	28,28
6	MTRXIN,	<pre>MATP99L,EQEXIN,SIL,/MFR0MDMI/,,/LUSET/N0MGG/0/0</pre>
· 7	ENDALTER	R
8	ALTER	30
9	EQUIV	MFRØMDMI,MGG/ \$
10	ENDALTE	R (
п	CEND	
12	TITLE =	DMAP ALTER, EXAMPLE5
13	METHØD	= 10
14	DISP =	ALL
15	BEGIN B	ULK

24

A STATE OF THE PARTY STATE OF STATE

BULK DATA FIELD

	1	2	3	4	5	6	7	8	9	10
16 17 18 19 20 21 22 23 24 25 26 27	CONROD CONROD DMIG DMIG EIGR +EIGIO GRID GRID GRID GRID MATI ENDDATA	1 2 MFRØMDMI MFRØMDMI 10 MAX 1 2 3 1	1 2 0 2 2 GIV 3.+7	2 3 6 1 2 0.0 1.0 2.0	1 1 1 1 0.0 1.0 0.0	1.0 1.0 2 2 0.0 0.0 0.0 0.0	1 2 2	1.5+3 1.5+3 123456 3456 123456		+E IG10

NOTES ON CARDS:

2. Use displacement approach.

- Select Rigid Format. 3.
- Replace line 28 of Rigid Format 3 (UM. 3.4.1) with the following DMAP 5. alter.
- 6. Use the structural operation MTRXIN (PM 4.57) to replace SMA2 (PM and create the matrix MFR@MDMI supplied by the user on DMIG Bulk Data cards. Sets flag NBMGG = 1 if that matrix is created, otherwise sets) flag to -1.
- Conclude current alter package.
 Initiate another alter package to be inserted following line 30 of Rigid Format 3 (without replacing line 30).

ORIGINAL PAGE IS OF POOR QUALITY

STRUCTURAL OPERATIONS

and the state of the

NOTES ON CARDS (cont'd):

- 9. Equivalence user supplied matrix (MFR9MDMI) to the NASTRAN mass matrix (MGG). This user supplied matrix will now be used in all subsequent operations which reference MGG. 10. Concludes current DMAP alter package. 11. Concludes Executive Control Deck. 12-15. Case Control deck for extracting normal modes and specification of

 - output requests.
- 18-20. Provides data for user supplied mass matrix referenced above by the name provided here (MFR9MOMI).

SPECIAL NOTE: The data set MATP99L used as input for MTRXIN (Card No. 6) is generated during the NASTRAN preface operation (PM 2.3.2) and is not an output from a previously specified DMAP instruction.

.

16. NASTRAN EXTERNAL INTERFACES

16.1 INTRODUCTION

NASTRAN is a true general purpose computer program. As such, it may be used to perform a wide range of structural, structural with nonstructural interactions, and many other problems with physical phenomenon that can be defined in matrix equation format. Because of NASTRAN's versatility, requirements often exist for the transfer of data between NASTRAN and other computer programs or between different organizations, each using NASTRAN.

Data transfer to and from NASTRAN may be performed using the following:

- Preprocessors for performing automatic data generation, data checking and updating, data conversion to NASTRAN format, grid point resequencing, partition vector generation, radiation view factor generation, etc.
- NASTRAN output as input to postprocessors which perform data scanning, transient response data conversion to generate acceleration input, plot file editing, report generation, shock spectrum analyses, atc.
- 3. NASTRAN DMAP and Substructure Control Deck commands for transfer of substructuring data tables and matrices from NASTRAN to magnetic tape or disk and vice versa for back-up storage or as input to NASTRAN at other facilities, even if the computers are different.
- 4. Special NASTRAN Bulk Data Deck cards for direct matrix and table input to represent special purpose and unusual model requirements, load, or response phenomenon.
- 5. Special matrix and output data card punching capability.

The data transfer capabilities to and from NASTRAN listed above are described in this section.

16.2 PRE- AND POSTPROCESSORS

Computer programs which interface as pre- and postprocessors with NASTRAN are available from government sources. Such special purpose programs have been developed to perform automatic bulk data generation, grid point resequencing, radiation view factor generation, and other data generation or output scanning and reporting tasks.

COSMIC, NASA's Computer Software Management and Information Center, is a clearinghouse where government-sponsored and maintained software is transferred from government agencies to the industry. COSMIC is responsible for distributing NASTRAN and the many NASTRAN-related pre- and postprocessor programs that have been developed. To obtain a catalog of programs available through COSMIC, write or call:

> COSMIC Suite 112, Barrow Hall University of Georgia Athens, Georgia 30602 Director: Dr. Harold Hale Telephone: (404) 542-3265

and a second water a state of the second
16.3 NASTRAN MODULES FOR DATA TRANSFER

=

NASTRAN provides the capability for users to conveniently and easily transfer data from. NASTRAN to user tapes or disk files. Access to this capability is provided through the seven DMAP modules listed below:

Module	Data Format	Operation
INPUTTI	GINØ	Read
INPUTT2	FØRTRAN	Read
Ø UTPUT1	GINØ	Write
ØUTPUT2	FØRTRAN	Write
BUTPUT3	"DMI" Card	Funch
таврсн	"DTI" Card	Punch
EXIØ	User Selected	Read and Write

The user must code a DMAP sequence or use a Rigid Format alter to make use of these utilities, except for EXIØ. The Substructure Control Deck commands SØFØUT, SØFIN, RESTØRE, DUMP, and CHECK may be used to execute EXIØ. Each module and its use is explained in the sections that follow.

16.3.1 NASTRAN GIND Files

The most efficient way to transfer NASTRAN data blocks from one run to another on the same type of computer, is through the use of INPUTT1 (Programmer's Manual, Section 4.98) and <u>#UTPUT1</u> (Programmer's Manual, Section 4.100). These modules use the NASTRAN read/write GIN**9** (Seneral Input/Output) routines. Instructions and examples for using INPUTT1 and <u>#UTPUT1</u> are presented in the User's Manual, Section 5.5.

16.3.2 NASTRAN FORTRAN Files

The NASTRAN utility modules INPUTT2 (Programmer's Manual, Section 4.99) and **BUTPUT2** (Programmer's Manual, Section 4.101) provide a general capability for interfacing NASTRAN with

16.3-1

NASTRAN EXTERNAL INTERFACES

other computer programs and pre- and postprocessors. These two utility programs read (or write) data blocks (tables or matrices) from tape or disk files using unformatted FØRTRAN READ and WRITE statements. Instructions and examples for using INPUTT2 and ØUTPUT2 are presented in Section 5.5 of the User's Manual.

16.3.3 NASTRAN Punch Files

â

and the second of the second
User's may create punched card records of NASIRAN matrices and tables using DMAP modules @UTPUT3 (Programmer's Manual, Section 4.102) and TABPCH (Programmer's Manual, Section 4.103). These records are useful for long-term storage of data in a physical format. Punched card files on magnetic tape are also a convenient way to transfer NASIRAN data between different types of computers. Note that if the data transfer is to be via magnetic tape, such data transfer from one make of machine to another will normally require assistance from computer support personnel at both machine sites.

GUTPUT3 and TABPCH generate punched card data in NASTRAN Bulk Data Deck card formats: DMI and DTI, respectively. These data card formats are presented in Section 2.4 of the User's Manual. Details regarding restrictions on their use are presented under "Remarks" for each bulk data card description.

Instructions for using ØUTPUT3 and TABPCH, with examples, are presented in Section 5.5 of the User's Manual. Note that to use DMI or DTI data blocks, an appropriate DMAP sequence or alter package must be written.

16.3.4 NASTRAN SOF Files

NASTRAN provides special capability for transfer of substructuring data files between different types of computers. This capability is accessed using the SØFØUT(EXTERNAL) and SØFIN(EXTERNAL) automated substructuring commands. More efficient operations are available if the data is to be transferred between computers of the same type. Use of these external transfer commands in the Substructure Control Deck automatically results in a DMAP call to module EXIØ (Programmer's Manual, Section 4.130). The external form of SØFØUT and SØFIN cause unformatted data records on the Substructure Operating File to be converted to formatted records, similar to punch or print records. These recrods may be saved on magnetic tape and, as formatted records, they can easily be read into NASTRAN on a different type of computer. Detailed instructions for using the SØFØUT and SØFIN commands are presented in Section 2.7 of the User's Manual.

16.3-2

NASTRAN MODULES FOR DATA TRANSFER

and a trapport of the trap and the second

When transferring substructure data with these commands, the user must consider tape standards and character sets of the two machines to be involved. For example, when creating a tape on a CDC computer for input to an IBM machine, the CDC user must request his output tape be in the EBCDIC character format. This request must be done using installation-dependent job control cards. The BCD character set would be specified if the input were to be either to CDC or to UNIVAC hardware. Note that external SØF files utilize a standard block size and should not be reblocked outside of MASTRAN.

17. INTRODUCTION TO NASTRAN APPLICATIONS

17.1 INTROCUCTION

Ξ

In earlie: sections of this manual general principles for modeling structures, controlling solutions, and operating in the NASTRAN environment are presented. The following sections contain more specific rules and guidelines particularized for the application of NASTRAN to each type of analysis. These discussions are augmented by annotated examples illustrating typical options. As a guide to identifying sections of interest to the reader, the paragraphs that follow outline how to plan for special applications of NASTRAN, and they summarize each in order of presentation in later sections.

These special applications are also discussed in order of decreasing frequency of useage and generality. The rules and guidelines presented in the first two sections, statics and dynamics, apply to the remaining special topics. Methods of analysis are presented in the latter sections because of their specialized nature and requirements for use by the experienced analysis utilizing the more sophisticated tools of NASTRAN.

17.1-1

SUMMARY OF APPLICATIONS

an out the second second second second second second second

17.2 SUMMARY OF APPLICATIONS

A brief summary of the applications capabilities of NASTRAN (cnd contents of the User's Guide sections which follow) is presented next.

17.2.1 Static Analysis

Basic procedures for finite element structural analysis using NASTRAN are described in Section 18. Many of the topics are fundamental, so they also apply to the subsequent sections. The subjects include definition of elements, constraints, singularities, and non-linearities. Interpretation of the output is also described. Key data cards and case control options are included in the discussions. Six sample problems are presented along with card-by-card descriptions of the input data.

17.2.2 Dynamic Analysis

Section 19 on dynamics examines the special modeling and analysis control options required for solutions in the time or frequency domain. Topics include identification and control of different dynamics analysis options, input options for mass and damping, and special options inclding use of transfer functions and non-linearities. Seven example problems are presented to illustrate some of the many modeling and solution alternatives.

17.2.3 Substructuring and Modal Synthesis

The automated multi-stage substructuring system in NASTRAN(AMSS) provides an efficient approach to the analysis of large-order structural models. The AMSS capability allows a structure to be conceptually subdivided into smaller, more manageable substructures which are initially modeled and checked in separate NASTRAN runs. Substructures are then assembled to construct a system mode. Operations for reducing matrix sizes, combining substructures, invoking solution procedures, and recovering and editing data are controlled directly by the user with simple linguistic commands. Solutions are obtained using existing NASTRAN Rigid Formats. This capability is described in Section 20.

17.2.4 Aeroelastic Analysis

NASTRAN structural models may be coupled with subsonic and supersonic aerodynamic models for aeroelastic flutter and gust response analysis. Structural grid points and aeroelastic degrees

17.2-1

INTRODUCTION FO NASTRAN APPLICATIONS

of freedom are coupled by automated spline methods which may be used for axial bodies, fixed surfaces, and moveable surfaces. The options for aerodynamic coupling include strip theory, mach box method, piston theory, and the doublet lattice approach.

Section 21 of the User's Guide contains a description of NASTRAN aeroelastic analysis capability. A description of how to use the aeroelastic capability is presented in the User's Manual, Section 1.11.

17.2.5 Heat Transfer Analysis

an an in the second
NASTRAN heat transfer analysis capability may be used to determine temperatures and fluxes, or to determine temperature inputs for structural problems. Steady state and transient problems can be solved, including heat conduction (with variable conductivity for static analysis), film heat transfer, and non-linear (fourth power law) radiation. A useful feature of NASTRAN heat transfer is that the same model can be used for both the heat transfer and structural response analyses if the same finite elements are appropriate.

Instructions for using NASTRAN heat transfer capability are presented in Section 22.

17.2.6 Cyclic Symmetry Modeling

Cyclic symmetry formulations may be used for many structures such as pressure vessles, rotating mechanisms, and antennas for space communication. Each of these structures consists of several, virtually identical segments evenly spaced around an axis of symmetry. The advantage to the cyclic symmetry method is that only one of the identical segments need be modeled. A solution is obtained from a minimum set of matrix equations which are on the order of only one segment. Two basic approaches, rotational and dihedral symmetry are provided by this capability.

NASTRAN cyclic symmetry capabilities are fully described in Section 23.

17.2.7 Hydroelastic Analysis

The NASTRAN hydroelastic modeling option provides the capability for analysis of coupled axisymmetric fluid and nonsymmetric finite element systems. Small motion analyses yielding normal modes and dynamic responses may be performed. Fluid compressibility and gravity effects may be included. Basic input cards and a typical sample problem are described in Section 24.

17.2-2

SUMMARY OF APPLICATIONS

17.2.8 Acoustic Cavity Analysis

The acoustic cavity analysis is designed to solve for resonant modes in slotted solid rocket engine cavities. The concepts of cyclic symmetry and axisymmetric elements are combined to obtain efficiency via reduced problem size. Required input and a sample problem are presented in Section 25.

17.2.9 Fully Stressed Design

The capability for fully stressed design provides a method in static analysis for optimizing structural properties for minimum size and/or uniform stress distributions. An iteration loop is provided to automatically update property data based upon internal stresses caused by several independent load cases. Convergence is achieved when all stresses or structural parameters reach their specified limits in at least one case. Input data and control parameters are described in Section 26. This section also includes a small comprehensive, example problem.

PLANNING FOR SPECIAL APPLICATIONS

17.3 PLANNING FOR SPECIAL APPLICATIONS

A complete analysis project requires more effort than simply defining the finite element model, prescribing loads and constraints, and executing NASTRAN. A complete plan for a large analysis project often includes pre-processing steps for data preparation, a pilot model study, grid point resequencing, and an extimation of the execution costs. For the actual NASTRAN executions, checkpoint/restart steps might be deliberately scheduled and controlled with DMAP commands. NASTRAN post-processors for stress-scanning and external plotting may be used to edit output data for ease of analyst interpretation.

Because of the many available methods for solving a particular problem with NASTRAN, the user may find it difficult to select the optimum solution path. For this reason the user should familiarize himself with all of the special NASTRAN options. With this knowledge he may define an analysis approach meeting his own particular needs. The purpose of the detail descriptions of special NASTRAN applications presented in the following sections is to familiarize the reader with the basic capabilities and required input data.

Sections 18 and 19, Statics and Dynamics, apply to nearly every NASTRAN analysis. Unique data requirements and modeling restrictions for the special NASTRAN capabilities are presented in Sections 20 through 26.

Specific requirements for data card formats are given in the NASTRAN User's Manual. The theoretical approaches and their limitations are presented in the Theoretical Manual.

18.1 INTRODUCTION

The static analysis capability in NASTRAN is designed to solve for steady-state structural response to point loads, pressures, inertial loads, thermal loads, or enforced displacements. Rigid Format 1 addresses constrained structures and Rigid Format 2 treats free-body structures under uniform acceleration. Nonlinear static solutions using differential stiffness require Rigid Format 4, buckling is analyzed using Rigid Format 5, and plasticity analysis is processed with Rigid Format 6.

The purpose of this section is to define options for basic finite element modeling and methods for obtaining solutions. Topics are organized according to the sequence in which the typical analyst would approach the problem. Basic methods of simulating a structure using grid points and finite elements are followed by procedures for defining constraints and loading conditions. Thus, the methods used in NASTRAN for controlling the solution and selecting output requests are summarized. Special purpose options and the nonlinear analysis methods are briefly discussed, followed by several examples of static analysis input data decks.

DEFINING THE STRUCTURAL MODEL

18.2 DEFINING THE STRUCTURAL MODEL

The idealized structural model in NASTRAN uses grid points (to which loads are applied and for which degrees of freedom are defined) and elements (which are connected at the points) as shown in Figure 1. Three general types of grid points are employed in static analysis:

- 1. Geometric Grid Point A point in three-dimensional space at which three components of translation and three components of rotation are defined. The motions of the structure at these points are the displacement coordinates which represent the degrees of freedom for which a solution is obtained. The location of the point, the direction of the applied loads, and the orientation of the translations and rotations may be defined in terms of local coordinate systems. The user defines each grid point on a GRID bulk data card.
- 2. Scalar Point A point to which only one degree of freedom is associated. These coordinates may represent actual displacements, displacement differences, or scalar quantities such as fluid pressure, electric current, voltage, or any other quantity required for a particular model. Such coordinates may be connected to one another and to grid points by means of constraint relationships or via scalar elements. The SPØINT bulk data card is used to explicitly define scalar points.
- 3. Axisymmetric Rings A point on the cross section of an axisymmetric structure, connected with axisymmetric elements. Displacement coordinates correspond to translation and rotation coefficients of a Fourier series describing deformation of the ring. RINGAX data cards define these points.

The structural element, be it a line, surface, or solid element, is a convenient parameterizing concept for specifying many properties of the structure, including its mechanical properties, material properties, mass distribution, and applied loads. In static analysis by the displacement method, stiffness properties are derived from the specified structural elements. Mass properties, used in the generation of gravity and inertia loads, may be specified either as properties of structural elements or as properties of grid points.

From a theoretical viewpoint, the formulation of a static structural problem for solution by the displacement method is completely described by the matrix equation

$$[K]{u} = {P}$$
.

(1)

18.2-1

Section Store



Figure 1. Topology of an idealized structural model.

DEFINING THE STRUCTURAL MCDEL

A REAL PROPERTY AND A REAL

In static analysis, the stiffness matrix [K] is calculated for all degrees of freedom using the finite elements. It is then reduced in order by the application of various constraints on the solution vector (u). In a parallel procedure, the load vectors {P} are generated from element and grid point bead date and are modified also via the same constraint relationships used for {u}. After the solution to Equation 1 is obtained for the displacements, {u}, a data recovery process produces requested displacements, forces, and stresses for each specified loading condition. The available options to the NASTRAN user for defining a finite element model and analysis procedures are described below.

18.2.1 Finite Elements

NASTRAN provides a library of more than 60 different finite element formulations. These range from simple scalar springs which connect single degrees of freedom, to three-dimensional solid elements connecting up to 32 grid points. Table 1 in Section 9 of this User's Guide lists all the different element data cards by name. Theoretical development of each element is given in Section 5 of the Theoretical Manual. A description of each element is presented in Section 1.3 of the User's Manual, while the actual data card formats are described in Section 2.4 of the User's Manual. NASTRAN elements are defined by the user with the following data card types:

- A connection card to define the grid or scalar points to be connected, the property number, and any necessary orientation date. The mnemonic for these Bulk Data cards contains the prefix C plus the element name. For example, the CBAR card defines the connections for the BAR element.
- 2. A property card to define the geometric data not implicitly given by the connected grid points and material identification numbers. Each property card is given a unique identification number which is referenced by any number of appropriate elements. The prefix P plus the element name is used to name the card, i.e., PBAR.
- 3. Material data cards to define elastic properties, mass density, thermal expansion data, and stress limits. The MATI, which i = 1, 2,..., define the material property data. Thermal and stress dependencies are defined by MATTI and MATSI data cards. Each material is given a unique identification number which may be referenced by any number of property cards.

18.2-3

STATIC ANALYSIS

As in most general purpose programs, certain exceptions exist to the general rules. For instance, some elements do not require reference to an associated property card or to material data.

The selection of the grid point locations and element types is made by the user based upon his experience, the desired accuracy of solution, the cost of execution, and the engineering objectives for the entire analysis. The following guidelines are useful in defining the element models:

- Select elements most suited to supporting the significant internal stresses. For instance, if bending stress is significant, plate elements would be indicated to support the out-of-plane applied loads.
- 2. The number of elements to be used for modeling a section of the structure is determined by the anticipated displacement and strain pattern. Each different element type approximates the true colution strain pattern with a particular function which may be a constant of a polynomial. The deviation between the actual strains and the approximated element strains determines the accuracy of solution.
- 3. Each type of element may not connect to all six degrees of freedom at a grid point. Any grid point displacement components which are not somehow "connected" by any attached elements must be constrained or they will cause matrix singularities.

18.2.2 Constraints and Boundary Conditions

a production of the second
್ಟ

Constraints are used in NASTRAN to specify known values and relationships for selected displacement coordinates. These constraints are used to specify fixed boundary conditions, to eliminate matrix singularities, to define rigid elements, and to support free body motion.

- Single-point constraints define the displacement at individual displacement degrees of freedom. The data cards used are SPC, SPC1, SPCADD, SPCD, and the seventh field on the GRID card. The constraints on the GRID card are permanent. The SPC, SPC1, and SPCADD data are selected by Case Control cards and may be changed for different subcases. The SPCD data defines enforced displacements via load requests.
- 2. Multipoint constraints are defined by user-supplied coefficients relating two or more displacements. This option is applicable to many modeling situations such as rigid

18.2-4

DEFINING THE STRUCTURAL MODEL

CONTRACTOR STORES

connections between two or more points, defining linkages and hinged joints and defining displacement differences as auxiliary degrees of freedom. The MPC and MPCADD Bulk Data cards are used to supply the input data. Note that:

- a. A dependent coordinate is generated for each data card. Since this coordinate is defined as dependent by the MPC equation, it may not be specified with any other constraint nor as a dependent coordinate in another MPC equation.
- b. Since no checks on the validity of the equations are performed, erroneous answers may result if input errors are made. For example, if a rigid linkage is defined using wrong distances between grid points, the structural response will simulate a local attachment to ground.
- 3. Rigid elements are a special automatic form for MPC data generation. The constraint coefficients are calculated from the geometry of the referenced grid points connected via the rigid element specified on CRIGD1, CRIGD2, CRIGD3, and CRIGDR cards. Since each rigid element generates a set of MPC equations, a corresponding set of dependent coordinates is generated. As with the MPC data, no other constraints may be applied to those dependent coordinates.
- 4. Rigid body supports are defined on a SUPØRT card, and they are used to supply temporary, nonredundant constraints on free body motion. They are required for inertia relief analyses using Rigid Format 2, and they are recommended for use in modal analysis of free bodies.
- 5. Matrix condensation, or Guyan reduction, is technically not a method of constraint, but it follows the same general path as the constraint sets described above. The data are supplied using @MIT, @MIT1, ASET, and ASET1 Bulk Data cards. The specified degrees of freedom for elimination by reduction are restricted to only nonconstrained coordinates. Matrix condensation is generally not recommended for use in static analyses for reasons of efficiency. The exception to this is when using the automated substructuring capability described in Section 20.

Errors in constraint data are not automatically corrected by NASTRAN. Conflicting constraint data, when detected, are treated as fatal errors.

STATIC ANALYSIS

Grid point singularity tests are performed to detect null columns in the stiffness matrix resulting from deficient constraints or element connectivities. A typical message contains:

1. The suspect grid point

and the standard and a many states and the states a

- 2. The order of the singularity
- 3. The combinations of grid point components that cause the singularity, arranged in order of severity.

The grid point singularity testing is limited to one grid point at a time and does not detect coupled singularities, such as a free body. The numerical tolerance is conservative and may issue singularity messages for components having small relative stiffnesses. It may also ignore implicit constraints on the displacement components associated with multipoint constraints or rigid element data.

18.2.3 Loading Constraints

Ì

Static loads in NASTRAN fall into four general categories -- directly applied loads, enforced element strains, inertial loads, and enforced grid point displacements. Each category is summarized below:

- Direct load options include concentrated forces and moments, pressures, and scalar loads. Data card names are FØRCE, MØMENT, PLØAD, SLØAD, and their variations, listed earlier in Table 1 of Section 4. The load data on each card must belong to a load set. Different load sets may be combined into one set with a LØAD Bulk Data card. Only those load sets called out in the Case Control Deck are applied to the model as load vectors in the solution phase.
- Enforced element strain options include temperature loads and enforced element deformations, e.g., caused by misfits in structural assembly. Only the R&D, TUBE, and BAR elements allow enforced deformations which are defined on the DEF&RM card.

All elements may be subjected to thermal expansion. The grid point forces and moments computed to prevent expansion are applied as loads on the model. Temperature loads are requested separately from, and are combined with the other applied loads called out in the Case Control Deck.

The temperature field required to define these loadings obey the following priorities:

18.2-6

DEFINING THE STRUCTURAL MODEL

A MARY CONTRACTOR OF A STATE OF A

- a. The default temperature for the entire structure is defined on a TEMPD card.
- b. Grid point temperatures, defined on TEMP cards, override the default.
- c. Element temperature fields containing gradients override the grid point data. Input data cards are TEMPRB, TEMPP1, TEMPP2. These data may also be used to define material coefficients for temperature-dependent materials.
- 3. Inertial loading options include gravity and centripetal forces. These loads, defined on GRAV and RFØRCE data cards, are calculated directly from the structural mass matrix. These data are requested by use of the Case Control LØAD card.
- 4. Enforced grid point displacements were discussed previously under Constraint Conditions in this section. The SPCD card is a special case in that it does not generate constraints and must be requested in a load set. Any component displacements defined on an SPCD card must be defined on SPC or SPC1 constraint data cards.

STATIC ANALYSIS CONTROL DATA

and the second design of the second
18.3 STATIC ANALYSIS CONTROL DATA

The NASTRAN Executive Control, Case Control, and parameter input via the Bulk Data Deck provide control options for directing the basic computations. Complete descriptions of these data are given in Sections 2.2, 2.3, and for the PARAM Bulk Data card in Section 2.4 of the User's Manual. Some of the basic controls are given below.

18.3.1 Executive Control Data

An Executive Control Deck example for static analysis is given below:

Card Input	<u>Comments</u>
ID STATICS, EXAMPLE	Identify job
APP DISP	Use displacement method
SØL 1,0	Select Rigid Format 1
TIME 5	Set internal time limit (CPU minutes)
CEND	End of deck

For more examples and other options on the above cards, see Section 6 of this guide. Other types of executive control cards define checkpoint/restart, request diagnostics, and provide DMAP input and alters.

18.3.2 Case Control Deck

The NASTRAN Case Control Deck is used to define loading conditions, to select constraints, to control printed output, and to request plots. A basic Case Control Deck example for static analysis is given below:

STATIC ANALYSIS

Card Input
TITLE = ANY CHARACTERS
SPC = 25
DISP = ALL
STRESS = ALL
SUBCASE 11
LØAD = 5
LABEL = FIRST CASE LABEL
SUBCASE 25
LØAD = 15
LABEL = SECOND CASE LABEL
BEGIN BULK

. The second s

Comments

Provide output page heading Select single-point constraint set Request output - displacements Request output - element stresses Denote first case Request first loading Provide output label Define second case Request second loading Provide output label End of deck

Other useful examples of case control requests are given in Section 8 of this guide. The options are summarized in Table 5.

At the end of the Case Control Deck, the Plot Control Deck may be added for requesting plots showing the deformed and undeformed static model. The descriptions of the options are given in the User's Manual, Section 4. A basic example of a plot request for the undeformed structure is given below:

Comments		
lotted		
)		

18.3.3 Control Parameters

Parametric control data are specified on PARAM bulk data cards. The format is:

PARAM NAME VALUE
STATIC ANALYSIS CONTROL DATA

and the second of the second
A complete list of parameter options for all Rigid Formats is given on the bulk data card description for PARAM in the User's Manual, Section 2.4, and for each Rigid Format in the User's Manual, Section 3.

4

ì

18.4 OTHER STATIC SOLUTIONS

The discussions above are primarily concerned with linear static analysis of user-defined structural configurations. Alternate analytical processes are available for special structural configurations and responses. These are summarized below.

18.4.1 Property Optimization

NASTRAN includes a method of design optimization for linear static analysis (Rigid Format 1) based on automation of a relatively simple strategy known as "fully stressed design." According to this concept, the cross-sectional properties of each structural element are changed at each design iteration to produce a limit stress (zero margin of safety) somewhere within the element, on the assumption that the loads carried by the element are unaffected by changes in its cross-sectional properties. The assumption is strictly true only for statically determinate structures. In indeterminate structures of low redundancy, the assumption is not much in error, so that a few repetitions of the algorithm will produce a stress distribution throughout the structure which has very nearly a zero margin of safety in every element, i.e., a "fully stressed" design. In structures of high redundancy, the procedures will converge more slowly (if at all), and modifications of the basic strategy may be required to achieve convergence.

The physical quantities involved in the design algorithm are: properties, stresses, and stress limits. The properties may include thicknesses, cross-sectional areas, or moments of inertia. Most NASTRAN elements have several independent properties. They also have several types of stresses and several places where stresses can be evaluated. The stress limits include those for tension, compression, and shear.

18.4.2 Cyclic Symmetry

Many structures, including pressure vessels, rotating machines, and antennas for space communications, are made up of virtually identical segments that are symmetrically arranged with respect to an axis. There are two types of cyclic symmetry: simple rotational symmetry, in which the segments do not have planes of reflective symmetry and the boundaries between segments may be generally doubly curved surfaces; and dihedral symmetry, in which each segment has a plane of reflective symmetry and the boundaries between segments are planar. In both cases, economic solutions for structural response are obtained by analyzing a subregion containing as few segments as possible.

18.4-1

Principles of cyclic symmetry call for reducing the analysis region to the smallest repeated section of the structure. Neither accuracy nor generality are lost in the process, except that the treatment is limited to linear relationships between degrees of freedom. The use of cyclic symmetry allows the analyst to model only one of many identical segments resulting in a large saving of computer time for most applications. Details of the procedures for applying cyclic symmetry are described in Section 23 of this guide and Section 1.12 of the User's Manual. Static analysis with cyclic symmetry is performed using Rigid Format 14.

18.4.3 Nonlinear Static Solutions

and the second
NASTRAN provides for two types of static nonlinearities, differential stiffness and material plasticity. The geometric stiffness effects treated by Rigid Formats 4 and 5 provide the user with a second order approximation to the nonlinear effects of large deflections.

Rigid Format 4 computes the differential stiffness matrix by an iterative technique and treats the new matrix as a load correction. The internal loads are thus not linearly related to the applied load. The iterations are terminated when the user's criteria are met to obtain a solution.

In Rigid Format 5, the applied loads are assumed to move with their points of application and they remain fixed in magnitude and direction. The "differential stiffness" matrix is computed from the internal element loads resulting from the static solution or from the previous iteration. Rigid Format 5 combines the "differential stiffness" matrix with the original stiffness matrix to solve an eigenvalue problem and obtain the critical load factor that would produce budkling. This critical load factor, or the eigenvalue, multiplied by the preload vector, gives the critical loading at which the structure would become unstable or buckle. The eigenvector obtained is the mode shape in which the structure would buckle.

Rigid Format 6, on the other hand, is used for applications involving nonlinear materials. That is, the material properties are assumed to be stress dependent. The loading is applied in "piecewise linear" increments. After each increment, a new stiffness matrix is generated based on the current state of stress in each element. This new matrix is used to solve for a corresponding increment of displacement and stress. These increments are accumulated to produce the final nonlinear results. Since each increment in load requires a new solution, the user is faced with a compromise between accuracy, which requires small increments, and efficiency, which suggests using the largest possible increments. All static load options are allowed except for those due to temperature and enforced element displacements.

18.4-2

OF POOR QUALITY

18.5 STATIC ANALYSIS EXAMPLES

This section illustrates the NASTRAN input data for simple static analysis. The application examples are extracted from input data used for problems discussed in the NASTRAN Demonstration Problem Manual. The user should refer to this manual for additional detail concerning the model and a discussion of the results. The specialized applications and the examples of how to use the DMAP commands, however, are not presented in the Demonstration Problem Manual.

18.5.1 Rigid Format 1 - Shell with Symmetry Boundary Conditions

This problem demonstrates the finite element approach to the modeling of a uniform spherical shell. A spherical coordinate system is chosen to describe the location and displacement degrees of freedom at the grid points. Triangular plate elements are used to provide a nearly uniform pattern. Two symmetric boundaries are used to analyze the structure with a symmetric pressure load. Further modeling detail is given in the Demonstration Problem Manual, Section 1.2. The model and its loads are pictured below.

The analysis is carried out in two parts. First, the solution is obtained with roller supports which produce primarily a membrane behavior. The second solution is a restart which solves the same problem but with a clamped edge support which induces bending moments at the boundary.

18.5.1.1 Input Cards - Unclamped Boundary and Symmetry Plots

Card	
0	NASTRAN FILES = (NPTP,PLT2)
1	ID DEM102, NASTRAN
2	DIAG 14
3	CHKPNT YES
4	TIME 5
5	APP DISPLACEMENT
6	
/	CENU
8	TITLE - SPHERICAL SHELL WITH PRESSURE LOADING, NO MOMENTS ON BOUNDARY
9	SUBTITLE = NASTRAN DEMØNSTRATIØN PRØBLEM NØ. 1-2-1
10	LØAD = 1
11	SPC = 2
12	PUTPUT
13	DISP = ALL
14	SPCF = ALL
15	STRESS = ALL
16	PLUT ID = NASTRAN DEMONSTRATION PROBLEM ND. 1-2-1
1/	WUIPUI (PLWI)
18	YLWIICK SU Maytmim Decommation 6 0
17	THAT THE DEFENSION O.C.

18.5-1

mine a sub- reares

 $= (\alpha^{(1)})^{1/2} + \delta^{(1)} + \delta^{(2)} \delta_{1} + \delta^{(2)} + \delta^{(2)} + \delta^{(2)} \delta_{1} + \delta^{(2)} + \delta^$

20 21 22 23 24 25 26 27 28 29 31 32 33 34 35 36 37 38 39 40	SET 1 IN SET 2 IN FIND PTITLE U PLOT LAB PTITLE D PLOT SET SET SET SET SET SET SET SET	CLUDE ELE - EDGES CLUDE, PL SCALE PR NDEFØRMED ELS SYMBØ EFØRMED S STATIC D PECTIVE P SCALE, Ø C, ØRIGIN 1, ØRIGIN 1, ØRIGIN 2, ØRIGIN 2, ØRIGIN 2, ØRIGIN 2, ØRIGIN 2, ØRIGIN 2, ØRIGIN	MENTS TRI AND CENTE (JTEL IGIN 1 STRUCTUR US 6 TRUCTURE EFØRMATIØ RØJECTIØN RIGIN 1000, L N 1000, SYM 1000 SYM 1000 SYM MATIØN 0, 1000, SH 1000, SH 1000, SH	A2 RLINE RE MO, SET 1, ABELS SYMBOLS 9, BOLS 9 SH BOLS 9 SH BOLS 9 SH BOLS 9 SH APE, SYMM APE, SYMM	T 1, ØRIG , VANT PØI , SHAPE, IAPE SYMME IAPE SYMME IAPE SYMME IETRY X, IETRY X, IETRY X,	IN 1, SHA NT, REGIØ TRY X, TRY Y, TRY XY	NPE, LABEL	S 35, 0.1,	0.9, 0.8	
	1	2	3	4	5	6	7	8	9	10
41 42 43	CØRD2S +CØR1 CTRIA2	2	0.0	.0 0.0 1	.0 6	.0 26	.0	.8	.1	+CØR1
44 45 46 47	GRDSET	1	2	24 90.	7.	.0	.0			
48 49 50	MAT1 PLØAD2 PLØAD2	1	3.+6 -1.0 -1.0	.0 1 31	.1666 2 32	3	4	5	6	
51 52 53	PLØTEL PLØTEL PTRIA2	50 68 31	26 13 1	1 17 3.		51 69	1 17	2 20		
54 55 56	SPC SPC1 +SPC1-2	1 20	26 346 23	12456 1	0.0 2	3	4	11	16	+SPC1-2
57 58 59	SPC1 +SPC1-1 SPC1	1 25 2	123456	5	10	15	19	22	24	+SPC1-1
60 61 62	SPC1 +SPC2-1	220	345 23 2345	1	2	3	4	11	16	+SPC2-1
63 64	SPC1 ENDDATA	2	12456	26	20					

18.5.1.2 Notes on Data Cards

i

0. Requests NPTP and PLT2 be disk files rather than tape.

1. NASTRAN run identifier.

- 2. Requests Rigid Format printout.
- 3. Checkpoint is requested for restart with clamped boundary conditions.

THE PARTY OF THE PARTY. OF THE PARTY OF THE

- 4. Limits Execution time to five minutes. Local installation job card should provide a maximum time of more than five minutes to allow NASTRAN to exit normally and punch the checkpoint dictionary.
- 5. The displacement approach is selected.

- 6. Rigid Format 1 is to be executed without looping, which allows NASTRAN to purge files when they are no longer required.
- 7. End of NASTRAN Executive Control Deck.
- 10. Load condition 1 is the only loading. Note, since this implies only one subcase, no subcase delimiter is required.
- 11. Selects single-point constraint set 2 from Bulk Data for symmetry boundary conditions and unclamped supports.
- 12-15. Specifies output for all grid point displacements, support reactions and element stresses.
- 16-18. Identifies plot header frame label and Stromberg-Carlson plotter.

.

- 19. Specifies the maximum displacement to be plotted to a scale of 6 inches relative to the 90-inch radius of the sphere.
- 20-23. Specifies element sets, scaling factors, and origin for plotting.
 - 24. Specifies plot title to appear on every plot frame generated.
 - 25. Plot first set of elements (set 1), undeformed and orthographic projection.
 - 27. Plot elements of set 1, deformed shape.
- 28-29. Calls for perspective plots and definition of plot parameters.
 - 30. Perspective plot with PLØTEL only, undeformed.
- 31-34. Perspective plot of all triangular plate elements fully developed for symmetry, undeformed (UM 4.2.2.3).
- 35-39. Perspective plot of all PLØTEL fully developed for symmetry, deformed.
- 41-42. Defines orientation of spherical coordinate system.
- 43-44. Triangular element connectivities.
 - 45. Sets default value 2 for the spherical coordinate system for output displacements at all grid points.
- 46-47. Grid point data in spherical coordinate system.
 - 48. Material properties data.
- 49-50. Pressure load data.
- 51-52. Fictitious line elements for plot purposes.
 - 53. Triangular element properties.
- 54-58. Single-point constraint set 1 including both clamped edge supports and symmetry boundary conditions. (Not selected in this run.)
- 59-63. Single-point constraint set 2 including unclamped edge supports and symmetry boundary conditions.
 - 64. Signals the end of Bulk Data.

18.5-3

Special note: The degrees of freedom constrained by the single-point constraint cards 54-63 are relative to a spherical coordinate system and not in the Basic X, Y, Z System.

10.5.1.3 Input Cards - Restart with Clamped Boundary

الى الوالياني الوقاة فالمحادية بعالاة التعالا متخلفة الامارين فالوالي

Card No. NASTRAN FILES = OPTP 0 ID DEM102A, RESTART 1 2 TIME 5 APP DISPLACEMENT 3 4 SOL 1,1 RESTART DEM102, NASTRAN, C2/04/76, 36000, 5 1, XVPS , FLAGS = 0, REEL = 1, FILE = 54 2, REENTER AT DMAP SEQUENCE NUMBER 7 6 7 98, REENTER AT DMAP SEQUENCE NUMBER 116 105 , FLAGS = 0, REEL = 1, FILE = 54 , FLAGS = 0, REEL = 1, FILE = 55 99, QG 100, XVPS 106 107 \$ END OF CHECKPOINT DICTIONARY 108 CEND 109 TITLE = SPHERICAL SHELL RESTART WITH CLAMPED BOUNDARY 110 SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 1-2A 111 ECH9 = B9TH 112 LGAD = 1113 SPC = 1114 **JUTPUT** 115 DISPLACEMENT = ALL 116 SPCFØRCE = ALL 117 ELFØRCE = ALL 118 STRESSES = ALL 119 BEGIN BULK 120 ENDDATA 121

18.5.1.4 Notes on Data Cards for Restart

5-108. Restart dictionary from previous run with unclamped boundary conditions.

- 112. Requests printout of both the current Bulk Data Deck as input (only one card in this case, ENDDATA) and the sorted printout of the full Bulk Data Deck kept from the previous checkpoint as modified by the current input.
- 113. Specifies the one pressure load condition.
- 114. Selects single-point constraint set 1 from Bulk Data for symmetry boundary conditions and clamped supports.

ORIGINAL PAGE IS

and the second


18.5-5

18.5.2 Rigid Format 1 - Nonsymmetric Bending of a Cylinder

This problem illustrates the application of the conical shell element and its related special data. This element uses the Fourier components of displacement around an axisymmetric structure as the solution coordinates. The geometry of the structure is defined by rings instead of grid points. Its constraints must be defined by the particular Fourier harmonics, and the loads must be defined either with special data or in a harmonic form. This element is used only for statics and inertia relief analysis and may <u>not</u> be used in conjunction with any of the other structural elements.

The structure to be solved is a short, wide cylinder with a moderate thickness ratio. The applied laods and the output stresses are pure uncoupled harmonics. The basic purpose of this problem is to check the harmonic deflections, element stresses, and forces. Further modeling and loading detail is given in the Demonstration Problem Manual, Section 1.5. The model and its applied edge moment loading is pictured below.



18.5-6

والدارية العومية فيتعادد والمراجع

18.	5.2.	1	Input	Cards	-	Axisymmetric	Mode1
-----	------	---	-------	-------	---	--------------	-------

```
Card
No.
  1
         ID DEM105, NASTRAN
         DIAG 8
TIME 24
  2
  3
         APP DISP
  4
  5
         SØL 1,1
  6
         CEND
  7
         TITLE = NØNSYMMETRIC BENDING ØF A CYLINDER ØF REVØLUTIØN
         SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 1-5-1
  8
  9
              LØAD = 15
 10
         AXISYM = CØSINE
         ØUTPUT
 11
              SET 1 = 5, 10, 15, 20, 25, 30, 35, 40, 45, 50, 100, 200
SET 2 = 1, 6, 11, 16, 21, 26, 31, 36, 41, 46, 50
 12
 13
              DISP = 1
 14
              FØRCE = 2
 15
 16
         HARMONICS = ALL
 17
         BEGIN BULK
```

	<u> </u>	2	3	4	5	D	7	8	9	10
18	AXIC	20								
19	CCONEAX	1	15	100	1					
20	CCONEAX	50	15	49	50				1	
21	MATI	15	91.0		.3	.5				
22	MOMAX	15	50	lo	157.0796		2 0		1	ļ
23	MOMAX	115	100	20	157.0796		1.0			
24	PCONEAX	15	15	11.0	15	.0833333	15	1 0	5	+Pr
25	+PC	10.0	0.5	10.	90.	180			1.0	
26	PØINTAX	200	100	1						
27	RINGAX	1		50.0	1.0			4		
28	RINGAX	100		50.0				1234		
29	ENDDATA									İ
		_	_1	1	1		L	L	1	1

18.5.2.2 Notes on Data Cards

- 10. Constrains the motions to be symmetric with respect to the X-Y plane.
- 14. Selects displacement output for rings specified in set 1, each harmonic will be output separately.
- 15. Selects element forces for all conical shell elements in set 2.
- 16. Output is to be printed for all harmonics of the applied load.
- 18. Required to define existence of an axisymmetric conical shell problem and specifies 20 harmonics to be used.
- 19-20. Conical shell element connectivities.
 - 21. Specifies material properties.
- 22-23. Specifies edge moment loading for each harmonic.

part in

24-25. Conical shell properties.

26. Defines the ID and location of a point on a RINGAX at which the moment can be applied. 27-29. Specifies location of axisymmetric rings.

Special note: Point 200 in set 1 references the PØINTAX location and obtains the sum of all displacement harmonics. This same PØINTAX could be used to input a concentrated moment at point 200 and NASTRAN would evaluate the appropriate harmonic coefficients.

4

18.5.3 Rigid Format 2 - Free Body Ring

This problem illustrates the use of inertia relief analysis to solve a free-body problem. In inertia relief, the structure is under constant acceleration due to the applied loads; the reactions to the applied load are due to the masses of the structure. Ficticious, nonredundant support points must be provided to define a reference system attached to the body. The displacements of the body are measured relative to the supported coordinates.

The basic problem is illustrated below. The structure consists of a spinning ring with a constant radial load applied to one point. The rotational velocity creates centrifugal loads and the point load causes inertial reactions. The actual dynamic motion of the whole structure is a cyclic motion of the center point coinciding with the rotation of the ring. The displacements measured by the inertia relief analysis, however, will be the static motion relative to the support point displacements. The displacements are defined in a cylindrical coordinate system $(u_1 = u_r, u_2 = u_{\phi}, u_3 = u_z)$. See the Demonstration Problem Manual, Section 2.1 for further modeling details and a discussion of the theoretical solution.



cereious suppor



teritoria e anteritoria

the same of

18.5.3.1 Input Cards - Inertial Relief

Card No.	
1	1D DEM201, NASTRAN
5	UIAG 8,14,21
3	IIME 5
4 r	APP DISPLACEMENT
2	SIDE 2,1
b	CEND
••	
é	TITLE & INERTIA RELIEF ANALYSIS OF A CIRCULAR RING
ĥ	LABEL = UNCENTRATED AND CENTRIFUGAL LOADS
10	SUBTITLE * NASTRAN DEMONSTRATION PROBLEM NO. 2-1-1
10	LØAU = 3
11	PUTPUT
14	$DISP = AL_i$
13	ØLØAD = ALL
14	SPCFØRCE = ALL
15	STRESSES = ALL
16	SET 1 - 1, 6, 7, 12 13 18 10 24
17	ELFORCE = 1
18	BEGIN BULK

BARAR		Te		· 1 · · · ·				9	10
CBAR		10	,		1.0	0.0	0.0	1	1
+B1 CBAR	24		-1.0	0.0	0.0	-1.0	0.0	i 0 0	+61
+824 CØRD2C	2	0	-1.0	0.0	0.0	-1.0	0.0	1	+B24
+CØRD	0.	9.0	0.0	10.0	0.0	0.0	10.0	1.0	+CØR
GRDSET		13	5	100.0	1.0	.0	.0		
GRID	25	?	11.0	0.0	0.0	l c	345		
LØAD MAT1	3	1.0	1.0	1	1.0	2	123456		
+MAT1 PARAM	100	200.	300,		.5				+MAT 1
PBAR	S S	19	1000.0	10.	10				
REARCE	1.0	1.0	-1.0	-1.0					+P5
SUPØRT	1	2	i	1	10.0	0.0	1.0		

18.5.3.2 Notes on Data Cards

- 5. Selects Rigid Format 2 without looping instructions to allow removal of files not requested.
- 10. Load set selection required for Rigid Format 2.
- 11. Requests for all grid points, the output of non-zero components of load.
- 16-17 Requests element forces for all elements selected in set 1.

Official

13

OUA

والمهرد فيجا بجا بجاها العاقية أالك وحود

- 19. Specifies beam properties and orientation default. The orientation is defined in terms of the displacement coordinate system of the BAR origin grid points, which in this case are cylindrical systems.
- 20-23. BAR element connectivities and offset vectors.
- 24-25. Definition of cylindrical coordinate system.
 - 26. Specifies a force applied at grid point 13 in coordinate system and single-point constraints.
- 28-29. Grid point definition.
 - 30. Static load combination of load sets 1 and 2.
 - 33. Sets a parameter to generate weight and balance information referenced to grid point 19.
- 34-35. BAR element properties.
 - 36. Defines a centrifugal force field for load set 2.
 - 37. Specifies fictitious supports for determinate reactions to be applied to constrain the free body motion.

ORIGINAL PAGE IS OF POOR QUALITY

STATIC ANALYSIS

18.5.4 Rigid Format 5 - Symmetric Buckling of a Cylinder

This problem demonstrates the use of buckling analysis to extract the critical loads and the resulting displacements of a cylinder subjected to axial loading conditions. The buckling analysis Rigid Format solves the statics problem to obtain the internal loads in the elements. The internal loads define the differential stiffness matrix $[K^d]$ which is proportional to the applied load. The load factors, λ_i , which cause buckling are defined by the equation:

$$(\lambda_{i}[K^{d}] + [K])\{u_{i}\} = 0$$

where [K] is the linear stiffness matrix. This equation is solved by the real eigenvalue value methods for positive eigenvalues λ_i . λ_i is the multiplying factor, which when applies to the pre-load will produce buckling. More than one mode of buckling can be extracted if so desired.

The problem (illustrated below) consists of a short, large-radium cylinder under a purely axial compression load. A section of arc 6 degrees is used to model the axisymmetric motions of the whole cylinder. See the Demonstration Problem Manual, Section 5.1 for the theoretical solution.



and the second states and the second states and



7

18.5.4.1 Input Cards - With Structural Plotting

Same and the second second

Card No.										
0 1 2 3 4 5 6	NASTRAN FI ID DEM501, DIAG 14 APP DISPLA SØL 5,1 TIME 16 CEND	ILES = PLI , NASTRAN ACEMENT	2							
7 8 9 101 112 1314 15 16 17 18 9 201 223 245 26 27 28 9 301 333 345 36 37 8 9 41 42 43 445 46	TITLE = S' SUBTITLE = SPC = LØAD = 100 ØUTPUT SET 1 SET 2 DISPLA SUBCASE 1 LABEL = S ØUTPUT ØLØAD SUBCASE 2 LABEL = B METHØ SUBCASE 2 LABEL = B METHØ SUBCASE 2 LABEL = B METHØ SUBCASE 1 PLØTID = ØUTPUT (PL PLØTID = ØUTPUT (PL PLØTID = ØUTPUT (PL PLØTID = ØUTPUT (PL PLØTID = ØUTPUT (PL PLØTID = S ØRTHØ MAXIN FIND PLØT STAT PLØT MØDØ ØRIGI SEGIN BUI	(MMETRIC E NASTRAN 1 1 1 1 1 1 1 1 1 1 1 1 1	33 10, 14, 13 , 74, 78 = 1 ALL 2 LUTION Ø	DF A CYLIP ATIØN PRØE B, 22, 26 IØN PRØBL IØN PRØBL IØN PRØBL IØN PRØBL IØN PRØBL IØN PRØBL IØN PRØBL IØN PRØBL IØN PRØBL	NDER BLEM NØ. 1 , 30, 34, .EM NØ. 5- DINT 2, LABELS , 3.0, .0, 200.0	5-1-1 38, 42, 1 1-1 5, SHAPE , ØRIGIN	46, 50, 54 1, SHAPE	4, 58, 62	`	
	1	2	٦	4	5	6	7	8	9	10
47	CØRD2C	100	0	25.0	.0	80.0	50.0	.0	80.0	+C#RD100
48 49 50 51 52	+CØRD100 CTRIA1 CTRIA1 EIGB +EIGB300	25.0 1 80 300 MAX	0.0 200 200 DET	0.0 1 33 .10	2 30 2.5	51 79 4	.0 .0 4	0	1.5E-05	+EIGB300

۰.

and a contained and

	<u> </u>	2	3	4	5	6	7	8	9	10
53	FØRCE	h	1	100	1.0+3	0.0	0.0	0.5		1
54	FORCE	1	33	100	1.0+3	0.0	0.0	-0.5	1	1
55	GRDSET	1				1	10.0	462		
56	GRID	11	1100	80.0	-3.0	-25 n	100	402		
57	GRID	79	100	80.0	1.5	22.5	100			
58	LØAD	100	11.0	1.897	1		1,00		ł	
59	MATI	400	1.+4							
60	PARAM	IERS	1							
61	PTRIAI	200	400	2.5	400	1 302				+DTOTA1+
62	+PTRIA1*	1.510	0.0		1.00	1				TFIRIAL
63	SEQGP	51	2.5	52	3.5	54	5 5	55	C E	
64	SEOGP	75	26.5	76	27 5	78	20 5	70	20.5	
65	SPC	50038	17	3	.0	1.0	23.3	/3	30.5	
66	SPC1	50037	11	lĩ	2	3	21	32	22	
67	SPCADD	1	50037	50038	-			52	33	
68	ENDDATA		1	1		1				
		L								

18.5.4.2 Notes on Data Cards

- 4. Selects Rigid Format 5 for buckling analysis.
- 9-10. Selects single-point constraint and preload sets from bulk data. Required to be specified above the subcase level.
 - 19. Subcase delimiter for linear static analysis of preload condition.
 - 23. Subcase delimiter for buckling analysis.
 - 25. Selects method for real eigenvalue extraction from bulk data. Output will be that selected above the subcase level.
 - 27. Specifies labeling information to identify plots.
- 28-29. Selects structural plotter package and Stromberg Carleson plotter.
 - 30. Defines set of all triangular elements (TRIA1) to be plotted.
 - 32. Calls for perspective projection.
 - 33. Defines orientation of model relative to view axes.
 - 34. Sets plotting parameters.
 - 35. Plots undeformed model with element set 1.
 - 37. Calls for orthographic projection.
 - 38. Specifies the maximum displacement to be plotted to a scale of 3 units relative to the 80-unit radius of the cylinder model.
 - 39. Sets plotting parameters.
 - 40. Calls for a deformed structural plot of the static preload deformations, Subcase 1.
- 41-42. Calls for plotting of all buckling mode shapes within the eigenvalue range of 0.5 to 3.0 without showing the undeformed model.
- 13-44. Defines the view angle and defines plotting parameters.

- 45. Calls for plotting of all buckling mode shapes within the range of 0.0 to 200.0 superimposed on a plot of the undeformed model.
- 47-48. Defines a cylindrical coordinate system.
- 49-50. Specifies the triangular element connectivities.
- 51-52. Provides the parameters for the determinant method of real eigenvalue extraction.
- 53-54. Defines the forces to be applied.

į

anna chaige an each ann an theatrainn an anna thathairtean anna anna

- 55-57. Defines grid point locations and default single-point constraints.
 - 58. Scales the applied load set 1.
 - 60. Calls for printing of residual vectors after static solution.
- 61-62. Triangular element properties.
- 63-64. Grid point sequencing to improve banding of the stiffness and differential stiffness matrices.
- 65-66. Single-point constraint set definitions.
 - 67. Union of single-point constraint sets.

and more services

18.5.5 Rigid Format 6 - Piecewise Linear Analysis with Rod Elements

This problem, as shown below, illustrates elastic-plastic deformation of a three-panel truss subjected to concentrated grid point loads. Piecewise linear analysis involves loading the truss in increments and recalculating the material properties for each element as a function of the element stresses for the last load increment. The new stiffness matrix is used to solve for the change in the deflection caused by a new increment in load. The sum of all these increments represents the final solution (TM 3.8). This problem, though a "standard" verification problem, is not described in the Demonstration Problem Manual.



18.5.5.1 Input Cards - Stress-Dependent Materials

Card <u>No.</u> 1 ID NØNLIN, TRUSS 2 DIAG 8 3 APP DISPLACEMENT 4 SØL 6,0 5 TIME 5 6 CEND

18.5-17

ł

7 T 8 L 9 10 11 12 Q 13 14 15 16 17 18 20 21 22 23 24 25 26 27 28 29 30 31 32 33	ITLE = PI ABEL = NA SPC = LØAD = PLCØEF UTPUT SET 1 = 1 OISPLA SF EL DISPLA SF EL DUTPUT PLØTTD = 1 OUTPUT(PLO PLØTTD = 1 OUTPUT(PLO CAMER VIEW 0 AXES MAXIM SET 1 INC FIND PTITLE = PLØT LABE PTITLE = PLØT STAT BEGIN BUL	ECEWISE L STRAN VEF 1 FICIENT = CFORCE = LFØRCE = LFØRCE = LFØRCE = LØAD = 1 NASTRAN VI ØT) ER SC A 3 0.0, 0.0, Z, X, Y UM DEFØRM LUDE RØD SCALE, ØR THREE PAN LS DEFØRMED IC DEFØRMED IC DEFØRMED	INEAR ANA RIFICATION = 4 , 5, 6, Th ALL ALL ERIFICATION 0.0 ATION 60.0 IGIN 1, SI EL TRUSS SHAPE ØVE ATION 0,1	ILYSIS ØF I PRØBLEM IRU 10 IRU 10 EN PRØBLE D ET 1 R UNDEFØR THRU 3, 1	A THREE-P M MED SHAPE MAXIMUM D	ANEL TRUS , EACH LØ EFØRMATIO	AD Ν 1.3, SH	APE, LABEL	.5	
	1	2	3	4	5	6	77	8	9	10
34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 57 58 960	CRØD CRØD CRØD CRØD CRØD CRØD CRØD CRØD	1 2 3 4 5 6 7 8 9 1 1 1 2 3 4 5 6 1 16.99235 2 18.82057 2 IRES 4 1 2 3	1 2 3 1 4 3 2 1 2 4 0 0 0 0 0 0 0 0 0 0 0 0 0	1 1 2 2 3 3 4 5 4 0 0 180.0 180.0 360.0 360.0 360.0 540.0 11.814744 100.0 2.0 7.97 21.16 11.77	2 3 3 4 4 5 5 6 6 120.0 240.0 .0 .0 .240.0 .0 .0 .0 .3 4801 .3 2.5	.0 .0 .0 .0 .0 .0 1.E04 200.0	-1.0 -1.0 0 0 0 0	2456 3456 3456 3456 3456 3456		*MAT1 *MAT-1
60 61 62 63 64 65 66 67 68	PRØD PRØD SPC TABLES1 *1001 *1002 *1003 ENDDATA	3 4 1 1000 0.0 4.182353 1.056149	2F-04 5F-03	11.77 7.97 1 2 0.0 10.0 13.0	.0 .0	1 2.091176 5.611675	2 6-4 08E -4	.0 6.273525 11.0	ENDT	*1001 *1002 *1003

18.5-18

and the state of the state of the state of the state of the state of the state of the state of the state of the

18.5.5.2 Notes on Data Cards

- 4. Selects Rigid Format 6 for piecewise linear analysis.
- 9-11. Only one subcase allowed (here it is implied by lack of subcase delimiter). Selects single-point constraint and load set from bulk data. Also selects PLFACT for increments to be applied to the load for each step of the analysis.
- 12-18. Specifies the desired printout of results. The results for elements with stress-dependent material properties will be printed out for each load increment, otherwise, only the cumulation stresses will be printed. All other selected output will be printed for the cumulation results at each load increment.
 - 23. Requests three blank frames be skipped before plotting the next request.
 - 30. Requests plot of undeformed structure for element set 1 (default to first defined set).
 - 32. Requests plot of deformed shape superimposed on undeformed plot for load increments 1 through 3 and defines the same deformation factor for all load increments.
- 51-52. Specifies material properties which are <u>not</u> stress dependent. Uses a double precision continuation card format.
- 53-54. Specifies material properties which are stress dependent.
 - 55. Specifies table reference for stress-dependent material property factors.
 - 57. Provides for three load increments.
- 58-61. Element properties.
- 64-67. Tabular definition of stress-dependent properties using double precision continuation card format. The abscissa (X-entry) represents the strain (E) and the ordinate (Y-entry) represents the corresponding stress. At each load increment NASTRAN recomputes Young's Modulus (E) to be applied for the next load increment (TM 3.8.3). The algorithms used differ with element type (TM 3.8.4).

19. DYNAMIC ANALYSES

19.1 INTRODUCTION

This section summarizes the procedures used in dynamic analyses of structures with NASTRAN. Nine Rigid Formats are provided for dynamic analyses of structures, and they are listed here:

Rigid Format <u>No.</u>	Description
3	Normal Modes Analysis
7	Direct Complex Eigenvalue Analysis
8	Direct Frequency and Random Response Analysis
9	Direct Transient Response Analysis
10	Modal Complex Eigenvalue Analysis
11	Modal Frequency and Random Response
12	Modal Transient Response Analysis
13	Normal Modes Analysis with Differential Stiffness
15	Normal Modes Analysis using Cyclic Symmetry

NASTRAN dynamic analyses are separated into three basic types:

- 1. Eigenvalue Extraction
- 2. Frequency and Random Response
- 3. Transient Response

Furthermore, analysis solutions for each type are obtained using either a direct equation solution or a normal modes transformation of the structural equations before solution.

A flowchart for NASTRAN dynamic analyses is presented in Figure 1. The flowchart also includes the names of key NASTRAN modules utilized in dynamic analyses.

Of the two approaches to problem formulation, the direct method simply uses the displacements at selected yrid points as dynamic degrees of freedom. The modes of structural vibration within a selected frequency range are used as degrees of freedom in the modal method. This approach results in a significant reduction in the number of degrees of freedom for solution, while maintaining response accuracy in the selected frequency range. However, an added step for eigenvalue extraction is introduced in the analysis.

DYNAMIC ANALYSES

The flow diagram in Figure 1 is simplified to the extent that it shows only the major operations that are performed. Complete descriptions of the sequences of module calls for all Rigid Formats are contained in Section 3 of the NASTRAN User's Manual. The functions indicated in Figure 1 are described in subsections of the Theoretical Manual as follows:

> Section 9.2 - Real Eigenvalue Analysis, READ Complex Eigenvalue Analysis, CEAD
> Section 9.3 - Dynamic Pool Distributor, DPD Direct Dynamic Matrix Assembler, GKAD Modal Dynamic Matrix Assembler, GKAM
> Section 9.4 - Dynamic Data Recovery, DDR
> Section 11. - Transient Response Analysis, TRD
> Section 12. - Frequency Response Analysis, FRRD Random Analysis Module, RANDØM

The inputs to NASTRAN for structural dynamic analysis have been organized to provide ease of transition from static analysis input requirements described in Section 18. All elements, constraints, and scalar point modeling options are available in the dynamic analysis Rigid Formats. However, the following basic differences must be considered:

- Longer running time results when the unreduced static model is used for a dynamic solution. Either the modal approach or the use of matrix condensation (ASET or GMIT data cards) with the direct approach is recommended to minimize the problem size.
- 2. Mass data are required.
- 3. Damping may be included using elements or proportionality factors.
- 4. Loads must be defined as functions of time or frequency.
- 5. Data appropriate to the method of solution for time steps, frequencies, and eigenvalue ranges are required, depending on the Rigid Format selected.

In addition to damping and mass modeling features provided for structural dynamic analysis, NASTRAN also provides general capabilities for modeling control systems, hydroelastic effects, and aeroelastic responses through the use of special elements and direct matrix input.

The following sections describe procedures and options used for various approaches to NASTRAN dynamic analysis. The first two sections contain the important topics of modeling mass and damping

19.1-2

RIGINAL PAGE IS

effects. The fourth section describes the modal and direct approach options available for formulating the solution matrices. A separate section is devoted to each of the individual types of solution: eigenvalue analysis, transient response, and frequency response. A summary of output selection procedures is presented, and this section is concluded by a set of annotated sample problems with input data listings.

à

19.2 MASS MODELING

Mass matrix terms for NASTRAN dynamic analyses are input using structural elements, special mass elements, or direct input matrix terms and transfer function data.

19.2.1 Inertia Properties of Structural Elements

All of the metric structural elements (rods, bars, shear panels, twist panels, plates, shells, and solid elements) may have uniformly distributed structural and nonstructural mass. Structural mass is calculated from material and geometric properties. The mass is assumed to be concentrated in the middle surface or along the neutral axis in the case of rods and bars, so that in-plane or in-line rotary inertia effects such as the torsional inertia of beams, are absent. Such effects can, of course, be assigned by the user to grid points. The masses of metric structural elements are transferred to the adjacent grid points at the option of the user by either of two methods, the Lumped Mass or Coupled ("consistent") Mass methods.

In the Lumped Mass method, the mass of an element is simply divided and assigned to surrounding grid points. Thus, for uniform rods and bars, one-half of the mass is placed at each end. For uniform triangles, one-third of the mass is placed at each correr. Quadrilaterals are treated as two pairs of overlapping triangles. Second mass moments are not computed with the Lumped Mass method. The virtues of the method derive from its simplicity.

In the Coupled Mass method, the mass matrix, computed for each structural element, includes off-diagonal coefficients that couple adjacent grid points. This "consistent mass method" is so named because in the element formulation the same assumed deformation shapes, which were used to develop the stiffness matrix terms, are also used to compute the inertial mass matrix terms. The relative accuracy of the two methods is discussed in Section 5.5.5 of the Theoretical Manual.

The options for coupled mass for individual element types (CPBAR, CPRØD, CPQUAD1, CPQUAD2, CPTRIA1, CPTRIA2, CPTUBE, CPQDPLT, CPTRPLT, CPTRBSC) are controlled by the PARAM card. The PARAM data card option, CØUPMASS, causes generation of coupled mass data for all the element types listed above. The default definition of the CØUPMASS parameter produces lumped mass matrix terms.

19.2.2 Concentrated Mass Elements

For modeling concentrated lumped masses, the NASTRAN elements (0NH1, C0NH2, and CMASSi(1 = 1,2,3,4) are used. Mass elements C0NM1 and C0NM2 are used to connect a mass to the six

19.2-1

ORIGINAL PAGE IS OF POOR QUALITY

degrees of freedom at a grid point. The CMASSi elements are used to connect inertia effects directly to scalar degrees of freedom or grid point components. (See Section 5.5.) of the Theoretical Manual for a discussion of the possible uses for these elements.)

DYNAMIC ANALYSES

19.2.3 Other Mass Options

Other useful options available in NASTRAN for mass matrix operations are:

and the second second and the second second second

- 1. A mass scale factor may be provided using the PARAM WTMASS data card. The mass matrix of the entire structure is scaled by this factor. Output from the structural weight summary are not scaled, resulting in mass printout in original units. The default value for WTMASS is 1.0.
- 2. A summary of the overall structural mass, center of gravity (CG), inertia, and principal axes orientation is controlled by the input parameter GRDPNT. If a grid point is specified, the data are calculated relative to that point in space. If zero is specified, the location of the CG and the inertia tensor are calculated relative to the origin of the basic coordinate system. (For details of the actual printout, see Section 1.3.9.4 of the User's Manual.)
- 3. Direct input mass matrix terms may be specified on DM1G bulk data cards. These are described in Section 19.4.



Figure 1. Simplified flow diagram for dynamic analysis by the displacement method.

۰.

DAMPING EFFECTS

and a second second and the second second second second second second second second second second second second

19.3 DAMPING EFFECTS

Damping effects are available via input options in all dynamic analyses except for real eigenvalue analysis (Rigid Format 3). The actual type of input damping data which may be used depends on the method of analysis formulation and solution defined by the Rigid Format. For example, in a modal formulation (Rigid Formats 10, 11, and 12), the equations of motion are normally uncoupled for solution of moda? coordinate response. Therefore, uncoupled modal damping functions may be used as input.

The damping options available in NASTRAN are shown in Table 1. The different types of damping are described below.

- <u>Viscous Elements</u> are defined on CVISC and CDAMPi bulk data cards. The VISC element connects two grid points and acts as a shock absorber. The DMAPi elements connect scalar degrees of freedom.
- 2. <u>Structural Damping</u> is proportional to stiffness matrix terms and is defined both on material (MATi) data cards and as an overall factor on the PARAM G data card. In direct complex eigenvalue analysis (Rigid Format 7) and frequency response (Rigid Format 8), the form of the matrix terms are imaginary stiffness matrix coefficients which approximate a hysterisis effect. In transient analysis, the damping matrix is scaled by frequency factors W₃ and W₄ on PARAM W3 and PARAM W4 data cards and are used as a viscous damping matrix.
- <u>Direct Input Damping matrices are defined using DMIG data cards</u>. Symmetric, unsymmetric, and complex matrix terms may be input. The only restriction is that real terms only must be used in transient analysis.
- 4. <u>Transfer Functions</u> may be used to provide viscous damping either explicitly, through the coefficients added to the damping matrix, or implicitly, through an attached control system model. TF bulk data cards are used for this purpose.
- 5. <u>Modal Damping Factors</u> are used in modal formulations to provide damping on the individual, uncoupled modal coordinates. The damping function $g(f) = 2\zeta$, where ζ is expressed as a percent of critical damping, is input on TABDMP1 data cards.

DYNAMIC ANALYSES

- - 46

1944年、中華に、1942年1月1日の1月1日

.

يوادي المحاسر

٦

Table 1. Available Types of Damping in the Dynamic Rigid Formats.

Type of	Rigid Format										
Damping	7	8	9	10	11	12					
Viscous El eme nts	X	x	x								
Structure (Hysterisis)	x	x									
Structur e (Viscous)			X								
Dir e ct Input	x	x	X	X	X	x					
Transfer Functions	x	x	x	X	X	X					
Modal Damping Factors				x	X	X					

19.3-2

SPECIAL DYNAMIC EFFECTS

19.4 SPECIAL DYNAMIC EFFECTS

In NASTRAN, dynamic analysis is performed via a basic quadratic format,

$$[Mp2 + Bp + K]{u} = {P}.$$
 (1)

Situations occur, rather frequently, where some of the properties of a dynamic system cannot be expressed directly as constant coefficients in a mass, damping, or stiffness matrix. If the analysis is formulated in the frequency domain, as are many problems in aeroelasticity and hydroelasticity, terms may occur that have other than constant, linear, or quadratic dependence on the operator p. These terms, however, may be expressed as frequency-dependent coefficients in the mass, damping, or stiffness matrices. This creates a certain awkwardness in the solution of these dynamic problems.

Several options, as outlined below, are provided for modeling nonstructural dynamic effects including transfer functions, direct input matrices, and nonlinear transient loads. For convenience, and to avoid conflicts with static analyses, additional degrees of freedom, called extra points (EP#INT data card), may be introduced for the dynamics Rigid Formats. (See Section 9.3 of the Theoretical Manual for details of the formulation equations.)

One example of the use of extra points is for structural/control system modeling. In order to facilitate the treatment of control systems, NASTRAN includes an input data format for the specification of transfer functions in the form

$$u_{e} = \frac{1}{b_{0} + b_{1}p + B_{2}p^{2}} \sum_{i} (a_{0}^{i} + a_{1}^{i}p + a_{2}^{i}p^{2}) u_{i}, \qquad (2)$$

which is interpreted by the program as a differential equation with terms in the rows of [N], [B], and [K] corresponding to u_e . u_e must be an extra point, but u_i can either be an extra point or any structural point contained in u_a .

Structural loads proportional to displacements at extra points are represented by adding terms in the stiffness matrix at the intersections of rows corresponding to the structural points and columns corresponding to the extra points. Direct input data cards are provided for this purpose. The superposition of all such terms, including those generated by transfer functions, is called the direct input stiffness matrix, $[K^2]$. The direct input mass and damping matrices are

19.4-1

DYNAMIC MALYSES

 $[N^2]$ and $[B^2]$. Terms in the direct input matrices may refer to extra points, to structural points, or to both. Since dynamic loads may also be applied to extra points, the means are available for the complete simulation of subsystems by means of extra points. Some applications to control system simulation are described in Section 14.2 of the Theoretical Manual.

In transient analysis, nonlinear effects may be modeled as an additional applied load vector $\{N\}$, whose elements are functions of the degrees of freedom (displacement or velocity). The vector, $\{N\}$, is added to the right side of the equation of motion and is treated in a similar but slightly different manner than the applied load vector during numerical integration.

In order to eliminate the cumbersome task of reducing nonlinear loads at each time step from the u_p displacement set to the solution set, it is required that the points to which the nonlinear loads refer belong to the solution set of coordinates. It is further required, for the same reason, that if a modal formulation is used, the points referenced by the nonlinear loads be members of the u_e set (extra points). Otherwise, two complete modal transformations (one for displacements and one for loads) would be required at each time step.

NASTRAN includes four different types of nonlinear elements. They are sufficient to generate many types of nonlinear relationships when used in conjunction with transfer functions and direct input matrices. These data are input on NØLINi bulk data cards.

en provinsión a como aco

19.5 DIRECT VS. MODAL FORMULATION

Both direct and modal formations for dynamic solution are available in NASTRAN. The direct method uses actual grid point, scalar point, and extra point degrees of freedom to define the solution matrices. The modal method transforms structural degrees of freedom into generalized coordinates representing displacements of the normal structural modes. The advantages of each method are summarized below.

The advantages of the Direct Formulation Method are:

- 1. No truncation of coordinates is involved. All modes participate, which provides more accuracy when a wide spectrum of frequencies is excited, such as for an impact problem.
- 2. All options for damping, initial conditions, and nonlinear effects are provided.

3. No extra transformations need to be performed on the load and displacement vectors. The advantages of the Modal Formulation Method are:

- Provides economic solutions when modal coordinates are uncoupled (no coupled, direct matrix input data are used), and/or a large number of time steps or frequencies are required.
- 2. The uncoupled transient solution algorithm performs an explicit integration with less error due to time step size.
- 3. The output of modal frequencies and mode shape vectors allows the user to identify significant modes and facilitates understanding of the dynamic responses.

From experience, the modal formulation has been the practical choice for most structural analyses involving low frequency excitation and a large number of time step or frequency solutions. The direct method is useful for special modeling requirements or when the modal formulation is otherwise inadequate.

With both methods, the use of matrix condensation (Guyan reduction) with the @MIT, @MIT1, ASET, or ASET1 data cards is a valuable tool. This option produces a static condensation on the stiffness matrix and a related, consistent transformation on the mass, damping, direct input, and load matrices (see Section 3.5.3 of the Theoretical Manual for details). The result is a set of smaller-order solution matrices for both real eigenvalue solutions and direct formulation response problems.

EIGENVALUE ANALYSIS

19.6 EIGENVALUE ANALYSIS

NASTRAN has provision for both real and complex eigenvalue analysis of dynamic systems. Real normal modes analysis is closely related to static analysis in that only undamped, real, symmetric matrices are used. Complex eigenvalue analysis is used to obtain general second-order eigenvalue solutions. Each type of analysis has several solution algorithm options, each with particular requirements and advantages. These are described next.

19.6.1 Normal Modes Analysis

The NASTRAN normal modes solution equation has the form:

$$\left[-\omega_{i}^{2}M_{aa} + K_{aa}\right] \{U_{aj}\} = \{0\}, \qquad (3)$$

where the mass, M_{aa} , and stiffness, K_{aa} , matrices correspond to the structural analysis displacement set after constraints and matrix condensation transformations have been applied. The matrices may represent free body structures. Obtaining the eigenvalues, ω_i^2 , and corresponding eigenvectors, $\{U_{ai}\}$, for Equation 3 comprises the basic solution for the analysis.

Data for control of real eigenvalue analysis are specified on the EIGR bulk data card. When free bodies are being analyzed, rigid body support coordinates may be specified on SUP#RT data cards. The purpose of the SUP#RT data are to instruct NASTRAN to calculate free body modes directly from the stiffness matrix, resulting in more accurate and economical solutions.

Different methods for extracting real eigenvalues may be selected using the EIGR bulk data card. Each method has unique characteristics, as described below:

1. The Givens-Householder or "GIV" method performs a Choleski decomposition on the mass matrix, multiplies the stiffness matrix by the inverse product, and performs a tridiagonal transformation on the results. The method extracts <u>all</u> eigenvalues in the order of highest to lowest frequency. The method is efficient when the matrix size is small enough (250) to be held in core. The method also requires a nonsingular mass matrix. The Guyan reduction procedure (@MIT and ASET data cards) is ideal for eliminating singular mass points, such as those associated with rotations, and for reducing the size of the matrices. The advantage of this method is that it can calculate a large number of modes efficiently. Its disadvantages are that it is most efficient for

19.6-1

DYNAMIC ANALYSES

Carline Charles and the second

smaller matrices, and it may produce numerical round-off errors for low frequency modes at or near zero frequency.

The Inverse Power or "INV" procedure performs "shifted-power method" vector iterations 2. to compute each normal mode. Each iteration consists of several matrix-vector transformations. Since a sum of the mass and stiffness matrix may be decomposed several times during the solution, these matrices should be either well-banded or small for efficiency. The method allows singularities in the stiffness and mass matrix. However, the sum of these matrices must not be singular.

An advantage of the inverse power method is that a modal constructed for a static solution will require little additional data for a normal modes solution. There are several disadvantages to this method, lengthy computer runs are required to obtain a large number of modes. Also, the user must estimate the region of the frequencies and the number of modes in the region, requiring some prior knowledge of the solution. If the user overestimates these values, the run could be costly. If he undervalues them, some modes may not be computed.

- The Fast Eigenvalue Extraction Routine, or "FEER" method provides a useful middle 3. ground between the GIV and INV methods. The method uses a Rayleigh-Ritz procedure with shape functions obtained from a power method. The reduced equations are solved using a tridiagonalization procedure which simultaneously orthogonalizes the shape functions. The mass matrix may be singular. Sparse and banded matrices are desirable. The execution cost for a moderate number of modes will generally be less than the other methods where the cost of the required matrix condensation is included for the Givens method.
- The Determinant, "DET," method has the same general characteristics as the usually more 4. efficient inverse power method. The main advantage of including both methods is the redundancy that is provided in case one method should fail for a particular analysis.

19.6.2 Complex Eigenvalue Analysis

The complex eigenvalue analysis technique in NASTRAN is of the form:

$$\left[M_{p}^{2} + B_{p} + K\right] \{Q\} = 0 .$$
 (4)

19.6-2

EIGENVALUE ANALYSIS

and the second design of the second second second second second second second second second second second second

The matrices N, B, and K may be complex and nonsymmetric. The degrees of freedom may include extra points and modal coordinates. In the direct method of formulation (Rigid Format 7), the general form of damping is used. In the modal method (Rigid Format 10), only direct input damping matrices and modal damping are allowed. The uncoupled modal solutions define structural characteristics prior to the introduction of damping and coupling to the extra points.

The control data for extracting the eigenvalues are provided on the EIGC bulk data card. Unlike real normal mode analyses, no special provisions are included for free body modes. As in normal modes analysis, the FEER, inverse power, and determinant eigenvalue extraction methods are available. However, the Upper Hessenberg, "HESS," method replaces the Givens method for complex eigenvalue extraction.

Complete descriptions of each NASTRAN eigenvalue extraction method are presented in the Theoretical Manual, Section 10.
19.7 FREQUENCY RESPONSE ANALYSIS

In frequency response analysis the input loading is defined as a series of sinusoidal loads, each operating at a unique frequency. The steady state response of the structure at each solution frequency is also assumed to be sinusoidal with the magnitude and phase angle being defined by a complex number. The general solution matrix equation is:

$$\left[-\omega^{2}[M] + i\omega[B] + [K]\right] \{u(\omega)\}e^{i\omega t} = P(\omega)e^{i\omega t}, \qquad (5)$$

where ω is the excitation frequency in radians of the input load {P}. All quantities including the output displacements, $u(\omega)$, may be complex numbers.

Figure 1 shows a simplified flow diagram for the frequency response module. The calculations are of two kinds: the generation of loads and the solution of the dynamic equations to obtain a displacement vector. In static analysis a number of automatic load generating subroutines (gravity load, pressure load, temperature-induced load, etc.) are employed in order to simplify the user's task of input data preparation. Similar schemes are not made available for dynamic analysis, mainly because of the great variety in the possible sources of dynamic loads. For frequency response analysis, the user is provided with a flexible input data format which provides a varying degree of generality versus simplicity.

In NASTRAN, the dynamic load vector that is used in a specific subcase, $\{P_j^C\}$, is constructed as a combination of component load sets, $\{P_i^k\}$.

$$\{P_j^{c}(\omega) = \sum_{k} S_{ck}\{P_j^{k}(\omega)\}, \qquad (6)$$

where S_{ck} is a factor defined on a DLØAD bulk data card that gives the proportion of component load k used in combined load set c. The advantage of this system is that it facilitates the examination of different combinations of loads from different sources. The form provided for specifying a component load set is:

$$\{P_{j}^{k}(\omega)\} = \{A_{jk}e^{i(\theta_{jk} - \omega\tau_{jk})} \cdot G_{k}(\omega), \qquad (7)$$

where A_{jk} , θ_{jk} , and τ_{jk} are tabulated coefficients defined on DAREA, DPHASE, and DELAY data cards, respectively, that may be different for each degree of freedom (j). A tabulated complex function

19.7-1

and the second second second second second second second second second second second second second second second

() <u>2</u>





19.7-2

- -----

FREQUENCY RESPONSE ANALYSIS

of frequency, $G_{k}(\omega)$, is provided (with two alternate forms on FLØAD1 and RLØAD2 data cards) to linearly interpolate to the frequencies, ω_{i} , at which solutions are requested. The form provided by Equation 7 is most useful for loads due to traveling waves.

Reaningful interpretation of the symbols in Equation 7 for such applications is that $G_k(\omega)$ is the pressure produced by the wave, that A_{jk} is the exposed area associated with the jth degree of freedom, and that τ_{jk} is the travel time required for the wave to reach the jth degree of freedom.

The addition of damping effects (Section 19.3) are recommended for the avoidance of singularities at the natural frequencies of the system.

έ

To define the results from these analyses, the other necessary input for frequency response analysis is a list of solution frequencies. These are specified on FREQI bulk data cards.

A complete description of NASTRAN frequency response analysis procedures is presented in Section 12 of the Theoretical Manual.

TRANSIENT RESPONSE ANALYSIS

ORIGINAL PAGE IS OF POOR QUALITY,

19.8 TRANSIENT RESPONSE ANALYSIS

Transient response analysis provides a direct simulation of the dynamic response of the system occuring due to initial conditions and time-varying excitation. The solution equation for transient response is:

$$[M]{a} + [B]{v} + [K]{u} = {P(t)} + {N(u)} + {N(u)}, \qquad (8)$$

where P(t) is a user-defined load function calculated at each time step, t. The resulting displacements (u), velocities (v), and accelerations (a), are calculated using either the direct method or the modal method. The nonlinear load vector $\{N\}$ is calculated using user-supplied functions of displacement and/or velocity.

The generation of transient loads can be a formidable task. The most general form of loading is one which has a different time history of load for each point in the structure. Even for a problem of moderate size (say 50 loaded points and 200 time intervals), the data set required to specify a general loading is large (50 x 200 = 10,000 entries). Thus, it is essential to provide data formats that accommodate special cases.

In NASTRAN, the applied transient load vector, $\{P_j^c\}$, that is used in a specific subcase is constructed as a combination of component load sets, $\{P_j^k\}$,

$$\{P_{j}^{c}(t)\} = \sum_{k} S_{ck}^{k}\{P_{j}^{k}(t)\}, \qquad (9)$$

where S_{ck} is a factor defined on DLØAD bulk data cards that gives the proportion of component load set k used in combined load set c. The advantage of this system is that it facilitates the examination of different combinations of loads from different sources.

Two separate forms are provided for specifying component load sets. In the first, or general form, TLGAD1 bulk data cards define the load function:

$${P_j}^k(t) = {A_{jk}}F_k(t - \tau_{jk}),$$
 (10)

where A_{jk} and τ_{jk} are tabulated coefficients that may be different for each loaded degree of freedom (j). The load coefficients A_{jk} may be static load vectors (FØRCE, MØMENT, PLØADI, etc.) in addition to the direct DAREA bulk data. A given table of coefficients (A or τ) may be

referenced by more than one component load set. A tabulated function of time, $F_k(t - \tau)$, is linearly interpolated between entries. The form provided by Equation 10 is particularly useful for laods due to traveling waves. In such applications, F_k represents the pressure produced by the wave, A_{jk} is the exposed area associated with the jth degree of freedom, and τ_{jk} is the travel time required for the wave to reach the jth degree of freedom.

In a second, for n of transient loading, the TLGAD bulk data card defines the function:

$$\{P_{j}^{k}(t)\} = \{A_{jk}\}(\bar{t})^{n_{k}} e^{\alpha_{k}\bar{t}} \cos(2\pi f_{k}\bar{t} + \phi_{k}), 0 \le \bar{t} \le T_{2k} - T_{1k}, (11)$$

$$\{P_{j}^{k}(t)\} = 0, 0 > \bar{t} \text{ and } \bar{t} > T_{2k} - T_{1k}, (11)$$

where

$$\bar{t} = t - T_{1k} - \tau_{jk}$$
 (12)

ORIGINAL PAGE IS

OF POOR QUALITY

The coefficients A_{jk} and τ_{jk} have the same meaning as they do in the general form, Equation 10. The six constants, T_{1k} , T_{2k} , n_k , α_k , f_k , and ϕ_k are entered on a separate data card for each load set. Although the appearance of Equation 11 is formidable, its application to a variety of special cases is straightforward. The time constants, T_{1k} and T_{2k} , define the time limits of non-zero load for a point at which the time delay τ_{jk} is zero. A square wave between T_{1k} and T_{2k} is specified by setting $n_k = \alpha_k = f_k = \phi_k = 0$. A sine wave with frequence f starting at $t = T_{1k}$ is specified by setting $n_k = \alpha_k = 0$, $f_k = f$ and $\phi_k = -\pi/2$. A polynomial function, $P = a + bt + ct^2 + ...$, is specified by combining a number of load sets with different integral values of n_k and $\alpha_k = \omega_k = \phi_k = 0$. Since blank entries are interpreted as zeros by the computer, the extra burden on the user due to the generality of Equation 11 is minimal in all of the above examples.

Other input data for transient analyses are initial conditions, time sueps, and nonlinear loads. These data are selected using Case Control. The TIC (Transient Initial Condition) data cards are used to identify initial displacements and velocities. Default values are zero for grid point degrees of freedom without explicit deformations. Initial conditions may be applied only to solution grid point or extra point displacement (not modal) coordinates.

19.8-2

TRANSIENT RESPONSE ANALYSIS

Shear a shear the second second

The TSTEP bulk data card is used to define different time intervals, each with different time step sizes. For each interval, the user defines the number of steps, the time increment, and the number of increments between each printed or plotted solution vector.

Å.

Nonlinear load data are input on NGLINI (1 = 1,2,3,4) bulk data cards. These are used to define external loads as functions of the current displacements or velocities. The degrees of freedom referenced by these data must be physical solution coordinates.

A thorough description of transient analysis procedures in NASTRAN is presented in the Theoretical Manual, Section 11.



EXAMPLES OF DYNAMIC ANALYSES

CONTRACTOR STREET

19.9 EXAMPLES OF DYNAMIC ANALYSES

÷.

K.

The following sample problems illustrate many of the options and controls used in typical dynamic analysis applications. The actual structural models are relatively small and simple for the purpose of illustration and for obtaining analytical and theoretical comparisons. However, nearly all the different types of bulk data are illustrated, along with many case control options such as subcases and plotting.

19.9.1 Rigid Format 3 - Normal Modes Analysis of a Plate

This problem demonstrates the solution for natrual frequencies of a large-order problem. The structural problem consists of a square plate with hinged supports on all boundaries. The 10x20 model, shown below, uses one half of the structure and symmetric boundary constraints on the mid-line in order to reduce the order of the problem and the bandwidth by one-half. The results of this analysis can be found in the Demonstration Problem Manual, Section 3.1. ORIGINAL PAGE IS OF POOR QUALITY

. .

44.2

•

11

10.5 F

. ...

DYNAMIC ANALYSES

;

. . .



19.9-2

EXAMPLES OF DYNAMIC ANALYSES

Sterne statutes securine and

Because only the bending modes are desired, the in-plane deflections and rotations normal to the plane are constrained. The inverse power method of eigenvalue extraction is selected and both structural mass density and nonstructural mass-per-area are used to define the mass matrix.

19.9.1.1 Input Cards - With Structural Plotting

. .

بذريانك

1

Card No.	
0	NASTRAN FILES = PLT2
1	ID DYNAMICS, NASTRAN
2	
3	APP UISPLACEMENT
4	
2	
0	
7	TITLE = VIBRATIONS OF A 10 BY 20 PLATE
8	SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 3-1
9	\$
10	SPC = 37
11	METHIDU = 3 \$ INV - ENCLIPSES I MIDUE - FINUS 3 RIDIDIS
12	
13	DUITUI CET 1 - 1 THON 11 24 THON 44 56 THON 66 70 THON 00 111 THON 121
14	SET 2 = 1 THUH 12 22 23 33 34 44 50 10KU 00,70 10KU 00,111 10KU 12!
16	99 100.110 THEN 121
17	DISPLACEMENTS = 1
18	SPCFORCE = 2
19	\$
20	PLØTID = NASTRAN DEMØNSTRATIØN PRØBLEM NØ. 3-1-1
21	ØUTPUT(PLØT)
22	PLØTTER SC
23	SET 1 INCLUDE PLØTEL
24	SET 2 INCLUDE QUADI
25	MAXIMUM DEFØRMATIØN 1.0
26	FIND SCALE, DRIGIN 10
27	PTITLE UNDEFORMED SHAPE
28	PLDT DRIGIN 10, SET 2, LABELS
29	FIND SCALE, BRIGIN 11
30	PITILE MODAL DEFORMATIONS
31	PLUI MUDUAL DEFURMATIONI, UNIGIN II, SHAPE
32	BEGIN BULK

CONSTRUCT ON THE OWNER 4 . . .

19.9-3

DYNAMIC ANALYSES OF POOR QUALITY

	1	2	3.	4	5		7	8	9	10
33 ····	COUADI	1 / 501 219	23	1	220	1300 and 231	12 232	.00	open Mirgeri Lini i Lini e	Maga Chine di Ku
35	EIGR	2 5 5	INV	.85 ^{%*}	. 89	Y ^a ang tuto	1 100 #1992 - 1	0	· · · · · ·	CSIMPL-3-
36 37 38	+SIMPL-1 EIGR +EIG1220	MAX 12 MAX	DET	.89	29.0	20	20	0	200 m - 1	+EIE1220
39 40 41	GRDSET GRID GRID	1 231	2 017	.0000 10.000	.000 20.000	.600	((126		5453 ⁻¹¹⁻³
42 13 44 45	HATI HATI PARAM PLØTEL	2 30000. GRDPNT 301	28000. 111 1	11	. 300	302			ing Maga Mga Sana Mga Sana	
46 47 48	PLIDTEL PQUAD1 +POUAD1	397 23 0,5	36 2 0.0	25 1.0	2	396 , 063333	25		6.04303	+PQUAD1
49 50 51 52	SCPc +31001H +31002H SPC1	37 67 155 37	5 78 166 35	1 89 177 221	12 100 188 222	23 111 199 223	34 122 210 224	45 133 221 225	144 226	+31001H +31002H +21001H
53 54	+21001H ENDDATA	227	228	229	230	231				

19.9.1.2 Notes on Data Cards

Marine 1841 1

- 4. Selects Rigid Format 3 for a Normal Modes Analysis.
- 11. Selects method of real eigenvalue extraction from Bulk Data.
- 13-18. Requests printout of results.
- 20-28. Sets up for orthographic projection plotting of the original model, undeformed.
- 29-31. Sets up for orthographic projection plotting of the mode shapes using the first element set (PLØTEL) for each of the three mode shapes generated.
- 33-34. Quadrilaterial plate element connectivities.
- 35-37. Specify methods for real eigenavlue extraction.
- 39-41. Grid point location and default single-point constraints.
- 45-46. Defines line elements for plotting.
- 47-48. Quadrilateral plate element properties.
- 49-53. Single-point constraints to define support conditions and symmetry boundary conditions.



19.9-4

EXAMPLES OF DYNAMIC ANALYSES

19.9.2 Rigid Format 7 - Damped Vibration of a String

This problem demonstrates the use of <u>direct</u> complex eigenvalue analysis and the various methods of supplying damping to the structure. The simulated model is a string under tension having uniform viscous and structural damping, as shown below. The stiffness due to tension is modeled with scalar springs, the mass is represented by scalar masses, and the viscous damping is provided by scalar dampers connected on one end to the points and fixed on the other end. The structural damping is provided by the scalar springs and an overall damping factor, G. The user is encouraged to refer to the Demonstration Problem Manual, Section 7.1 for a detailed discussion of the theory and background to this problem.







19.9.2.1 Input Cards - Scalar Elements

Card

ORIGINAL PAGE IS OF POOR QUALITY

30

<u>.</u>							
1 2 3 4 5	ID DEM701 TIME 8 APP DISP SØL 7,1 CEND	, NASTRAN	I				
6 7 9 10 11 12	TITLE = (SUBTITLE CMETH ØUTPUT SET 1 DISP BEGIN BUL	COMPLEX E1 = NASTRAN HOD = 1 \$ I = 1,51,1 = 1 _K	GENVALUE I DEMONSTI FEER 101,151,2	S ØF A 500 RATIØN PR 01,251,30	D CELL STI DBLEM ND. 1,351,401	RING 7-1-1 ,451,501	
	1	2	3	4	5	6	1

	1	2	3	4	5	D	/	0	7	IV
13 14 15 16 17 18 19 20 21 22 23 24 25	CDAMP3 CDAMP3 CELAS3 CELAS3 CMASS3 CMASS3 EIGC +CFEER PARAM PDAMP PELAS PMASS ENDDATA	60002 60499 1 400 40002 40499 1 -1.0 G 481 101 301	401 401 101 301 301 FEER 15.0 .05 6.283185 1.0+07 10.0	2 499 0 499 2 499 MAX	0 0 2 500 0 10.0	60500 2 500 40500	401 101 101 301 1.0-5	500 2 500 500	0 3 0 0	+CFEER
	L						the second second second second second second second second second second second second second second second se			

19.9.2.2 Notes on Data Cards

- 4. Selects the Direct Complex Eigenvalue Analysis approach.
- 8. Selects the FEER method of complex eigenvalue extraction from Bulk Data.
- 9-11. Requests printout of displacements for all grid points in set 1 which will be printed for each mode shape, real components only.
- 13-14. Scalar damper connectivities.
- 15-16. Scalar spring connectivities.
- 17-18. Scalar masses.
- 19-20. Specifies parameters and method for complex eigenvalue extraction.
 - 21. Defines a uniform structural damping coefficient.
- 22-24. Defines scalar element properties.



19.9-6

19.9.3 Rigid Format 10 - Rocket Guidance and Control

λŶ.

This problem, although a simplified model, contains all of the elements used in a linear control system analysis. The flexible structure, shown below, consists of three sections: two sections are constructed of structural finite elements; the third section is formulated in terms of its modal coordinates. A sensor is located at an arbitrary point on the structure and connected to a structural point with multipoint constraints. The measured attitude and position of the sensor point is used to generate a voltage to control the gimbal angle of the thrust nozzle. The nozzle control is in itself a servomechanism consisting of an amplifier, a motor, and a position and velocity feedback control. The nozzle produces a force on the structure due to its mass, acceleration, and the angle of thrust. The motion of any point on the structure is dependent on the elastic motions and the large-angle effects due to free-body rotations.

The user should refer to the Demonstration Problem Manual, Section 10.1 for the discussion of the control system, the associated transfer function parameters, and other modeling details.





19.9.3.1 Input Cards - DMAP Alter, Transfer Function and Resequencing

Card No.	
1 2 3 4 5 6 7 8 9	ID DEM1001, NASTRAN DIAG 14 TIME 5 APP DISPLACEMENT SØL 10,1 ALTER 103 MATGPR GPLD,USETD,SILD,PHIA//C,N,H/C,N,A \$ ENDALTER CEND
10 11 13 14 15 16 17 18 19 20 21	TITLE = COMPLEX EIGENVALUE ANALYSIS - ROCKET CONTROL SYSTEM SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 10-1-1 LABEL = FLEXIBLE STRUCTURE CASE MPC = 101 METHOD = 2 TFL = 20 CMETHOD = 11 OUTPUT SET 1 = 1,100,101,1010 THRU 1090 SVECTOR (SORT1,PHASE) = ALL DISPLACEMENT (SORT1,PHASE) = 1 BEGIN BULK

		2	3	4	5	6	7	8	9	10
22 23 24	BARØR CBAR CRAP	1	10	1	2	.0	10.0	.0	1	T
25 26 27	CELAS4 CELAS4	1001 1003	2.026+7	1001 1003	16	1002 1004	32.42+7 518.7+7	1002 1004		
28 29	CMASS4 CMASS4 CØNM2	2003 101	2.5+3 2.5+3 1	1001 1003	2001 2003 3333.3	2002 2004	2.5+3 2.5+3	1002 1004	2002 2004	
30 31 32	EIGC +FC	119	19 DET	MAX	833.3					+EC
33 34	EIGP EIGR	11	0.0 INV	0.0	2	10.0	6	6		
35 36	+E2 EPØINT	MASS 1010	1011	1030	1040	1050	1060	2	1080	+EZ
37 38 39	GRUSEI GRID GRID	1 101	116.176	0.0	0.0	0.0		1345		
40 41 42	MATT MPC MPC	1 3 100	10.4+6 16 11	4.0+6 6 6	-1	1001	6	.0628318		+161
43 44 45	MPCADD Pafiim Param	101 GRDPNT LMODES	100 101 4	3				-1.0		
46 47 48	PBAR PBAR SEOGP	10 20 100	1 1 10.5	4.0+2 2.0+2	6.0+4 2.0+4 7.5	6.0+4 2.0+4			l	
49 50	SPOINT SUPORT	1001 101	1002 2	1003 101	1004 6	2001	2002	2003	2004	
52 53 54	TF +TX ENDDATA	20 20 100	1080 6	2 -4.25+6	.0	.0	50.0 8.5+4			+T6 +TX
			ί Ι	. 1	, ,					1

19.9-8

EXAMPLES OF DYNAMIC ANALYSES

WALKS STORE THE PERSON AND SHE STORE AND SHE

19.9.3.2 Notes on Data Cards

- 5. Selects the Modal Complex Eigenvalue Analysis approach.
- 6-8. DNAP alter package to print out the grid point list (GPLD), u-set degrees of freedom (USETD), scalar index list (SILD), and the normal modes matrix for the a-set displacements.
- 13. Selects a multipoing constraint set from Bulk Data.
- 14. Selects a method for real eigenvalue extraction from Bulk Data.
- 15. Selects the transfer function (TF) set from Bulk Data.
- 16 Selects the method for complex eigenvalue extraction from Bulk Data.
- 17-20. Request printout of SØLUTIØN set displacements and displacements for structural grid points of set 1 for each mode shape in magnitude/phase angle format.
- 22-24. Beam element connectivity.
- 25-28. Scalar element connectivity.
- 29-30. Concentrated mass data.
- 31-35. Parameters for methods of eigenvalue extraction. Card 33 defines poles in complex plane that are used iwth the associated EIGC parameters.
 - 36. Definition of extra points used to define control system parameters.
- 37-39. Defines location of grid points and default single point constraints.
- 40-43. Multipoint constraint equations used to represent modal coordinates of model between grid points 16 and 19.
 - 44. Requests weight and balance printout.
 - 45. Defines number of lowest modes to be used in this modal formulation.
- 46-47. Beam element properties.
 - 48. Specified resequencing of grid points 100 and 101.
 - 49. Scalar point definition (need not be defined on SPØINT card if number is used with scalar element definition, e.g., cards 25-28).
 - 50. Defines degrees of freedom for use in computing determinate reactions for constraining rigid body motion.
- 51-53. Specification of transfer function parameters.

19.9.4 Rigid Format 8 - Direct Frequency Response of a Plate

This problem illustrates the use of the direct method of determining structural response to steady-state sinusoidal loads. The steady-state response of the structure at each frequency is calculated in terms of complex numbers which reflect the magnitudes and phases of the results.

The particular model for this study is a square plate composed of quadrilateral plate elements, as shown below. The exterior edges are supported on hinged supports and symmetric boundaries are used along x=0 and y=0. The applied load is sinusoidally distributed over the panel and increases with respect to frequency. For further details on the input and results, see the Demonstration Problem Manual, Section 8.0.



19.9-10

A STATE OF A STATE OF A STATE OF A STATE OF A STATE OF A STATE OF A STATE OF A STATE OF A STATE OF A STATE OF A

19.9.4.1 Input Cards - Frequency Dependent Loading

Sent and an entrance

a destrict All other has

Card No. 1 ID CMSO1X10, NASTRAN APP DISPLACEMENT 2 SØL 8,1 3 4 TIME 12 5 CEND TITLE - FREQUENCY RESPONSE OF A SQUARE PLATE 6 SUBTITLE - NASTRAN DEMONSTRATION PROBLEM NO. 8-1-1 7 SPC = 37 8 9 DLDAD = 8FREQUENCY = 8 10 11 *DUTPUT* SET 1 = 1,4,7,11,45,55,78,88,111,114,117,121 DISPLACEMENT (SØRT2,PHASE) = 1 SPCFØRCE (SØRT2,PHASE) = 1 12 13 14 15 **BEGIN BULK**

	1	2	3	4	5	6	7	8	9	10
16 17	COUAD1 COUAD1	1 109 27	23 23	109	2 110	13 121	12 120	.00 .00 2 500000	-01	
10	DAREA"	37		109		3		2.4471748	SE-02	
20	FREG	8	0.0	8.0	9.0	10.0	11.0	126		
22	GRID	1 121		.0 10.0	.0 10.0	.0 .0				
24	MATI	8	3.0+7		.300	6666667			13 55715	
26	RLDADI	8	37	_	0	1			10.00/10	
27. 28	SPC1 +41001H	37	4 8	1 9	2 10	3	4	5	0	
29	SPC1	37	35	111	112	113	114	115	116	+21001H
30 31	TABLEDI	11	118	113	120	121				+11
32 33	+T1 ENDDATA	0.0	10.0	100.0	40.0	ENDT				

19.9.4.2 Notes on Data Cards:

- 3. Selects Direct Frequency and Random Response Analysis approach.
- 9. Selects the RLBAD1 frequency response dynamic loading condition from Bulk Data.
- 10. Required to select from Bulk Data the set of frequencies to be solved.
- 13. Request the displacements for grid points of set 1 to be printed in magnitude/phase format, sorted by grid point.
- 14. Requests support forces to be printed at grid points of set 1 to be printed in magnitude/phase format, sorted by grid point.
- 16-17. Quadrilateral element connectivities.
- 18-19. Specifies the dynamic load scale factor and point of application in double precision format.
 - 20. Defines set of frequencies for solution. Output will also be printed at these frequencies because no ØFREQUENCY card is used.

-

21-23. Provides gridpoint locations and default single point constraints.

24-25. Defines material and element properties.

THE REAL PROPERTY AND INCOME.

26. Defines frequency dependent dynamic load function.

27-30. Specifies support and symmetric boundary single point constraints.

31-32. Defines table of frequency dependent coefficients referenced, in this case, by RLBAD1 (field 6).

EXAMPLES OF DYNAMIC ANALYSES ORIGINAL PAGE IS OF POOR QUALITY

19.9.5 Rigid Format 11 - Modal Frequency and Random Response of a Beam

This problem demonstrates the frequency response solution of a structure using an uncoupled modal formulation. The structural degrees of freedom are the uncoupled modal displacements. This problem also illustrates various methods of applying frequency response loads which may be input as complex numbers and added together for each subcase.

The structure to be solved consists of a beam with simple supports, as shown below. Included in the structural model is a "general element" representing the first two cells of the ten-cell beam. The applied load for the three subcases are applied to the center three points on the beam. The random analysis data consists of a flat power spectral density function ("white noise") for the three loading subcases. The first subcase spectral density is correlated to the third subcase spectral density, simulating two interdependent probability functions. The XY-plotter capability is used for output. For further details and theory, see the Demonstration Problem Manual, Section 11.1.



19.9.5.1 Input Cards - "Complex" Loading and Random Response Output

No. NASTRAN FILES = (PLT2,NPTP) 0 ID DEMIIOI, NASTRAN 1 2 DIAG 14 CHKPNT YES 3 DISPLACEMENT 4 APP 5 11,3 SOL 6 TIME 7 ALTER 112 MATPRN PHIA...,// \$ 8 9 ENDALTER 10 CEND

Card

ž

· 我们有一个人们,这次不是我们的你们都没有了这个的情况,就是我们的是我的的是我的的时候,我们有一个人,不是一个人们,不是我们的是我们的吗?"""

5. Î.

•

, **`**,

Card No. 11 MAXLINES = 50000 TITLE = FREQUENCY AND RANDOM RESPONSE OF A 10 CELL BEAM 12 SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 11-1-1 13 14 SPC = 1115 METHOD = 2 FREQUENCY = 503 16 RANDOM = 11 17 SDAMPING = 11 18 DUTPUT 19 20 21 SET 2 = 5,10SET 6 =6 22 SET 10 = 6,11 23 24 DISP(SORT2, PHASE) = 10 ACCELER(SORT2, PHASE) = 10 25 26 $\beta L \beta A D = 6$ ELFORCE(SORT2, PHASE) = 2 27 SUBCASE 1 LABEL = THREE POINTS LOADED WITH TWO SETS 28 29 DLOAD = 50630 SUBCASE 2 31 LABEL = ONE POINT LOADED WITH TWO SETS AND TIME DELAYS 32 DLØAD = 507 33 SUBCASE 3 34 LABEL = ONE FOINT LOADED WITH TWO TABULAR LOADS 35 DLGAD = 51036 37 PLOTID = NASTRAN DEMONSTRATION PROBLEM NO. 11-1-1 **DUTPUT(XYDUT)** 38 39 PLOTTER SC. 40 CAMERA = 3SKIP BETWEEN FRAMES = 1 41 42 XGRID LINE = YES 43 YGRID LINE = YES 44 XLOG = YES 45 YLØG = YES 46 XTITLE = FREQUENCY (HERTZ) 47 **YTITLE = S** 48 TCURVE = POWER SPECTRAL DENSITY OF POINT 6 DISPLACEMENT 49 XYPLØT, XYPRINT DISP PSDF / 6(T3) 50 S 51 TCURVE = POWER SPECTRAL DENSITY OF POINT 6 ACCELERATION 52 XYPLOT ACCELERATION PSDF / 6(T3) 53 \$ 54 XLOG = NO 55 YLØG = NØ 56 XTITLE = TIME LAG (SECONDS) 57 YTITLE = R 58 TCURVE = AUT#C#RRELATION FUNCTION FOR POINT 6 DISPLACEMENT 59 XYPLØT, XYPRINT DISP AUTØ / 6(T3) **BEGIN BULK** 60

....

EXAMPLES OF DYNAMIC ANALYSES

Sharen and a star of the star

10.54236

Sec. 1

ł

and the second second second

r

l

A DATE AND A DESCRIPTION OF A DESCRIPTIO

	1	2	3	4	5	• 6	7	8	9	10
61	CBAR	3	1	3	4	20.	.0	1.	1	
62	CBAR	10	1	10	11	20.	.0	1.	1	
63	CONM2*			1				5.34604-3	3	
65	-mi (man/2+	12		2		1		1 069209	2	1112
66	*M2	0. 1		ō. 1				1.003200	6	
67	CON42*	13		3				5.34604-3		#13
68	*M3		_	_						
69	DAREA	2	5	5	-100.	e	2	50		
71	DAREA	2	7	3	50.	5 7	5	100		
72	DAREA	3	6	3	100.	·	J			
73	DAREA	510	6	3	1.0					
74	DELAY		6	3	.5555-2					
/5	DLOAD	500	¦.	1.	5	1.	07			
77	DLOAD	510	2.0	1.0	5101	1.0	5102			
78	DPHASE	1	6	3	30.					
79	DPHASE	5102	6	3	-30.0		_			
80	EIGR	2	INV	40.	1000.	3	5			+EG
82		MASS 508	0	5.0	40					
83	GENEL	1101	.0	2	1	2	3	2	5	+1
84	+1	3	1	3	3	3	5	-	-	+2
85	+2	UD		1	1]	3	1	5	*30
80 87	*30 *31	2	5_9	.8904493	5-8	0.		0.	c .	*31
88	*40	-2.31696-	•6	0.		7.723200	5-6	-2.31696	-6	+41
89	+41	2.31696-6	5	Ŏ.		-6.95088	04-6	2.31696-	5 Š	*50
90	*50	1.7808987	7-8	0.		0.		24.71424	1-6	*51
91	*51	-9.26784-	-6	4.633920	3-6					+60
92 93	+70	3	1.0		0.0	0.0	1.0	-2.0	0.0	+/0
94	+80	0.0	0.0	1.0	0.0	0.0	0.0	1.0		TOU
95	GRDSET							246		
96	GRID	1		.0	.0	.0				
9/ 08	GRIU		10 446	20.	.0	.0				
99	PARAM	GRDPNT	0	4.70		.2323-3				
100	PARAM	LMODES	4							
101	PBAR		1	21.189	.083	.083				
102	PANDOS	11			.5					
104	RANDPS	ii	2	2	1.0		11			
105	RANDPS	ii	3	3	.5		ii			
106	RANDT1	11	100	.0	.1 •					
107	RLBADI	101	510		6102	5101	5100			
109	RI GAD2	5	2		5102	1	5102			
110	RL BAD2	6	3		1	li	2			
m	RL MAD2	7	3	1		1				
112	SPC			13			13			
114	TARDMP1			15			3			
115	+DAMP	0.0	0.0	50.0	0.02	ENDT				
116	TABLED1	11								+TAUU
117	TABLED	0.0	ր.	100.	μ.	ENDT				
110	+TAD21	6.0	30.	100	30	ENDT				+TAUZ1
120	TABLEDI	5101							1	+TAD30
121	+TAD30	0.0	75.0	10	75.0	ENDT				

	<u></u>	22	3	4	5	6	7	8	9	10
122 123 124 125 126 127	TABLED1 +TAD31 TABRND1 +TR +TR2 ENDDATA	5102 0.0 11 -1.0 101.0	50.0 0.0 0.0	100. 0.0 ENDT	50.0 100.0	ENDT 100.0	100.0	100.0	0.0	+TAD31 +TR +TR2

19.9.5.2 Notes on Data Cards

A start and the start and the

- 3. Checkpoint is requested to be taken.
- 5. Selects modal approach for Frequency and Random Response analysis removing the looping and mode acceleration computations.
- 7-9. DMAP alter to obtain a printout of the mode shapes for the analysis set of displacements generated during the real eigenvalue analysis. (This output is not otherwise available with this Rigid Format.)
- 11. Specifies a user limit in NASTRAN for a maximum of 50,000 lines of output.
- 14. Selects a set of single point constraints from Bulk Data.
- 15. Selects the method to be used for real eigenvalue extraction required to get the appropriate modal coordinate data for this medal approach.
- 16. Selects the set of frequencies required for solution of the frequency response analysis.
- 17. Selects the power spectral density factors (RANDPS) and autocorrelation function time lag (RANDT1) from Bulk Data required for the random response analysis.
- Defines the equivalent of structural damping as a function of frequency in modal formulation problems (TM 9.3.4).

÷

ì

- 19-26. Specifies output sets for displacements, accelerations, loads and element forces. Note that the loads will be output for all frequencies at each grid point (SØRT2) and in real/imaginary format. Because SØRT2 was specified for other output quantities it overrides the SØRT1 default for loads. Also note that requests for output of the results of the Random Response analysis are part of the XY plot output package.
- 27-29. Defines the first dynamic load to be analyzed.
- 30-32. Defines the second dynamic load to be analyzed.
- 33-35. Defines the third dynamic load to be analyzed.
- 37-40. Sets up for XY plots using Stromberg Carlson requesting both film and paper inserting one blank frame between each plot.
 - 42. Requests drawing of vertical grid lines at each tic mark.
 - 43. Requests drawing of horizontal grid lines at each tic mark.
- 46-47. Provides axis labeling information.
 - 48. Provides label to identify curve being plotted.
 - 49. Requests both plotted and printed output of the z-displacement power spectral density function for grid point 6.
 - 51. Provides label to identify next curve to be plotted.
 - 52. Requests plotted output of the z-acceleration power spectral density function for grid point 6.
- 54-55 Requests scalar distribution of tic marks for both the X and Y axes.
- 56-58 Provides titling information for next curve to be plotted.

19.9-16

EXAMPLES OF DYNAMIC ANALYSES

- 59. Requests both plotted and printed output of the autocorrelation function for the zdisplacement at grid point 6.
- 61-02. Defines beam element connectivities.

Sal Line

- 63-68. Defines concentrated mass elements in double precision format.
- 69-73. Specified dynamic load scale factor sets and the points of application.
 - 74. Provides for the dynamic load time delay to be referenced by the RL#AD2 (SID=7 in field 4) card number 111.
- 75-77. Specifies the dynamic load combinations to be applied.
- 78-79. Provides for the dynamic load phase lead to be referenced in field 5 of the RL#AD2 and RL#AD1 cards 125 and 123, respectively.
- 80-81. Specified parameters for real eigenvalue extraction.
 - 82. Specified frequency range and intervals for use in frequency response analysis $(f_i = 0.0, 5.0, 10.0, \dots, 200.0)$ or 40 increments.
- 83-94. Specifies general element matrix data (TM 5.7).
- 95-97. Defines grid point locations and default single point constraints.
 - 98. Defines material properties.
 - 99. Requests output for weight and balance.
 - 100. Requests the four fundamental modes to be used as modal coordinates.
 - 101. Beam element properties.
- 102-105. Specification of input power spectral density function.
- 106-111. Frequency response dynamic load functions.
- 112-113. Single point constraint sets (set 11 was selected).
- 114-115. Frequency dependent damping coefficient table referenced by SDAMP in Case Control.
- 116-123. Tabular definition of frequency dependent load coefficients referenced by RLØAD1 and RLØAD2 cards.
- 124-126. Tabular function of power spectral density function referenced by RANDPS cards.

19.9.6 Rigid Format 9 - Direct Transient Analysis with Direct Matrix Input

This problem demonstrates the capability of NASTRAN to perform a transient analysis on a system having nonsymmetric stiffness, damping, and mass matrices. The problem is formulated with direct matrix input without the use of elements. It illustrates the use of time step changes, selection of printout intervals, application of loads, initial conditions, and a simple curve plot package. For details of the problem formulation and the results, see the Demonstration Problem Manual, Section 9.1.

19.9.6.1 Input Cards - Direct Matrix Input and XYØUT Plotting

an a substantiation of the second statement of the second s

a,

E,

Card No. 0 **NASTRAN FILES = PLT2** ID DEM901, NASTRAN 1 2 DIAG 14 APP DISPLACEMENT 3 SPL 9,1 4 TIME 5 5 6 CEND 7 TITLE = TRANSIENT ANALYSIS WITH DIRECT MATRIX INPUT SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 9-1-1 8 9 TSTEP = 3210 IC = 3211 DLDAD = 32K2PP=KCOMP 12 M2PP=MCBMP 13 14 B2PP=BCOMP 15 DUTPUT 16 SVELD = ALL DISP(SORT2) = ALL 17 PLBAD(SPRT2) = ALL 18 PLOTID = NASTRAN DEMONSTRATION PROBLEM NO. 9-1-1 19 **BUTPUT(XYBUT)** 20 21 PLØTTER SC 22 CAMERA = 3 SKIP BETWEEN FRAMES = 1 23 24 TCURVE = * * * * EPDINT DISPLACEMENT(INCHES) * * * * 25 XTITLE = TIME(SECONDS) 26 \$ 27 YVALUE FRINT SKIP = 1 28 XDIVISIONS = 25 XVALUE PRINT SKIP = 1 * * * * * * * * * * * FULL FRAME PLOTS * * * * * * 29 30 \$ * 31 YGRID LINES = YES YGRID LINES = YES 32 33 34 YDIVISIONS = 22 \$

1

Card No.		
35	YTITLE = EPØINT 10	DISPLACEMENT *INCH*
36	XYPLØT DISP / 10(T1)	
37	\$	
38	YDIVISIØNS = 20	
39	YTITLE = EPØINT 11	DISFLACEMENT *INCH*
40	XYPLOT DISP / 11(T1)	
41	\$	
42	YTITLE = EPØINT 12	DISPLACEMENT *INCH*
43	XYPLØT DISP / 12(T1)	
44	\$	
45	YTITLE = EPØINT 13	DISPLACEMENT *INCH*
46	XYPLØT DISP / 13(T1)	
47	BEGIN BULK	

and the second states in the

	1	2	3	4	5	6	7	8	9	10
48	DAREA	1	10		=1.5	11		-1.0		
49	DAREA	li	112		-13.5	113		36.0		
50	DELAY	1	10		11.0	lii	11.0			1 1
51	DELAY	1	12		11.0	13		1.0	Į I	1 1
52	DMIG	BCOMP	0	1	li	2		'		1 1
53	DMIG	BCOMP	lii l	Ó			0	-15.0	l i	+BC1
54	+BC1	11	0	30.0		112	lõ	-15.0	i 1	
55	DMIG	BCØMP	12	0		lii -	ŏ	-24.0		+BC2
56	+BC2	12	0	48.0		13	lo	-24.0	ł	
57	DMIG	BCOMP	13	0		12	l o	-2.0		+BC3
58	+BC3	13	lo I	4.0						
59	DMIG	KC ØMP	0	1	1	2		1	ļ	
60	DMIG	KCØMP	10	0	1	10	0	2000.		+KC1
61	+KC1	11	0	-1000.					1	
62	DMIG	DCOMP	12	0	•	11	0	-100.8		+KC2
63	+KC2	12	0	200.0		13	0	-100.0	(1
64	DMIG	KCOMP	13	0	}	12	0	-20.0	i l	+KC3
65	+KCC	13	C	40.0				[]		i I
66	DMIG	MCOMP	0	1	1	2		í '		
67	DMIG	MCOMP	10	0	.	10	0	20.0		+MC1
68	+MC1	11	0	-10.0				j l		
69	DMIG	MCOMP	11	0	1	18	0	-1.5		+MC2
/0	+MC2	11	0	3.0	i	12	0	-1.5		1
/1	DMIG	MCOMP	12	0			0	-4.0	l I	+MC4
12	HHU4	12	0	8.0		13	0	-4.0		i l
15	TADLED	10	111	12	13			l I		
74	TABLEDI	10				20				+ <u>11</u>
75			0.0	0.0	0.0	.00	1.0	100.0	1.0	+T2
70		ENUI	1.0	[1.0.0		{		i I
79		32			0.0	10.0	i I			
70	TIC	32			.0			[]	1	
80	TIC	32	12		.0					
81	TIMADI	32		1	-10.0	i		1		
82	TSTEP	32	10	025	2	1 1		ł I	1	
83	+51		100	015	5	i I				+51
84	FNDDATA		100	.012	,				i I	
•		1		1 1		1		1	t I	1

19.9.6.2 Notes on Data Cards:

- 4. Selects the direct approach to Transient Analysis.
- 9. Selects the integration time steps from Bulk Data (TSTEP)

.

10. Selects the initial conditions from Bulk Data (TIC).

- 11. Selects the dynamic loading function from Bulk Data (TLØAD1).
- 12-14. Specifies the direct matrix input for stiffness, mass and damping respectively.
- 15-18. Specifies output requirements for all degrees of freedom in the model. Notice that because one output quantity specifies SØRT2, all time steps for each degree of freedom, the SVELØCITY request will default also to SØRT2.
- 19-23. Sets up for plotting with the XY-curve plotter.

24-25. Labeling information.

- 27. Plot the value for every other tic mark along the Y-axis.
- 28. Provide 25 divisions along the X-axis.

an and a second second second second second second second second second second second second second second seco

29. Plot the value for every other tic mark along the X-axis.

31-32. Plot grid lines for each tic mark in both directions.

33. Provide 22 divisions along the Y-axis.

35. Title for Y-axis.

36. Requests plot of displacement for extra point 10 versus time.

38-39. Provide 20 divisions and new title for Y-axis.

- 40. Requests plot of displacement for extra point 11 versus time.
- 42. Use same 20 divisions and new title for Y-axis.
- 43. Requests plot of displacement for extra point 12 versus time.
- 45-46. Repeat 42 and 43 for plot of displacements for extra point 13.

48-49. Dynamic load factors applied to extra points.

50-51. Delay factors for each extra point loading.

- 52-58. Direct matrix input at extra points for damping.
- 59-65. Direct matrix input at extra points for stiffness.

66-72. Direct matrix input at extra points for mass.

- 73. Provides identification for extra points (single degree of freedom per point).
- 74-76. Table of dynamic load coefficients versus time referenced by TLØAD1 card (field 6).
- 77-80. Specifies time dependent loading function.
 - 81. Specifies time dependent loading function.
- 82-83. Specifies two sets of time step intervals for integration of the transient response and user specifies output interval.

SPECIAL NOTE: No structural elements were used in this problem. All degrees of freedom were identified with separate extra points.

19.9.7 Rigid Format 12 - Modal Transient Response of a Beam

This problem demonstrates the transient analysis of a free-body using the integration algorithm for uncoupled modal formulation. A hundred-cell beam model is shown below. Modal damping is included at a function of natural frequency. The omitted coordinate feature was used to reduce the number of degrees of freedom to an analysis set equivalent to eleven grid points. Both structural and curve plots are requested. For further details and the results, see the Demonstration Problem Manual, Section 12.1



19.9.7.1 Input Cards - Both Structural and XYOUT Plotting

Card No.	
0	NASTRAN FILES = PLT2
1	ID DEM1201, NASTRAN
3	APP DISPLACEMENT
4	SØL 12,3
5	TIME 10
6	CEND
7	TITLE = TRANSIENT ANALYSIS OF A FREE ONE HUNDRED CELL BEAM
8	SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 12-1-1
9	DLOAD = 516
10	SDAMP = 15
11	TSTEP = 516
12	METHOD = 2
13	ØUTPUT
14	SET 1 = 1, 26, 51, 75, 100
15	SET 2 = 1, 26, 76
16	DISPLACEMENT = 2
17	STRESS = 1
18	PLØTID = NASTRAN DEMØNSTRATIØN PRØBLEM NØ. 12-1-1
19	ØUTPUT (PLØT)
20	PLØTTER SC

Card	
No.	
01	CET 1 THEILINE DAD EVELUNE ONTO DETNIE 1 THEIL O 11 THEIL 20
21	SET I INCLUDE BAK EXCLUDE GRID PDINIS I INKU 9,11 INKU 20
22	22 IMRU 30,32 IMRU 40,42 IMRU 50,52 IMRU 60,62 IMRU 70
23	72 IMRU 80,82 IMRU 90,92 IMRU 100
24	MAXIMUM DEFORMATION 2.0
25	SIERES PROJECTION
26	FIND SCALE, DRIGIN 100, VANIAGE PDINI, SEI I
2/	PLUS SEL I, UKIGIN 100, SHAPE, LABELS
28	PLDT TRANSIENT DEFURMATION I, TIME JUIZ, U.U.3,
29	MAXIMUM DEFORMATION U./O, SET I, DRIGIN TOU, SMARE
30	PRIHOGRAPHIC PROJECTION
31	FIND SCALE, BRIGIN I, REGION 0.0,0,0,1,0,0,5
32	FINU SCALE, PORIGIN Z, REGIPT U.U.U.S.I.U.I.U
33	PLDI IKANSIENI UEPDKMAIIDMI, IIME .UIZ, .UIO
34	MAXIMUM UEFUKMALIUN I.U,
30	SELL, DRADATE,
30	SEI I, BRIGIN Z, VEUIDR Z
37	DITDUT / VANT)
30	
39	CAMEDA - 2
40	CATERA - J CVID DETUEEN EDAMES - 1
41	JAIP DEIWEEN FRANKES - I
42	YADIN LINES - YES
43 AA	$\frac{1}{10}$
45	
45	YVAI HE PRINT SKIP = 1
47	VVALUE PRINT SKIP = 1
48	XTITLE = TIME (SECONDS)
49	YTITIE = DISP * INCH *
50	TCURVE = * * * * * * * G R I D 5] * * * * * * * * *
51	XPLOT, XYPRINT, DISP RESP / 51 51 (T3)
52	TCURVE = * * * * * * * G R I D 1 O 1 * * * * * * *
53	XYPLOT XYPRINT DISP RESP / 101(T3)
54	YTITLE = ACCELERATION
55	TCURVE = * * * * * * G R I D 5] * * * * * * * * * * *
56	YXPLOT, XYPRINT, ACCE RESP / 51(T3)
57	TCURVE = * * * * * * G R I D 1 0 1 * * * * * * * *
59	YVDIAT YVDDINT ACCE DESP / 101(T3)

58 XYPLØT,XYPRI 59 BEGIN BULK

3

	1	2	3	4	5	6	7	8	9	10
60 I	BARØR CBAR	1	17	1	2	10.0	.0	100.0	1	
62 (63 (64)	CBAR CØNM2 +M1	100 20	17 1	100	101 10.0					+M]
65 66	DAREA	1 2	101 INV	3 .0	100. 1500.	5	6			+EG
67 68 69	+EG GRDSET GRID	MASS 1		.00	.00	.00		1246		
70 71 72	GRID MAT1 +MAT1	101 1 111.11	10. 4+ 6	20.00 4.+6	.00	.00 .2523-3				+MAT1
73 74 75	MITI +201 PARAM	53 98 GRDPNT	2 99 0	3 100	4	5	6	7	8	+100

19.9-22

.

EXAMPLES OF DYNAMIC ANALYSES

	1	2	3	4	5	6	7	8	9	10
77 78 79 80 81 82 83 83	PBAR +PBAR SUPØRT TABDMP1 +TD11 TLØAD2 TSTEP ENDDATA	17 1.111 15 10.0 516 516	1 -1.111 3 .01 1 104	1. 1 100. .00139	.083 5 .1 1	.083 3000. .0	.1 .1	ENDT 60.		+PBAR +TDI1

19.9.7.2 Notes on Data Cards

- Selects the modal approach to Transient Analysis without looping and without modal acceleration computations.
- 9-11. Selects dynamic load function, frequency dependent equivalent coefficients for structural damping (TM 9.3.4) and integration time step intervals from Bulk Data.
 - 12. Selects method for real eigenvalue extraction.
- 13-17. Specifies desired output for displacements and element stresses.
- 18-20. Sets up for structural plotting.
- 21-23. Defines elements and grid points to be plotted.
 - 24. Specifies the maximum deformation to be plotted on a scale of 2.0 relative to the length of the beam (I 20.0). Note, must come before the FIND card.
 - 25. Requests the following plots be in stereo.
 - 26. Sets up plot parameters.
 - 27. Plot element set 1 as undeformed structure.
 - 28. Plot transient deformations in the time range from 0.012 to 0.013 without the underlay of the undeformed model. Note that only one subcase is implied even though no subcase delimiter is present. Also, the maximum deformation will be plotted to 76 percent of originally defined scale for all mode shapes.
 - 30. Sets up now for orthographic projection.
 - 31. Sets parameters for a plot on the lower half page.
 - 32. Sets parameters for a plot on the upper half page.
- 33-36. Requests two plots of transient response within the time range of 0.012 to 0.016. The deformed shape is plotted in the lower half of the plot page and the z-component vectors are plotted in the upper half of the plot page.
- 38-41. Sets up for XY curve plotting.
- 42-43. Requests plots of grid lines both in X and Y directions.
- 44-45. Specifies ten intervals in both X and Y directions.
- 46-47. Requests values be plotted for every other tic mark.
- 48-50. Axes and curve labeling informations.
 - 51. Requests both plotted and printed output of the z-displacement response at grid point 51.
- 52-53. New curve title and same output request for grid point 101.

 $C \cdot s$

- 54-58. New axis and curve label information and a request for both plotted and printed output of the z-acceleration response at grid points 51 and 101.
- 60-62. Beam default parameters and element connectivities.
- 63-64. Concentrated mass inertial properties.

19.9-23

- 65. Dynamic load factor and point of application.
- 66-67. Specified parameters for real eigenvalue extraction.
- 68-70. Gridpoint location and default single point constraints.
- 71-72. Material properties.

يني المحيد ما يتحدث المحدث المح

- 73-74. Specifies degrees of freedom to be omitted by Guyan reduction.
- 75-76. Requests weight and balance output and specified first six fundamental modes be used as modal coordinates.
- 77-78. Beam properties.
 - 79. Provides fictitious components for computing determinate reactions for constraining free body acceleration.
- 80-81. Defines frequency dependent coefficients for equivalent structural damping to be selected from Case Control (SDAMP).
 - 82. Specifies transiant response dynamic load.
 - 83. Specified integration time step intervals and the interval for which output is generated.

20.1 INTRODUCTION

-

Ĭ

Substructuring is an analytical technique used to facilitate the solution of structural problems by subdividing the structural models into smaller, more manageable components. The most elementary component, or <u>basic</u> substructure, is modeled separately just as any finite element model would be. These basic substructures are combined to build more complex substructures which, in turn, can be progressibely combined with other substructures in stages to eventually arrive at the final desired solution model. Once the solution model is analyzed, the results at each stage of the combination process may be recovered until, ultimately, the detailed solution data are recovered for each of the original basic substructures. In effect, substructuring is an extension of basic finite element theory itself whereby the usual simple beam, plate, and solid elements are replaced by basic substructures which themselves may be viewed as components of even more complex substructure.

To provide maximum program flexibility, both manual and automated approaches to substructuring are available. The manual approach requires user-generated DMAP alters and car be used in all Rigid Formats except Piecewise Linear Analysis. The procedures for single-stage manual substructuring are discussed and illustrated with a complete and fully annotated example of the input in Section 1.10 of the User's Manual.

The following discussion is concerned entirely with automated multi-stage substructuring, which provides the following features:

- 1. Simple commands to control execution and data recovery at all stages of analysis.
- 2. Automatically generated DMAP alters.
- 3. Automated procedures to control and maintain the extensive data files required.
- 4. Data storage on single direct access file.
- 5. Data transfer between IBM, CDC, or UNIVAC computers at any stage in the analysis.
- 6. No restrictions on grid point and element numbering.
- 7. Modeling only one of two or more identical substructure components.
- 8. Substructure reduction to model coordinates (modal synthesis).

To effectively employ this automated substructuring capability of NASTRAN for static and dynamic analyses, the user should gain an overall understanding of the basic program design concepts, the data base on which it operates, and the control functions provided. These topics are discussed in the sections which follow.

20.1-1

SUBSTRUCTURING AND MODAL SYNTHESIS

Complete examples of substructuring analysis with listings and explanations of input data cards are included in Section 20.19. The new user would benefit greatly by executing these examples as a first test of the automated substructuring system.

A complete description of all aspects of automated substructuring is presented in the User's Manual, Section 1.10.2. In that section, many details of using the SDF, interpreting substructuring diagnostic printout, etc., are presented and thus not presented here in the User's Guide. A detailed description of the substructuring control cards and a summary of pertinent bulk data cards is provided in Section 2.7 of the User's Manual. A detailed description of each of these bulk data cards is included alphabetically along with all other bulk data cards in the User's Manual, Section 2.4. The basic theory of automated substructuring is presented in Section 4.6 of the Theoretical Manual. The special transformations for modal synthesis are developed in Section 18 of the Theoretical Manual.

It should be noted that cyclic symmetry is available as an alternate formulation for substructuring structures with rotational or dihedral symmetry. This capability is described in the User's Guide, Section 23.

3

BASIC CONCEPTS

20.2 BASIC CONCEPTS

Automated substructuring analysis is available for use with NASTRAN Rigid Formats 1, 2, 3, 8, and 9. This provides capability for static analysis, static analysis with inertial relief for unsupported structures, normal modes, frequency response, and transient response analyses. The capability allows an unlimited number of substructures to be combined and/or reduced in any sequence desired. Each substructure is represented by its mass, stiffness, and damping matrices. A reduction in size or condensation of these matrices is accomplished using the Guyan reduction technique or reduction to normal or complex modal coordinates.

Although the NASTRAN substructuring system may be very effectively used for small and moderate size problems, several features were designed and implemented to accommodate very large problems. The most important of these features is the automated data base management system used to maintain the Substructure Jperating File (SØF) on which all pertinent matrix and substructural loading data and associated control files are stored. This SØF carries all the information needed from run to run throughout a substructuring analysis.

Processing of a substructuring analysis is subdivided into three phases:

Phase 1: Initial generation of individual basic matrices.

- Phase 2: Substructure matrix reduction and assembly, solution of the assembled substructure, and recovery of substructure displacements and reaction forces.
- Phase 3: Completion of the analysis with conventional selective output for each individual basic substructure.

A NASTRAN substructuring run generally proceeds in the following manner: First, several separate Phase 1 executions are performed, one for each basic substructure. Second, one or more Phase 2 executions may be performed. In a Phase 2 run, any number of substructure reductions and/or combinations, resulting in higher level (more complex) pseudostructures, may be performed. Phase 2 processing may be halted at any stage of the model assembly and then continued in a subsequent Phase 2 execution. The results at each step in the operation are stored in the SØF so as to be available for subsequent execution. The final steps of a Phase 2 operation would be the solution step for the highest level structure and the data recovery steps with limited output capability (displacements, forces of constraint, modal energies, and applied loads only) for any lower level substructure. Complete and detailed

20.2-1

ORIGINAL PAGE IS OF POOR QUALITY

SUBSTRUCTURING AND MODAL SYNTHESIS

a survey how to a

المرد بالمرجوع المرجوع والمعرف والمرجوع

data recovery for the basic substructures must be obtained by separate Phase 3 executions, one for each basic substructure. This level of data recovery may include any or all of the NASTRAN output normal for a non-substructure snalysis.

All Phase 2 operations except the solution step may also be performed in a Phase 1 execution. However, the user must remember that the first step of a Phase 1 execution must involve the creation of a new basic substructure.

Automated substructuring allows each basic substructure to be defined independently. This concept is represented by three key features of the system.

- 1. There are no restrictions as to duplication of grid point or element identification numbers, load sets, individual coordinate systems, etc. All data for a given substructure is associated with an assigned unique name for that structure. The only data restriction is one of proper modeling, i.e., common boundaries require grid points to be located in a consistent manner relative to one another for each connecting substructure.
- 2. No substructure may appear as a component of another substructure more than ence: and no degrees of freedom within a substructure may be connected ("combined") to other degrees of freedom in that same substructure except by multipoint constraints imposed at the solution step operation or in Phase 1 creation of the substructure.
- 3. All pertinent substructure data are stored on the SØF, an expandable direct access file. This file may be selectively edited and/or dumped to tape and transmitted to another user who may have need of the data. Provision is made for automated tape conversion among CDC, IBM, and UNIVAC computers to facilitate such data transmittal between different users. Use of this file is described in Section 1.10.2 of the User's Manual.

Control of the automated substructuring system is obtained through the use of linguistic commands, similar to those of Case Control. These commands are placed in the Substructure Control Deck shown in Figure 1. This Substructure Control Deck is input between the Executive Control and Case Control Decks.

Each substructure control command is sutomatically translated into appropriate DMAP ALTER cards to augment the requested Rigid Format sequence. The user may also include his own DMAP

20.2-2



and a second a special of the state state of the state of the second state of the second state of the second st

in the second second

. . . .

1

Figure 1. Substructuring Input Data Deck.

20.2-3

SUBSTRUCTURING AND MCDAL SYNTHESIS

· · · · · · ·

and a start start was a start of the start starting and a start of the start start start start start of the sta

ALTER commands, or he may modify a previously defined DMAP sequence. A description of how the user may interface with this NASTRAN-generated substructuring DMAP is presented in Section 2.7.2 of the User's Manual. Listings of the DMAP ALTER's generated by each substructure command are presented in Section 5.9 of the User's Manual. Descriptions of the corresponding modules provided for substructuring are found in the NASTRAN Programmer's Manual.
SUMMARY OF OPERATIONAL COMMANDS

20.3 SUMMARY OF OPERATIONAL COMMANDS

User control of the automated multi-stage substructuring system is obtained via the Substructure Control Deck commands. The key terms used to describe these commands and their functions are defined in Table 1. A summary of the substructuring command options is presented in Table 2. Some of these commands require specific bulk data cards which are listed for each reference in Table 3. The user should also refer to Section 2.7 of the User's Manual for a complete description of the Substructure Control Deck commands and to Section 2.3 for detailed descriptions of the corresponding bulk data cards.

20.3.1 The Substructure Analysis Tree

The operation and control functions of automated substructuring analysis are best illustrated and explained using the "tree" structure presented in Figure 1. This figure defines the geneology of all the component substructures used in building the final model. Basic substructures are created at the Phase 1 level. Substructures "A," "B," and "E" are shown in solid boxes indicating they were formed from actual data deck submittals and are physically different models. The dotted boxes are called "image" substructures and are the result of the EQUIVALENCE operation, described next.

20.3.2 Equivalence Operation

To reduce the potential for input error and to simplify the bookkeeping tasks, all specific references to loadings and grid points for connections, boundary sets, constraints, etc., are made with respect to the <u>basic</u> substructure name only. No component substructure name may be used more than once while building the solution structure. If the same component substructure is to be used more than once, the EQUIVALENCE operations should be used to assigne unique names to all substructures comprising that component. The dotted boxes in Figure 1 are called "image" substructures and are the result of an EQUIVALENCE operation rather than an actual Phase 1 data deck submittal. The EQUIVALENCE operation defines a new substructure which is a duplicate of an existing substructure and automatically creates all equivalent lower level component substructures which are used to cre. a higher level, or more complex substructure. Thus, space is saved on the data files by eliminating storage of redundant matrix data. A four-bladed propeller, for example, could be seen to consist of four identical components and, hence, only one need

SUBSTRUCTURING AND MODAL SYNTHESIS

A

W. Bert & St. St. St. St.





i er

20.3-2

SUMMARY OF OPERATIONAL COMMANDS

be explicitly modeled. The other three blades could be defined solely by using the EQUIVA-LENCE command.

The image substructures exist in name only. Note in Figure 1 that the names of the image structures are identical to the equivalent parent structure, with the exception of a prefix character. The new names would be created automatically by NASTRAN with the use of the PREFIX subcommand to EQUIVALENCE. These new prefixed names would then be used to reference the appro0 priate component substructure as if it were created independently.

From the user point of view, <u>all</u> substructures shown in Figure 1, with either solid or dotted boxes, are separate and distinct substructures. They may have different applied loads, boundary conditions, and responses. For example, though only A, B and E represent actual Phase 1 executions, Phase 3 data recovery executions may be made for A, B, E, XA, XB, YA, YB, YXA, YXB, and YE, each of which generally would have different results.

Substructure names are allowed no more than eight alphanumeric characters. Notice in the EQUIVALENCE operation shown in Figure 1, the required subcommand PREFIX generates an additional character which is placed ahead of the existing name as a prefix to the parent substructure name. Care must be taken with successive EQUIV operations to monitor the growth of image substructrue names so as not to exceed the eight-character limit. If the limit is exceeded, the right-most character will be truncated. Therefore, it is possible to inadvertently create duplicate substructure names as more prefixes are added. It is recommended, therefore, that the entire tree structure for the analysis be prepared ahead of time to help avoid these problems. This preplanning will also be an invaluable aid to the task of data preparation and proper sequencing of the individual steps in the analysis.

20.3.3 Combine Operation

The C@MBINE command (see Table 2) with its numerous subcommands, offers flexibility in the assembly of substructures into a higher level substructure. The C@MBINE capability allows component substructures to be translated, rotated, and/or symmetrically transformed via mirror image transformation for proper positioning in space.

For example, the right wing of an aircraft is first modeled and an EQUIVALENCE operation is performed to define an identical duplicate wing. Then, in the CØMBINE operation, a SYMTRANSFØRM is applied so that the wing now appear as the actual left wing (a mirror image of the right wing), and a TRANSFØRM is applied to properly position it on the left side of the aircraft. Caution is advised in that the symmetry transformation (SYMTRAN) is <u>always</u> applied to the component in its own basic coordinate system before the usual translation and rotation (TRANS) for final positioning. The mathematical implications of these commands are discussed in Section 4.6 of the Theoretical Manual.

Substructures which may change due to design iterations should be combined with other structures as late in the sequence of CØMBINE operations as possible. This will minimize the cost of creating a new solution structure. Also, if the design iteration changes are minor and their impact on other substructures in the model can be neglected, then RECØVER operations (described later) need be performed only from the lowest level of substructure affected by the changes. Frequently, these design changes can be evaluated using only the Phase 3 recovery calculation. Of course, care must be taken to maintain compatibility with the degree of freedom list defining the solution displacement vector. That is, the boundary grid points and connections should not be changed.

Careful ordering of the list of substructure names for any COMBINE command is advised for the purposes of controlling the bandwidth of the resulting matrices. For example, the commands

> COMBINE A, B, C, D NAME = ABCD

yield substructure ABCD with matrix row and column sequences defined by:

A INTERIOR AB BOUNDARY

SUMMARY OF OPERATIONAL COMMANDS

С	INTERIOR
AC	BOUNDARY
BC	BOUNDARY
ABC	BOUNDARY
С	INTERIOR
etc.	

Thus, control of matrix bandwidth is possible with the CØMBINE command just as it is at the component level with grid point sequencing.

Another factor to consider in ordering the substructure names in the CØMBINE command is the sum of matrix multiply times required for execution. To minimize the time required for the CØMBINE matrix multiply operations the list of substructures, A, B, C, etc., should be smallest model to largest model (using matrix size as the criterion).

20.3.4 Reduce Operation

The REDUCE command causes a Guyan reduction to be performed on an existing substructure. The user specifies which degrees of freedom are to be retained using the BDYC and BDYS (or BDYS1) bulk data cards provided. The degrees of freedom retained are all called boundary degrees of freedom although they need not ever appear on the boundary with another substructure. Obviously, all degrees of freedom eventually needed for boundary connections must be retained, i.e., they must not be reduced out. However, care must be taken to retain in this boundary set all the appropriate degrees of freedom needed to represent the dominant displacement patterns for accurate calculation of eigenvalues and eigenvectors for normal modes analyses.

20.3.5 MREDUCE or CREDUCE Operation

The MREDUCE and CREDUCE commands provide a modal synthesis capability to automated multistage substructuring. With these commands the user defines boundary degrees of freedom to identify degrees of freedom retained as physical coordinates. The remaining degrees of freedom are replaced by a smaller set of normal (MREDUCE) or complex (CREDUCE) generalized modal coordinates. MREDUCE may be used when real symmetric mass and stiffness matrices are used to define the model. CREDUCE provides a general model reduction capability when damped modes are desired or complex or unsymmetric matrices are present.

s. 1. S. Landerson (Service Construction)

30.00

The user may also define constraints for the structure to be applied only for the purpose of calculating the modes. BDYC and BDYS (or BDYS1) bulk data cards are used to define these degrees of freedom and are requested by the subcommand FIXED.

Note that for both the REDUCE and MREDUCE substructure commands, the damping matrices, B and K4, and the load vectors, P, are transformed to the reduced set of coordinates. The reduced substructures may be processed with any of the other substructure operations. However, substructures generated with the complex modal reduction, CREDUCE, may not be processed with any commands requiring real arithmetic, namely REDUCE, MREDUCE, or SØLVE with Rigid Formats 1, 2, 3, or 9.

20.3.6 Solve Operation

As many EQUIVALENCE, CØMBINE, REDUCE, MREDUCE, or CREDUCE commands as desired may be used in one or more Phase 1 or Phase 2 executions. However, only one SØLVE command is allowed in any single Phase 2 execution, and the SØLVE command is not allowed in Phase 1 executions. As indicated in the definitions of Table 1, the SØLVE command requests a solution for structural response to applied static loads (Rigid Formats 1 and 2), the calculation of normal modes (Rigid Format 3), or structural response to frequency dependent or time loads (Rigid Formats 8 and 9) of the substructure named in the command.

20.3.7 Recover Operations

The RECØVER command is used to recover the solution data for successively lower level substructures. Only the displacements, forces of constraint, modal energies, and applied loads can be selectively output for any component substructure with this command. The BRECØVER command is then used in a Phase 3 execution to obtain all the detail response output normally provided by NASTRAN for each desired <u>basic</u> substructure. The command, MRECØVER, is used to recover mode shape data for modal reduced substructures.

20.3.8 Mot Operations

Capability for creating structural model plots is provided in substructuring in either Phase 1, 2, or 3. Undeformed plots may be developed in Phase 1 or 2, and deformed structural plots of individual substructures may be generated in Phase 3. XY response plots may be generated in either Phase 2 or 3.

20.3-6

Original page is Of poor quality

SUMMARY OF OPERATIONAL COMMANDS

madel for a 200 parent de construire en encore

Phase 1 plotting may be performed in two ways. First, all structural plot requests in the Case Control Deck will be executed for the basic substructure model data submitted. Second, any Substructure Control Deck commands of the form

PLØT name

will cause all Case Control Deck plot requests to be executed for the named substructure. Plot executed for the named substructure. Plot data are generated in this case only if the SAVEPLØT = N command was used in the Phase 1 executions for those basic substructures which form the named substructure. The SAVEPLØT = N command causes plot set N to be saved for that substructure. Note, only one plot set may be saved for any substructure.

Undeformed structural plots are generated in Phase 2 under the following conditions:

- The SAVEPLØT command was used in the Phase 1 runs and plot sets were defined in the Case Control Deck for those runs.
- 2. The PLOT command is used in the Phase 2 Substructure Control Dack.
- 3. Case Control Deck cards for structural plots are present.

xy response plots may be generated in Phase 2 under the following conditions:

- 1. The Phase 2 run must include a RECØVER operation (normally follows a SØLVE command).
- 2. The Rigid Format must be either 8 or 9
- 3. The Case Control Deck XY plot requests must reference only the scalar degrees of freedom associated with the SØLVEd structure.

Structural plots and XY plots are generated in Phase 3 runs using Case Control Deck cards in the same way as for non-substructuring analyses. All output requested with these Case Control Deck cards is associated with the particular substructure involved in the Phase 3 run. No Substructure Control Deck PLØT commands are used in Phase 3.

Examples of plotting with substructure analysis are presented in Section 20.4 following.

At a second second

Table 1. Definitions of Substructure Terminology

Basic Substructure	- A structure formulated from finite elements in Phase 1.
Boundary Set	- Set of degrees of freedom to be retained in a reduce operation.
Combine Operation	- Merge two or more structures by connecting related degrees of freedom. The matrix elements for connected degrees of freedom are added to pro- duce the combined structure matrices, and the substructure load vectors are processed and stored for subsequent combination at solution time.
Component Substructure	 Any basic or pseudostructure comprising a part of an assembled sub- structure.
Connection Set	 Set of grid points and their component degrees of freedom to be con- nected in adjoining structures.
Equivalence Operation	 The creation of a secondary substructure equivalent to a primary substructure. Also creates image substructures back to the basic substructure level.
Image Substructure	 A substructure equivalent to an existing component substructure. May have different applied loads and/or solution vectors but had identical stiffness and mass matrices. Image substructures are automatically created as a result of an equivalence operation.
Phase (1, 2, or 3)	 Basic steps required for multi-stage substructure processing with NASTRAN - creation, combination, reduction, solution and recovery, and data recovery.
Primary Substructure	 Any basic substructure or any substructure resulting from a combine or reduce operation.
Pseudostructure	- A combination of component substructures.
Reduce Operation	 Structural matrix and load vector Guyan or modal reduction process to obtain smaller matrices.
Secondary Substructure	- A substructure created from an equivalence operation.
SØF	 Substructure Operating File. Contains all data necessary to define a structure at any stage, including solutions.
Solution Structure	- The resulting substructure to be used in the solve operation.
Solve Operation	 To obtain solutions using the present structural matrices and user- defined input data.

SUMMARY OF OPERATIONAL COMMANDS

and a start and a second and a second and a second s

Table 2. Summary of Substructure Commands

. . .

•

A. Phase and Mode Control

# SUBSTRUCTURE NAME*	- Defines execution phase (1, 2, or 3) - Specifies Phase 1 or Phase 3 substructure name (not used in Phase 2)
	- Requests plot data be saved in Phase 1
SAVELEN	Definer matrix options (K. B. K4, M. P. or PA)
P PT I P NS	- Der mes mach in operand (nor ca DerGa, STEP)
RUN	- Limits mode of execution (DRT, op, DRTop, Star)
# ENDSUBS	- Terminates Substructure Control Deck

B. SOF Controls

50

1

-

ì

SØF	- Assigns physical file for storage of the Spr
PASSWORD*	- Protects and ensures access to correct file
CARANT OF SAFIN	- Copies SØF data to or from an external file
DACTTIAN	- Specifies initial position of input file
NAMES	- Specifies substructure name used for input
ITEMS	- Specifies data items to be copied in or out
SOFPRINT	- Prints selected items from the SØF
DUMP	- Dumps entire SØF to a backup file
RESTORE	- Restores entire SØF from a previous DUMP operation
CHECK	- Checks Cuntents of external file created by SØFØUT
DELETE	- Deletes out selected groups of items from the SØF
	- Edits out selected groups of items from the SØF
DESTRAY	- Destroys all data for a named substructure and all the substructures
	which it is a component

C. Substructure Operations

COMBINE	- Combines sets of substructures
NAME*	- Names the resulting substructure •
TOLERANCE*	- Limits distance between automatically connected grids
CONNECT	 Defines sets for manually connected grids and releases
BUTPUT	- Specifies optional output results
COMPONENT	- Identifies component substructure for special processing
TRANSFORM	 Defines transformations for named component substructures
SYMTRANSFORM	- Specifies symmetry transformation
SEARCH	- Limits search for automatic connects
EQUIV	- Creates a new equivalent substructure
PREFIX*	 Prefix to remame equivalenced lower level substructures

Mandatory Control Cards

* Required Subcommand

1

.

Table 2. Summary of Substructure Commands (continued)

C. <u>Substructure Operations (continued)</u>

÷*

.

REDUCE	- Reduces substructure matrices
NAME*	- Names the resulting substructure
BØUNDARY*	- Defines set of retained degrees of freedom
RSAVE	 Indicates the decomposition product of the interior point stiffness matrix is to be saved on the SØF
ØUTPUT	- Specifies optional output requests
MREDUCE	- Reduces substructure matrices using a normal modes transformation
NAME*	- Names the resulting substructure
BØUNDARY*	- Defines set of retained degrees of freedom
FIXED	- Defines set of constrained degrees of freedom for modes calculation
RNAME	 Specifies basic substructure to define reference point for inertia relief shapes
RGRID	 Specifies grid point in the basic substructure to define reference point for inertia relief shapes. Defaults to origin of basic sub- structure coordinate system
METHØD	- Identifies IEGR Bulk Data card
RANGE	- Identifies frequency range for retained modal coordinates
NMAX	 Identifies number of lowest frequency modes for retained modal coordinates
ØLDMØDES	- Flag to identify rerunning problem with previously computed modal data
ØLDBØUND	- Flag to identify rerunning problem with previously defined boundary set
USEMØDES	- Flag to indicate modal data have been input on bulk data
ØUTPUT	- Specifies optional output requests
RSAVE	 Indicates the decomposition product of the interior point stiffness matrix is to be stored on the SØF
CREDUCE	 Reduces substructure matrices using a complex modes transformation
NAME*	- Names the resulting substructure
BOUNDARY*	- Defines set of retained degrees of freedom
FIXED	- Defines set of constrained degrees of freedom for modes calculation
COHTEM	- Identifies EIGC Bulk Data card
RANGE	 Identifies frequency range of imaginary part of the root for retained modal coordinates
NMAX	 Identifies number of lowest frequency modes for retained modal coordinates
ØLDMØDES	- Flag to identify rerunning problem with previously computed modal data
GPARAM	- Specifies structural damping parameter
ØUTPUT	- Specifies optional output requests
RSAVE	 Indicates the decomposition product of the interior point stiffness matrix is to be stored on the SØF

* Required Subcommand

SUMMARY OF OPERATIONAL COMMANDS

• • •

Table 2. Summary of Substructure Commands (continued)

C. <u>Substructure Operations (continued)</u>

•

h

MRECOVER	- Recovers mode shape data from an MREDUCE or CREDUCE operation
SAVE	- Stores modal data on SØF
PRINT	- Stores modal data and prints data requested
SØLVE	 Initiates substructure solution (statics, normal modes, frequency response, or transient analysis)
RECOVER	- Recovers Phase 2 solution data
SAVE	- Stores solution data on SØF
PRINT	- Stores solution and prints data requested
BRECOVER	- Basic substructure data recovery, Phase 3
PLØT	- Initiates substructure undeformed plots

κ.

Table 3. Substructure Bulk Data Card Summary

Bulk Data Used by Substructure Commands REDUCE, MREDUCE, and CREDUCE ٨.

BDYC	 Combination of substructure boundary sets of retained degrees of freedom or fixed degrees of freedom for modes calculation
BDYS	- Boundary set definition
BDYSI	- Alternate boundary set definition
Bulk Data Used	by Substructure Command COMBINE
CONCT	 Specifies grid points and degrees of freedom for manually specified connectivities - will be overridden by RELES data
CONCTI	- Alternate specification of connectivities
RELES	 Specifies grid point degrees of freedom to be disconnected - overrides CØNCT and automatic connectivities
GTRAN	- Redefines the output coordinate system grid point displacement sets
TRANS	 Specifies coordinate systems for substructure and grid point transformations

C. Bulk Data Used by Substructure Command SØLVE

=; \

8.

LOADC	 Defines loading conditions for static analysis
MPCS	- Specifies multipoint constraints
SPCS	- Specifies single-point constraints
SPCS1	- Alternate specification of single-point constraints
SPCSD	 Specifies enforced displacements for single-point constraints
DAREAS	- Specifies dynamic loadings
DELAYS	- Specifies time delays for dynamic loads
DPHASES	- Specifies phase lead terms for dynamic loads
TICS	- Specifies transient initial conditions

ديونه والمربين بهمانها والومان وال

20.4 SUBSTRUCTURE ANALYSIS EXAMPLES

Two example problems are presented to assist the user in setting up decks for an actual analysis. The first is a structural analysis using equivalent substructures, CDMBINE and REDUCE operations, and substructure plotting. The second example illustrates the use of modal synthesis techniques in automated substructuring.

20.4.1 Problem 1 - Substructure Analysis Using Rigid Format 1 and 2

Figure 1 shows two basic substructures, Table and LEGS. Note that these structures have different basic coordinate systems, as shown in the figure. Figure 1 shows a combined structure which is assembled from the two basic substructures. All of the data decks used to generate and analyze the composite structure are listed in Tables 1-4. These include the data for the Phase 1 generation of the basic substructures, the Phase 2 assembly of the complete structure, its solution and data recovery, and the Phase 3 data recovery. Tables 5-8 present a detailed description of each of the data decks used in this analysis.

20.4.2 Problem 2 - Dynamic Analysis of a Truss Using Automated Modal Synthesis and Rigid Format 9

20.4.2.1 Description

This problem illustrates the automated substructuring and modal synthesis procedures which provide accurate, efficient solutions in dynamic analysis. Each component substructure is reduced to modal and boundary degrees of freedom prior to the substructure combine operation. The combination structure, formulated in terms of the component modes, is also reduced to modal degrees of freedom for solution by the transient analysis Rigid Format.

20.4.2.2 Input

The geometry for the two basic substructures is shown in Figure 2. All members are RBD elements. All grid points are constrained to include only in-plane displacements. The basic input data and the substructure control data are described below.

and a second second second

strates in concerns.



a. Phase 1 Basic Substructures.



b. Phase 2 Combined Substructure.

Figure 1. Example Problem 1.



Boundary Points

¢

1.44.



Figure 2. Nine-Cell Truss Basic Substructures.

Manufacture Conversion

*

20.4-3

1. Parameters

a = 4 0	(typical frame width)
h = 30	(typical frame height)
A = 0.3	(cross section of members)
$E = 10^{6}$	(Young's Modulus)
$P = 2.5 \times 10^{-3}$	(density)

a carrier a statement of the statement of the statement of the statement of the statement of the statement of the

2. Constraints

 $u_z = \theta_x = \theta_y = \theta_z = 0$ all points (Boundary conditions are applied only during solution.)

3. Loads

 $P_{y42} = 10^3$ (load on substructure BBASIC)

4. Transient Loads

 $U_{y42} = .143 @ t = 0$ (initial condition) $P_{y42} = 10^3 0 < t < .12$ (load history) 0 t > .12

The basic run sequence and substructure operations are shown in Figure 3. Each box defines a component substructure. Each trapezoid defines output data resulting from a substructure operation. Four separate runs are performed as described below.

- <u>Runs 1 & 2</u> The NASTRAN Phase 1 operations formulate the finite element matrices and basic static loads.
- 2. <u>Run 3</u> In Phase 2 the basic substructures are reduced to modal coordinates and combined together. Another modal synthesis reduction is performed on the combination and the resulting eigenvectors are printed.
- 3. <u>Run 4</u> In the second Phase 2 operation, the transient analysis is performed on the reduced structure using the SØLVE operation. The transient output data is transformed back to the original grid point definitions with the RECØVER command.
- 4. Listings of the four runs are presented in Tables 9-12.

ton-

and the second sec

٦



Figure 3. Substructure Formulation Tree and Solution Sequence.

. ج

20.4-5

.

× .

20.4.2.3 Results

è

and the second second second and the second second second second second second second second second second second

For assessing the accuracy of the modal synthesis, the whole structure was run in Rigid Format 3 to determine natural frequency and mode shapes. This procedure eliminates the effects of finite element errors common to both methods.

Natural frequencies for the combined system are shown in Table 13 along with the error ratios of the difference. Note that the lowest frequency component mode omitted from the analysis was 197.2 Hz. Below this frequency, the resulting modes are excellent.

The transient response of the resulting structure is illustrated in Figure 4 for a typical point in the system. Note that very little excitation of the higher frequency modes occurred.

ار داری این در ده . .





And the second second

20.4-7

•

.

Table 1. Phase 1 Data Deck for Subroutine TABLE

and the second second second second second second second second second second second second second second second

Card No. 0 NASTRAN FILES = NPTP 1 ID TABLE RASIC

Same Party and

المراجع والمحمد المراجع والمراجع المراجع المراجع المراجع المراجع والمراجع والمراجع والمراجع والمراجع والمراجع

1 2 3 4 5 6	ID TABLE APP DISE SOL 2,0 TIME 1 CHKPNT CEND	E,BASIC P,SUBS YES								
7 8 9 10 11 12 13	SUBSTRUCTURE PHASE1 PASSWØRD=PRØJECTX SØF(1)=SØF1,250,NEW NAME=TABLE SAVEPLØT=1 SØFPRINT TØC ENDSUBS									
14 15 16 17 18 19	TITLE=T/ LØAD=2 ØUTPUT(1 SET 1=AI PLØT BEGIN BI	ABLE, PHA PLØT) LL ULK 2	SE ØNE 3	4	5	6	7	8	9	10
20 21 22 23 24 25 26 27 28 29 30 31 32 33 33 33 35	CCUAD2 CTRIA2 CTRIA2 FØRCE FØRCE GRID GRID GRID GRID GRID GRID GRID GRID	3 1 2 2 2 1 2 3 4 5 6 7 1 2 1	2 1 3 4 3.+7 1	5 1 3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	6 2 4 10.0 10.0 0.0 7. 0.0 7. 0.0 7. 3	4 4 1 -1.0 -1.0 5. 5. 0.0 0.0 -5. -5. 4.3	3	123456	-	

Table 2. Phase 1 Data Deck for Substructure LEGS

Card No.										
0 1 2 3 4 5 6	NASTRAN 1D LEGS, APP DISP SØL 2,0 TIME 1 CHKPNT Y CEND	FILES=(NF BASIC , SUBS ES	PTP,INP3)							
7 8 9 10 11 12 13 14 15 16	SUBSTRUCTURE PHASE1 PASSWORD-PROJECTY SOF(1)=SOF4,7500 NAME=LEGS SAVEPLDT=1 SOFOUT INP3 POSITION=REWIND NAME=LEGS EDIT(32) LEGS ENDSUBS									
17 18 19 20 21 22	TITLE-LE LØAD-1 ØUTPUT(F SET 1 PLØT BEGIN BL	IGS PHASE PLOT) =All JLK 2	ØNE	4	5	6	7	8	9	10
23 24 25 26 27 28 29 30 31 32 33 34 35	CBAR CBAR CBAR FØRCE FØRCE GRID GRID GRID GRID GRID GRID MAT 1 PBAR ENDDATA	1 2 3 1 1 1 2 3 4 5 1 1	1 1 1 4 3.+7	1 3 4 0.0 5. 5. 0.0 100. 1.0	2 2 3 2.0 10. 10. 0.0 0.0 100. .3 50.	5 5 5 3.0 0.0 0.0 0.0 0.0 0.0 0.0 4.3 100.	.0 .0 10.	4.0 4.0 123456	2 2 2 2	

Table 3. Phase 2 Data Deck

Card No. ID SUBSTR, PHASE2 1 2 APP DISP, SUBS SPL 1.0 3 4 TIME 1 5 DIAG 23 CEND 6 SUBSTRUCTURE PHASE2 7 PASSWORD=PROJECTX 8 SØF(1)=SØF1,250 ØPTIØNS=K,M,P 9 10 SØFØUT, INP4, TAPE 11 POSITION-REWIND 12 NAME=TABLE 13 SOFPRINT TOC COMBINE LEGS, TABLE 14 15 16 NAME-SIDEA T9LER=(.001 17 PUTPUT=1,2,7,11,12,13,14,15,16,17 COMPONENT LEGS 18 19 20 TRANS=10 EQUIV SIDEA, SIDEB 21 22 23 PREFIX=B COMBINE SIDEA, SIDEB 24 NAME=BIGTABLE 25 TØLER=0.001 OUTPUT=1,2,7,11,12,13,14,15,16,17 COMPONENT SIDEB 26 27 28 SYMT=Y **REDUCE BIGTABLE** 29 NAME = SMALTABL 30 31 BOUNDARY=100 BUTPUT=1,2,3,4,5,6,7,8 32 33 SØFPRINT TØC 34 PLOT SMALTABL SOLVE SMALTABL RECOVER SMALTABL 35 36 37 PRINT BIGTABLE 38 SAVE BTABLE 39 SØFPRINT TØC 40 ENDSUBS 41 TITLE=PHASE TWO SUBSTRUCTURE 42 DISP=ALL SPCF=ALL 43 44 **ØLØAD=ALL** 45 SPC=10 SUBCASE 1 46 LGAD=10 47 48 SUBCASE 2 49 L#AD=20 OUTPUT (PLOT) 50 SET 1=ALL 51 PLØT BEGIN BULK 52 53

A REPORT OF THE REPORT OF THE REPORT OF THE REPORT OF THE REPORT OF THE REPORT OF THE REPORT OF THE REPORT OF T

·

www.gogoodaa

Table	3.	Phase	2	Data	Deck	(cont'd)
Table	3.	rnase	۲	ha ra	DECK	(

Card No.

_

E. ______

<u>nu.</u>			_	•	E	٨	7	8	9	10
54 55 56 57 58 59 60 61 62 63 64 65 66 67	1 BDYC +A BDYS1 BDYS1 BDYS1 LOADC LOADC LOADC SPCS1 SPCS1 SPCS1 SPCS1 TRANS +B ENDDATA	2 100 10 20 10 20 10 10 10 10 10 10 0.0	3 LEGS TABLE 4 123456 123456 1.0 1.0 BLEGS BTABLE LEGS TABLE 8.0	4 20 10 1 2 LEGS TABLE 123456 4 123456 4 .0 -5.0	BLEGS BTABLE 3 6 3 1 2 2 1 2 1 7.0	20 10 4 1.0 3 3 3 -5.0	5 B ^I EGS BTABLE 4 4 3.0	1 2 5 5 11.0	1.0 1.0 -5.0	+A +B

20.4-11

Table 4. Phase 3 Data Deck

Card No. 0 NASTRAN FILES = 0PTP 1 ID TABLE, BASIC 2 APP DISP, SUBS 3 SOL 1,0 4 TIME 1 5 RESTART TABLE, BASIC (Restart deck) 6 CEND 7 SUBSTRUCTURE PHASE3 8 PASSWORD=PRØJECTX 9 SØF(1)=SØF1,250 10 BRECØVER BTABLE 11 ENDSUBS 12 TITLE=PHASE THREE FØR REFLECTED TABLE 13 DISP=ALL 14 ØLØAD-ALL 15 SPCF=ALL 16 STRESS=ALL 17 SUBCASE 1 18 SUBCASE 2 LØAD=2 19 20 BEGIN BULK 21 ENDDATA

بالمرجع العاريان المترج المعود ومربون

The second second second statement of the second second second second second second second second second second

Table 5. Description of Phase 1 Data Deck for Substructure TABLE

Card

-۱

- No. Refer to Table 1 for input cards described below.
- 1-6 Standard NASTRAN Executive Control Deck except the 'SUBS' option is selected on the APP card.
 - 7 First card of Substructure Control Deck. Phase 1 is selected.
 - 8 Password protection on the SØF is 'PRØJECTX'.
 - 9 The SØF consists of one physical file with an index of one. (Indices must begin with one and increase sequentially.) The name of the file is 'SØF1' and it has a maximum size of 250,000 words. The file is to be initialized. (Internal pointers will be set to indicate that the SØF contains no data.)
- 10 The basic substructure to be generated will be identified by the name TABLE.
- 11 Plot set 1 will be saved on the SØF for performing plots of the combined structure in Phase 2.
- 12 Print a table of contents for the SØF. This includes a list of all substructures and their data items.
- 13 End of Substructure Control Deck.
- 15 Selects the load to be saved on the SØF for use in Phase 2. Note that multiple loads may be saved by using multiple subcases. In addition to external static loads, thermal loads and element deformatic to to add may be selected.
- 16-18 Plot control cards are required if the SAVEPLØT subcommand is used in the Substructure Control Deck. These cards are used to define the plot sets for Phase 2 plotting. It is not necessary that a plot tape be set up in Phase 1.
- 19-35 Standard NASTRAN Bulk Data Deck. These cards define the mathematical model of the basic substructure.

Table 6. Description of Phase 1 Data Deck for Substructure LEGS

Card

No. Refer to Table 2 for input cards described below.

and the second state of the second state of

- 1-6 Standard NASTRAN Executive Control Deck except the 'SUBS' option is selected on the APP card.
 - 7 First card of the Substructure Control Deck. Phase 1 is selected,
 - 8 Password protection on the SØF is 'PR@JECTY'.
 - 9 The SØF consists of one physical file with an index of one. (Indices must begin with one and increase sequintially.) The name of the file is 'SØF4' and it has a maximum size of 250,000 words. The file was used previously as on SØF (Phase 1 for TABLE).
- 10 The basic substructure to be generated will be identified by the name LEGS.
- 11 Plot set 1 will be saved on the SØF for performing plots of the combined structure in Phase 2.
- 12-14 After substructure LEGS has been generated and saved on the SØF, it is copied out to user tape INP3.
 - 15 All data items for substructure LEGS are removed from the SØF. (The substructure name remains in the SØF directory, however.)
 - 16 End of Substructure Control Deck
 - 18 Selects the load to be saved on the SØF for use in Phase 2. Note that multiple loads may be saved by using multiple subcases. In addition to external static loads, thermal loads, and element deformation loads may be selected.
- 19-21 Plot control cards are required if the SAVEPLØT subcommand is used in the Substructure Control Deck. These cards are used to define the plot sets for Phase 2 plotting. It is not necessary that a plot tape be set up in Phase 1.
- 22-35 Standard NASTRAN Bulk Data Deck. These cards define the mathematical model of the basic substructure.

والمراجع و

Table 7. Description of Phase 2 Data Deck

Card No. Refer to Table 3 for input cards described below.

- 1-6 Standard NASTRAN Executive Control Deck <u>except</u> the 'SUBS' option is selected on the APP card. DIAG 23 requests an echo of the automatic DMAP alters generated.
 - 7 First card of the Substructure Control Deck. Phase 2 is selected.
- 8,9 These cards specify the same SØF used in Phase 1 for substructure TABLE.
- 10 The card causes matrix operations to be performed on stiffness, mass, and load matrices. The default for Rigid Format 1 is stiffness and loads only. However, Rigid Format 2 was selected in the Phase decks. This caused all three matrix types to be generated in Phase 1.

11-13 Basic substructure TABLE is copied from the SØF to user tape INP4.

- 14 Print the SØF table of contents.
- 15-20 Perform an automatic combination of substructures TABLE and LEGS. The resultant combined pseudostructure will be named SIDEA. The tolerance for connections is 0.001 units. Detailed output is requested (see Substructure Command CØMBINE). The basic coordinate system for substructure LEGS is transformed according to transformation set 10 in the Bulk Data.
- 21,22 Create a new secondary substructure SIDEB which is equivalent to SIDEA. This operation causes image substructure BLEGS and BTABLE to be generated.
- 23-28 Perform an automatic combination of substructures SIDEA and SIDEB. The resultant combined pseudostructure will be named BIGTABLE. The tolerance for connections is 0.001 units. Detailed output is requested. The basic coordinate system for pseudostructure SIDEB is symmetrically transformed about the XZ plane, identified by Y, the axis normal to the plane (sign change for all 'Y' degrees of freedom).
- 29-32 Perform a matrix reduction on the matrices of substructure BIGTABLE. The resultant reduced pseudostructure will be named SMALTABL. The retained degrees of freedom are selected in boundary set 100 in the Bulk Data. Detailed output is requested.
 - 33 Print the SØF table of contents.
 - 34 Plot pseudostructure SMALTABL. The plot control cards in the Case Control Deck are used to define the plots.
 - 35 Perform a static solution of pseudostructure SMALTABL. The constraint sets and loads selected in the Case Control Deck are used.
- 36-38 Recover the displacements of substructures BIGTABLE and BIABLE from the solution of SMALTABLE and save them on the SØF. Also, print the results for substructure BIGTABLE. The output requests in the Case Control Deck are referenced when the PRINT subcommand is invoked.
 - 39 Print the SØF table of contents.
 - 40 End of the Substructure Control Deck.
- 42-44 Case Control output requests. Referenced by the PRINT subcommand of the RECØVER command.
- 45-49 Constraint and load set selections are referenced by the SØLVE command.
- 50-52 Plot control cards are referenced by the PLØT command.

en con

3

Table 7. Description of Phase 3 Data Deck (continued)

Card

Ì

No. Refer to Table 3 for input cards described below.

- 54-58 These Bulk Data cards define the boundary set of retained degrees of freedom which was selected in the REDUCE operation (cards 29-32).
- 59-64 These cards define the loads and constraints selected in the Case Control Deck for the substructure SØLVE operation.
- 65,66 These cards define the transformation which is applied to the basic coordinate system of substructure LEGS in the first COMBINE operation (cards 15-20).

The second second second second second second second second second second second second second second second se

Table 8. Description of Phase 3 Data Deck for Substructure BTABLE

Card

No. Refer to Table 4 for input cards described below.

. .

- 1-6 Standard NASTRAN Executive Control Deck <u>except</u> the 'SUBS' option is selected on the APP DISP card. "Card" 5 is actually the Restart deck punched out in Phase 1 for substructure TABLE.
 - 7 First card of the Substructure Control Deck. Phase 3 is selected.
- 8,9 These cards specify the same SØF used in Phase 2.
- 10 This card causes the data for the image basic substructure BTABLE to be copied from the SØF to GINØ data blocks. The data can then be used for data recovery operations, i.e., deformed structure plots, stresses, etc.
- 11 End of Substructure Control Deck.
- 13-16 Output requests for Phase 3 data recovery.
- 17-19 The Subcase Definitions in Phase 3 must be identical to those used in the SØLVE operation in Phase 2. SPC and MPC constraints in Phase 3 must be the same as those used in Phase 2. Load sets selected in Phase 3 must correspond to those selected in Phase 2 for each subcase. However, load sets selected in Phase 2 which do not exist for this particular or basic substructure can not be selected in Phase 3. See Section 1.14.4 for a more detailed discussion of the Phase 3 Case Control Deck.

ł

Table 9. Phase 1 Data Deck for Substructure ABASIC

NASTRAN NLINES=35, BUFFSIZE=500 ID DEMONT, RUNT APP DISP, SUBS SØL 2,0 TIME 1 CEND SUBSTRUCTURE PHASE1 SØF(1) = FT10,500,NEW PASSWORD = MDLSYN NAME = ABASIC SØFPRINT TØC ENDSUBS TITLE = DEMO PROBLEM 165A MODAL SYNTHESIS OF A NINE-CELL TRUSS SUBTITLE = RUN 1 - BASIC SUBSTRUCTURE A LOAD = 980 \$ 1 G ACCELERATION IN -Y DIRECTION **BEGIN BULK** CRØD CRØD CRØD CRØD CROD CRØD CRØD CRØD CRØD CRØD CRØD CRØD CRØD CRØD CROD CRØD CROD CRØD CRØD CRØD CRØD CRØD CRØD CRØD CRØD CROD CROD CROD CRØD CRØD CRØD CROD CROD CRØD 41-CROD CRØD CROD GRAV 980.0 0.0 -1,0 0,0 GRDSET GRID 0.0 -30.0 0.0 GRID 0.0 0.0 0.0 GRID 0.0 30.0 0.0 GRID 40.0 -30.0 0.0 GRID 40.0 0.0 0.0 GRID 40.0 30.0 0.0

20.4-18

and the second second second second second second second second second second second second second second second

March Indep

Table 9. Phase 1 Data Deck for Substructure ABASIC (continued)

010	21		80.0	-30.0	0.0
GKID	21		90.0	0.0	0.0
GRID	22		00.0	0.0	
GRID	23		80.0	30.0	0.0
CRID	31		120.0	-30.0	0.0
OKID	22		120 0	0.0	0.0
GRID	32		160.0	~~~~~	0.0
GRID	33		120.0	30.0	0.0
6910	Â1		160.0	-30.0	0.0
CRID	42		160.0	0.0	0.0
GRID	76		160.0	20.0	0 0
GRID	43		100.0	30.0	0.0
GRID	51		200.0	-30.0	0.0
CRID	52		200.0	0.0	0.0
OKID	6.5		200 0	30 0	0.0
GRID	23		200.0		2 5 2
MAT1	1	10.0+6		0.3	2.3-3
DDO	1	1	0.3		
CNDATA	•	•	• • •		
LINUATA					

_

-1

20.4-19

Table 10. Phase 1 Data Deck for Substructure BBASIC

NASTRAN NLINES=35,BUFFSIZE=500										
APP DI	APP DISP SIRS									
SØL 2	0									
TIME										
CEND										
SUBSTR	RUCTURE	PHASE 1								
SØF(1)) = FT10	,500								
PASSW	PASSWORD = MOLSYN									
NAME *	* RRV21(•								
	uni iputi Po									
TITLE	= DEMA	DOGRIS	1 1650 MG			NTHE CELL	TDUCC			
SUBT 11		IN 2 - !	RASIC SURS		FR	MINE-CELL	18033			
LØAD	= 980 1		CELERATIO	N IN -Y	DIRECTION					
BEGIN	BULK		,		or correct					
CRØD	1	1	1	2						
CRØD	2	1	2	3						
CRØD	11	1	11	12						
CRØD	12	1	12	13						
CRØD	21	1	21	22						
CROD	22	I	22	23						
CROD	31	1	31	32						
CROO	32 41	1	32	33						
CROD	42	1	42	42 43						
CROD	in	i	1	11						
CRØD	112	i	2	12						
CRØD	113	1	3	13						
CRØD	121	1	11	21						
CRØD	122	1	12	22						
CRØD	123	1	13	23						
CROD	131	1	21	31						
CDAD	132	1	22	32						
CROD	133	1	23	33						
CRØD	142	i	32	42						
CROD	143	i	33	43						
CRØD	211	1	2	11						
CRØD	212	1	2	13						
CRØD	221	1	12	21						
CRØD	222	j	12	23						
CROD	231	1	22	31						
	232	1	22	33						
CROD	241	i	32	41						
GRAV	980	1	980.0	43	-10	0 0				
GRDSET			300.0	0.0	-1.0	0.0	3456			
GRID	1		0.0	-30.0	0.0		0,00			
GRID	2		0,0	0.0	0.0					
GRID	3		0.0	30.0	0.0					
GRID	11		40.0	-30.0	0.0					
GRID	12		40.0	0.0	0.0					
GRID	21		40.0 20.0	30.0	0.0					
GRID	22		80,0 80,0	-30.0	0.0					
GRID	23		80.0	30.0	0.0					
GRID	31		120.0	-30.0	0.0					
GRID	32		120.0	0.0	0.0					
GRID	33		120.0	30.0	0.0					
GRID	41		160.0	-30.0	0.0					

a the second second second second second second second second second second second second second second second

 a grange and a signature of the

Table 10. Phase 1 Data Deck for Substructure BBASIC (continued)

GRID	42		160.0	-30.0	0.0
GRID	43		160.0	30.0	0.0
MATI	1	10.0+6		0.3	2.5-3
PROD	1	1	0.3		
ENDDATA					

1

γ÷

Table 11. Phase 2 Deck for COMBINE and MREDUCE Operations

NASTRAN NLINES= 35, BUFFSIZE= 500 ID DEMOM1, RUN3 APP DISP, SUBS SØL 3,0 TIME 5 DIAG 8,19 CEND SUBSTRUCTURE PHASE2 SPF(1) = FT10,500PASSWORD = MOLSYN ØPTIØNS K,M,P SØFPRINT TØC MREDUCE ABASIC NAME MRAB **BØUNDARY 5** FIXED 5 METHØD 9 ØUTPUT 1,5,6,9,10 SØFPRINT TØC MREDUCE BBASIC NAME MRBB **BØUNDARY 4** FIXED 4 METHØD 9 ØUTPUT 1,5,6,9,10 SØFPRINT TØC COMBINE MRAB, MRBB NAME MCOMB TØLERANCE 0.001 ØUTPUT 2,7,12 CØMPØNENT MRBB TRANSFORM 40 SØFPRINT TØC MREDUCE MCOMB NAME RTRUSS **BØUNDARY 42** FIXED 9 METHØD 90 NMAX 18 ØUTPUT 1,5,6,9,10 SØFPRINT TØC ENDSUBS TITLE = DEMØ PRØBLEM 165A MOØAL SYNTHESIS ØF A NINE-CELL TRUSS SUBTITLE = RUN 3 - MØDAL REDUCTIØNS AND CØMBINE ØPERATIØNS LABEL = USE 7 MODES PER COMPONENT AND 18 MODES OF COMBINATION **BEGIN BULK** BDYC **BBASI** 4 2 BDYC 5 ABASIC 1 BDYC ABASIC 2 9 BDYC 42 ABASIC 2 BBASIC 42 BDYS1 1 12 1 2 3 51 52 53 2 BDYS1 12 3 2 1 BDYS1 42 2 2 7 10000.0 GIV +E1 EIGR 9 0.0 +E1 MAX EIGR 90 GIV 0.0 10000.0 20 +E2 MAX +E2 TRANS 40 200.0 0.0 Ü.0 200.0 0.0 1.0 +T] +T1 201.0 0.0 0.0 ENDDATA

a se tha name a suite

٠

Table 12. Phase 2 Deck for SØLVE and RECØVER Operations

NASTRAN NLINES=35, BUFFSIZE=500 ID DEMONT, RUN4 APP DISP, SUBS SOL 9,0 TIME 5 CEND SUBSTRUCTURE PHASE2 SPF(1) = FT10,500PASSWORD = MOLSYN OPTIONS K,M,P SØFPRINT TØC SOLVE RTRUSS RECOVER RTRUSS PRINT ABASIC UIMPROVE RANGE 0,0,0,4 ENERGY ALL PRINT MRAB PRINT BBASIC UIMPROVE RANGE 0,0,0,4 ENERGY ALL PRINT MRBB SØFPRINT TØC ENDSUBS TITLE = DEMØ PRØBLEM 165A MØDAL SYNTHESIS ØF A NINE-CELL TRUSS SUBTITLE = RUN 4 - TRANSIENT RESPØNSE ANALYSIS LABEL . REDUCED TO MODAL DEGREES OF FREEDOM SET 1 = 7 THRU 13 SPC = 123 DLGAD = 101 IC = 522 TSTEP = 40 DISP = ALL SDISP(SØRT2) = 1 BEGIN BULK DAREAS 980 BBAS1C 2 2 1.0+3 LØADC 980 1.0 ABASIC 980 1.0 PARAM G 0.05 PARAM ₩3 0.01 SPCS1 123 ABASIC 12 1 2 3 TICS 522 BBAS1C 2 2 0.1 TLØAD2 101 980 0.4 12.0 TSTEP 40 40 2.0-2 1

and the second sec

of the second second

ENDOATA

· · · · · · · · · · · · ·

٠

•

- 20 M -

- T

j

i	Mode No.	Frequency (H _z)	% Difference From Full Model		
	ı	3.596	. 0085		
	2	17.564	. 0012		
	3	28.492	. 0639		
	4	39.507	.0011		
	5	61.099	. 0050		
	6	80.280	. 0066		
	7	84.454	. 379		
8 9		98.898	. 039		
		111.764	. 018		
10		123.348	. 345		
	11	127.556	. 548		
12		130.359	. 155		
	13	134.922	2.220		
	14	153.606	.174		
15		162.019	2.98		
	16	180.321	5.27		
17		200.702	14.89		
			1		

Table 13. NASTRAN Natural Frequencies and Differences for 20 Degrees of Freedom Mode Synthesis.
21. AEROELASTIC ANALYSIS

21.1 INTRODUCTION TO NASTRAN AEROELASTICITY

Aerodynamic analysis, like structural analysis, is based upon a finite element approach. The finite elements are strips or boxes for which there are aerodynamic forces. There are two major points to be considered. The aerodynamic elements, even for rather complex vehicles, tend to be in regular arrays. Thus, while NASTRAN has the provision to generate arrays of structural elements, it is also desirable to generate arrays of aerodynamic elements. In particular, the aerodynamic elements for lattice methods are arrays of trapezoidal boxes whose sides are parallel to the airflow. These should be described simply by defining properties of the array (panel). The grid points defining the structure usually will not coincide with the grid points defining the aerodynamic elements. Provision has been made to generate equations of constraint between the two sets of grid points. The geometry interpolation is a key feature, since it allows the choice of structural and aerodynamic elements to be based upon structural and aerodynamic considerations separately.

Aerodynamic forces are generated via the flow surrounding the structure. The theory produces a matrix defining the forces upon the structure in terms of the deflections of the structure. State-of-the-art methods which involve interactions between aerodynamic elements are available only for sinusoidal motion. Phase lags occur between the motions and the forces, thus the matrices are complex. Furthermore, these complex matrices depend upon parameters of the flow, namely reduced frequency (ratio of frequency to velocity) and Mach number (ratio of velocity to speed of sound). Such a matrix, if computed by an interaction theory such as the doublet lattice method, will be expensive to produce. The most effective method to evaluate the matrix for a large number of parameter values is to compute the matrix for a few selected ones, and then interpolate to others. This parametric interpolation is an automatic feature of the solution modules for modal aeroelastic analysis.

21.1-1

DATA INPUT AND GENERATION

a contraction and a second of the

ORIGINAL PAGE IS OF POOR QUALITY

21.2 DATA INPUT AND GENERATION

Aerodynamic elements are strips, boxes or bodies which represent the structure for computation of aerodynamic forces. These elements, like the structural elements, are defined by their geometry, and their motions are defined by degrees of freedom at associated aerodynamic grid points. Often requirements of the aerodynamic theory will dictate the geometry of the boxes. For example, the doublet lattice theory requires trapezoidal boxes with their parallel edges along the streamlines. Also, if one box lies in the wake of another, their edges must be on the same two streamlines. Aerodynamic elements and grid points will be generated to reduce the labor of the user (many less data cards will be required) and to help insure that theoretical requirements are met.

Aerodynamic calculations are made in a cartesian aerodynamic coordinate system. By the usual convention the flow is in the positive x direction. The basic coordinate system was not chosen, since it would place an undesirable restriction upon the description of the structural model. Any NASTRAN cartesian system may be specified, and flow will be defined in the direction of its positive x axis. All aerodynamic calculations are made initially in the aerodynamic coordinate system. All element and aerodynamic grid point data, computed initially in the basic coordinate system, will be converted to the aerodynamic coordinate system. The global (displacement) coordinate system of the aerodynamic grid points will have its Tl direction in the flow direction. T3 is normal to the eleuent for boxes, and parallel to the aerodynamic Z direction in the case of bodies. Coordinate system data are generated for the aerodynamic grid points.

The grid points are physically located at the centers of the boxes and body elements. Permanent constraints are generated for the unused degrees of freedom. A second set of grid points, used only for undeformed plotting, is located at the element corners. All six degrees of freedom associated with each grid point in this second set are permanently constrained. Grid point numbers are generated based upon the element identification number. For any panel, the external grid point numbers for the boxes start with the panel identification number and increase consecutively.

Aerodynamic degrees of freedom, along with the extra points, are added after the structural matrices and modes have been determined. This introduces the following displacement sets:

up Aerodynamic box and body degrees of freedom

u_{ca} Permanently constrained degrees of freedom associated with aerodynamic grid points

 u_{ns} White of u_{n} (physical) and u_{sA}

 \mathbf{u}_{DA} Union of \mathbf{u}_{k} and \mathbf{u}_{DS} (physical and aerodynamic)

The set u_{pA} replaces u_p as the set available for output at grid, scalar and extra points.

21.2-1

STRUCTURAL-AERODYNAMIC INTERCONNECTIONS

21.3 STRUCTURAL-AERODYNAMIC INTERCONNECTIONS

The interpolation between the structural and aerodynamic degrees of freedom is based upon the theory of splines (Figure 1). High aspect ratio wings, bodies, or other beamlike structures should use linear splines. Low aspect ratio wings, where the structural grid points are distributed over an area, should use surface splines. Several aplines can be used to interpolate to the boxes on a panel or elements on a body; however, each point can refer to only one spline. Any box or body element not referenced by a spline will be "fixed" and have no motion. For any point, expecially a control surface degree of freedom, a linear relationship (like an MPC) may be specified.

For all types of splines, the user must specify the structural degrees of freedom and the aerodynamic points involved. The structural points, called the g-set, can be specified by a list or by specifying a volume in space and determining all the grid points in the volume. The degrees of freedom retained at the grid points include only the normal displacements for surface splines. For linear splines, the normal displacement is always used and, by user option, torsional rotations or slopes may be included. The global transformation at structural points is automatical y applied for surface and linear splines.

The SPLINE1 data card defines a surface spline. This can interpolate for any "rectangular" subarray of boxes on a panel. For example, one spline can be used for the inboard end of a panel and another for the outboard end. The interpolated grid points (k-set) are specified by naming the lowest and highest acrodynamic grid point numbers in the area to be splines. The two methods for specifying the grid points use SET1 and SET2 data cards. A parameter DZ is used to allow some smoothing of the spline fit. If $\mathcal{L} = 0$ (the usual value), the spline will pass through all deflected grid points. If DZ > 0, then the spline (a plate) is attached to the grid deflections via springs, which produce a smoother interpolation that does not necessarily pass exactly through any of the points. The flexibility of the springs is proportional to DZ.

The SPLINE2 data card defines a linear spline. As can be seen from Figure 1, this is really a generalization of a simple spline to allow for interpolation over an area. It is similar to the method often used by aeronautical engineers who assume that an airfoil chord is rigid. The portion of a panel to be interpolated and the set of structural points are determined in the same manner as with SPLINE1. A NASTRAN coordinate system must be supplied to determine the axis of the splint. Since the spline has torsion as well as bending flexibility, the user may specify

21.3-1

AEROELASTIC ANALYSIS

the ratio of flexibilities; the default value for this ratio is 1.0. The attachment flexibilities, D_z , $D_{\theta x}$, and $D_{\theta y}$, allow for smoothing, but usually all values are taken to be zero. An exception would occur if the structural model does not have slopes defined, in which case the flexibility DTHX must be infinite; the convention DTHX = -1.0 is used in this case. When used with bodies, there is no torsion and the spline axis is along the body.

There are certain cases with splines where attachment flexibility is either required or should not be used. The following special cases should be noted.

- 1. Two or more grid points, when projected onto the plane of the element (or the axis of a body) may have the same location. To avoid a singular interpolation matrix, a positive attachment flexibility must be used.
- With linear splines, three deflections with the same spline y-coordinate would overdetermine the interpolated deflections since the perpendicular arms are rigid. A positive DZ is needed to make the interpolation matrix nonsingular.
- 3. With linear splines, two slopes (or twists) at the same y-coordinate would lead to a singular interpolation matrix. Use DTHX > 0 (or DTHY > 0) to allow interpolation.
- 4. For some modeling techniques, i.e., those which use only displacement degrees of freedom, the rotations of the structural model are constrained to zero to avoid matrix singularities. If a linear spline is used, the rotational constraints should not be enforced to these zero values. When used for panels, negative values of DTHX will disconnect the slope, and negative values of DTHY will disconnect the twist. For bodies, DTHY constrains the slopes since there is no twist degree of freedom for body interpolation. For a linear spline, if all of the structural points lie on a straight line, the use of infinite (negative DTHX or DTHY) rotational flexibility results in a kinematically unstable idealization.

For linear splines used with wings, the parameter DTØR should be selected as a representative 'ue of EI/GJ.

21.3-2

STRUCTURAL-AERODYNAMIC INTERCONNECTIONS



OF POOR QUALITY

Figure 1. Splines and their coordinate systems.

AERODYNAMIC MODELING

21.4 AERODYNAMIC MODELING

Aerodynamic elements define the interaction between the structure and an airflow. Since the elements usually occur in regular arrays, the connection cards are designed to specify arrays. The grid points associated with the elements in an array are generated within the program. Spline methods are used to interpolate for aerodynamic grid point deflection in terms of structural points.

For every aerodynamic problem, basic parameters are specified on the AER® bulk data card. A rectangular aerodynamic coordinate system must be identified. The flow is in the positive X-direction in this system. The use of symmetry (or antisymmetry) is recommended to analyze symmetric structures, to simulate ground effects, or to simulate wind tunnel walls. Any consistent set of units can be used for the dimensional quantities.

The types of elements available are shown in Table 1. Every CAERØi element must reference a PAERØi data card, which is used for additional parameters. Lists of real numbers are sometimes required, which are given on AEFACT lists. These lists may include division points (for unequal box sizes) are parameter values.

21.4.1 Modal Flutter Analysis

The purpose of modal flutter analysis is to study the stability of an aeroelastic system with a minimum number of degrees of freedom. A prerequisite to modal flutter analysis is the calculation of an aerodynamic matrix with a transformation to modal coordinates. This operation is often very costly and care should be taken to avoid unnecessary computations. One method is to compute the modal aerodynamic matrix at a few Mach numbers and reduced frequencies and interpolate to others. Matrix interpolation is an automatic feature of the flutter rigid format. The MKAER@1 and MKAER@2 data cards allow the selection of parameters for the aerodynamic matrix calculation on which the interpolation is based.

The method of flutter analysis is specified on the FLUTTER bulk data card. The FLUTTER caruis selected in case control by an FMETHØD card. Three methods of flutter analysis are available; K, KE and PK. These are shown in Table 2.

AEROELASTIC ANALYSIS

21.4.1.1 The K-Method

The K-method allows looping through three sets of parameters: density ration $(\rho/\rho_{ref}; \rho_{ref})$ is given on an AER9 data card); Mach number m; and reduced frequency k. For example, if the user specifies two values of each, there will be eight loops in the following order.

LOOP (CURVE)	DENS	MACH	REFREQ
1 2 3 4 5 6 7 8	1 2 1 2 1 2 1 2	1 1 1 2 2 2 2	1 2 2 1 1 2 2

Values for the parameters are listed on FLFACT bulk data cards. Usually, one or two of the parameters will have only a single value. Caution: do not set up a large number of loops; it may take an excessive time to execute.

A parameter VREF may be used to scale the output velocity. This can be used to convert from consistent units (e.g., in/sec) to any units the user may desire (e.g., knots), determined from $V_{out} = V/V_{REF}$. Another use of this parameter is to compute the flutter index, by choosing $V_{REF} = b\omega_{\theta} \sqrt{\mu}$.

21.4.1.2 The KE-Method

The KE-method is similar to the K method. By restricting the option, the KE-method is a mure efficient K-method. The two major restrictions are that no damping (B) matrix is allowed and no eigenvector recovery is made. This means that the KE-method is not suitable for a control system, but it is a good method for producing a large number of points for the classical V-g curve. The KE-method also sorts the data for plotting. A plot request for one curve gives all the reduced frequencies for a mode while a similar request in the K-method gives all of the modes at one k value.

AERODYNAMIC MODELING

المريا المحرجين المحوي ومناطق والمراجع

21.4.1.3 The PK-Method

The PK-method treats the aerodynamic matrices as frequency dependent springs and dampers. A frequency is estimated and the eigenvalues are found. From an eigenvalue, a new frequency is found. The convergence to a consistent root is very rapid. The major advantage of the method is that the damping values obtained at subcritical flutter conditions appear to be more representative of the physical damping. Another advantage occurs when the stability at a specified velocity is required since many fewer eigenvalue analyses are needed to find the behavior at one velocity.

The input data for the PK-method also allows looping, as in the K method. The inner loop of the user data is velocity, with Mach number and density on outer loops. Thus, the effects of varying any or all of the three parameters on one run is possible.

21.4.2 Aeroelastic Response Analysis

The purpose of the modal aeroelastic response analysis is to study the behavior of an aeroelastic system resulting from applied loads and gusts. One rigid format can solve frequency response, random response, and transient response problems. The capability includes control systems (using NASTRAN Extra Points and Transfer Functions), multiple loading conditions (with SUBCASES), and rigid body modes.

The input data Geck is the same as for the flutter analysis, except for load requests and output selection. The point loads are applied with Standard RLØAD (frequency response) or TLØAD (transient response) data cards. For gust fields, which are only implemented for the Doublet-Lattice/Body Aerodynamic theory, the vertical stationary gust velocity can be specified with either RLØAD or TLØAD cards. In this manner, the response to either random or time-dependent gusts may be obtained.

For random response analysis, the power spectral density of the load must be supplied. For gusts, either the Von Karman or the Dryden formula can be selected. The output power spectral density is requested by the XYØUT Case Control cards. The r.m.s. value and N₀, the expected frequency, are automatically printed when PSDF information is requested.

The user must supply the basic flight conditions. The velocity is specified by the AER® data card, while Mach number and dynamic pressure (q) are supplied on PARAM bulk data cards.

21.4-3

AEROELASTIC ANALYSIS

55 A816

i

The damping must be modal damping. Ordinarily, a modal viscous damping is assumed, as in the NASTRAN model dynamic rigid format. A parameter KDAMP = -1 can be used to substitute modal structural damping, the modal stiffness is multiplied by $[1+ig(\omega)]$.

ORIGINAL PAGE IS OF POOR QUALITY

÷

Ņ

AERODYNANIC MODELING

H. M. Landa

Туре	Doublet Lattice Panel	Lifting Body (Interference)	Mach Box Surface	Strip Theory	Piston Theory
Data Cards	CAER®1 PAER®1	CAE RØ2 PAE RØ2	CAERØ3 PAERØ3	CAER 9 4 PAER 9 4	CAEROS Paeros
Mach Number	Subsonic	Subsonic	Supersonic	All regimes	Hypersonic
Symmetry Options	2 planes y = 0 z = 0	2 planes y = 0 z = 0	l plane required	None	None
Interaction	Panels in the	and bodies same group	Boxes on one surface	None	None
Comments			One or two control surfaces	Control surface allowed. User may vary parameters.	A strip theory, coefficients from piston or Van Dyke theory. Control surface
Interconnection to Structure	Box centers	Slender body element centers	User specified locations	Strip 1/4-chord	Strip 1/4-chord
Displacement Components Used at Connection Points	3,5	3,5 z-bodies 2,6 y-bodies	3	3,5 No control 3,5,6 Control	3,5 No control 3,5,6 Control

Table 1. Aerodynamic Elements.

Table 2. Flutter Analysis Methods.

ł

/ 	"K"	"КЕ"	"PK"
Structural Matrices	K (complex) B (somplex) M (complex)	K (complex) M (complex)	K (real) B (real) M (real)
Aerodynamic Matrices	M (complex)	M (complex)	K (real) B (real)
User Input Loops	p-density m-Mach number k-reduced frequency	p-density m-Mach number k-reduced frequency	p-density m-density V-velocity
Output	V-g curve Complex modes Displacements Deformed plots	¥~g curve	V-g curve Complex modes Displacements Deformed plots
Method	Compute roots for user input p, m, k.	Compute roots for user input ρ , m, k. Reorder output so a "curve" refers to a mode.	For each ρ , m, V, iterate on each root to find consistent results. (Details in the Theoretical Manual.)
Eigenvalue Method	Several methods available, selected by user via CMETHØD in case control.	Complex Upper* Hessenberg	Real Upper* Hessenberg

* No CMETHØD card is used.

27.5 MODAL FLUTTER ANALYSIS EXAMPLE

This problem illustrates the use of the aeroelastic analysis to determine flutter frequencies and mode shapes for an untapered wing having 15° sweep and an aspect ratio of 5.34 as shown in Figure 1.

21.5.1 <u>Input</u>

Bulk data cards used include CAERØ1, PAERØ1, SPLINE2, SET1, AERØ, MKAERØ1, FLUTTER, and FLFACT as illustrated in User's Manual Section 1.11.

21.5.2 Theory

Reference 1 specifies the reduced frequency k = .1314 (p.17), frequency ratio $\omega/\omega_{\alpha} = 0.51$ (p.35) and torsion frequency $\omega_{\alpha} = 1488$ (p.17).

The flutter velocity is found from

$$V = \frac{b\omega}{k} = \frac{\frac{REFC}{2} \times \omega_{\alpha} \times \omega_{\omega}}{\omega_{\alpha}} = 5980 \text{ in/sec,}$$
(1)

where REFC is the reference length input on the AERØ bulk data card.

The flutter frequency is found from

$$f = \frac{\omega_{\alpha} \times \frac{\omega}{\omega_{\alpha}}}{2\pi} = 121 \text{ Hz}$$
 (2)

21.5.3 <u>Results</u>

The results obtained are compared with both theoretical results using the modified strip analysis method and with experimental results. The flutter velocity is in good agreement. (See Figure 2.)

Frequencies are automatically output while mode shapes used in the modal formulation are obtained using an ALTER to the Rigid Format following the Real Eigenvalue Analysis Module.

Mode shapes for all points in the model may be obtained by checkpointing the problem using the Normal Mode Analysis (Rigid Format 3) and subsequently restarting using the Aeroelastic Analysis.

1000

à

۰.

ν.

21.5.4 Driver Deck and Sample Bulk Data

С.

ì

Card No.										
0 1 2 3 4 5 6 7 8 9 10 11	NASTRAN ID UMF APP SØL TIME DIAG ALTER MATGPR ENDALTEI DIAG 18 CEND	FILES-UM DEM1002 1978 AERØ 10,0 10 14 94 \$ GPL,USE R \$	IF 1, NASTRAN 100210 T, SIL, PHI	A//C,N,FE	/C,N.A \$					
12 13 14 15 16 17 18 19 20 21 22 23 24 25 26 27 28	TITLE = SUBTITLE LABEL = ECHØ = E SPC = 1 METHØD = CMETHØD ØUTPUT() XTITLE = YTTITLE YBTITLE TCURVE = CURVEL IN XYPAPERP BEGIN BU ENDDATA	AERØELAS E = NASTR/ K VALUES BØTH = 10 = 20 = 30 (YØUT) = DAMPLIN = FREQUEN = V-G AND ESYMBØL = - LØT VG / ULK	TIC FLUTT AN DEMØNS .200(*) NG (G) NCY (F) V-F DATA = -1 1(G.F) 2(ER ANALYS TRATIØN P .167(0) PØINTS (G,F) 3(G	IS ØF A F RØBLEM NØ .143(1)	IFTEEN DE . 10-2-1 .125(2)	GREE SWEP .111(3) 5(G,F)	T WING .100(4)		
	ACDA	2	3	4	5	6	77	8)	10
	CAEROI +CAIOI CBAR CMASS2	101 -1. 1	1.3+4 1 26795 1 2.8-6	2.0706 1 0.0 1	1.145-7 6 2.0706 2	4 -1. 0.0	5.45205 0.0	0.0	1 2.0706 1	+CA101
	CORDER	1	0.0-0	0.0	0.0	0.0	0.0	0.0	h.	+01
	Data	20	HESS	MAX						+FC
	EIGR	10	GIV	.3	1.1		6			
	FL FACT	I MAX	067							TER
	FLUTTER	30	KE	1	2	1		,		
	GRDSET		1				li	126		
	MATI	li	10 4+6	0.0	.0	0.0				
	MKAERØI	.45		5.3.0		2.01-4				
	+MK PAFRA1	0.0001	1.1	.2						THK
	PARAN	COUPMASS	1		1					
	PARAM	LMODES	3			1	1			
	PBAR SET)	100	R	7.175-2	9.83-6	1	36.8-6		l	
	SPC1	n	345		111				1	
	SPLINE2	100	101	ion	124	100		,	Ι.	1
	+SP	0.0	0.0	1			0.0	1.		+SP

21.5-2

MODAL FLUTTER ANALYSIS EXAMPLE

.

يريه وهو م كور مت





Figure 1. Fifteen degree sweep model.

AEROELASTIC ANALYSIS



V, in/sec

Figure 2. V-g results for fifteen degree sweep model.

AEROELASTIC RESPONSE EXAMPLE

A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR A

21.6 AEROELASTIC RESPONSE EXAMPLE

This example illustrates the use of the aeroelastic response analysis to perform frequency, random, and transient response calculations for a structure excited by aerodynamic loadings. This problem is also discussed in Section 1.11.5 of the User's Manual.

The structural model represents a wing and aileron configuration as shown in Figure 1. For this demonstration problem, the aileron is locked and the fuselage to which the wing is attached is a rigid body represented by grid point 11. Only out-of-plane motions are retained in the model. The wing is modeled with GENEL data defining the flexibility matrix, [Z], and a free-body matrix, [S]. The aileron also is modeled as a rigid body with the hinge line at point 8. The vertical flap deflection at point 12 is defined by an MPC equation.

The aerodynamic model consists of 42 doublet lattice aerodynamic boxes, forming one coupled group as shown in Figure 2. Three CAERØ1 aerodynamic elements are used to define the areas of uniform mesh on the wing. The aerodynamic degrees of freedom, implicitly defined by the CAERØ data, are coupled to the structure with surface splines defined on SPLINE2 data cards.

21.6.1 Frequency Response Input

This example illustrates the frequency response analysis for a smooth gust shape and generates spectral density output plots for a random gust magnitude.

1. Parameters:

V = 5183.2	(Airstream velocity)
M = 0.62	(Airstream mach number)
$p = 1.1468 \times 10^{-7}$	(Air density)
g = 0.06	(Structural damping)

2. Constraints:

 $\Theta_y = \Theta_z = 0$ Grid 11 (No fuse lage isolation $u_x = u_y = \Theta_z = 0$ All Grids $\Theta_x = \Theta_y = 0$ All Grids except 11 and 12

3. Load:

 $V_g = \frac{8360}{2}$ (1 - cos 2mt) (t < 1) Gust Velocity

· · · ·

21.6.2 Frequency Response Results

Ì

Figures 3 and 4 show plots of the fuselage plunge and aierlon displacement as real and imaginary functions of frequency.

21.6.3 Frequency Response Driver Deck and Sample Bulk Data

and the second second

Card No. 0 NASTRAN FILES=UMF 1 ID DEM11031, NASTRAN 2 UMF 1978 110310 3 APP **AERØ** 4 SØL 11,0 5 TIME 3 6 CEND 7 TITLE = JET TRANSPORT WING DYNAMIC ANALYSIS SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 11-3-1 8 Q LABEL = SYMMETRIC RESPONSE , STIFF AILERON 10 ECHØ = BØTH11 12 JET TRANSPORT WING EXAMPLE MØDEL DESCRIPTION \$ 13 Ś SYMMETRIC RESPONSE TO A RANDOM 14 GUST WITH A STIFF AILERON \$ 15 Ś. 16 SPC = 14 \$ SYM , NØ PITCH 17 MPC = 1METHOD = 10 \$ GIVENS 18 19 SDMAP = 200020 FREQ = 4021 RANDOM = 1031 \$ EMPIRICAL PSDF 22 ØUTPUT 23 24 SØLUTIØN RANDOM ANALYSIS USING \$ 25 DØUBLET-LATTICE METHØD AERØDYNAMICS \$ 26 AT MACH. NO. OF .62 \$ 27 \$ 28 SET 1 = 1 , 2 , 12 \$ 29 SET 2 = 1 , 9 THRU 12 , 1040 SET 3 = 1130 31 SET 4 = 1001, 1022, 1023, 1040, 1041 \$ 32 SDISP(IMAG) = 133 DISP(IMAG) = 234 SPCF(IMAG) = 335 AERØF = 4 36 SUBCASE 1 37 LABEL = RANDOM GUST ANALYSIS GUST - 3002 33 39 \$ 40 PRODUCES XY PAPER PLOTS OF MODAL AND GRID POINT DISPLACEMENTS s AND WING ROOT BENDING MOMENTS 41 \$ 42 s \$ FREQ RESP PACKAGE (COMPLEX NUMBERS) 43 OUTPUT(XYOUT) CURVELINESYMBOL = 1 44 45 XTITLE = FREQUENCY(HERTZ) JET TRANSPORT , FREQUENCY RESPONSE

· .

AEROELASTIC RESPONSE EXAMPLE

46	YTITLE	= MØDAL I	DEFLECTION	N				
47	TC	URVE = FII	RST MØDE	(PLUNGE)				
48	XY	PAPERPLØT	SDISP	/ 1(T1R	м), 1(т)	(P)		
49	TC	URVE =	SECOND	MODE (WI	NG BENDING			
50	XY	PAPERPLØT	SDISP	/ 2(T)R	4) . 2(T) I	(P)		
51	TC	URVE =	TWELFT	H MODE (A	ILERØN	,		
52	XY	PAPERPLØT	SDISP	/ 12(T)R	H) 12(T)	IP)		
53	YTITLE	= PHYSIC	AL DEFLEC	TIGN	•/] •=(••	•••		
54	TCI	URVE =	WING (3/4 CHØRI) . 1/4 CH	IARD STA	458)	
55	XYI	PAPERPLØT	DISP /	10(T3RM	10(13)	P) 9(T	(RM) Q(1	(415)
56	TCI	URVE =	FUSELA	GE PLUNGE	,,	, , , , , , , , , , ,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	5117
57	XYI	PAPERPLØT	DISP /	11(T3RM), 11(731	(P)		
58	TCUF	RVE =	AILERØN	DEFLECT 1	N	,		
59	XYPA	NPERPLØ T	DISP /	12(R2RM)	, 12(R21P	•)		
50	TCUR	RVE =	AERØDYNA	MIC BØX N	EAR TIP .	РІТСН		
51	XYPA	PERPLØT	DISP /	1040(R2RM	1), 1040(R2IP)		
62	TCUR	RVE =	WING RØP	T BENDING	MØMENT			
53	YTITLE =	ROTATION	IAL CONSTR	AINTS				
D4 c c	XYPA	PERPLØT	SPCF /	11(R3RM)	, 11(R3IP)		
00	> RANE	MOM ANALYS	IS ØUTPUT	REQUESTS		•		
D0 57	X111	LE = FREQ	UENCY (HE	RTZ)	JET TRAN	SPØRT , R	ANDOM ANA	LYSIS
0/ 50	ILUR	(VE =	POWER SP	ECTRAL DE	NSITY FUN	CTIØN	-	
00		LE = FUSE	LAGE PLUN	GE (11T3)	, PSDF ,	GUST LØA	D
77 70	X12A	PERPLOI	DISP PS	DF	/ 11(T3)	- •	
70 71		LE = WING	IP DISP	LACEMENT	(9T3)	, PSDF ,	GUST LØA	D
72	×124	TE - UINC	DISP PS	DF	/ 9(T3)	-	
72		LE = WING	KØØI BEN	DING MØME	NT (11R3)	, PSDF ,	GUST LØA	n
74	RECTN DH	PERPLYDI .	SPLF PSD	ł	/ 11(R3)		
75	FNDDATA	LK						
•	LIDUATA				٩			
	<u> </u>	2	3	4	5	6	7	8
	AEFACT	1	0.0	.09	.21	33	45	56
	+AE1	.74	ł	1			. 75	. 50
		11	lasca	1101 000				

.

AFFACT	Ti	Tala	<u> </u>		··		0	9	10
HAFT	74	0.0	.09	.21	.33	.45	.56	. 66	T+AE1
AFRØ	li'	0260	121 222	1 1460 7	.				
CAFRØI	1001	1000	131.232	1.1468-/	11			1	SYM
+CAN1	78 75	1000		225			4	1	+CA01
CELAS2	13	5142671	112	225.	35.	500.	0.	[100.	
CMASS2	2	13967 2	112	5			1	[
CONMI	lī		12	5		Í			
+51	17400	1	1		4 37.7		1		+51
CØRD2R	1	1	0		4.3/+/				+52
+C1	-1.	0	10.0	10.0	0.0	0.0	0.0	[-1.	+C1
DAREA	9999	lii	1	1			1	1	
EIGR	110	lgiv		1		1		[DUMMY
+EIGR	MAX	[0.0	'.		12			+EIGR
FREQ1	40	0.0	.25	30			ł		
GENEL	432		1	3	2	2			
+01	4	3	5	3	6	3		3	+01
GRID	ון		20.25	90.	Ŭ	1	12455	3	+02
GUST	3002	3002	1.1962-4	0.0	8360		12450	1	
MKAERØI	.62							1	
+MK	. 02	0.10	0.50			ľ			THE
MPC	1	12	3	-1.0	8	3	1 5		10001
+MPC1		7	3	-0.5	12	5	33 25	1	THELI
PALROI	1000					-	00.20		1
PARAM	GUSTAER	1				1	1		
PARAM	LMODES	12					l .		1 1
PARAM	MACH	.62							
PARAM	0	4.00747				}			
KAKAK		_00259071							

AEROELASTIC ANALYSIS

RANDPS	1031	1	1	1.		1032			
RLBADI	3002	1 3 3 3 3	-		1004				
SETI	114		THRU	11					
SPC	14		45						
SPLINE1	104	1022	1026	1039	15				
SPLINE2	101	1001	1001	1021	14	0.0	2.	0.	+SP1
+SP1	-1.0	-1.0				1			
SUPERT	111	3				[
TABDHP1	2000				r.				+12000
+T2000	0.	.06	10.	.06	ENDT				1.
TABLED1	1004			1		1	· ·		+T1004
+T1004	0.	0.	.01	1.	10.	1.	ENDT		
TABRND1	1032	1		ł		Į			+001
+001	0.00	2.8708+0	0.25	1.2641+0	0.5 0	4.7188-1	0.75	2.3080-1	+002

21.6.4 Transient Response Input

This example illustrates the transient response solution using a Fourier transform of the frequency response solution.

1. Parameters:

V = 5183.2	(Airstream velocity)
M = 0.62	(Airstream mach number)
ρ = 1.1468 x 10 ⁻⁷	(Air density)
g = 0.06	(Structural damping)

2. Constraints:

 $\begin{array}{l} \Theta_{\mathbf{y}} = \Theta_{\mathbf{z}} = 0 & \text{Grid 11 (no fuse lage isolation)} \\ \mathbf{u}_{\mathbf{x}} = \mathbf{u}_{\mathbf{y}} = \Theta_{\mathbf{z}} = 0 & \text{All Grids} \\ \Theta_{\mathbf{x}} = \Theta_{\mathbf{y}} = 0 & \text{All Grids except 11 and 12} \end{array}$

3. Load:

 $V_g = \begin{cases} 8360 & (t < 1.0) \\ -16720 & (t > 1.0) \end{cases}$ Gust Velocity

21.6.5 Transient Response Results

Figures 5 and 6 are the same quantities as shown in Figures 3 and 4 for the transient solution

a set inter a

`

ł

```
Card
No.
 0
     NASTRAN FILES=UMF
 1
     10
               DEM11032,NASTRAN
     UMF
               1978
                      110320
 2
     APP
               AERØ
 3
     SOL
 4
               11,0
 5
     TIME
               3
     CEND
 6
 7
     TITLE = JET TRANSPORT WING DYNAMIC ANALYSIS
     SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 11-3-2
 8
     LABEL = SYMMETRIC RESPONSE . SQUARE EDGE GUST . TRANSIENT ANALYSIS
 9
     ECHØ = BØTH
10
 11
12
      $
               MØDEL DESCRIPTIØN
                                           JET TRANSPORT WING EXAMPLE
13
     $
                                       SYMMETRIC RESPONSE TO A SQUARE
14
     $
                                       EDGE GUST WITH A STIFF AILERON
15
      $
16
          SPC = 14 $ SYM , NØ PITCH
17
          MPC = 1
ì8
          METHØD = 10 $ GIVENS
19
          SDAMP = 2000
20
          FREQ = 40
21
          TSTEP = 41
22
      $$$$$$ TWELVE MODES AND FORTY TWO BOXES
                                                    AERØ CALC THREE K VALUES
          GUST = 1011 $ SQUARE
23
24
          DLØAD = 9999 $ NEEDED TØ FØRCE APPRØACH TRANSIENT GUST
25
      ØUTPUT
26
27
      $
              SÓLUTIÓN
                                       TRANSIENT ANALYSIS USING
28
     $
                                       DØUBLET-LATTICE METHØD AERØDYNAMICS
29
      $
                                       AT MACH NØ. ØF 0.62
30
      $
31
          SET 1 = 1 , 2 , 12 $
          SET 2 = 1 , 9 THRU 12 , 1040
32
          SET 3 = 11
33
34
          SDISP = 1
          DISP = 2
35
          SPCF = 3
36
37
      $
38
     $
                    PRØDUCES XY PAPER FLØTS ØF MØDAL AND GRID PØINT DISPLACEMENTS
39
      $
                AND WING ROOT BENDING MOMENT TIME HISTORIES
40
      S
41
      ØUTPUT(XYØUT)
                       $ TRANSIENT PACKAGE (REAL NUMBERS)
          CURVELINES (MBØL = 1
42
43
          XTITLE = TIME(SECØNDS)
                                        JET TEANSPORT . SQUARE GUST
44
          TCURVE = FIRST MODE (PLUNGE)
      YTITLE = MØDAL DEFLECTIØN
45
46
          XYPAPERPLØT
                         SDISP /
                                    1(11)
47
          TCURVE = SECOND MODE (WING BENDING)
          XYPAPERPLØT
48
                         SDISP / 2(T1)
          TCURVE = TWELFTH MØDE (AILERØN)
XYPAPERPLØT SDISP / 12(TI)
49
50
51
      YTITLE = PHYSICAL DUFLECTION
52
          TCURVE = WING ( 3/4 CHØRD , 1/4 CHØRD , STA 458 )
53
                          DISP / 10(T3), 9(T3)
          XYPAPERPLOT
```

AEROELASTIC ANALYSIS

,

2

TCURV XYPAP TCURV XYPAP	E = FUSEL ERPLØT E = AILER ERPLØT	AGE PLUNG DISP / ØN DEFLEC DISP /	E 11(T3) TIØN 12(R2)						
TCUR XYPA	VE = AERØI PERPLØT	DYNAMIC B DISP /	ØX NEAR T 1040(R2)	IP , PIT	CH				
YTITLE =	ROTATION	AL CONSTR	AINTS					•	
XYPA	VE = WING PERPLOT	SPCF /	UING MØME 11(R3)	NI					
BEGIN BU	LK								
ENDUATA									
1	2	3	4	5	6	7	8	9	10
AEFACT	1	0.0	.09	.21	. 33	. 45 1	.56	.66	+AE1
+ALI	. /4	0.000					1		
ALKU CACDAI	1001	8360.	131.232	1.1468-7	11	1_			SYM
LCADI	1001		0	205			4		+CAØ1
	10.75	5142671	10.	225.	35.	500.	0.	100.	1
CMASS2	2	13967 2	12	5					
CONMI	lī	11	12	15					1.61
+51	17400.				4 37+7				152
CØRD2R	1		0.0	0.0	0.0	0 0		1.1	+01
+C1	-1.	0.	0.			0.0		1	1.01
DAREA	1001	12	5	5142671.		1			
EIGR	10	GIV	0.0	1.		12		<u> </u>	+EIGR
+EIGR	MAX		ţ					}	
FREQT	40	0.0	.25	39					1
GENEL	432		11	3	2	3	3	3	+01
+01	4	3	5	3	6	3	7	3	+02
GRID		1,000	20.25	90.			12456		}
MMAEDAI	62	1000	1.	0.0	8360.			1	
A MK	02	0.10	0.50			ł		1	+MK
MPC	1.02	12	0.50	1 1 0			1	[
+MPC1	1'	7	3	-1.0	0	3	1.5		+MPC1
PAERØI	1000		5	-0.5	12	12	33.25		
PARAM	GUSTAFR	1						1	
PARAM	IFTM	ò						1	
PARAM	LMØDES	12				1			
PARAM	MACH	. 62			1				
PARAM	Q	4.00747							1
PARAM	WTMASS	.0025907						1	
SETI	14	1	THRU	11					
SPC	14		45					1	
SPLINE	104	1022	1026	1039	15				1
JPLINEZ			1001	1021	14	0.0	2.	0.	+SP1
TOROT	111	-1.0							
TARDMD1	2000	J							
+12000	0.	06	10	06	ENDT				+T2000
TABLEDI	1003		10.	.00	נאטו				
+T1003	0.	1.	1.	1	11	1_1	2	1_1	+11003
TLØAD1	1000	1001	••		1003	- ' •	12.	-1.	1 11003
TSTEP	41	40	.1	1			1	1	

AEROELASTIC RESPONSE EXAMPLE

. . . .

. . .

REFERENCES

 Yates, E. C. and R. M. Bennett, "Use of Aerodynamic Parameters From Nonlinear Theory in Modified-Strip-Analysis Flugger Calculations for Finite-Span Wings at Supersonic Speeds; NASA TN D-1824, July 1963. .

•

ù.

Ar or pra



Figure 1. Structural definition of the transport wing.

+ - ALATING

A. Same

Ì



Figure 2. Aerodynamic element model.

۴

21.6-9

AEROELASTIC ANALYS S







Ē





AEROELASTIC ANALYSIS

Ì





AEROELASTIC RESPONSE EXAMPLE

Ì



Figure 6. Aileron displacement versus time at grid point 12.

22. HEAT TRANSFER ANALYSIS

22.1 INTRODUCTION TO NASTRAN HEAT TRANSFER

NASTRAN heat flow capability may be used either as a separate analysis to determine temperatures and fluxes, or to determine temperature inputs for structural problems. Steady state and transient problems can be solved, including heat conduction (with variable conductivity for static analysis), film heat transfer, and nonlinear (fourth power law) radiation.

The heat flow problem is similar, in many ways, to structural analysis (Figure 1). The same model can be used for both analyses if the same finite elements are appropriate. The same grid points, coordinate systems, elements, constraints, and sequencing can be used for both problems. There are several differences, such as the number of degrees of freedom per grid point and the methods of specifying loads, boundary film heat conduction, and nonlinear elements. For heat flow problems, the only unknown at a grid point is the temperature (cf. structural analysis with three translations and three rotations), and hence, there is one degree of freedom per grid point. Additional grid or scalar points are introduced for fluid ambient temperatures in convective film heat transfer. If radiation effects are included or the conductivity of an element is temperature dependent, the problem becomes nonlinear (cf. structural analysis with temperature dependent materials which only requires looking up material properties and computing thermal loads.)

As in structural analysis, the choice of a finite element model is left to the analyst. Temperature distributions can be output in a format which can be used as input to structural problems. Heat flow analysis uses the special cards shown in Table 1 and many structural NASTRAN Bulk Data cards. These include (where i means there is more than one type): CBAR, CDAMPi, CELASi, CHEXAI, CIHEXI, CØMRØD, ØORDII, CQDŁM, CQDPLT, COUADI, CRØD, CTETRA, CTRAPRG, CTRIAI, CTRIAG, CTRMEM. CTUBE, CVISC, CWEDGE, DAREA, DELAY, DLØAD, DMI, DMIG, EPØINT, GRDSET, GRID, LØAD, MPC, MPCADD, NØLINI, OMITI, PARAM, Piii (for elements requiring properties), PLØTEL, SEQGP, SLØAD, SPCI, SPCADD, SPØINT, TABLEDI, TF, TLØADI, and TSTEP.

An explanation of the DMAP statements and information on Case Control Deck requirements for each Rigid Format, relevant to heat transfer analysis, are presented in the User's Manual, sections 3.17, 3.18, and 3.19. A detailed consideration of theory and applications of heat transfer may be found to the publication NASTRAN Thermal Analyzer--Theory and Application Including a Guide to Hedeling Engineering Problems, Hwa-Ping Lee, NASA Report TM X-3503, April 1977.

22.1-1

22.2 HEAT TRANSFER ELEMENTS

The basic heat conduction elements are the same as NASTRAN structural elements. These elements are shown in the following table:

Heat Conduction Elements					
Туре	Elements				
Linear	BAR, RØD, CØNRØD, TUBE				
Membrane	TRMEM, TRAI1, TRAI2, QDMEM, QUAD1, QUAD2				
Solid of Revolution	TRIARG, TRAPRG				
Solid	TETRA, WEDGE, HEXA1, HEXA2, IHEX1, IHEX2, IHEX3				
Scalar	CELASI, CDMAPI				

A connection card (Cxxx) and, if applicable, a property card (Pxxx) is defined for each of these elements. These heat conduction elements are composed of constant gradient lines, triangles, and tetrahedra. Linear elements have a constant cross-sectional area. The offset on the BAR is treated as a perfect conductor (no temperature drop). For the membrane elements, the heat condition thickness is the membrane thickness. The bending characteristics of the elements do not enter into heat conduction problems. The solid of revolution element, TRAPRG, has been generalized to accept general quadrilateral rings (i.e., the top and bottom need not be perpendicular to the z-axis for heat conduction. The quadrilaterals are composed of overlapping triangles, and the wedges and hexahedra from subtetrahedra. Scalar spring elements are used for transient analysis temperature constraints and scalar damping elements are used to add thermal mass. Gradients and fluxes may be output by requesting ELFØRCE.

Thermal material conductives and heat capacities are given on MAT4 (isotropic) and MAT5 (anisotropic) Bulk Data cards. Temperature dependent conductivities are given on MATT4 and MATT5 Bulk Data cards, which can only be used for nonlinear static analysis. The heat capacity per unit volume is specified as the product of density and heat capacity per unit mass (ρC_p). lumped conductivities and thermal capacitance may be defined by the CELAS1 and CDAMP1 elements, respectively.

HEAT TRANSFER ANALYSIS

a a suite de la companya de la companya de la companya de la companya de la companya de la companya de la comp

-

A special element (HBDY) defines an area for boundary conditions. There are five basic types called P\$INT, LINE, REV, AREA3, and AREA4. A sixth type, ELCYL, is for use only with QVECT radiation. The HBDY is considered an element, since it can add terms to the conduction and heat capacity matrices. Hence, there is a CHBDY connection and PHBDY property card. When a film heat transfer condition is desired, film conductivity and heat capacity per unit area are specified on MAT4 data cards. The ambient temperature is specified with additional points (GRID or SP\$INT) listed on the CHBDY connection card. See Figure 2 for geometry. The several types of power input to the HBDY elements can be output by the ELF\$RCE request.

Radiation heat exchange may be included between HBDY elements. A list of HBDY elements must be specified on a RADLST Bulk Data card, and the emissivities are specified on the PHBDY cards. The Stefan-Boltzmann constant (SIGMA) and absolute reference temperature (TABS) are specified on PARAM Bulk Data cards. Radiation exchange coefficients (default is zero) are specified on RADMTX Bulk Data cards.

Grid point temperatures for thermal stress analysis (static structural analysis) are specified on TEMP bulk data cards. If the heat conduction model is the same as the structural model with the same grid, connection and property cards, the temperature cards for the structural analysis can be directly output from the heat conduction analysis. The output request in Case Control is THERMAL (PUNCH), and the format of the card is exactly that of a double field TEMPP data card.

22.3 CONSTRAINTS AND PARTITIONING

Constraints are applied to provide boundary conditions, represent "perfect" conductors, and provide other desired characteristics for the heat transfer model.

Single point constraints are used to specify the temperature at a point. The grid or scalar points are listed on SPC or SPC1 Bulk Data cards, not GRDSET or GRID cards, because in heat transfer analysis there is but one degree of freedom per point. The component on the data card must be "O" or "1", which declares the degree of freedom to be in the us set. The method of specifying temperature is dependent upon the problem type. In linear statics analysis, the SPC or SPC1 card is used to constrain grid points at a fixed temperature. In nonlinear statics analysis, the SPC or SPC1 card is used to designate the grid point ID which is to be constrained. The actual value of the temperature is indicated on a TEMP card, selected by TEMP(MATERIAL) in the Case Control deck. In transient analysis, the SPC or SPC1 card may be used to fix the temperature of a grid point only when the temperature is zero. When the temperature is non-zero a large conductive coupling to a "ground" at absolute temperature must be defined. From the structural relationship F=Kx, the thermal analogy is made where K is the conductive coupling, F is an applied load, and x is the fixed temperature. In this case, x is adjusted to the desired temperature by defining the spring constant, K, of a CELASi element, which is connected to "ground," and a load, F, which is applied to the grid point in question. The numerical value of K should be several orders of magnitude greater than the numerical value of the conductances prescribed for the rest of the model.

Multipoint constraints are linear relationships between temperatures at several grid points and are specified on MPC cards. The first entry on an MPC card will be in the u_m set. The type of constraint is limited if nonlinear elements are present. If a member of set u_m touches a non-linear (conduction or radiation) element, the constraint relationship is restricted to an "equivalence." The term "equivalence" means that the value of the member of the u_m set will be equal to one of the members of the u_n set (a point not multipoint constrained). Those points not touching nonlinear elements are not so limited. The user will be responsible for satisfying the equivalence requirement by having only two entries on the MPC data card.

22.3-1

22.4 THERMAL LOADS

Thermal "loads" may be boundary heat fluxes or volume heat addition. As in the case of structural analysis, the method of specifying loads is different for static and transient analysis. The HBDY element is used for boundaries of conducting regions. Surface heat flux input can be specified for HBDY elements with QBDY1 and QBDY2 data cards. These two cards are for constant and (spatially) variable flux, respectively. Flux can be specified with the QHBDY data card without reference to an HBDY element. Vector flux, such as solar radiation, depends upon the angle between the flux and the element normal, and is specified for HBDY elements with the QVECT data card. This requires that the HBDY element orientation be defined on the CHBDY card. Volume heat addition into a conduction element is specified on a QVBL data card.

Static thermal loads are requested in Case Control with the LØAD card. All of the above load types plus SLØAD can be requested. Transient loads are requested in Case Control with a DLØAD card to select TLØAD time functions. Transient thermal loads may use DAREA (as in structural transient), and/or the QBDY1, QBDY2, QHBDY, QVECT, QVØL, and SLØAD cards. The resultant thermal load will be the sum of all loads applied. This means the LØAD SIDs and DAREA SIDs must be the same when referenced on a TLØADi card.

٠,

STATIC ANALYSIS

22.5 STATIC ANALYSIS

Linear static analysis uses APProach HEAT SOLution 1. The Rigid Format is the same as that used for static structural analysis. This implies that several loading conditions and constraint sets can be solved in one job by using subcases in the Case Control deck.

Nonlinear static analysis uses APProach HEAT, SOLution 3. This specially designed Rigid Format will allow temperature dependent conductives of the elements, nonlinear radiation exchange and a limited use of multipoint constraints. There is no looping for load and constraints, and the solution is iterative. The user can supply values on PARAM Bulk Data cards for:

MAXIT (integer)Maximum number of iterations (default 4).EPSHT (real)ε convergence parameter (default .001).TABS (real)Absolute reference temperature (default 0.0).SIGMA (real)Stefan-Boltzmann radiation constant (default 0.0).IRES (integer)Request residual vector output if positive (default -1).

The user must supply an estimate of the temperature distribution vector $\{u^1\}$. This estimate is used to calculate the reference conductivity plus radiation matrix needed for the iteration. $\{u^1\}$ is also used at all points in the u_s set to specify a boundary temperature. The values of $\{u^1\}$ are given on TEMP Bulk Data cards, and they are selected by TEMP(MATERIAL) in Case Control.

Iteration may stop for the following reasons:

- 1. Normal convergency: ϵ_T < EPSHT, where ϵ_T is the per unit error estimate of the temperatures calculated.
- 2. Number of iterations > MAXIT.
- 3. Unstable: $|\lambda_1| < 1$ and the number of iterations > 3, where λ_1 is a stability estimator.
- 4. Insufficient time to perform another iteration.

The precise explanations and definitions are given in the NASTRAN Theoretical Manual, Section 8.4. Error estimates ϵ_p , λ_1 , and ϵ_T for all iterations may be output with the Executive Control card DIAG 18, where ϵ_p is the ratio of the Euclidian norms of the residual (error) loads to the applied loads on the unconstrained degrees of freedom.

22.5-1

TRANSIENT ANALYSIS

22.6 TRANSIENT ANALYSIS

ż

Transient analysis uses APProach HEAT SOLution 9. This Rigid Format may include conduction film heat transfer, nonlinear radiation, and NASTRAN nonlinear elements. Extra points are used as in structural transient analysis. All points associated with nonlinear loads must be in the solution set. Loads may be applied with TLØAD and DAREA cards as in structural analysis. Also, the thermal static load cards can be modified by a function of time for use in transient analysis. If the static load data is used to define a transient load, the static load set identification is referenced on the TLØAD card in the DAREA field. Loads are requested in Case Control with DLØAD. Initial temperatures are specified on TEMP Bulk Data cards and are requested by IC in Case Control. Previous static or transient solutions can easily be used as initial conditions, since they can be punched in the correct format. An estimate of the temperature $\{u^1\}$ is specified on TEMP Bulk Data cards for transient with radiation, and is requested by TEMP(MATERIAL). The parameters available are:

TABS (real)	Absolute reference temperature (default 0.0).
SIGMA (real)	Stefan-Boltzmann radiation constant (default 0.0).
BETA (real)	Forward difference integration factor (default .55).
RADLIN (integer)	Radiation is linearized if positive (default -1).

22.7 HEAT TRANSFER EXAMPLE

The sample problem below illustrates the capability of NASTRAN to solve heat conduction problems. Linear solid state conduction through a washer is solved in two ways: (1) using solid elements and (2) using ring elements. The washer, shown in Figure 3, has a radial heat conduction with the temperature specified at the outside and a film heat transfer condition at the inner edge. Due to symmetry about the axis and the assumption of negligible axial gradients, the temperature depends only upon the radius.

22.7.1 Input

The first NASTRAN model is shown in Figure 4. The solid elements (HEXA1, HEXA2, WEDGE, and TETRA) and boundary condition element (HBDY, type AREA4) are used. The conductivity of the material is specified on a MAT4 card. Temperatures are specified at the outer boundary with SPC cards. Punched temperature output is placed on TEMP bulk data cards suitable for static analysis. Another variation of the problem is shown in Figure 5. Solid of revolution elements (TRIARG and TRAPRG) and boundary condition element (HBDY, type REV) are used. The conductivity of the material and the convective film coefficient are specified on a MAT4 card. The CHBDY card refere res a scalar point at which the ambient temperature is specified using an SPC card. An SPC1 c2⁻¹ is used to constrain the temperature to zero degrees at gridpoints on the outer surface.

22.7.2 Theory

The mathematical theory for the continuum is simple, and can be solved in closed form. The differential equation is

$$\frac{1}{r}\frac{\partial}{\partial r}\left(rk\frac{\partial U}{\partial r}\right)=0.$$
 (1)

The boundary conditions are

$$-k\frac{\partial U}{\partial r} = H(U_a - U) \text{ at } r = r_1, \qquad (2)$$

and

$$U = 0$$
 at $r = r_2$. (3)

22.7-1
The solution is

$$U(r) = \frac{HU_{a}}{(k/r_{1}) + Hn(r_{2}/r_{1})} n(r_{2}/r)$$

= 288.516 n(2/r)

.

22.7.3 Results

A comparison of theoretical temperatures with the NASTRAN results is shown in Table 2.

22.7.4 Driver Deck and Sample Bulk Data

an a provinsi provinsi na pratika na provinsi p

Note: Only one data card of each type is listed to illustrate its use.

Card No.

.

0 1 2 3 4 5 6	NASTRAN F ID UMF TIME APP SØL CEND	ILES = U DEM1121, 1977 1 HEAT 1,1	MF NASTRAN 11210							
7 8 9 10 11 12 13 14 15 16 17 18	TITLE = L SUBTITLE LABEL = S ØLØAD = A SPCFØRCES THERMAL(P ELFØRCE = SUBCASE 1 LABEL = T SPC = 351 LØAD = 25 BEGIN BUL	INEAR ST = NASTRA @LID ELE LL = ALL RINT, PUN ALL 23 EMPERATU 1 K	EADY STA' N DEMØNS' MENTS,SUI CH) = ALI RE SPECII	TE HEAT CØ TRATIØN PR RFACE FILM FIED AT ØU	INDUCTIØN IØBLEM NO I HEAT TR	THROUGH 1-12-1 ANSFER DARY	A WASHER			
_	· · · · · · · · · · · · · · · · · · ·	<u> </u>	<u> </u>		5	6	7	8	9	10
19	CHEVAL	701	702	AREA4	1	12	112	101		
21	+SOL1	113	112	1	2	13	12	101	+SOL1	
22	CHE XA2	2	200	2	3	14	13	102	103	+5412
23	+SPL2	114	113						1.00	- oper
24			0	.0	0.	.0	.0	0.	100.0	+CØRD11
26	CTETRA	3	200	104	114	12	102			
27	CWEDGE	8	200	4	5	15	103	105	115 .	
28	GRDSET						iii	100	113	
29	GRID	1		1.0	0.	0.				
30		1 400	11.0	1	[1		1	1
- 51	PARAM	TRES	11	1						·

÷

a an sis for the addate of the second addates for the

1

22.7-2

HEAT TRANSFER EXAMPLE

ري. بير جيونيه، وريو مريم مري

	1	2	3	4	5	6	7	8	9	10
33 34 35	QBDY1 SEQGP SPC	251 12 351	288.5 1.1 11	701 13 1	2.1	14 22	3.1 1	15 .0	4.1	

a tan tan at tan a

36 ENDATA

Notes on Cards for the Solid Element Solution

Card

=.

No.

- 12 This card in the Case Control Deck requests that the temperatures of all points be printed on paper and punched on cards.
- 19 The CHBDY card defines a boundary element which is used for heat flux, thermal vector flux, convection and/or radiation.
- A positive integer value of the IRES parameter in a HEAT statics problem will cause the printing of all residual vectors. 31
- 32 The PHBDY card defines the properties of the HBDY element.
- The QBDY1 card defines a uniform heat flux into HBDY elements. To be used in a static run, this card must be requested in the Case Control Deck (LOAD=SID). 33

Card

No.							
0	NASTRAN F	ILES≐UMF					
1	ID	DEMI122,N	ASTRAN				
2	UMF	1977 1	1220				
3	APP	HEAT					
4	SØL	1,0					
5	TIME	10					
6	CEND						
-							
/	TITLE = L	INEAR STE	ADY STATE	CONULITO	N THROUGH	AWASHER	
8	SUBILILE	= NASIKAN	DEMONSIK	ALLUN PRO	BLEM NW.	1-12-2	
.9	LABEL = K	ING ELEME	NIS, FILM	HEAT TRA	NSFER		
10							
11	SPORTAD = A	LL - 811					
12	SPURAKUE	F ALL					
13	THERMAL (PRINI,PUN	ich) = ALL				
14	ELFUKUE =	ALL					
10	37C = 330	v		•			
10	DEGIN DUL	N.					
	1	2	3	4	5	6	7
17	CHRDY	14	100	REV	1	12	
18	+HBDY14	23	23				
19	CTRAPRG	17	Ā	5	16	15	.0
20	GRID	li	·	l i.o	lõ	.õ	
LV	1				1 · *	• •	

13

11

352

1	12				+HBDY14	
16 .0	15 .0	.0	200			
2.1 488.5 22	14	3.1	15	4.1		

8

9

10

29 ENDDATA

MAT4

PHBDY

SEQ GP

SPC SPC1

SPCADD

SPOINT

TEMPO

21

26

27

28

1 200

100

12

352 351

350

23

201

.10

300

1.1

23

351

.0

1

22.7-3

.

.

Notes on Cards for the Ring Element Solution

Card

a car artest

No.

13 This card in the Case Control Deck requests that the temperatures of all points be printed on paper and punched on cards.

..

- 17 The CHBDY data card defines a boundary element which is used for heat flux, thermal vector flux, convection and/or radiation.
- 22 The PHBDY card defines the properties of the HBDY element.

28 The TEMPD card defines a default temperature for all grid points of the structural model which have not been given a temperature on a TEMP card. Temperature sets must be selected in the Case Control Deck (TEMP=SID) to be used by NASTRAN.

HEAT TRANSFER EXAMPLE

Table 1. Bulk Data Ca.ds of Special Reference to Heat Transfer Analysis

a new confinence 🖬 🛶

Ļ

and a star water and a

ŝ

CHBDY	Heat Boundary Element
MATT3	Material Temperature Dependence
MATT4	Material Temperature Dependence
MATT5	Material Temperature Dependence
PHBDY	Property of Heat Boundary Element
QBDY 1	Boundary Heat Flux Load
QBCY2	Boundary Heat Flux Load
QHBDY	Boundary Heat Flux Load
QVECT	Thermal Vector Flux Load
qvol	Volume Heat Addition
RADLST	List of Radiation Areas
KADMTX	Radiation Matrix
TABLEMI	Material Property Table
TABLEM2	Material Property Table
TABLEM3	Material Property Table
TABLEM4	Material Property Table
TEMP	Grid Point Temperature Field
TEMPAX	Axisymmetric Temperature
TEMPD	Grid Point Temperature Field Default

22.7-5

HEAT TRANSFER ANALYSIS

بر العقور ، د شهر د

r(radius)	Theoretical Temperatures	NASTRAN Temperatures (Solids)*	NASTRAN Temperatures (Rings)*
1.0	199.984	202.396	199.932
1.1	172.486	173.904	172.448
1.2	147.381	148.833	147.355
1.3	124.288	124.783	124.269
1.4	102.906	102.852	102.894
1.5	83.001	82.913	82.992
1.6	64.380	64.306	64.375
1.7	46.889	46.832	46.886
1.8	30.398	30.356	30.397
1.9	14.799	14.773	14.798
2.0	0.000	0.000	0.000

Table 2. Comparison of Theoretical and NASTRAN Temperatures for Heat Conduction in a Washer

* These are the average temperatures at a radius.

Ì

22.7-6

HEAT TRANSFER EXAMPLE

hallen som sen and sen and sen and sense and the sense of

16.2.4

ż

-) -

. · ·

and and in the second second



Figure 1. Thermal model diagram.

22.7-7

HEAT TRANSFER ANALYSIS

Type = POINT

an an an the second second second second



The unit normal vector is given by $\bar{n} = \bar{V}/|\bar{V}|$, where \bar{V} is given in the basic system at the referenced grid point (see CHBDY data card, fields 16-18).

Type = LINE



The unit normal lies in the plane of \vec{V} and \vec{T} , is perpendicular to \vec{T} , and is given by $\vec{n} = (\vec{T} \times (\vec{V} \times \vec{T}))/|\vec{T} \times (\vec{V} \times \vec{T})|$.

Type = ELCYL



Type = REV

The unit normal lies in the x-z plane, and is given by $\bar{n} = (\bar{e}_y \times \bar{T})/|\bar{e}_y \times \bar{T}|$. \bar{e}_y is the unit vector in the y direction.

Type = AREA3 or AREA4

The unit normal vector is given by $\bar{n} = (\bar{T}_{12} \times \bar{T}_{1x})/|\bar{T}_{12} \times \bar{T}_{1x}|$, where x = 3 for triangles and x = 4 for quadrilaterals.

Figure 2. HBDY element orientation (for QVECT flux).

 \overline{T}_{1x} G_1 \overline{T}_{12} G_2

22.7-8

HEAT TRANSFER EXAMPLE

. .

a rate of the state

and the second second second second

.

_

___`**`**

_

_

-



Figure 3. Washer Analyzed in Heat Conduction Demonstration Problem.

Lander Second

e en c



Figure 4. Elements and Grid Points.

HEAT TRANSFER EXAMPLE

;

=

Ξ \ -



T TRIARG elements Q TRAPRG elements U_a = 488.5 at left end U_a = 0.0 at right end

Figure 5. Section of a pipe, modeled with ring elements.

23.1 CYCLIC SYMMETRY MODELS

Many structures, including pressure vescels, rotating machines and antennae for space communications, are made up of virtually identical segments that are symmetrically arranged with respect to an axis. There are two types of cyclic symmetry as shown in Figures 1 and 2: simple rotational symmetry, in which the segments do not have planes of reflective symmetry and the boundaries between segments may be general doubly-curved surfaces; and dihedral symmetry, in which each segment has a plane of reflective symmetry and the boundaries between segments are planar. The use of cyclic symmetry allows the user to model only one of the identical substructures which will result in a large saving of data preparation and computer time for most problems. Table 1 summarizes the Bulk Data cards of special relevance to cyclic symmetry. The theoretical treatment for cyclic symmetry is given in Section 4.5 of the Theoretical Manual.

The total model consists of N identical segments which are numbered consecutively from 1 to N. The user supplies a NASTRAN, model for one segment using regular elements and standard modeling techniques, except that, grid points are not permitted on the polar axis. All other segments and their coordinate systems are automatically rotated by the program to equally spaced positions about the polar axis. The boundaries must be comformable, i.e., the segments must coincide. This is easiest to insure if a cylindrical or spherical coordinate system is used, but such is not required. CTYPE on the PARAM card is used to specify either rotational symmetry of dihedral symmetry. The number of segments, N, in the structural model is specified also on the PARAM card, NSEGS. As indicated in Figure 2, dihedral symmetry provides solutions for each segment and its reflected image. This requires application of both symmetric and antisymmetric boundary conditions.

In rotational symmetry the basic transformation equation between the structure segments n = 1, 2, etc., and the harmonic indices k = 0, 1, 2, etc., is

$$u^{n} = \bar{u}^{0} + \sum_{k=1}^{KMAX} [\bar{u}^{kc} \cos(n-1)ka + \bar{u}^{ks} \sin(n-1)ka]$$
(1)

where

 u^n is any displacement. load, stress, etc., on the nth segment (n = 1,2...NSEGS), \bar{u}^0 . \bar{u}^{kc} , \bar{u}^{ks} are the corresponding cyclic coefficients used in the solution which define the entire structure,

23.1-1

CYCLIC SYMMETRY MODELING

k is the cyclic index (i.e., KINDEX),

KMAX is the limit (KMAX $\leq \frac{N}{2}$) of k. (If all values of k are used, the transformation is exact),

and

 $a = \frac{2\pi}{NSEES}$ is the circumferential angle for each segment.

In the case of dihedral symmetry, the repeated segment may be divided into two half-segments divided by a plane of symmetry. The reason for using dihedral symmetry is to reduce the size of the model by one half. The solution is obtained for symmetric motions (S) and antisymmetric motions (A) of the right half segment which is the one to be modeled by the user. Thus, for each cyclic index, K, four coefficients are obtained defining the variable, n, i.e., $\bar{u}^{ks,S}$, $\bar{u}^{kc,S}$, $\bar{u}^{ks,A}$. In the right hand segment the terms are added

Right side:
$$\bar{u}^{ks} = \bar{u}^{ks,S} + \bar{u}^{ks,A}$$
 (2)

In the left hand mirror image the antisymmetric solution is subtracted.

Left Side: $\vec{u}^{ks} = \vec{u}^{ks}, S_{-\vec{u}}^{ks}, A$ (3)

The two boundaries are called sides 1 and 2. In the case of rotational symmetry, side 2 of segment n is connected to side 1 of segment n+1, as shown in Figure 1. In the case of dihedral symmetry, side 1 is on the boundary of the segment and side 2 is on the plane of symmetry for the segment, as shown in Figure 2. In either case the grid point numbers on sides 1 and 2 must be specified on the bulk data card, CYJØIN.

No restrictions are placed on the use of the single point constraint, the multipoint constraint, or the SMIT feature of NASTRAN. However, the constraints are assumed to be the same for each segment. Constraints between segments are automatically applied to the degrees of freedom, which are not otherwise constrained, at grid points specified on CYJBIN bulk data cards. The SPCD bulk data card may be used to vary the magnitude of enforced displacements for each of the segments.

For purposes of minimizing matrix bandwidth, the equations of the solution set are normally sequenced with the cosine terms alternating with the sine terms. The user may request an alternate sequence on the PARAM card, CYSEQ, which orders all cosine terms before all sine terms. The latter may improve efficiency when all of the interior points have been omitted.

STATIC ANALYSIS

23.2 STATIC ANALYSIS

Static loads are applied to the structural model in the usual way. NASTRAN assumes that separate subcases are defined for each segment. If there are multiple loads, a set of sequential ordered subcases, one subcase per load, must be defined for each segment in Case Control. In effect, it is as each segment of the model is treated as a separate structure with its own load. All segments, of course, must have the same number of consecutive subcases, one for each static load applied to the whole model. In the case of rotational symmetry, there will be a number of subcases equal to the number of segments in the structural model times the number of loading conditions. In the case of dihedral symmetry, there will be twice as many subcases as for rotational symmetry because of the two symmetric components. If there is more than a single loading condition, the number of loading conditions must be specified on the PARAM card, NLGAD.

When loadings are specified in this manner, by physical segment, the data reduction will also be done in terms of the physical segments. All the normal output is available by segment and loading including structure plots plus the symmetrical components of displacement, internal forces and stresses.

An alternate procedure for specifying the static loads may be used if the transformed values of the forcing functions are known. In this case, the transformed values of the loads are specified directly on the usual loading cards. The PARAM card, CYCIØ, must be included in the Bulk Data Deck to indicate that cyclic transform representation rather than physical segment representation is being used for the static loads. If this option is used, the subcases must be ordered according to the symmetrical components with the cosine cases preceding the sine cases for each symmetrical component. The output quantities will also be presented in terms of the symmetric components.

As indicated in the Theoretical Manual, Section 4.5, the cyclic symmetry uses a finite Fourier transformation. Hence, the use of cyclic symmetry procedures does not introduce any additional approximations beyond those normally associated with finite element analysis. In the case of static analysis, a shortened approximate method may be used where the maximum value of the harmonic index is specified on the PARAM card, KMAX. The default procedure is to include all harmonic indices. The use of a smaller number of harmonic indices is similar to truncating a Fourier series. The stiffness associated with the higher harmonic indices tends to be large, so that these components of displacements tend to be small.

23.2-1

CYCLIC SYMMETRY MODELING

The **D**HIT feature may be used to remove all degrees of freedom at internal grid points without any loss of accuracy. Since this reduction is applied to a single segment prior to the symmetry transformations, it can greatly reduce the amount of subsequent calculation.

.*

5.35

141

• :

Sector Sector

NORMAL MODES ANALYSIS

Note a second of

23.3 NORMAL MODES ANALYSIS

....,

When dihedral symmetry is involved in normal modes analysis, only the modes for the symmetrical components $\bar{u}^{kC,S}$ and $\bar{u}^{kS,A}$ are obtained. The modes for the other two terms are identical and correspond to a one-segment rotation of the structure.

The Fourier transform solutions are performed separately for each harmonic index. The harmonic index for each solution is specified on the PARAM card, KINDEX. The standard restart procedures can be used to calculate vibration modes for additional harmonic indices. Note, however, that no provision is made to recover solution data in terms of the physical segment.

The **MIT feature is used in the usual way to reduce the** size of the analysis set and involves the usual approximations. The SUPBRT card for free bodies cannot be used with cyclic symmetry.

CYCLIC SYMMETRY EXAMPLE

and the second of the

23.4 CYCLIC SYMMETRY EXAMPLE

An example to illustrate the handling of cyclic symmetry by Rigid Format 14 is presented. Another problem processed by Rigid Format 15 may be found in the NASTRAN Demonstration Problem Manual.

23.4.1 Description

A constant thickness circular plate with six radial stiffeners and a central hole, shown in Figure 3 is analyzed using dihedral symmetry. The plate is subjected to a uniform pressure load applied over a 60° segment of the plate.

The finite element model is shown in Figure 4. The stringers are 60° apart but only 30° of the structure needs to be modeled when using the dihedral symmetry option. There are 12 subcases since these are two half segments in a 60° segment and only one loading condition. The CYJØIN bulk data card defines those points in the moddle of the segment (SIDE 2) and those points on the boundary between segments (SIDE 1).

23.4.2 Input

Parameters:

 $R_0 = 1.0$ (outside radius) $R_i = .14$ (inside radius) t = .01 (plate thickness) a = .06 (height and width of stiffeners) $E = 10.6 \times 10^6$ (modulus of elasticity) v = .325 (Poisson's ratio)

Boundary Conditions:

 $U_r = U_{\theta} = \theta_z = 0 \text{ (all points)}$ $U_z = \theta_r = 0 \text{ (along } r = 1.0\text{)}$

Applied loads::

```
Pressure = 200.0 between 0 = 60° and 120°
```

Cyclic symmetry parameters:

CTYPE = DRL KMAX = 2 NSEGS = 6 NLOAD = 1

23.4.3 Results

Ĩ,

The structure can be analyzed using rotational symmetry or dihedral symmetry described here and the results will be identical. The results for the normal displacements are given in table 1 for r = 0.46.

23.4.4 Driver and Sample Bulk Data

Note: Only one data card of each type is listed to illustrate its use.

Card <u>No.</u> 1 2 3 4 5 6	ID APP SØL TIME DIAG CEND	DEM14011, NA DISPLACEMENT 14,0 5 8	STRAN	
7 8 9 10 11 12 13 14 15 16 17	TITLE = SUBTITLE LABEL = SPS = 10 ØUTPUT ØLØAD = DISP = A SUBCASE LABEL SUBCASE	STATIC ANALYS = NASTRAN DEI DIHEDRAL CYLC ALL ALL LL 1 = SEGMENT 1 2	IS ØF A CIRCULAR PLA MØNSTRATIØN PRØBLEM I IC SYMMETRY RIGHT	TE 19. 14-1-1
18 19 20 21 22	LABEL SUBCASE LABEL LØA SUBCASE	² = SEGMENT 1 3 = SEGMENT 2 4D = 102 4	LEFT RIGHT	
23 24 25 26	LABEL + LØA SUBCASE LABEL	= SEGMENT 2 ND = 102 5 = SEGMENT 3	ELEFT	OF POOR QUALITY
27 28 29 30 31	SUBCASE LABEL SUBCASE LABEL SUBCASE	= SEGMENT 3 7 = SEGMENT 4 8	I LEFT I RIGHT	
32 33 34 35	LABEL SUBCASE LABEL SUBCASE	= SEGMENT 4 9 = SEGMENT 5 10	I LEFT 5 RIGHT	
36 37 38 39 40	SUBCASE LABEL SUBCASE LABEL	= SEGMENT 6 1] = SEGMENT 6 12 = SEGMENT 6	5 RIGHT 5 LEFT	•

(- 50-

CYCLIC SYMMETRY EXAMPLE

No.										
	<u> </u>	22	3	4	5	6	7	88	9	10
42 43 44 45 46 47 48 49 50 51	CBAR CNGRNT CØRD2C +C1 10 CVJØIN GRDSET GRID MAT1 PARAM	1 10 1 1. 1 1 10 1 CTYPE	1 11 0 .0 10 C 1 10.6+6 DRL	10 .0 .11 10 1.0	20 .0 21 20 .0 .325	.0 .0 20 30 .0 2.59-4	.0 .0 40 12.9-6	1. .0 50	1 1. .60	+C1 CYC SYM CYC SYM
52 53 54 55 56 57	PBAR +PB1 PLØAD2 SPC1 SPCADD ENDDATA	1 .0 102 110 101	1 .03 200. 123 46 110	1.8-3 .03 10 10 112	5.4-7 .0 20 11	5.4-7 .03 30 12	1.0-6 .03 40	.03 50	03	+PB1

Data Deck Notes

ì

Card

No.

- 3 The SØL card references Rigid Format 14, which is a special set of DMAP instructions for static analysis using cyclic symmetry.
- 47 The CYJØIN card is used only for cylcic symmetry problems. It defines the boundary points of a segment of a structural model.
- 51 The CTYPE parameter is required for Rigid Formats 14 and 15. The value of the parameter defines the type of cyclic symmetry.

•

L-6

23.4-3

CYCLIC SYMMETRY MODELING

Table 1. Bulk Data Cards of Special Relevance to Cyclic Symmetry

CYJDIN Boundary points of a segment

- PARAM(CTYPE) Defines the type of cyclic symmetry (required)
- PARAM(NSEGS) Number of identical segments in the model (required)
- PARAM(NLBAD) Number of static loading conditions (optional in static analysis)
- PARAM(CYCI9) Form of input and output data (optional in static analysis)
- PARAM(CYCSEQ) Procedure for sequencing the equations in the solution set (optional)
- PARAM(KMAX) Maximum value of the harmonic index (optional in static (analysis)
- PARAM(KINDEX) Specified a single value of the harmonic index (required in normal modes analysis)

a second design and a second

	DIHE	DI HE DRAL METHOD				
•	Subcase	Grid	Value			
0	1	30	1.365			
15	1	31	1.379			
30	1 2	32 32				
45	2	31	1.412			
6U	23	30 30	1.430			
75	3	31	1.464			
90	3	32 32	1.484			
105	4	31				
120	4 5	30 30	i 1,430			
135	5	31	1.412			
150	5 6	32 32	1.396			
165	6	31	1.379			
180	6 7	30 30	1.365			
195	7	31	1.359			
210	7 8	32 32	1.354			
225	8	31	1,349			
240	89	30 30	1.345			
255	9	31	1.344			
270	9 10	32 32	1.345			
285	10	31	1.344			
300	10 11	.10 .30	1.345			
315	11	31	1.349			
330	11 12	32 32	1.354			
345	12	31	1.359			
360	12	30	1.365			

Taule 2. Displacements of Circular Plate Under Pressure Load at r = 0.46

CYCLIC SYMMETRY MODELING

a a second and second a



Conformable Interface

1. The user models one segment.

والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع

- 2. Each segment has its own coordinate system which rotates with the segment.
- Segment boundaries may be curved surfaces. The local displacement coordinate systems must conform at the joining points. The user gives a paired list of points on Side 1 and Side 2 which are to be joined.

Figure 1. Rotational symmetry.



- The user models one-half segment (an R segment). The L half segments are mirror images of the R half segments.
- Each half segment has its own coordinate system which rotates with the segment. The L half segments use left hand coordinate systems.
- Segment boundaries must be planar. Local displacement systems axes, associated with inter-segment boundaries, must be in the plane or normal to the plane. The user lists the points on Side 1 and Side 2 which are to be joined.

Figure 2. Dihedral symmetry.

23.4-7

....

ì



Figure 3. Circular plate with stiffeners.

CYCLIC SYMMETRY EXAMPLE

۲Ì



Figure 4. Finite element model.

24. HYDROELASTIC ANALYSIS

24.1 SOLUTION OF THE FLUID MODEL

The NASTRAN hydroelastic option allows the user to solve a wide variety of fluid problems having structural interfaces, compressibility, and gravity effects. A complete derivation of the NASTRAN model and an explanation of the assumptions are given in the Theoretical Manual, Section 16.1. The input data and the solution logic have many similarities to those of a structural model. The Bulk Data cards of special relevance to hydroelastic analysis are listed in Table 1. The standard normal modes analysis, transient analysis, complex eigenvalue analysis, and frequency response solutions are available with minor restrictions. The differences between a NASTRAN fluid model and an ordinary structural problem are due to the physical properties of a fluid, which are:

- The independent degrees of freedom for a fluid are the Fourier coefficients of the pressure function (i.e., "harmonic pressures") in an axisymmetric coordinate system. The indpendent degrees of freedom for a structure are typically displacements and rotations at a physical point in space.
- 2. Much like the structural model, the fluid data will produce "stiffness" and "mass" equivalent matrices. Because they now relate pressure and flow instead of displacement and force, their physical meaning is quite different. The user may not apply loads, constraints, or sequencing, directly on the fluid points involved. Instead, the user supplies information related to the boundaries and NASTRAN generates the correct constraints, sequencing, and matrix terms internally. Indirect methods, however, are available to the user for utilizing the internally generated points as normal grid or scalar points. See Section 24.4 for the identification code.
- 3. When a physical structure is to be connected to the fluid, the user supplies a list of fluid points and a related list of special structural grid points and NASTRAN produces the unsymmetric matrix terms which define the actual physical relations. A special provision is included in NASTRAN for the structure which has planes of symmetry. The user may, if he wishes, define only a section of the boundary and solve his problem with symmetric or antisymmetric constraints. The fluid-structure interface will take the missing sections of structural boundary into account.

24.1-1

HYDROELASTIC ANALYSIS

4. Because of the special nature of fluid problems, various user convenience options are absent. The fluid elements and harmonic pressures may not be included in the structural plots at present. Plotting the harmonic pressures versus frequency or time may not be requested directly. Because mass matrix terms are automatically generated if compressibility or free surface effects are present, the weight and C.G. calculations may not be correct and should be avoided if fluid elements are present. Also, the Inertia Relief Rigid Format uses the mass matrix to produce internal loads and if fluids are included, the special fluid terms in the mass matrix may produce erroneous results.

In spite of the numerous differences between a NASTRAN structural model and a NASTRAN fluid model, the similarities allow the user to formulate a model with a minimum of data preparation and to obtain efficient solutions to large order problems. The similarities of the fluid model to the NASTRAN structural model are:

1. The fluid is described by points in space and finite element connections. The locations of the axisymmetric fluid points are described by rings (RINGFL) about a polar axis, much like the axisymmetric conical shell. The rings are connected by elements CFLUIDi) which have the properties of density and bulk modulus of compressibility. Each fluid ring produces internally a series of NASTRAN scalar points which represent "harmonic pressures" p^n and p^{n*} . The pressure function $p(\phi)$ is defined by the following equation:

$$p(\phi) = p^{0} + \sum_{n=1}^{N} p^{n} \cos n\phi + \sum_{n=1}^{N} p^{n*} \sin n\phi \qquad 0. < N < 100$$

where the set of harmonics 0, n and n* are selected by the user. If the user desires the output of pressure at specific points on the circular ring, he may identify them as "pressure points" (PRESPT) by giving a point number and an angle on a specified fluid ring. The output data will have the values of pressure at the angle, ϕ , given in the above equation. The output of free surface displacements normal to the surface (FREEPT) = also available at specified angles, ϕ .

2. The input da to NASTRAN may include all of the existing options except axisymmetric structural element data. All of the existing case control options may be included

SOLUTION OF THE FLUID MODEL

with the addition of source fluid case control requests. The case control card option "AXISYM=FLUID" is necessary when any harmonic fluid degrees of freedom are included. All of the structural element and constraint data may be used (but not connected to RINGFL, PRESPT, or FREEPT fluic points). The structure-fluid boundary is defined with the aid of special grid points (GRIDB) which may be used for any purpose that a structural grid point is presently used.

3. The output data options for the structural part of a hydroelastic model are unchanged from the existing NASTRAN options. The output values for the fluid will be produced in similar format as the displacement vectors with modifications for the harmonic data. Printed output for the fluid may include both real and complex values. Pressures and free surface displacements, and their velocities and accelerations, may be printed with the same request (the case control request PRESSURE=SET is equivalent to DISP=SET) as structural displacements, velocities and accelerations. Structural plots are restricted to GRID and GRIDB points and any elements connected to them. X-Y plot and Random Analysis capabilities are available for FREEPT and PRESPT points if they are treated as scalar points. The RINGFL point identification numbers may not be used in any plot request. Instead, the special internally generated points used for harmonics may be requested in &-Y plots and Random Analysis. See Section 24.4 for the identification numbering code. No element stress or force data is produced for the fluid elements. As in the axisymmetric conical shell problem, the case control request ARM@NICS = N is used to select up to the Nth harmonic for output.

24.1-3

HYDROELASTIC INPUT DATA

24.2 HYDROELASTIC INPUT DATA

A number of special NASTRAN bulk data cards are required for fluid anlaysis problems. These cards are compatible with structural NASTRAN data. A brief description of the uses for each bulk data card follows.

AXIF

This card controls the formulation of the axisymmetric fluid problem. It is required card if any of the subsequent fluid-related cards are present. The data references a fluid-related coordinate system to define the axis of symmetry. The gravity parameter is included on this card rather than on the GRAV card because the direction of gravity must be parallel to the axis of symmetry. For convenience, the values of density and bulk elastic modulus may also be input on this card in the event that these properties are constant throughout the fluid. A list of harmonics and the request for the nonsymmetric (sine) coefficients are included on this card to allow the user to select any of the harmonics without producing extra matrix terms for the missing harmonics. A change in this list, however, will force the RESTART logic to start at the beginning of the problem.

RINGFL

The geometry of the fluid model about the axis of symmetry is defined with the aid of these data cards. The RINGFL data cards serve somewhat the same function for the fluid as the GRID cards serve for the structure. In fact, each RINGFL card will produce a special grid point internally for each of the various harmonics selected on the AXIF data card. These internally generated points may not be connected directly to normal NASTRAN structural elements (see GRIDB and BDYLIST data cards), nor may constraints be applied directly to RINGFL fluid points.

CFLUIDi

The data on these cards are used to define a volume of fluid bounded by the referenced RINGFL points. The volume is called an element and logically serves the same purpose as a structural finite element. The physical properties (density and bulk modulus) of the fluid element may be defined on this card if they are variables with respect to the geometry. If a property is not defined, the default values on the AXIF card are assumed. Two connected circles (RINGFL) are used to define fluid elements containing the axis of symmetry. A choice of three or four points is available in the remainder of the fluid.

24.2-1

HYDROELASTIC ANALYSIS

GRIDB

ين فاد بينيو، المحدث عن تطعور بالعديات الدين ال

This card provides an alternate to the GRID card by defining of structural grid points indirectly as a location on a fluid ring. It is used to identify the structural grid point with a particular RINGFL fluid point for hydroelastic problems. The particular purpose for this card is to force the user to place structural boundary points in exactly the same locations as the fluid points on the boundary. The format of the GRIDB card is identical to the format of the GRID card except that one additional field is used to identify the corresponding RINGFL point. The GRDSET card, however, for GRIDB data does not supply defaults.

If the user desires, he may use GRIDB cards without a fluid model. This is convenient in case the user wished to solve his structural problem first and to add the fluid effects later without converting GRID cards to GRIDB cards. The referenced RINGFL point must still be included in a boundary list (BDYLIST), see below, and the AXIF card must always be present when GRIDB cards are used. (The fluid effects are eliminated by specifying no harmonics on the AXIF card.

FREEPT, PRESPT

These cards are used to define points on a free surface for surface displacement output and pressure output for points in the fluid. No constraints may be applied to these points. Scalar elements and direct matrix data may be connected to these points, but since the generalized displacements actually represent pressures, the physical meaning of the elements will be different than in the structural sense.

FSLIST, BDYLIST

The purpose for these cards is to allow the user to define the boundaries of the fluid with a complete freedom of choice. The FLIST card defines a list of fluid points which lie on a free surface. The BDYLIST data is a list of fluid points to which structural GRIDB points are connected. Points on the boundary of the fluid for which SPC, BDYLIST or FSLIST data are not defined are assumed to be rigidly restrained from motion in a direction normal to the surface.

With both of these lists the sequence of the listed points determines the nature of the boundary. The following directions will aid the user in producing a list.

 Draw the z axis upward and the r axis to the right. Plot the locations of the fluid points on the right hand side of z.

24.2-2

- 2. If one imagines himself traveling along the free surface or boundary with the fluid on his right side, the sequence of points encountered is used for the list. If the surface or boundary touches the axis, the word "AXIS" is placed in the list. "AXIS" may be used only for the first and/or last point in the list.
- 3. The free surface must be consistent with static equilibrium. With no gravity field, any free surface consistent with axial symmetry is allowed. With gravity, the free surface must be a plane perpendicular to the z axis of the fluid coordinate system.
- Multiple free surface lists and boundary lists are allowed. A fluid point may be included in any number of lists.

Figure 1 illustrates a typical application of the free surface and structural boundary lists.

FLSYM

This card allows the user the option of modeling a portion of the structure with planes of symmetry containing the polar axis of the fluid. The first plane of symmetry is assumed at $\phi = 0.0$ and the second plane of symmetry is assumed at $\phi = 360^{\circ}$ /M where M is an integer specified on the card. Also specified are the types of symmetry for each plane, symmetric (S) or antisymmetric (A). The user must also supply the relevant constraint data for the structure. The solution if performed correctly only for those harmonic coefficients that are compatible with the symmetry conditions as illustrated in the following example for quarter symmetry, M = 4.

Candon	Plane	Plan	e
series	1	S	٨
Cosine	S	0,2,4,	1,3,5,
	A	none	none
Sine	S	none	none
(*)	A	1,3,5,	2,4,6,

DMIAX

These cands are used for Direct Matrix Input for special purposes such as surface friction effects. They are equivalent to the DMIG cards, the only difference being the capability to

24.2-3

specify the harmonic numbers for the degrees of freedom. A matrix may be defined with either DMIG or DNIAX cards, but not with both.

and the strength of the second

ورد وريود

.)

=' **;**

and a state of a state of an art bound to the state of the state of

lan gerega - Saraha sina - Sagar **gi kiki** mananisi sa mikar - Saraha - Saraha

A BARAN

RIGID FORMATS

24.3 RIGID FORMATS

The characteristics of the fluid analysis problems which cause restrictions on the type of solution are:

- 1. The fluid-structure interface is mathematically described by a set of unsymmetric matrices. Since the first six Rigid Formats are restricted to the use of symmetric matrices, the fluid-structure boundary is ignored. Thus, for any of these Rigid Formats the program solves the problem for a fluid in a rigid container with an optional free surface and an uncoupled elastic structure with no fluid present.
- 2. No means are provided for the direct input of applied loads on the fluid. The only direct geans of exciting the fluid is through the structure-fluid boundary. The fluid problem may be formulated in any Rigid Format, however, only some will provide non-trivial solutions.

The suggested Rigid Formats for the analysis of axisymmetric fluids and the restrictions on each are described below:

Rigid Format No. 3 - Normal Modes Analysis

The modes of a fluid in a rigid container may be extracted with a conventional solution request. Free surface effects with or without gravity may be accounted for. Any structure data in the deck will be treated as a disjoint problem. (The structure may also produce normal modes.) Normalization of the eigenvectors using the PØINT option will cause a fatal error.

Rigid Format No. 7 - Direct Complex Eigenvalue Analysis

The coupled modes of the fluid and structure must be solved with this Rigid Format. If no damping or direct input matrices are added, the resulting complex roots will[®] be purely imaginary numbers, whose values are the natural frequencies of the system. The mode shape of the combination may be normalized to the maximum quantity (harmonic pressure or structural displacement) or to a specified structural point displacement.

Rigid Format No. 8 - Direct Frequency and Random Response

This solution may be used directly if the loads are applied only to the structural points. The use of overall damping (parameter g) is not recommended since the fluid matrices derived would not have physical significance. Output restrictions are listed on page

24.3-1

HYDROELASTIC AHALYSIS

Rigid Format No. 9 - Direct Transient Response

Transient analysis may be performed directly on the fluid-structure system if the following rules are followed.

- 1. Applied loads and initial conditions are only given to the structural points.
- All quantities are measured relative to static equilibrium. The initial values of the pressures are assumed to be at equilibrium.
- 3. Overall structural damping parameters ω_3 and g should not be used.

Rigid Formats 10, 11, and 12 - Modal Formulations

Although these Rigid Formats may be used to solve fluid dynamics problems, their practicality is limited. The modal coordinates used to formulate the dynamic matrices will be the normal modes of both the fluid and the structure solved as uncoupled systems. Even though the range of natural frequencies would be typically very different for the fluid than for the structure, NASTRAN will select both sets of modes from a given fixed frequency range. The safest method with the present system is the extraction of all modes for both systems with the Tridiagonalization Method. This procedure, however, results in a dynamic system with large full matrices. The Direct Formulation would be more efficient in that case. At present, the capability for fluid-structure boundary coupling is not provided with Rigid Formats 10, 11, and 12. However, the capability may be provided by means of ALTERs using the same logic as in the direct formulations.

HYDROELASTIC DATA PROCESSING

24.4 HYDROELASTIC DATA PROCESSING

The fluid related data cards submitted by the user are processed by the NASTRAN preface module to produce equivalent grid point, scalar point, element connection, and constraint data card images. Each specified harmonic, N, of the Fourier series solution produces a complete set of special grid and connection card images. In order to retain unique internal identification numbers, the user identification numbers are encoded according to the algorithm below:

RINGFL points:

NASTRAN grid Id. = User ring Id. + 1,000,000 x I_N

where

 $I_N = N + 1$ cosine series $I_N = N + 1/2$ sine series

CFLUIDi connection cards:

NASTRAN element Id. = User element Id. + 1000 x $I_{\rm N}$

where $\boldsymbol{I}_{\boldsymbol{N}}$ is defined above for each harmonic $\boldsymbol{N},$

For example, if the user requested all harmonics from zero to two, including the sine(*) series, each RINGFL card will produce five special grid cards internally. If the user's identification number (in field 2 of the RINGFL data card) were 37, the internally generated grid points would have the following identification numbers:

Harmonic	Id.		
0	1,000,037		
1+	1 ,50 0,037		
1	2,000,037		
2*	2,500,037		
2	3,000,037		

These equivalent grid points are resequenced automatically by NASTRAN to be adjacent to the original RINGFL identification number. A RINGFL point may not be resequenced by the user.

The output from matrix printout, table printout, and error messages will have the fluid points labeled in this form. If the user wished, he may use these numbers as scalar points for Random Analysis, X-Y plotting, or for any other purpose.

HYDROELASTIC ANALYSIS

In addition to the multiple sets of points and connection cards, the NASTRAN perface also may generate constraint sets. For example, if a free surface (FSLIST) is specified in a zerogravity field, the pressures are constrained by NASTRAN to zero. For this case the internally generated set of single point constraints are internally combined with any user defined structural constraints and will always be selected automatically.

If pressures at points in the fluid (PRESPT) or gravity-dependent normal displacements on the free surface (FREEPT) are requested, the program will convert them to scalar points and create a set of multipoint constraints with the scalar points as dependent variables. The constraint set will be internally combined with any user defined sets and will be selected automatically.

The PRESPT and FREEPT scalar points may be used as normal scalar points for purposes such as plotting versus frequency or time. Although the FREEPT values are displacements, scalar elements connected to them will have a different meaning than in the structural sense. Any restrictions on their motion will negate their basic definition.

HYDROELASTIC SAMPLE PROBLEM

24.5 HYDROELASTIC SAMPLE PROBLEM

24.5.1 Description

Section 24.5.2 lists the data deck for a sample hydroelastic prolbam. Figures 1 and 2 describe the problem and list the parameters of the model. The relatively small number of grid points were chosen to keep the problem simple, not for accuracy. Structural data cards are included in their standard formats. Explanations for key Case Control and Bulk Data cards are given in the following notes:

- The "AXISYM=FLUID" card is necessary to control the constraint set selections and the output formats for a fluid problem. It must appear above the subcase level.
- 2. "DISPLACEMENT=" and "PRESSURE=" case control cards are pseudonyms. "DISP=ALL" will produce all structure displacements, free surface displacements, and all fluid pressure values in the output. The "HARMONICS=" control is a limit on the harmonic data and has the same function as in an axisymmetrical concial shell problem.
- 3. The AXIF card defines the existence of a hydroelastic problem. It is used to define overall parameters and select the harmonic degrees of freedom, N=1 and N=2.
- 4. The RINGFL cards define the five points on the fluid cross section.
- 5. The CFLUIDi cards are used to define the volume of the fluid as finite elements connected by the RINGFL points. Since parameters ρ and B are missing, the default values on the AXIF card will be used.
- 6. The FSLIST card is used to define the free surface at z = 10.0. The density factor, p, is placed on the card in this case. If blank, the default value on the AXIF card is used.
- The fluid-structure boundary is defined on the BDYLIST card. The AXIF default density is used.
- 8. The GRIDB cards define the structure points on the fluid boundary. Points #3 through #6 are connected to the #2 fluid ring. The rotation in the r direction ("4" in field 8) is constrained.

24.5-1
HYDROELASTIC ANALYSIS

9. The fact that one-quarter symmetry was used for the structure requires the use of the FLSYM card. Symmetric-Antisymmetric boundaries indicate that only the cosine terms for the odd harmonics will interact with the structure. If Symmetric-Symmetric boundary conditions were chosen on the FLSYM data card, only the even harmonics of the cosine series would interact with the structure.

24.5.2 Driver Deck and Bulk Data Listing

ì

policy we are however,

Card No. 1 ID HYDRO, USER APP DISP 2 7,0 3 SØL TINE 4 5 CEND TITLE = SAMPLE HYDR@ELASTIC PROBLEM 6 7 SUBTITLE = EIGENVALUE ANALYSIS WITH FLEXIBLE BOUNDARY 8 AXISYM=FLUID 9 SPC = 3CMETHOD = 1 10 11 BUTPUT DISP = ALL 12 HARMONICS = ALL 13 14 ELFORCE = ALL BEGIN BULK 15

.

				BULK	DATA FIE	LD				
	1	2	3	4	5	6	7	8	9	10
16	AXIF	2	32.2	0.03	NØ					+AX
17	+AX	1	3							
18 19	CORD2C +CØ	2 1.0	0.	0. 0.	0.	0.	0.	0.	1.0	+CØ
20 21 22	RINGFL RINGFL RINGFL	1 7 13	4.0 4.0 4.0		10.0 5.0 0.0	2 8	8.0 8.0		10.0 5.0	
23 24 25 26	CFLUID2 CFLUID2 CFLUIC3 CFLUID4	101 102 103 104	1 7 7 1	7 13 8 2	13 7	8				
27	FLIST	0.03	AXIS	1	2					
28	BOYLIST		2	8	13					

HYDROELASTIC SAMPLE PROBLEM

.

......

......

4 78 Pr. - 100

	<u> </u>	22	3	4	5	6	7	8	9	10
29 30 31 32 33 34 35 36 37 38 39 40	GRIDB GRIDB GRIDB GRIDB GRIDB GRIDB GRIDB GRIDB GRIDB GRIDB GRIDB	3 4 5 6 9 10 11 12 14 15 16 17			0.0 30.0 60.0 90.0 0. 30. 60. 90. 0. 30. 60. 90.		222222222222222222222222222222222222222	4	2 2 2 8 8 8 8 8 13 13 13 13 13	
41 42 43 44 45 46 47 48 49 50 51	CQUAD2 CQUAD2 CQUAD2 CQUAD2 CQUAD2 CQUAD2 CQUAD2 PQUAD2 PQUAD2 MAT1 SPC1 SPC1 SPC1	10 11 12 13 14 15 11 12 3 3 3 3	11 11 11 11 11 11 12 10.6+6 246 135 135	3 4 5 9 10 11 0.5 3 6 14	9 10 11 14 15 16 0.3 9 12 15	10 11 12 15 16 17 0.05 14 17 16	4 5 6 10 11 12			
52	FLSYM	4	\$	A						
53 54 55	EIGC +EI ENDDATA	1 0.	INV O.	MAX O.	5.	3.	2	2		+EI

Table 1. Bulk Data Cards of Special Relevance to Hydroelastic Analysis

AXIF	Indicates the existence of an axisymmetric fluid analysis.
BDYLIST	Defines the boundary between a fluid and a structure.
CFLUIDi (i=2,3 or 4)	Defines three types of fluid elements.
DMIAX	Direct input matrix terms for axisymmetric fluid or structure problems.
FLSYM	Allows fluid boundary matrices to conform to structural symmetry definitions.
FREEPT	Defines points on the surface of a fluid for surface displacement output.
FSLIST	Defines the fluid points which lie on a free surface boundary.
GRIDB	Defines structural grid points as locations on a fluid ring.
PRESPT	Defines pressure points in the fluid for pressure data output.
RINGFL	Defines a point on a fluid ring.

24.5-3

HYDROELASTIC ANALYSIS





Fluid: Density, ρ = 0.03 Bulk Modulus, B = ⇔ Gravity, g = 32.2

Structure: Thickness, = 0.5Density, $\rho = 0.05$

Figure 1. Sample hydroelastic problem.

24.5-4

HYDROELASTIC SAMPLE PROBLEM

3

المعصومية المعالي المتعامية



FSLIST: : 36, 37, 38, 39, 40 BDYLIST #1: AXIS, 22, 23, 29, 36 BDYLIST #2: 40, 35, 28, 21, 14, 7, 6, 5, 4, 3, 2, 1, AXIS

Figure 2. Examples of boundary lists.

25. ACOUSTIC CAVITY ANALYSIS

25.1 MODELING OF THE ACOUSTIC CAVITY

The NASTRAN structural analysis system is used as the basis for acoustic cavity analysis. Many of the structural analysis options such as selecting boundary conditions, applying loading conditions, and selecting output data are also available for acoustics. The data cards specifically used for acoustic cavity analysis are listed in Table 1 and described below. The card formats are exhibited in Section 2.4 of the User's Manual. Their purposes are analogous to the use of structural data cards. A gridwork of points is distributed over the longitudinal cross section of an acoustic cavity and finite elements are connected between these points to define the enclosed volume.

The points are defined by GRIDF data cards for the axisymmetric central fluid cavity and by GRIDS data cards for the radial slots. The GRIDF points are interconnected by finite elements via the CAXIF2, CAXIF3, and CAXIF4 data cards to define a cross sectional area of the body of rotation. The CAXIF2 element data card defines the area of the cross section between the axis and two points off the axis (the GRIDF points may not have a zero radius). The CAXIF3 and CAXIF4 data cards define triangular or quadrilateral cross sections and connect three or four GRIDF points respectively. The density and/or bulk modulus at each location of the enclosed fluid may also be defined on these cards.

The GRIDS points in the slot region are interconnected by finite elements via the CSLØT3 and CSLØT4 data cards. These define finite elements with triangular and quadrilateral crosssectional shapes respectively. The width of the slot and the number of slots may be defined by default values on the AXSLØT data card. If the width of the slots is a variable, the value is specified on the GRIDS cards at each point The number of slots, the density, and/or the bulk modulus of the fluid may also be defined individually for each element on the CSLØT3 and CSLØT4 cards.

The AXSLØT data card is used to define the overall parameters for the system. Some of these parameters are called the "default" values and may be selectively changed at particular cross sections of the structure. The values given on the AXSLØT card will be used if a corresponding value on the GRIDS, CAXIFi, or CSLØTi is left blank. The parameters ρ (density) and B (bulk modulus) are properties of the fluid. If the value given for bulk modulus is zero, the fluid is considered incompressible to the program. The parameters M (number of slots)

25.1-1

ACOUSTIC CAVITY ANALYSIS

and the second second second second second second second second second second second second second second second

and W (slot width) are properties of the geometry. The parameter M defines the number of equally spaced slots around the circumference with the first slot located at $\phi = 0^{\circ}$. The parameter N (harmonic number) is selected by the user to analyze a particular set of acoustic modes. The pressure is assumed to have the following distribution.

$p(r,z,\phi) = p(r,z) \cos N\phi$

If N = 0, the breathing and longitudinal modes will result. If N = 1 the pressure at ϕ = 180° will be the negative of the pressure at ϕ = 0°. If N = 2, the pressures at ϕ = 90° and ϕ = 270° will be the negative of that at ϕ = 0°. Values of N larger than M/2 have no significance.

The interface between the central cavity and the slots is defined with the SLBDY data cards. The data for each card consists of the density of the fluid at the interface, the number of radial slots around the circumference, and a list of GRIDS points that are listed in the dequence in which they occur as the boundary is traversed. In order to ensure continuity between GRIDF and GRIDS points at the interface, the GRIDF points on the boundary between the cylindrical cavity and the slots are identified on the corresponding GRIDS data cards rather than on GRIDF cards. Thus, the locations of the GRIDF points will be exactly the same as the locations of the corresponding GRIDS points.

Various standard NASTRAN data cards may be used for special purposes in acoustic analysis. The SPC1 data card may be used to constrain the pressures to zero at specified points such as at a free boundary. The formats for these cards are included in Section 2.4 of the User's Manual. Dynamic load cards, direct input matrices, and scalar elements may be introduced to account for special effects. The reader is referred to Sections 1.4 and 1.5 of the User's Manual for instruction in the use of these cards.

25.1-2

ASSUMPTIONS AND LIMITATIONS

25.2 ASSUMPTIONS AND LIMITATIONS

The accuracy of the acoustic model will be dependent on the selection of the mesh of finite elements. The assumption for each element is that the pressure field has a linear variation over the cross section and a sinusoidal variation around the axis in the circumferential direction. In areas where the pressure gradient changes are large, such as near a sharp corner, the points in the mesh should be placed closer together so that large changes in flow may be defined accurately by the finite elements. The shape of the finite elements plays an important part in the accuracy of the results. It has been observed that long narrow elements produce disproportionate errors. Cutting a large square into two rectangles will not improve the results whereas dividing the square into four smaller squares may decrease the local error by as much as a factor of ten.

The slot portion of the cavity is limited to certain shapes because of basic assumptions in the algorithms. The cross section of the cavity normal to the axis must have a shape that is reasonably well defined by a central circular cavity having equally spaced, narrow slots. Various shapes are shown in Figure 1 in the order of increasing expected error. It is recommended that shapes such as the cloverleaf and square cross section be analyzed with a full three-dimensional technique. The assumption of negligible pressure gradient in the circumferential direction within a slot is not valid in these cases.

The harmonic orders of the solutions are also limited by the width of the slots. The harmonic number, N, should be no greater than the number of slots divided by two. The response of the higher harmonics is approximated by the slot width correction terms discussed in the NASTRAN Theoretical Manual, Section 17.1.

The output data for the acoustic analysis consists of the values of pressure in the displacement vector selected via the case control card "PRESSURE = i." The velocity vector components corresponding to each mode may be optionally requested by the case control card "STRESS = i," where i is the set number indicating the element numbers to be used for output, or by the words "STRESS = ALL." The "SET=" card lists the element or point numbers to be output.

Plots of the finite element model and/or of the pressure field may be requested with the NASTRAN plot request data cards. The central cavity cross section will be positioned in the XY plane of the Basic Coordinate System of NASTRAN. The slot elements are offset from the XY

25.2-1

ACOUSTIC CAVITY ANALYSIS

.

_

plane by the width of the slot in the +Z direction. The radial direction corresponds to Z and the axial direction corresponds to the Y direction. Pressures will be plotted in the Z direction for both the slot points and the central cavity points. The Case Control data cards for plotting are documented in the User's Manual. The PLØTEL elements are used for plotting the acoustic cavity shape. The plot request card "SET n INCLUDE PLØTEL" must be used where n is a set number.

ACOUSTIC CAVITY SAMPLE PROBLEM

25.3 ACOUSTIC CAVITY SAMPLE PROBLEM

25.3.1 Description

Section 25.3.2 contains a listing of the data cards used as a simple example of acoustic cavity analysis. The problem to be solved is illustrated in Figure 2. The model was subdivided into only ten finite elements in order to limit the number of data cards. For reasonable engineering accuracy, this model should be subdivided into at least four times that number of elements.

Each data card in the listing is given a number on the left side. The following is a brief description of each card:

Card(s)

- 1-5 Each data card in the Executive Control Deck has the format of a request word and a selection separated by blanks or a comma. The ID card is first, the CEND card is last, but the intermediate cards may appear in any order. The user may put any pair of words on the ID card for identification purposes. In this particular case Rigid Format number 3 (SQL 3,0) was chosen which is Normal Modes analysis. A limit of two minutes CPU time was set (TIME 2).
- 6-7 The TITLE= and SUBTITLE= cards may contain any list of letters and numbers following the (=) sign. This list will appear on the first two lines of each output page.
- 8 The method of eigenvalue extraction is selected with the METHØD= data card. The number 11 refers to the identification number of an EIGR bulk data card which appears below as card 32 and 33.
- 9-11 A simple output request is illustrated with these cards. PRES=ALL will result in printout of all pressures at the GRIDF and GRIDS points. STRESS=ALL will result in the printout of all velocities in the elements. This printout will occur for all extracted eigenvectors. Selected points or elements can be printed via the SET card described in the User's Manual.
- 12 The BEGIN BULK card denotes the beginning of the Bulk Data Deck. The Bulk Data Deck cards may occur in any order. Putting these cards in alphabetic sort will save NASTRAN sorting time in large problems, however.

25.3-1

ACOUSTIC CAVITY ANALYSIS

13 In this problem all the parameters except slot width w_d are constant throughout the volume. The data values on the AXSLØT card will be used whenever a corresponding entry in the following cards is blank.

بالتباريق الأبار للالتابة وتوهيه وتعربون الماهم وترز

- 14-20 The location of points on the slot are defined with these cards. Cards 14, 16, 18 and 20 serve a dual purpose by defining a GRIDS point identification number in field 2 and a GRIDF point identification number in field 6. The two types of points thereby are forces to have the same locations at the interface.
- 21,22 The location of points within the axisymmetric fluid cavity are described by the GRIDF card. No points are allowed to have a zero or negative radius.
- 23-31 These cards describe the elements shown in Figure 2. Each element is given a unique identification number and a list of the connected GRIDS or GRIDF points. Since the parameters ρ and B are constants, these fields are left blank so the values on the AXSEØT card will be used.
- 32,33 The EIGR card is used to define parameters for eigenvalue extraction (resonant frequencies). More than one of these cards may appear. The method to be used is selected with the METHØD= data card in the Case Control Deck (card 8). With this particular card we selected the Givens Tridiagonalization method (GIV) with a desired number of three (Nd = 3) output mode shapes. The modes will be normalized such that the maximum pressure is 1.0 (NØRM=MAX). These two cards illustrate a continuation card.
- The SLBDY card defines the boundary between the slot and the central cavity. Both the density (ρ) and the number of radial slots (M) are blank so the AXSLØT defaults are used, i.e., $\rho = 1.2 \times 10^{-7}$ and M = 4. Only four GRIDS points are on the boundary so a continuation card is not necessary. Field 8 being blank signifies the last entry.
- 35. The ENDDATA card is required to denote the end of the bulk data. Any following cards will be ignored by NASTRAN.

25.3-2

.

ł

25.3.2 Driver Deck and Bulk Data Listing

Ç,

ł.

εì

Card Ng.										
1 2 3 4 5	ID ACOUS APP DISP SOL 3,0 TIME 2 CEND	,CAY				Executiv	ve Control	Cards		
6 7 8 9 10 11 12	TITLE - SUBTITLE METHDO - ØUTP BEGIN BU	ACOUSTIC = FIRST 11 UT PRES = STRESS LK	CAVITY E HARMONIC ALL = ALL	XAMPLE PR	KOBLEM	Case Cor	ntrol Data	a Cards		
	ı	2	3	4	5	6	7	8	9	10
13	AXSLOT	1.2-7	21.0	1	T	4				
14 15 16 17 18 19 20	GRIDS GRIDS GRIDS GRIDS GRIDS GRIDS GRIDS	2 3 5 6 8 9 12	4.0 8.0 4.0 8.0 4.0 8.0 4.0	0.0 .0 4.0 4.0 8.0 8.0 1.2+1	0.2E 01 1.0 2.0 1.0 2.0 1.0 2.0	1 4 7 11				
21 22	GRIDF GRIDF	10 13	2.0 2.0	12.0 1.4E1						
23 24	CSLØT4 CSLØT4	1 2	2 5	3 6	6 9	5 8				
25	CSLØT3	3	8	9	12					
26 27 28 29	CAXIF2 CAXIF2 CAXIF2 CAXIF2 CAXIF2	4 5 6 9	1 4 7 10	4 7 10 13						
30 31	CAXIF3 CAXIF3	7 8	7 10	10 11	11 13					
32 33	EIGR +AB	ו 11 11 אאג	GIV				3			+AB
34	SLBDY			12	8	5	2			
35	ENDDATA							<u> </u>		

ACOUSTIC CAVITY ANALYSIS

a same sono in

and the sugar

× -

1

Table 1. Bulk Data Cards of Special Relevance to Acoustic Cavity Analysis

AXSLØT	Contains the harmonic index and default values for several several
CAXIFi (1=2,3 or 4)	Defines an axisymmetric element which connects i fluid active
CSL#Ti (i=3 or 4)	Defines evenly spaced radial slots.
GRIDF	Defines a scalar degree of freedom for harmonic applicate of
GRIDS	Defines a scalar degree of freedom with a two-dimensional level.
SLBDY	Defines a list of slot points which lie on an interface between an axisymmetric fluid and a set of evenly spaced radial slots.

日本を言いていたい

ACOUSTIC CAVITY SAMPLE PROBLEM

.



SMALL ERRORS

LARGE ERRORS

.



ACOUSTIC CAVITY ANALYSIS



Parameters:

Density: $\rho = 1.1463 \times 10^{-7} \text{ lb-sec}^2/\text{in}^4$ Bulk Modulus: $B = \rho a^2 = \gamma RT = 20.59 \text{ lb/in}^2$ Harmonic: N = 1Number of slots: M = 4Note: Consistent Dimensional Units must be used.

FINITE ELEMENT MODEL:



Figure 2. Description of example problem.

25.3-6

26. FULLY STRESSED DESIGN ANALYSIS

26.1 FULLY STRESSED DESIGN THEORY

The fully stressed design option is part of the static analysis Rigid Format for structural analysis. The theoretical basis is presented more fully in the Theoretical Nanual. The Bulk Data cards of special relevance to fully stressed design are listed in Table 1. Essentially, however, the process begins by performing a static analysis for all loading conditions using the initial values for all element properties. A new property, P_2 , will be scaled such that

$$P_2 = P_1 \left[\frac{\alpha}{\alpha + (1 - \alpha)\gamma} \right]$$
⁽¹⁾

where P_1 is the current property value of γ is an iteration factor with a default value of unity. The scale factor, α , is defined as follows:

$$\alpha = Max \left(\frac{\sigma}{\sigma_{\ell}}\right) , \qquad (2)$$

where σ is a stress value and σ_g is a stress limit. The maximum value of α is taken for all loading conditions. Values of γ smaller than unity limit the property change in a single iteration, and thereby tend to improve the stability of the process. The maximum change in any property is limited by

$$K_{\min} < \frac{P_2}{P_1} < K_{\max}$$
 (3)

where P_i is the initial value of property and K_{\min} and K_{\max} are user-supplied limits.

Convergence is achieved by completing the user-specified number of iterations, by having all selected element properties reach the user-specified limits, or by satisfying the following convergence criteria:

$$\frac{|\sigma - \sigma_{\ell}|}{\sigma_{\ell}} < \varepsilon$$
 (4)

where c is a user-supplied convergence limit.

26.1-1

APPLICATION

26.2 APPLICATION

Functional modules (@PTPR1 AND @PTR2) are provided to automatically adjust the properties based on maximum stress levels, and to control the number of design iterations based on usersupplied convergence criteria. All elements using a common property are sized together, i.e., a plate with uniform thickness remains uniform. If the user wishes to scale the properties for each element separately, each element must have its own property card. After a sufficient number of iterations, the element properties will be adjusted to the minimum values necessary to carry the prescribed loads. The following actions are required by the user to utilize the fully stressed design capability:

- The user must select stress output in the Case Control Deck for all elements that will participate in the fully stressed design.
- All required stress limits must be specified on the structural material cards associated with element properties that will participate in the fully stressed design.
- 3. The property optimization parameters must be specified on the bulk data card P@PT. This card contains user-specified values for the maximum number of iterations, the convergence criteria (ε), the iteration factor (γ), and output options to print and/ or punch the calculated values of the element properties.
- 4. The property optimization limits $(K_{min} \text{ and } K_{max})$ must be specified on the PLIMIT bulk data card if the user wishes to limit the maximum and minimum values of the element properties.

The detailed definitions of the scale factors for each of the element types are given in Table 1. The symbols σ_t , σ_c , and σ_s represent the limiting stress values in tension, compression and shear, given on the structural material cards. All of the properties listed for each element are scaled in the same way, i.e., both are the area and torsional constant for the RØD are modified using the same scale factor. Some bulk data cards which are of special importance to this type of problem are listed in Table 2.

SAMPLE PROBLEM

26.3 SAMPLE PROBLEM

An example is presented here to illustrate the fully stressed design technique using Rigid Format 1.

26.3.1 Description

A flat plate with a reinforced hole in the center is optimized for stresses due to a uniform end load. Restrictions on the minimum thickness are maintained. The plate is shown in Figure 1, and the finite element idealization is illustrated in Figure 2. This problem has been investigated by G. G. Pope (Reference 1). Due to symmetry, only one quadrant is modeled. Due to the membrane load, all rotations and normal displacements are constrained. The QCMEM and TRMEM elements are used for the plate and RØD elements for the reinforcement around the hole.

The problem demonstrates several features unique to fully stressed design capability to NASTRAN.

- Elements with no limits on the range of the property change, i.e., the RØD has no PLIMIT data.
- 2. Elements with a lower limit on the property optimization card. All membrane elements are required to have a resultant thickness which must not be less than a minimum thickness. This minimum is determined from the thickness obtained when the plate without a hole is subjected to an end load at a prescribed stress limit.
- 3. Elements whose stress is not inspected but being in an area of nearly uniform stress have their properties changed due to another element's stress. Element 7 has no stress request but does have the same property identification number as element 17. This type of optimization can save computer time at the expense of a design that may not be truly optimized.
- A property whose value depends on the maximum stress of two elements. Elements 5
 and 15 have the same property card. This option may be necessary if insufficient
 core is allocated.
- 5. Temperature dependent stress limits for material 3.

26.3-1

FULLY STRESSED DESIGN ANALYSIS

6. Using one stress limit only. The membrane elements use the maximum principal shear only which is 1/2 the major principal stress allowed. This stress limit was chosen to model better the octahedral limit uses in Reference 1.

The rod elements use only the tension and compression stress appropriate to the given property, namely, area.

7. An additional load case that was not included in the fully stressed design because a stress request was not made. The second subcase may be considered a displacement verification of this load case.

26.3.2 Input

1. Parameters

l = 30.0 in	(total length)
w = 20.0 in	(total width)
d = 10.0 in	(hole diameter)
t _o = 3.348 in	(initial plate thickness)
$A_0 = 1.674 \text{ in}^2$	(initial rod cross sectional area)
E = 30.x10 ⁶ psi	(modulus of elasticity)
v = 0.3	(Poisson's ratio)
t = 1.0 in e	(lower limit for plate thickness corresponding to a 25.0x10 ³ maximum principal stress)

2. Boundary conditions:

on y = 0 plane, $u_y = 0$ (symmetry) on x = 0 plane, $u_x = 0$ (symmetry) all points $u_z = \theta_x = \theta_y = \theta_z = 0$ (permanent constraints)

3. Loads:

First subcase: uniform load, $F_{10} = 25.0 \times 10^3$ lb/in Second subcase: at grid points 69 and 79, $F_{12} = 1000.0$ lb at grid point 78, $F_{12} = -2000.0$ lb

(contact load on rim of hole - displacement check only)

SAMPLE PROBLEM

and a star second a second second second second second second second second second second second second second

26.3.3 Theory

The mathematical programming technique used in Reference 1, from which the example problem was taken, differs from the one used in NASTRAN or described above and in the Theoretical Manual, Section 4.4. The two techniques might be expected to give similar results when the same model is used. However, Reference 1 employs elements which allow varying properties and stresses, while NASTRAN elements allow only constant properties and stresses. Somewhat different geometry is used in the NASTRAN model to include the use of quadrilateral elements for illustration. Additional features of the NASTRAN model are discussed in items 3, 4 and 5, Section 26.3.1 above.

26.3.4 Results

The optimization process in this problem is terminated at five iterations. The initialweight-to-final-weight ratio is 2.70, compared to Pope's result of 2.63. Tables 3 and 4 show the optimized nondimensional properties of the elements around the arch. Note that the results from reference 1 are averaged to provide an equivalent constant property element for comparison.

FULLY STRESSED DESIGN ANALYSIS

26.3.5 Driver Deck and Sample Bulk Data

Le president

Access of the second

```
Card
No.
 0
     NASTRAN FILES=(PLT2)
               DEM1161,NASTRAN
  1
      ID
      APP
 2
               DISPLACEMENT
  3
      SOL
               1,0
      TIME
  4
               9
      DIAG
  5
               8
  6
      CEND
 7
      TITLE = FULLY STRESSED DESIGN OF A PLATE WITH A REINFORCED HOLE
 8
     SUBTITLE = NASTRAN DEMONSTRATION PROBLEM NO. 1-16-1
     LABEL = TEMPERATURE DEPENDENT MATERIALS.
 9
10
          TEMPERATURE(MATERIALS) = 3000
          SPC = 11
11
12
          DISPLACEMENT = ALL
13
     SUBCASE 10
14
          LABEL = DESIGN CASE - UNIFORM END LOAD
15
          SET 111 = 1 THRU 105 EXCEPT 7
16
          STRESS = 111
17
          LOAD = 10
18
     SUBCASE 12
19
          LABEL = CHECK CASE - CONTACT LOAD AT NOZZLE.
20
          LØAD = 12
21
     PLOTID = NASTRAN DEMONSTRATION PROBLEM NO. 1-16-1
22
     ØUTPUT(PLØT)
23
     PLØTTER SC
24
          SET 1 = 1, 7, 38, 61, 69
          SET 2 = INCLUDE ELEMENTS QDMEM, TRMEM
25
26
               MAXIMUM DEFØRMATIØN 0.8
27
               AXES Z, X, Y
28
               VIEW 0.0, 0.0, 0.0
29
          FIND SCALE, ØFIGIN 12, SET 1
     PTITLE = ARCH MODEL
30
31
     PLOT SET 2, ORIGIN 12 LABEL
     PTITLE = DEFLECTION VECTORS FOR BOTH LOADS AND EACH ITERATION
32
     PLØT STATIC DEFØRMATIØN SET 2, ØRIGIN 12, VECTØR RXY, SYMBØL 7
FIND SCALE, ØRIGIN 12, SET 1, REGIØN 0.0, 0.0, 0.6, 1.0
33
34
     PTITLE = ARCH MODEL REFLECTED ABOUT VERTICAL AXIS
35
     PLOT SET 2. ØRIGIN 12, SYMMETRY X, SET 2, ØRIGIN 12
36
37
     BEGIN BULK
```

SAMPLE PROBLEM

ł

4

:::****

Ξ. - \

	ı	2	3	4	5	6	7	8	9	10
38 39 40 41 42 43 44 45 46 47 48 9 50 51 52 53 55 55 56 57 58	CQDHEM CRDD CTRNEM FØRCE GRDSET GRID HAT1 +CØNST HATT1 PARAM PLINIT PØPT PQDMEM PRØD PTRMEM SPC1 TABLEM1 +TAB-N1 TENPD ENDDATA	1 101 11 10 1 1 2 GRDPNT QDHEM 5 1 101 11 11 222 1. 3000	1 30.E06 0 .2986858 .04 1 3 1 1 12.5E3 80.	11 48 13 -10. 12.5E3 222 .95 3.348 1.674 3.348 9 10.	13 49 11 .3125E5 15. .3 1 2 19 12.5E3	3 102 21 .0 .283 THRU YES 29 ENDT	1 102 1.0 61 39	49 .0 3456 70.0	59	+CØNST +MATT1 FSD FSD +TAB-M1

FULLY STRESSED D'SIGN ANALYSIS

Element	Stress Value Used	Scale Factor (α)	Properties Changed
RØD Tube	Axial Tension (σ ₁) Axial Compression (σ ₂) Torsion (τ)	$\operatorname{Max}\left(\frac{\sigma_{1}}{\sigma_{t}}, \frac{\sigma_{2}}{\sigma_{c}}, \frac{\tau}{\sigma_{s}}\right)$	Area (A) Torsional Constant (J)
BAR	$\begin{array}{llllllllllllllllllllllllllllllllllll$	$\operatorname{Max}\left(\frac{\sigma_{a1}}{\sigma_{t}}, \frac{\sigma_{b1}}{\sigma_{t}}, \frac{\sigma_{b1}}{\sigma_{t}}, \frac{\sigma_{a2}}{\sigma_{c}}, \frac{\sigma_{b2}}{\sigma_{c}}\right)$	Area (A) Torsional Constant (J) Moments of Inertia (I ₁ , I ₂ , I ₁₂)
TRMEM QDMEM	Principal Tension (σ ₁) Principal Compression (σ ₂) Maximum Shear (τ _m)	$Max\left(\frac{\sigma_{1}}{\sigma_{t}}, \frac{\sigma_{2}}{\sigma_{c}}, \frac{\tau_{m}}{\sigma_{s}}\right)$	Thickness (t)
TRPLT QDPLT TRBSC	Same as Above (Fiber Distances z ₁ & z ₂)	Same as Above	Moment of Inertia (I)
TRIA1 QUAD1	Same as Abov e	Same as Above	Moment of Inertia (I) Membrane Thickness (t _l)
TRIA2 QUAD2	Same as Above	Same as Above	Thickness (t)
SHEAR	Maximum Shear (τ _m)	$\frac{\tau_{m}}{\sigma_{s}}$	Thickness (t)

Table 1. Scale Factors for Fully Stressed Design

ORIGINAL PAGE IS OF POOR QUALITY

OF POOR QUALITY

26.3-6

SAMPLE PROBLEM

••••

· · · ·

ORIGINAL PAGE IS OF POOR QUALITY

ć

Table 2. Bulk Data Cards of Special Relevance to Fully Stressed Design

FØRCE	Defines a static load at a grid point.
MATI	Defines the material properties for linear, temperature-independent, isotropic materials.
MATTI	Specifies table references for material properties which are temperature dependent.
PLIMIT	Defines the maximum and minimum limits for ratio at new property to original property.
PØPT	Defines the basic parameters and existence of a property optimization analysis.
TABLEMI	Defines a tabular function for use in generating temperature- dependent material properties.

TEMPD Defines a default temperature for all grid points at the structural model which have not been given a temperature on a TEMP card.



=;

ng metalo iko Ki Lina Riman

1 A 24 C 4

FULLY STRESSED DESIGN ANALYSIS

Element	Original t/t	Noference 1 Average t/te	NASTRAN t/t _e
37	3.348	1.24	1.00
38	3.348	1.00	1.04
39	3.348	1.00	1.00
46	3.348	2.10	1.14
47	3.348	1.34	2.00
57	3.348	3.32	1.34
59	3.348	3.19	4.40
67	3.348	4.58	5.47
68	3.348	3.26	1.00
59	3. 348	4.52	5.49

Table 3. Optimized Nondimensional Thickness Comparisons.

Element	Original A/dt _e	Reference 1 Average A/dt	NASTRAN A/dte
101	. 1674	.0249	.00716
102	. 1674	.0238	0.0 effective
103	. 1674	. 0636	.05019
104	. 1674	. 1880	. 1839
105	. 1674	. 3540	. 3287

Table 4. Optimized Nondimensional Area Comparisons.



Ř.

ł



Figure 1. Plate with reinforced hole.

FULLY STRESSED DESIGN ANALYSIS

. 3



Figure 2. Finite element wodel.

FILMED MAR 12 1982

#